

Mcpartlan, David ORCID: https://orcid.org/0000-0003-1445-8613 (2023) Young researchers in schools: a participative action research study into the efficacy of a whole school mental health strategy. Doctoral thesis, University of Cumbria.

Downloaded from: http://insight.cumbria.ac.uk/id/eprint/7274/

Usage of any items from the University of Cumbria's institutional repository 'Insight' must conform to the following fair usage guidelines.

Any item and its associated metadata held in the University of Cumbria's institutional repository Insight (unless stated otherwise on the metadata record) may be copied, displayed or performed, and stored in line with the JISC fair dealing guidelines (available <u>here</u>) for educational and not-for-profit activities

provided that

- the authors, title and full bibliographic details of the item are cited clearly when any part of the work is referred to verbally or in the written form
 - a hyperlink/URL to the original Insight record of that item is included in any citations of the work
- the content is not changed in any way
- all files required for usage of the item are kept together with the main item file.

You may not

- sell any part of an item
- refer to any part of an item without citation
- amend any item or contextualise it in a way that will impugn the creator's reputation
- remove or alter the copyright statement on an item.

The full policy can be found <u>here</u>.

Alternatively contact the University of Cumbria Repository Editor by emailing insight@cumbria.ac.uk.

Young researchers in schools: A participative action research study into the efficacy of a whole school mental health strategy

This thesis is submitted for the degree of Doctor of Philosophy

by

David Simon McPartlan

to the

University of Cumbria, (Institute of Health)

in August 2023

The word count for this thesis is 81,650 words and it is within the permitted maximum.

Signature D. M. P. .

Abstract

In England between April and June of 2021, 190,271 young people were referred to mental health services, an increase of 134% since June 2020 (Local Government Association, 2022). Since 2014, schools have been expected to support young people's mental health needs (Department for Education, 2014).

This qualitative work critically explores the efficacy of a whole school mental health strategy, in a comprehensive academy in England. As participative action research (PAR), this approach involved self-selecting young people aged 16-18 years old to collaborate with me as a young research team (YRT). They worked closely with the study participants, aged 12-15 years, who volunteered from the pupil premium cohort (The Department for Education, 2022). A weekly cycle of meetings between myself, the YRT and participants took place, providing qualitative data. This research focuses on a school mental health strategy and new approaches to young people's participation in school decision-making.

My findings and contributions to knowledge are divided into two sections. Firstly, I present findings that indicate a school mental health strategy requires trusting staff / young people relationships to be successful. As a further contribution, I suggest relationships are viewed through a nanosystems lens (Rudasill et al., 2018), so schools can start to address this issue. My second area contributing to new knowledge is how this unique methodology has enabled the development of youth participative dialogic action research (YPDAR). I have discovered how using YPDAR can positively impact young people and school character. YPDAR benefits young people as attachment-like relationships may develop between young researchers and participants. In addition to improving young people's socio-emotional skills, this research can boost their confidence, empowerment, agency and trust in the school. School character also benefits, as YPDAR requires a power shift from school to young people, strengthening relationships and the development of trust between them.

Table of Contents

<u>ABST</u>	RACT	I
<u>ACKN</u>	IOWLEDGMENTS	VII
DECL	ARATION OF AUTHORSHIP	VIII
<u>GLOS</u>	SARY:	IX
ABBR	REVIATIONS:	X
<u>PROL</u>	OGUE	1
<u>PART</u>	1 SETTING THE SCENE	4
<u>СНАР</u>	PTER 1: INTRODUCTION AND RESEARCH RATIONALE	<u>5</u>
1.1	YOUNG PEOPLE THEIR MENTAL HEALTH AND SCHOOLS	
1.2	THE DEVELOPMENT OF A WHOLE SCHOOL MENTAL HEALTH STRATEGY	
1.3	YOUNG PEOPLE AS RESEARCHERS	
1.3.1		
1.3.2		
1.4	RESEARCH AIMS	
1.5	RESEARCH, KNOWLEDGES AND VOICE	10
<u>CHAP</u>	PTER 2: RESEARCH CONTEXT	11
2.1	INTRODUCTION	11
2.2	A HISTORY OF EDUCATION POLICY AND HOW IT HAS IMPACTED YOUNG PEOPLE'S MENTAL HEALTH	11
2.2.1	Post-war welfare state	11
2.2.2	FROM A POST-WAR VISION OF WELFARE TO THE BEGINNINGS OF COMPETITION	13
2.2.3	COMPETITION AND MARKETISATION OF EDUCATION	-
2.2.4	THE CHANGING SCHOOL SYSTEM AND ITS IMPACT ON VALUES	17
2.3	SCHOOL PURPOSE	
2.4	SCHOOL CHARACTER AND HOW IT MAY INFLUENCE OUTCOMES FOR YOUNG PEOPLE	
2.4.1	OVERVIEW	
2.4.2	WHAT DO I MEAN BY THE TERM 'SCHOOL CHARACTER'?	
2.4.3	VALUES AND CLIMATE AS SUBSETS OF SCHOOL CHARACTER	
2.5		
2.5.1	BRONFENBRENNER'S ECOLOGICAL SYSTEMS THEORY (EST)	
2.6	CHAPTER SUMMARY	34

|--|

3.1	CHAPTER INTRODUCTION	. 36
3.2	WHAT IS MENTAL HEALTH?	. 36
3.3	MENTAL HEALTH IN THE CONTEXT OF THIS RESEARCH	. 38
3.4	MENTAL HEALTH AND SCHOOL APPROACHES IN 2021	.41
3.4.1	OVERVIEW	.41
3.4.2	THERAPEUTIC EDUCATION AND HOW IT IS VIEWED	.42
3.5	UNDERSTANDING ADOLESCENCE AND LINKS TO MENTAL HEALTH	.43
3.5.1	ATTACHMENT AND HOW IT CAN IMPACT ADOLESCENCE	.45
3.5.2	NEUROPERSON MODEL	.48
3.6	ADOLESCENCE AND BRAIN DEVELOPMENT	. 51
3.6.1	Societal approaches to mental health: From biomedical to biopsychosocial	.51
3.6.2	MEDICAL ADVANCES AND THEIR CONTRIBUTION TO AN UNDERSTANDING OF MENTAL HEALTH	. 52
3.7	APPLYING BRAIN DEVELOPMENT RESEARCH TO THIS THESIS	. 53
3.7.1	Adolescence and social cognition	. 53
3.7.2	Adolescence and self-concept	. 54
3.7.3	Social media and the adolescent mental health debate	. 56
3.8	CHAPTER SUMMARY	. 57
СНАР	TER 4: THE IMPACT OF AGENCY, POWER, IDENTITY AND CAPITAL FOR YOUNG PEOPLE	
	VG RESPONSIBILITY	.59
<u></u>		
		-0
4.1		
4.2	POWER, YOUNG PEOPLE AND SCHOOLS	
4.3	HOW POWER IMPACTS RELATIONSHIPS	
4.4	AGENCY AND HOW IT IS IMPACTED BY POWER RELATIONSHIPS	
4.5	SENSE OF SELF	
4.6	THE INFLUENCE OF POWER ON EPISTEMIC AGENCY IN SCHOOLS	
4.7	EPISTEMIC INJUSTICE	
4.8	OVERCOMING EPISTEMIC INJUSTICE	
4.9	IDENTITY, WHAT IT IS AND HOW IT IMPACTS UPON YOUNG PEOPLE AND THEIR ABILITY TO TAKE RESPONSIBIL 74	ITY
4.10	SOCIAL AND LEARNER IDENTITY IN EDUCATION	. 76
4.10.	1 VIEWING YOUNG PEOPLE IN SCHOOLS THROUGH A LENS OF HABITUS AND CULTURE CAPITAL	.77
4.10.	2 НАВІТUS	. 78
4.10.	3 CULTURE CAPITAL	. 80
4.11	CHAPTER SUMMARY	. 81
СНАР	TER 5: RESEARCH METHODOLOGY	. 82
5.1	INTRODUCTION	07
5.1 5.2	Social constructivism	-
	CRITICAL COMMUNICATIVE METHODOLOGY (CCM)	
5.3		
5.4		
5.4.1		
5.4.2	90	
5.4.3	THE DEVELOPMENT OF ITERATIVE ACTION RESEARCH	
5.4.4	WHY I CHOSE YOUTH PARTICIPATIVE ACTION RESEARCH	. 95
5.5	THE BARRIERS TO YOUNG PEOPLE'S PARTICIPATION	
5.6	YOUNG PEOPLE'S RIGHTS	100

5.7	KNOWLEDGE – ACCESSING IT, CREATING IT OR CO-PRODUCING IT?	
5.8	SCHOOL CONTEXT	
5.8.1	MY POSITIONALITY WITHIN THE RESEARCH	
5.8.2	HOW IS ROBUST THINKING ENSURED WITHIN THE PROCESS OF ACTION RESEARCH?	
5.8.3		
5.9	REFLEXIVE APPROACHES	
5.10	HOW THE REFLECTIVE AND REFLEXIVE APPROACHES HELPED ME NAVIGATE THE MESS	
5.11	RESEARCH ANALYSIS	
5.12	CHAPTER SUMMARY	
PART	2. FROM THEORY TO ACTION	
<u></u>		
CUAR		110
	PTER 6: RESEARCH METHODS	
6.1	CHAPTER INTRODUCTION	
6.2	THE RECRUITMENT OF YOUNG PEOPLE FOR THE RESEARCH	
6.2.1	PARTICIPANT YOUNG RESEARCH TEAM AND RECRUITMENT	
6.3	YOUNG PEOPLE AS COLLABORATORS IN RESEARCH DESIGN	
6.4	THE INFLUENCE OF VIVENCIA, PRAXIS AND CONSCIENTISATION ON THE METHODS	
6.5	FROM DATA COLLECTION TO ANALYSIS	
6.5.1	DETAILED RESEARCH ANALYSIS	
6.6	TRIANGULATION OR CRYSTALLISATION?	
6.7	WHO IS THE ANALYST?	
6.8	SUMMARY	
СПУЦ	PTER 7: ETHICAL CONSIDERATIONS	140
7.1	CHAPTER INTRODUCTION	
7.2	SEEKING ETHICAL APPROVAL	
7.2.1		
7.2.2		
	THE SIGNIFICANCE OF ETHICAL CONSIDERATIONS TO THIS STUDY	
7.4	ETHICAL CONSIDERATION OF WORKING IN A SOCIAL CONSTRUCTIVIST PARADIGM	
7.5	THE IMPORTANCE OF THE RESEARCH RELATIONSHIPS	
7.5.1	- ,	
7.5.2		
7.5.3		
7.6	ETHICAL VOICE IN THIS RESEARCH	
7.6.1		
7.6.2		-
7.7		
RESEA	USING A PATIENT AND PUBLIC INVOLVEMENT (PPI) APPROACH TO ENABLE YOUNG PEOPLE	O SHAPE THE
	USING A PATIENT AND PUBLIC INVOLVEMENT (PPI) APPROACH TO ENABLE YOUNG PEOPLE T RCH	
7.8		
	RCH RELATIONAL ETHICS	
7.8	RCH Relational ethics Fundamental principles of relational ethics	
7.8 7.8.1	RCH Relational ethics Fundamental principles of relational ethics Time and contextual experience	
7.8 7.8.1 7.8.2	RCH RELATIONAL ETHICS FUNDAMENTAL PRINCIPLES OF RELATIONAL ETHICS TIME AND CONTEXTUAL EXPERIENCE THE NATURE OF DIALOGUE WITHIN RELATIONAL ETHICS	
7.8 7.8.1 7.8.2 7.8.3	RCH RELATIONAL ETHICS FUNDAMENTAL PRINCIPLES OF RELATIONAL ETHICS TIME AND CONTEXTUAL EXPERIENCE THE NATURE OF DIALOGUE WITHIN RELATIONAL ETHICS	
7.8 7.8.1 7.8.2 7.8.3 7.9	RCH RELATIONAL ETHICS FUNDAMENTAL PRINCIPLES OF RELATIONAL ETHICS TIME AND CONTEXTUAL EXPERIENCE THE NATURE OF DIALOGUE WITHIN RELATIONAL ETHICS	

<u>CHAI</u>	PTER 8: FINDINGS	<u> 164</u>
8.1	CHAPTER INTRODUCTION	
8.1.1		-
8.2	FINDINGS	
	ESSIVE SCHOOL STRUCTURES ENCOURAGE TEACHERISM, WHICH HAS A DETRIMENTAL EFFECT ON	
	/YOUNG PERSON'S RELATIONSHIPS	
017111		
C 11 A F		474
	PTER 9: DISCUSSION, LIMITATIONS AND CHALLENGES	1/1
9.1		
9.2	THE PROCESS	
9.2.1		
9.2.2		
9.2.3		
9.2.4		
9.2.5		
9.2.6		
9.3	METHODOLOGICAL INNOVATION	
9.3.1	INTRODUCTION	198
9.3.2	DIALOGUE AND CRITICAL COMMUNICATIVE METHODOLOGY	199
9.3.3	HABERMAS' THEORY OF COMMUNICATIVE ACTION	200
9.3.4	IMPACT OF A REFLEXIVE AND REFLECTIVE APPROACH WITH THE YRT	202
9.3.5	RESEARCH IMPACT ON SCHOOL	204
9.3.6	YOUNG PEOPLE RESEARCHING WITH YOUNG PEOPLE	205
9.4	A YPDAR FRAMEWORK FOR SCHOOLS	209
9.4.1	YPDAR ANALYSIS	211
9.4.2	RELATIONAL ETHICS IN YPDAR	215
9.5	APPLYING YPDAR FOR USE IN SCHOOLS	218
9.5.1	A MODEL FOR YOUNG PEOPLE'S PARTICIPATION IN SCHOOL RESEARCH	220
9.6	LIMITATIONS AND CHALLENGES	222
9.6.1	LIMITATIONS	222
9.6.2	CHALLENGES	223
CHAR	PTER 10: FUTURE RESEARCH, RECOMMENDATIONS AND CONCLUSION	225
10.1	FUTURE RESEARCH, RECOMMENDATIONS AND IMPACT	
	Conclusion	
10.3		221
<u>EPILC</u>	DGUE	229
BIBLI	OGRAPHY	231

Table of Figures

FIGURE 1: CONCEPTUALISATION AND CATEGORIZATION OF SCHOOL CLIMATE (WANG & DEGOL, 2016)	29
FIGURE 2: BRONFENBRENNER NESTED MODEL OF ECOLOGICAL SYSTEMS (GUY-EVANS, 2020)	32
FIGURE 3: SYSTEMS VIEW OF SCHOOL CLIMATE	
FIGURE 4: TRANSDOMAIN MODEL OF HEALTH (MANWELL ET AL., 2015, FIGURE 4.)	
FIGURE 5: THREE PART SOCIO-EMOTIONAL MENTAL SKILLS	
FIGURE 6: DOMAINS OF SOCIO-EMOTIONAL SKILLS AND THEIR CONTEXTS	50
FIGURE 7: HOW POWER INFLUENCES YOUNG PEOPLE'S AGENCY, IDENTITY AND CAPITAL	60
FIGURE 8: CONCENTRIC METHODOLOGY MODEL	83
FIGURE 9: ACTION RESEARCH CYCLE	92
FIGURE 10: STRESS WHEEL ERROR! BOOKMARK NOT	
FIGURE 11: LADDER OF CHILDREN'S PARTICIPATION (HART, 1992)	
FIGURE 12: IN EMPOWERING CHILDREN AND YOUNG PEOPLE: PROMOTING INVOLVEMENT IN DECISION-MAKING (TRESEDE	, ,
FIGURE 13: COLLABORATIVE SELF-REFLECTION TOOL	108
FIGURE 14: RESEARCHER SUMMATIVE REFLECTIONS	110
FIGURE 15: THE UCHICAGO FRAMEWORK. TAKEN FROM 'A FRAMEWORK OF OUTCOMES 2.0' (MCNEIL, ET AL., 2019)	111
FIGURE 16: METHODOLOGICAL TOOL	120
FIGURE 17: SCHOOL CLIMATE HEXAGO SUMMARY EXERCISE	126
FIGURE 18: SCHOOL CLIMATE AND MENTAL HEALTH TOOL	127
FIGURE 19: FRAMEWORK DEVELOPED BY THE RESEARCHER TAKEN FROM YRT HEXAGON EXERCISE	128
FIGURE 20: YRT FRAMEWORK MODEL 1 ADAPTED FROM RESEARCHER MODEL FIG. 19	130
FIGURE 21: YRT FRAMEWORK MODEL 2 ADAPTED FROM RESEARCHER MODEL FIG 19	131
FIGURE 22: FINAL QUIRKOS ANALYSIS DIAGRAM	133
FIGURE 23: QUALITATIVE ANALYSIS HELP SHEET	134
FIGURE 24: YOUNG RESEARCH TEAM PARTICIPATIVE ACTION RESEARCH CYCLE	135
FIGURE 25: TEACHERISM DEFINITION	146
FIGURE 26: INABILITY TO TAKE RESPONSIBILITY: A DEFICIT-BASED TOOL	176
FIGURE 27: TAKING RESPONSIBILITY: AN ASSET-BASED TOOL	
FIGURE 28: A CONCEPTUAL FRAMEWORK FOR EMOTIONAL WELLBEING IN SCHOOLS	192
FIGURE 29: SCHOOL MENTAL HEALTH MODEL	196
FIGURE 30: DIALOGUE DESCRIPTION	200
FIGURE 31: YPDAR FRAMEWORK FOR SCHOOLS	210
FIGURE 32: YPDAR ANALYSIS FRAMEWORK	215
FIGURE 33: A RELATIONAL ETHICS FRAMEWORK FOR YPDAR IN SCHOOLS	218
FIGURE 34: YOUNG PERSON LEVEL OF PARTICIPATION IN SCHOOL RESEARCH (AFTER TRESEDER, 1997)	
FIGURE 35: YOUNG PEOPLE'S PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL OF PARTICIPATION IN SCHOOL MODEL (INCORPORATING ASPECTS) OF LUNDY'S MODEL (INCO	PATION
(Lundy et al., 2020))	221

Table of tables

TABLE 1: SHIFTS, RUPTURES AND STATE (ADAPTED FROM BELL 2021)	14
TABLE 2: ADVANTAGES AND DISADVANTAGES OF YRT SELECTED METHODS	121

Acknowledgments

I thank William Howard School for allowing me to conduct my research there. In particular, my heartfelt thanks go to my inspirational young research team and the participants, without whom this work would not have been possible. Thanks also go to the school staff, in particular Tracy Swan and Cecilia Sawdon, who have been invaluable support over the past two years.

I want to thank my friends and family for their support and patience: Pauline, my wife; my four children Tom, Dan, Harry and Lewis; as well as my late parents, whom I have to thank for the values they instilled in me; I know they would be incredibly proud of me. In addition, a mention needs to go to Tino who has been a loyal and constant companion during this voyage of discovery. Also, gratitude to the many friends I have leaned on over the past three years.

I want to thank my supervisory team for their patience and hard work in cajoling, leading and prompting me in the most positive ways imaginable. Whilst late to the party, Tracy Hayes offered insight, wisdom and unwavering support. Karen Morris has seen me through from my interview at the start to finish with steadfast faith in my ability to get it done. Special thanks to Kaz Stuart, without whom this PhD would not have happened. She was my inspiration; persuading me to apply and supporting me along every step of the road. Thanks to all three of you for teaching an 'old dog some new tricks', and thanks for believing in me.

A big thank you to Marcus Byron who created most of my illustrations and whose cartoons helped illuminate my ideas and theories.

Finally, let this thesis inspire all those young people who like me when at school, were told they were 'no good' and would not succeed.

"They will always tell you that you can't do what you want to do, but you can do what you want to do. You just have to believe in yourself. The system is to bring you down, but you can rise up."

Bob Marley

Declaration of Authorship

I, David Simon McPartlan declare that this thesis is my own work and has not been submitted in substantially the same form for the award of a higher degree elsewhere.

Glossary:

- Academy: a state school funded directly by the government and run by an academy trust.
- Academy trust: a not-for-profit company which runs and is responsible for the performance of a single or group of academies.
- Adolescence: a developmental stage starting with puberty and ending when an individual maintains a stable, independent position in society.
- **Emotional wellbeing:** a positive mental state where an individual's basic needs are met to the point where they have a sense of purpose and are able to achieve personal goals and participate within the school community.
- **Epistemic agency:** the ability and motivation to refine and alter one's belief-forming methods and practices.
- **Mental health:** an individual's psychological condition which, like physical health, is something that all young people experience and can impact upon how they feel and behave.
- Mental illness /Mental health disorder: mental illnesses are medically-classified signs and symptoms that are often complex and multi-faceted. They require medical diagnosis and treatment to enable young people to live productively both in school and in the wider community.
- **Participants**: volunteer young people, aged 12-15 years old from the school's free school meal cohort who met with the young research team to explore the research themes.
- **Pupil premium cohort:** a group of young people, designated as disadvantaged and qualifying for extra school-funding to improve their educational outcomes.
- **School character:** the interaction between cultural, physical, emotional, relational and social aspects of school environment: how the school feels.
- School mental health: an aspirational concept with a focus on ensuring that all young people are in a position to achieve to their potential. It encompasses education, and support for young people and their families, as well as signposting to external services where appropriate.
- Social constructivism: the view knowledge is created through social interactions.
- **Teachersism:** the dysconscious act that enables adults to have power over young people in schools.

- Young research team: co-researcher volunteers taken from the school's 16–18-yearolds.

Abbreviations:

- Action research (AR)
- Child and adolescent mental health services (CAMHS)
- Chief executive officer (CEO)
- Collaborative self-reflection tool (CSRT)
- Critical communicative methodology (CCM)
- Coronavirus (COVID)
- Ecological systems theory (EST)
- Grant maintained schools (GMS)
- Local education authority (LEA)
- Local management of schools (LMS)
- Magnetic resonance imaging (MRI)
- National Health Service (NHS)
- Office for Standards in Education (OFSTED)
- Participative action research (PAR)
- Public and patient initiative (PPI)
- Social and emotional attitudes to learning (SEAL)
- Stage-environment fit theory (ESFT)
- Special educational needs and disabilities (SEND)
- Special interest group (SIG)
- Systems view of school climate (SVSC)
- Teaching and learning (T&L)
- Transformative student voice (TSV)
- United Kingdom (UK)
- United Nations Convention on the Rights of the Child (UNCRC)
- United States of America (USA)
- University of Chicago Consortium of School Research (CCSR)
- World Health Organisation (WHO)
- Youth participative action research (YPAR)
- Youth participative dialogic action research (YPDAR)
- Young research team (YRT)

Prologue

As an individual who had a 'chequered' academic career, this PhD has been an illuminating journey of self-discovery. From a very early age, school for me was a social space where academia was somewhat of an inconvenience. My earliest school experiences were notable for the fact that teachers were keen to tell me that I was 'lazy' and 'not academic'. This linked to the fact that corporal punishment was the preferred way of control in schools and this resulted in my negative mindset around education. However, I always enjoyed the social side of school and, despite the physical punishments I received, little deterred me from my ambition to become a teacher. Scraping through O-levels, A-levels and a Bachelor of Education degree to gain entry to the next step on this road, resulted in me starting a long career as a teacher. My initial path as a geography teacher soon gave way to responsibility posts in the pastoral and welfare side of teaching; a Deputy head of year, a Head of year, a Director of learning and finally the Pastoral assistant headteacher of a large secondary comprehensive school. On reflection, this should be no surprise to me as my values have always been about caring for others and looking after those who may find life a challenge. My values are directly linked to inclusivity, integrity, honesty and trust; building relationships is key to who I am and how I behave. My responsibilities in my final years as a teacher reflected this, as I led a large pastoral team in school as we built systems that linked with outside agencies such as social care, police and youth offending, to do our best to support young people and their families.

A meeting in school in 2017 between myself, a student who was returning from a short-term school exclusion, and a parent is where this PhD journey started. As I explored the behaviour of the young person with the parent (an ex-student who had also been excluded) and we discussed how the young person was experiencing mental health issues, I realised that the behaviours of the parent. when at school, and the child were of a similar nature. Whilst as a pastoral team we were acutely aware of the increase in mental health-related school issues, it was not until I was sitting in this meeting that I realised that we were possibly dealing with intergenerational mental health issues and an alternative approach, by the school, was required. My final major contribution to the school thereafter was the design and implementation of a whole school mental health strategy. The strategy aimed to enable young people to take responsibility for their mental health within a supportive school structure. This PhD is the next step in this journey, as the main focus of this research is to explore how effective the mental health strategy has been. As I will go on to explain below, the research has been less than straightforward as it has developed beyond its initial focus. My desire to include the thoughts of young people in this research has seen it develop beyond just investigating the school's mental health strategy, to exploring how adults can best effectively consult with young people in schools. I believe that if we want to encourage young people to improve

their own lives, we need to partner with them so that they can explore the issues that impact them, and so support them to find solutions.

Another fundamental consideration related to who would be the participants in the study. My research aims were clear but I also wanted to ensure that whichever young people were involved in the research also had an opportunity to benefit from it. I chose to focus on the group of young people who are designated by the Pepartment for Education as 'pupil premium' (The Pepartment for Education, 2022). Schools receive funding for young people who are placed in this category for a number of reasons, but mostly because their parents qualify for financial support. My thought process was that, as many of these young people are from an economically-disadvantaged background, they may benefit from access to the project. Forthwith, I will refer to them as participants in this thesis.

The school in which this research took place is a secondary academy situated in the small market town of Brampton in the north east of Cumbria. It is a larger-than-average comprehensive school with approximately 1400 students from the age of 11 to 18 years. The school has a largely rural catchment that extends north to the Scottish border, east into Northumberland, and takes from the Eden valley and the city of Carlisle. William Howard ¹ was formed in 1980 with the amalgamation of the town's grammar school and the secondary modern. It became an academy in 2012 and is now the academy lead for the Cumbria Education Trust.

The school is predominantly made up of white British heritage students with a minority coming from another ethnic background. Whilst the proportion of students who are designated as disadvantaged (pupil premium) is below average, the number of students with special educational needs and disabilities (SEND) is above average. The reason for this is that the school has a special needs resource base that caters for up to nine young people. I would describe the school as a true comprehensive as it takes from some of the poorest wards in the country, as well as extremely affluent parts.

During the most recent Office in Standards for Education (OFSTED) inspection, in May 2019, the school was judged to be 'Good'. It was described as having a "positive, welcoming culture" and as being "highly inclusive", where "Pupils from a wide geographic area and different backgrounds come together well". The inspection report also explained that "Pupils feel safe and well looked after", and

¹ Ethics approval allows for the school to be named, as principles of confidentiality rather than anonymity were accepted by the school.

that they are "well-mannered and behave well". Behaviour in the school was described as "good", as "exclusions are decreasing for most pupils as behaviour improves". Curriculum design was identified as meeting "the needs of pupils and the local labour market" and was "enhanced by a wide range of extra-curricular activities and leadership opportunities" (Ofsted, 2019). Whilst never perfect documents, this, in my experience, would suggest it was a fair summary of the nature of a school; one which has worked hard to become an inclusive environment for the local young people in its catchment area. Whilst school context is important, wider historical concerns in relation to mental health also need to be highlighted.

This research started in October 2019 as I embarked on my PhD. The empirical research was due to commence in the autumn of 2020, however the onset of the Coronavirus (COVID) pandemic held this work up as schools were initially closed and then re-opened with restrictions. After a failed attempt at starting the research online (young people were reluctant to engage with this approach), we started meetings and data collection in March of 2021, and this continued until the end of the school year. The findings of the research were presented to the school in the autumn of 2021.

The thesis is divided into 3 parts, each of which is further split into chapters. Each part of the thesis is proceeded by a brief synopsis outlining its purpose and how the chapters contribute towards it.

Part 1 Setting the scene

The first part of this thesis comprises five chapters and grounds the research by giving a context to it. Part one starts with a short introductory chapter developing the research rationale and is followed by chapter two which frames the research context in relation to a history of education, school purpose, and school character. Chapter three takes a detailed look at adolescence and mental health. This leads to chapter four which explores the role power plays in young people's school life and how this has the potential to impact their agency, identity and accumulation of capital. I conclude part one with chapter five, a methodology chapter detailing the philosophy of the research.

Chapter 1: Introduction and research rationale

1.1 Young people their mental health and schools

Child and adolescent mental health is one of the most significant issues of the 21st century, so much so that the government have appointed a Mental Health Ambassador to champion this cause (Prime Ministers Office, 2021). Between 2004 and 2017 there have been increasing numbers of children being referred by schools to mental health specialists, prescribed anti-depressants, presented in accidents and emergency due to mental health concerns, and reporting lower general wellbeing (NHS Digital, 2018). In 2017, it was estimated that 850,000 young people had diagnosable mental health disorders (DfE/DoH, 2017), with a further rise since. Between April and June of 2021, there were 190,271 referrals to young people's mental health services, an increase of 134% on the same period the previous year (Local Government Association, 2022). With the country emerging from the pandemic, concerns are growing, and latest figures show increases in probable mental health issues for children and young people between 2017 and 2020 from 10.8% to 16% (Newlove-Delgado et al., 2021). There is now an expectation that schools become involved in supporting young people's mental health needs (DfE/DoH, 2017).

The mental health schools' agenda is gaining traction due to the recognition of the increasing numbers of young people who are experiencing mental health issues, such as those demonstrated above. The government have initiated mental health support for young people in schools through the introduction of a named staff member with responsibility for mental health in schools, and the introduction of the NHS Mental Health Support Team Pilot in schools (Department of Health and Social Care & Department for Education, 2017). After a successful pilot, this has since been developed further (NHS England, 2021). Unfortunately, the government are advocating a deficit clinical mental health model (Glazzard and Stones, 2021) identifying young people with mental health needs as the problem; the solution is the use of mental health support teams providing interventions for them. Whilst this is a major push by government, they also advocate other school initiatives. These include support for young people and their families; monitoring of young people's mental health; taking measures to reduce stigma around mental health; and a curriculum promoting the understanding of mental health. The focus of this work needs coordination and is most

effectively achieved through whole school strategies (Bostwick and Glazzard, 2018). Schools' roles within the mental health structure are still ill-defined. However, it is accepted that school staff are not trained mental health practitioners and, as such, cannot treat mental illness or mental health disorders. However, with leadership, guidance and support they can educate, monitor, signpost, and support both young people and their families, so young people have every opportunity to fulfil their potential both in school and beyond (Bostwick and Glazzard, 2018).

In the prologue, I described how the foundation of the whole school mental health strategy grew from my experiences as a teacher where I saw the impact of the school on young people. I have listened to young people complain about schools piling on the pressure, using aspirational targets to improve performance. However, there is evidence that by using performance rather than mastery goal structure, schools are more likely to increase behaviour issues as well as raise levels of depression (Wang, 2009; Olivier et al., 2022). During my career as a teacher, there were many times when I came across young people who felt that they did not belong; whilst the school tried to develop an inclusive and welcoming environment, there were a number who found it difficult to fit in. They were often the ones in trouble, the ones who did not want to come to school or those who had ill-defined health and wellbeing issues. The research into belonging and emotional health supports this view as those with lower feelings of belonging report lower emotional health (Freeman et al., 2009). To compound the issue further, this problem is accentuated for those who come from a more disadvantaged background. In her book Miseducation, Reay (2017) makes powerful arguments that the present education structures are based on wealth that reinforces rather than addresses a hierarchical class structure. As set out in section 2.2.4 the values at play are neoliberal ones and related to the market and all the problems that this creates; ultimately, as we have seen, those who are most in need are likely to be left behind. Reay tells us that the system works against those from the working classes; they are at a disadvantage from day one at school and it becomes increasingly more difficult to catch up as their education journey continues. The reoccurring theme is that some young people do not fit into our school environments, and often poor staff relationships are at the core. Positive interpersonal relationships between groups of young people, as well as between young people and teachers, correlate with higher life satisfaction (Suldo et al., 2013). They are often more able to cope in school as well as having greater optimism about school as a whole (Ruus et al., 2007).

There are clear links between what happens in schools and how young people feel about themselves (Ruus et al., 2007). The connections between these are complex and sometimes confusing. My values are closely linked to my work in schools leading pastoral teams. This is one of the reasons I believe the research should have young people at the centre and why I designed it in a way that enabled them to be young researchers.

1.2 The development of a whole school mental health strategy

Having spent over 35 years teaching, and most of that in the area of pastoral work, my last substantial piece of work in schools was to design and implement a whole school strategy for mental health. This strategy was based upon educating both young people and staff about mental health: what it was, how one could look after it and where to go for support. The basis of the strategy was to not only provide support but to encourage young people to self-refer to staff in school. Leaving teaching two years into the strategy implementation, I started my PhD intent on working with young people in school to research how effective this strategy was.

1.3 Young people as researchers

My aim from the beginning of the research was to ensure it was a participatory project. I was aware there were differing levels of young people's participation (Mercer, 2002; Cook-Sather, 2020) in educational research. This was something that also gave me the opportunity to explore how my teaching career may impact my work as a researcher (see section 5.8.1). As I allude to throughout this thesis, my past school role would often impact how I framed and reflected upon the research. My values were often challenged, compromises were called for and I was, on occasion, required to make uncomfortable decisions. This is one of the reasons I was drawn towards participatory action research (PAR) as I wanted to help young people by improving their access to school support. The partnerships I developed with them during this research led me to the conclusion that, with guidance, they could become change agents in the school. There was also a part of me that felt as if I was making amends for my past in schools by involving them as co-researchers in this research.

Being aware that young people are often used as objects of research within a school setting (Erickson & Christman, 1996; Wöhrer & Höcher, 2012), my intention was to ensure that this

research was conducted from the young person's perspective; something gaining credence although not commonly adopted (Noffke & Somekh, 2008; Brady, 2017). Accordingly, prior to initiating the data collection phase for my thesis, I visited the school and conducted a patient and public initiative (PPI) exercise, something I explore in greater detail in section 7.7. The purpose of the PPI meetings was to gather the thoughts of young people so that they could help shape the research. One of their recommendations, which I adopted, was for the data collection to be completed by sixth-formers, whom they believed the younger participants would talk to more readily. In this respect, I agree with Moules & Kirwan (2005) when they say young people are more likely to open up to their peers than they are to adults. I intended to get a critical view of the whole school mental health strategy from a young person's perspective. If they were more comfortable talking to someone closer to their own age, they were more likely to give insightful and authentic answers.

1.3.1 Pupil premium cohort as the participants

Another fundamental decision taken at this point was my invitation to young people from the pupil premium cohort to collaborate as participants in the research. Coming from the most economically disadvantaged section of society, my aim was to give volunteers an opportunity to contribute to and benefit from this unique school research. This decision was founded on the basis that research tells us poverty results in poorer outcomes both in health (Marmot et al., 2020) and education (Hirsch, 2007; Mowat, 2015). Hirsch suggests young people in a similar demographic to the participants in my study feel a lack of control over their learning, and therefore become reluctant learners. Whilst this is complex, deep-set and often linked to factors outside of school, education still plays a key role in this area. Furthermore, young people from poorer backgrounds are more likely to lack confidence in school. Mowat (2015) suggests they can feel anxious, sad, frustrated, and angry about the school experience, something often compounded by discriminatory behaviour by teachers. What both Mowat and Hirsch's research also suggests is that work with these young people is more effective when they feel more involved in the decision-making process about their own futures.

Hirsch (2007) believes that young people's engagement in school is linked to confidence and school relationships. Mowat (2015), however, suggests that schools need to value young people for who they are and treat them with unconditional positive regard (Rogers, 1957). In my teaching career, I would observe young people from this demographic arrive in school

keen and eager to impress, but gradually over time withdraw into their shells. Their performance nose-dived and many became either invisible or disaffected. Whilst many different approaches were attempted, solutions were rarely identified and we were often left scratching our heads. I, therefore, felt I had an opportunity to support young people from this cohort and explore whether they could benefit from being involved in a research project such as this.

1.3.2 6th formers as a Young Research Team

After the PPI exercise, which suggested 6th formers work with me as co-researchers, the decision to ask for 6th-form volunteers to make up a young research team (YRT) was also founded on a number of other principles. Firstly, as the oldest students in the school they are often looked up to by other year groups, and the school promotes them as being aspirational role models for the younger years. Secondly, they have a small number of free periods allowing them to withdraw participants from lessons and not encroach on their lesson or social time. My experience as a teacher told me in order to engage young people as fully as possible it was important not to impinge on social time, whilst I was also aware school staff were less likely to complain if the YRT were not missing lessons. An added bonus was participants would have the opportunity to miss the occasional lesson. As a rural school, the majority of young people are bused to and from school, therefore after-school meetings are difficult for many to attend. My final, and possibly most important, reason for identifying this cohort as young researchers was their age. Many of them have been at the school for either five or six years and it is this experience of the institution (its workings, its systems, and idiosyncrasies) I wanted to tap into; the research needed to be viewed through the eyes of these experts and social actors (Cowie & Khoo, 2017).

1.4 Research aims

I therefore decided upon the following research aims:

- To explore the efficacy of the whole school mental health strategy.
- To ensure the research is informed by young people in the school.
- To develop ways of collaborating with young people, which ensures findings are not influenced by the researcher's previous position in the school.

1.5 Research, knowledges and voice

This was a participative action research project intent on involving young people as fully as possible, being practice-based research. It was designed with sixth-form volunteers as co-researchers, and a YRT working alongside the researcher and collecting qualitative data from a younger cohort of students (participants). This was therefore a youth participative action research (YPAR) project.

The outcomes of the research were twofold. The findings about the school's mental health strategy focussed on three major areas; relationships with staff, the impact of working alongside the YRT and the school curriculum. The second major outcome was related to the methodology that I used. Learning from critical communicative methodology, I synthesised this with YPAR to develop a process with dialogue at its core: something I call youth participative dialogic action research (YPDAR).

The "knowledges" (Martinez-Vargas, 2022) that I draw upon within this thesis are many. I differentiate between the singular term knowledge and instead use the plural term knowledges as they are relational, cultural and specific to individuals from different backgrounds. My experience as a teacher and school leader has given me an educational perspective on the work. In addition to this I became a researcher, a process enabling me to think and reflect differently, giving me a complimentary view to that of a teacher. The combination of these two perspectives enables a noteworthy, alternative and unique view of the research area. The YRT have used their lived experience, as have the younger participants, to contribute to the generation of knowledges. In addition to this, the YRT had numerous conversations with school staff whose indirect input cannot be discounted. My writing, therefore, takes on many forms, as it is generally split between the first and third person. There are also occasions where I will highlight the voices by using different texts as follows:

My reflective voice is used when extracting from my reflective diary

- Young people's voice is used when I draw from their verbal or written reflections to me

Chapter 2: Research context

2.1 Introduction

This chapter explores how education policy since the end of World War Two has seen distinct shifts impacting the mental health of young people in schools. The policy direction, in particular over the past 40 years, has resulted in the marketisation of schools that, in turn, has seen a transformation in both school values and subsequently how they are run. This leads me to briefly touch upon school purpose, as I use this as a lens to explore how this has impacted young people's experience of school. I draw on school climate and culture literature to develop my understanding of what I call school character. The final section investigates how ecological systems theory (EST) relates to this work. Of particular interest is how their application can help understanding of the relationship between young people, the school environment and mental health.

2.2 A history of education policy and how it has impacted young people's mental health

In this section, I provide the context for developing a whole school mental health strategy within education in the United Kingdom (UK). I will explore the macro-scale historical educational policy perspective in the UK since 1944.

As is common in the study of history, we use the past to understand what is happening now; this is undoubtedly true in UK education. I, therefore, begin with a brief history of education policy, before concentrating on a more detailed look at recent influences. From the 1960s onwards, the alternating Labour/Conservative/Labour/coalition and Conservative governments set a path that has framed mental health within education settings. The following section will demonstrate why I believe the rise in the challenges of young people's mental health is at least partially a result of the new capitalist age of education. This is characterised by a situation whereby schools are encouraged to compete against each other and are judged objectively using exam results as the overriding criteria. Thus, success in schools is about how well someone achieves and not how well they feel.

2.2.1 Post-war welfare state

The 1944 Education Act was conceived during the war by the then President of the Board of Education (to be Secretary of State for Education), R.A. Butler. There was a feeling amongst the country's population that Britain needed to be a better post-war place, and secondary education for all was required (Barber, 2014). The new Act did not disappoint as the main thrust was the delivery of free maintained education for all, with a tripartite secondary system that saw grammar schools alongside technical and secondary schools. The 11 plus attainment test, with elements of intelligence testing incorporated, was introduced to allocate young people grammar and other schools (Ball, 2021). Whilst the intention was to have technical schools for the cohort below those selected for grammar, many local authorities chose not to open them. With some areas seeing a pass rate of as little as 12%, the majority of young people went to secondary modern schools. This new system aimed for universal and free education, not based on class or wealth but on merit. However, what transpired were different types of schools for different kinds of students, and the idea of parity of esteem was never actually achieved (Ball, 2021). The following 20 years saw the growth of the new free and universal maintained sector of schools. This saw a rise in pupil numbers from 5.5 million in 1947 to 9.1 million in 1967; the number of teachers doubled during the same period. Whilst the school leaving age did rise to 15 years of age in 1948, it took until 1972 for this to go up to 16 years old, the age at which it stands today. This meant that of those pupils who did not pass their 11 plus exam and gain entry to grammar schools, 80% went to secondary modern schools (Barber, 2014), often leaving school with little or no certification (Ball, 2021). Even as late as the 1960s, only one in eight young people in secondary modern schools recorded Olevel passes. This, and the introduction of the Certificate of Secondary Education (CSE), designed for less academic young people, reinforced the two-tier system of schooling in the UK (Ball, 2021). However, for the first time in the country's history young people were guaranteed a secondary education, eventually up to 16 years of age, something previously reserved only for the wealthy in society. It thus represented an attempt by a Labour government to bring about class equality through universal education.

Whilst the 1944 Education Act was introduced by Churchill's coalition government, the 1945 Labour government implemented the post-war agenda following Keynesian social welfare principles (Ellison, 1996). The Labour manifesto was about welfare reform and gained support from a population who had endured years of austerity during World War Two (Tomlinson, 1998). Instead, there was a promise of employment for all on the back of the 'social democratisation of the political system' (Jessop, 1980), including the nationalisation

of coal (the National Coal Board) and iron and steel (The Iron and Steel Corporation of Great Britain). 1948 also saw the formation of the NHS, providing free healthcare for all; education was, therefore part of this welfare revolution (Stoye, 2018). These welfare reforms were designed to ensure all society had jobs and access to free healthcare. The aim was to raise everyone's living standards and break away from the class-driven society of the past. This was possibly something that, as I discuss below, was not quite as straightforward as assumed.

2.2.2 From a post-war vision of welfare to the beginnings of competition

The Wilson Labour governments of the 1960s continued the socialist agenda within the education sector with the introduction of Circular 10/65 (DES, 1965), which declared the government's objective to end selective education and introduce comprehensive schools. This was non-statutory guidance and, as such, local education authorities were only asked to submit plans for comprehensive schooling in their area; it was not made a matter of law (Ball, 2021). However, whilst there was an intention to get rid of grammar schools, there was no clear vision of what comprehensive education should look like. This, therefore, resulted in reproducing a version of grammar school education within comprehensive schools (Ball, 1981; Riseborough, 1981; Reynolds & Sullivan, 1987).

Other issues were also at play, as the Black Papers (Ball, 2021) became influential within educational policy from 1969 to 1977. They were a series of publications written by rightwing commentators criticising comprehensive and progressive education, calling for a return to traditional educational values. The Black Papers attacked on three fronts. Firstly, they believed what was going on in schools was responsible for both the decline in academic standards and a decline in Britain itself; many commentators blamed the education system for the recession at the time (Ball, 2006). Not only was there no clear evidence for the academic decline (HMSO, 1975), but more and more youngsters were leaving school with qualifications than ever before (Wright, 1977). Secondly, teachers themselves were blamed for being politically motivated, as captured by Thornbury (1978, pp. 136-7) when describing English teachers who "indoctrinated themselves and their classes in attitudes critical to the police, local government bureaucracy... and employers". The final accusation suggested behaviour in classrooms and on the streets was declining, and this was also the responsibility of schools; juvenile and, in particular street crime was on the rise. A symbolic crusade or moral enterprise, championed by the Right in the form of the Black Papers, to prevent

disaster by returning to traditional values was undertaken. This was supported by teacher bashing newspapers such as the *Daily Mail, The Sun* and *The Star* and television programmes such as the BBC Horizon's Lesson for the Teacher. The message was clear that schools were out of control and that teachers could not be trusted; parents were encouraged to believe that they were now in a position to wrestle control for themselves (Ball, 2006).

There was also another agenda during this time. Margaret Thatcher, the leader of the Conservative government, was determined to fight socialism. Her championing traditional family values were set against a view of declining morality within UK society. Not only did we have teachers who were revolutionaries (Cox, 1981), but we also had unions perceived as the enemy within. The racial politics surrounding issues of immigration was set against the jingoism and rediscovery of Nation that found voice, and some would say was exploited, by the Falklands War (Ball, 2006). In a newspaper article (The Times, 21 October 1974), Keith Joseph, who was thought to be the architect of Thatcherism, highlighted further the ills of society including the rise in delinquency, hooliganism and teenage pregnancy, as well as gang culture in schools. Society's decline was further promoted daily in the newspapers and on television as schools were portrayed negatively as progressive institutions under the control of left-wing teachers. Through education policy, however, concerned parents were soon to have a voice and choose which school was best for their children (Ball, 2021). This was part of the rising conservative neoliberal ideology that promoted freedom of choice, market forces and quality through competition. These ideals would be seen to drive education policy for at least the following 40 years (Ball, 2021).

The changes in policy between a welfare-oriented education system and a neoliberal one can be seen as a broader reflection of societal change. This was not the first time this had taken place, and was not the last. It has been summarised in table form as follows:

Shift	Rupture	State	Key dates	
1870-1944	1870-1944			
Industrial	Break with resistance to	Modern (or	1902 Education Act	
revolution	state education and	interventionist) state	Established a Board of	
	welfare		Education	
1944-76		I		
Post war	Move to welfare state	Welfare state	1944 Education Act	
growth of	education		Established the principles of	
economy and			universal and free education	
middle classes				
1976-97				
Economic	Break from emerging	Neoliberal state	1980 & 1988 Education Act	
crises	national comprehensive		Included giving and extending	
including mass	systems and end of		choice to parents	
unemployment	professional autonomy		1988 Education Reform Act	
	for teachers		Included the devolution of	
			school budget control from	
			LEAs to schools.	
			1992 Creation of OFSTED	
1997-2022				
Knowledge	End of national systems	Competition state	2000 Learning and Skills Act	
economy, basic	locally administered		Introduced the City	
and hi-skills			Academies Programme	
			2010 Academies Act	
			Rapid expansion in academy	
			numbers	

So far, I have briefly explored the post-war period that incorporated a welfare approach to education policy and looked at how a neoliberal enterprise gradually replaced this during the Thatcher years. Whilst this policy history gives a context of how societal events influenced educational structures, there has been a seismic shift in what is expected from schools since the late 1990s. This has dramatically impacted how young people's mental health in schools is now perceived, something I discuss later in this chapter. The following section explores the development of the competition state and how this has impacted our educational institutions' value base. I intend to then show how this, in turn, had a negative knock-on effect to how schools respond to the wellbeing of young people.

2.2.3 Competition and marketisation of education

Whilst the ideology of the competition state was set during Thatcher's early years in government, the irony is it was the Wilson-Callaghan government of 1974-1979 who largely prepared for the educational direction until the end of the century and possibly beyond (Lowe, 2004). The Labour government were beset with social, political, and economic problems that saw the beginning of the end for the socialist ideologies dominating education post 1945 (Ball, 2021). By the late 1970s and early 1980s, a Conservative government came into power and there was a clear change in direction as Keith Joseph (Secretary of State for Education) was talking about 'market solutions' and 'a paternalistic inspectorate' (Knight, 1990); around the same time performance based appraisal for teachers was also under discussion (Simon, 1991).

In the first years of the Conservative government the work started with the aim of weakening the power of the local education authority (LEA) and strengthening the voice of the parent. This was done by giving parents freedom to choose their child's school, as well as providing guaranteed parent places on governing bodies (Ball, 2021). City technology colleges, the first independent state schools free from LEA control, were introduced in 1986. The 1988 Education Reform Act introduced local management of schools and grant-maintained schools. Schools no longer needed their LEAs but could commission their own services. If these both hastened the journey towards school autonomy then the introduction of OFSTED certainly raised both teacher and school accountability (Gillard, 2018). National and school performance targets were introduced in 1992, and were then used by OFSTED to judge, and sometimes praise but often condemn schools (Leckie & Goldstein, 2016). Schools were

entering an era of performativity which, as I will go on to discuss, had an inevitable impact on young people as exam results became the measure of a school's performance.

Competition was very real, both nationally and locally, as league tables were produced that pitted local schools against each other (Ball, 2021). Schools used their budgets to advertise and market their institution as pupil numbers were all important; school funding largely being reliant upon the numbers enrolled. A school struggling with exam results was likely to suffer on a number of fronts. A poor OFSTED rating would mean that other local schools could attract 'their' pupils; as competitor school's numbers wert up and they became richer, then a 'failing' school's numbers would fall. A fall in numbers meant a reduction in revenue and, in turn, would impact the quality a school could offer; I witnessed local schools finding themselves in freefall. The Conservative government first introduced performance tables in 1992, and encouraged a public discourse around good or bad schools based on performativity. Labour built on this with their agenda around failing schools, subjecting them to a greater number of inspections, interventions and sometimes closure (Ball, 2017). This pattern continued through successive governments as a way of forcing schools to move from local authority control, and has led to the introduction and growth of Academies and Academy chains, now multi-million pound businesses in their own right (Ball, 2017).

2.2.4 The changing school system and its impact on values

My experience in education has demonstrated that the very purpose of schools is at the heart of the changes that have taken place over the past 40 years or so. When, in the early 1980s, I entered the teaching profession I did so to help young people. It was a vocation and, as I progressed through my career, I helped young people prosper through improving their life chances. Assessments are the most obvious example of this, and I took great pride in the results that students achieved. As time moved on, I followed a pastoral route as a Head of Year and then Assistant Headteacher with responsibility for all welfare issues. This involved working with families and young people who often found school challenging. It also meant partnership working with outside agencies including children's services, the police and youth offending services. My priorities had changed and the focus of my vocation became those who were identified as being vulnerable. My values have always been at the heart of my work, be it my past teaching career or now my research. In contrast, and during this same time span, an enterprise culture in education became pervasive as governments built upon the work of previous administrations. The culture of competition had seeped into, and subsumed, the fundamental meaning of what I see schools as being about. As single schools have been subsumed into multi-million-pound academy trusts, then so their values have changed as a matter of necessity. As corporate culture has taken over, schools are now economic institutions needing to compete with market forces in order to survive; their employees are required to tow the party line. From personal experience and discussion with colleagues working across the sector, I have seen good and honest professionals push at both legal and moral boundaries when it comes to assessments. In some instances there was evidence of professionals turning a blind eye to illegal exam practice because they knew there was potential for an unravelling that risked the success, status and standing of the school. I have met curriculum leaders who were advised by LEA advisors to raise their students' moderated marks and so improve grades; the headteacher agreed with the advice and his expectation was that the advice should be followed! In my experience the pressure of accountability through the market has certainly eroded the value base within the profession. Schools need students to perform and achieve prescribed levels of exam results. If results are not at a required standard, schools and their staff come under great pressure and, on occasions, staff may lose their job. As such I agree with Ball (2006, p.11) when he says: "Value replaces values, except where it can be shown that values add value".

As policy pushed schools from the timeworn bureaucratic post-war ways into a proposed exciting entrepreneurial future, a new type of school leader was required. Gone were the traditional academics who managed the schools, and they were soon to be replaced with quality-driven professionals who would market the school, attract greater pupil numbers, improve results, and hold their staff to account. Thatcher (1995) talked about 'dependency culture' and the problems created by the welfare state. Her stated aim was to replace this culture with the value of self; this is reflected in what was happening in schools. Thatcher attacked this dependency culture in favour of her enterprise culture (Adkins, 2017). Young people in schools who may fall into this dependency category are those who, through no fault of their own, may not be able to help themselves. The young person with mental health problems I mention in the prologue, whose parents and possibly grandparents had mental health problems, is typical of the ones who are left behind in our system. In far too many

schools, they are not the priority, as the focus is on exam results, and thus their problems are too great to be overcome without substantial financial support.

The values once centring around young people now focussed on the market. It was all about image, exam results and competition for market share; schools now focussed on self-interest in order to thrive or, on occasions, just survive. The values underpinning a school were now set by individual headteachers, many of whom were under increasing pressure in this new competitive world where the very survival of schools was at stake (Grace, 1996). Many headteachers chased the promise of money by becoming independent of LEAs. All schools had to improve exam results as this was the new currency of success; a bright student who could add value in this respect was one that was therefore valued by schools.

What I witnessed at times was however a tragedy. Many young people who needed the greatest support often missed out on potentially beneficial school places as their background or demographic was not seen as something that would enhance the school.

The reform of education in the 1980s took the power away from the local authority. Headteacher appointments were made by them, they had a school advisory team and they also had representation on a school's governing body. Whilst schools were allowed to get on with things in their own way, the local authority had a window into what they could see and what they were doing. With reform, autonomy slowly became real as school leaders set the agenda for schools. No longer was there a local representative able to observe what was going on in the school, and instead monitoring from the national government started creeping in. Lipsky (1980) introduced the idea of 'street-level bureaucrats', who are public sector workers often with high ideals and strong ambitions; a description that fits many of the headteachers that I have worked for. He argued that these public servants, although governed from above, were in fact policy makers in their own right as they need to make daily decisions about how they worked in their own specific environment. As the focus in schools was shifted from a traditional education to an exam and goal-based performance structure, so school leaders needed to move with the times. As street-level bureaucrats, they were now able to set their own agendas and, as such, the traditional value base in schools changed forever. So, whilst schools are monitored more rigorously than ever, it is also true to say that staff within them continue to have agency to do things in their own way. The values, set by a chief executive officer (CEO) of an academy trust or by the headteacher of a school, sets the

ethos of the institution. I argue that values are also crucial to school character (see section 2.4) as they can often dictate how staff and young people interact, and how both groups may feel about the school. I suggest that schools with a zero-tolerance approach towards behaviour will have a very different feel to a school espousing restorative values. Amongst the other areas that contribute to a school's makeup are the physical environment and government policy; the character of a school is about how all these separate areas interact to influence the thoughts and feelings of the individual students that attend.

Within this new world, values took centre stage. Whilst the policy drive, from above, is about economics, market share and self-interest, I believe an individual school's first priority should be the young people they serve. The balance between what is best for the school and what is best for the wider community of young people it may attract is a delicate one. Whilst this new breed of headteacher was focussing on school survival, there were some young people who would not be seen as enhancing the aims of such schools. During my teaching career, I came across several schools where young people with social emotional and mental health difficulties (Department for Education, 2015) were seen as disadvantageous to their image and exam results. Why then would these schools want to spend time and money attracting them? Once the school finance would be required to provide services to look after them, which budget would that money come from? The underlying messages from the government and OFSTED were the only currency that counted was exam results. The introduction of targets and school league tables filtered down to create a culture of performativity. This was taken to extremes by teachers who were in fear of failure; not of their students but of reaching the ever-increasing and continuously stretched targets. This had little to do with child-centred education (O'Neill & Adams, 2012).

Parents then, as now, were in a position where they have to choose a school. Whilst some would send their children to the local school, others look for the best one. They have a limited pool of information on which to base their decision; this includes OFSTED reports largely based on exam results; criteria that many within the teaching profession saw as erroneous. Evidence suggests that parental choice increased the inequality gap in education (Ball, 2021). Simply put, the more affluent parents could afford to bus their children to a better school and the less affluent could not (Ball, 2021). As mentioned above, the result was often that some schools started to contract in size, and as their budget started to decrease the quality of services they could offer would be reduced. The lack of funding meant that often

the first things to be impacted were support services; the thing that many of these young people needed most was welfare support. A colleague of mine, when faced with the prospect of some of her support staff being made redundant, once reported that the headteacher had said to her "You are therefore asking me to get rid of teachers so you can keep your support staff"? The implication was that it was either teachers or support staff who would lose their jobs. The choices were stark but, ultimately, the result was often a poorer education for the most vulnerable in the most deprived areas.

Within the context of results and quality-driven headteachers, schools under threat from OFSTED, and classroom teachers threatened by targets, what space is there for individual students and the potential problems that they bring? In my experience finances are spent to enhance measurable outcomes such as school image, and also to improve exam results. However, the soft outcomes, such as wellbeing and mental health, are rarely measured and are therefore likely to be low down on any priority spending list. One of the criteria parents look for in a school for their child is high academic standards (Reyes & Due, 2015), and schools know this and will do what they need in order to achieve their goals.

For the majority of my career, I was lucky enough to work in a school that did prioritise pastoral care and the welfare of young people. Whilst I always sought increased funding for my team, money was spent on support staff and services so that we were able to look after young people and support their families. However, I was also acutely aware that we were the lucky ones, as the majority of my colleagues in other schools did not have the same priorities. Far too many schools developed into inward-looking institutions that had been swept along by the drive to grow what they had at the expense of others. The child-centred welfare agenda had been overtaken by the neoliberal agenda of the market and self. This is something that started its journey during the 1970s, and is still now having a significant impact on schools and, more importantly, on the life of some of our most vulnerable young people. Not only has the support for young people been removed, but there has also been increasing pressure on them to perform; this has resulted in a downturn in young people's mental health and also, ironically, on their exam results (Smith, et al., 2021).

The implications of the policy shift in education resulted in other subtle changes in our schools. With the focus on examinations, I experienced schools discouraging young people's participation in out-of-the-class trips and visits. This narrowed young people's experiences

along with their opportunities to develop social and cultural capital. In addition to this, the drive for improved examination results has seen a push for greater teacher control in schools that has led to the increasing implementation of zero-tolerance behaviour policies; something I have seen at first hand. This strengthens the power imbalance that favours teachers in schools and undermines some young people's school experience. This is a complex area overlapping with what the purpose of school is. As I discuss in the following section, school purpose is itself complex and ill-defined, but is an area I need to address as it adds context for my thesis.

2.3 School purpose

On a simple level, as someone who has been involved in education for nearly 60 years, I have always seen its purpose as being to educate the country's children. Young people often get but one chance at school; if the school gets it wrong, it has let them down or failed them. This could include underachievement relating to qualifications or, more broadly, be in relation to preparedness for life in society. With increasing experience of education, I have developed more nuanced views and believe that education has a moral purpose to make a difference and bring improvements to people's lives (Elliott, 2012). This, in turn, links to the changing value base I have addressed in the section above, as it raises ethical questions about school leaders responding to policy change. As such, I agree with Ball (2006, p.11) when he says "Ethical reflection is rendered obsolete in the process for goal attainment, performance improvement and budget maximisations".

Looking more widely, the purpose of education and schools is something that has been debated for many years, from ancient times when figures such as Plato and Aristotle wrote about it to our more recent history when Dewey wrote that it was about teaching children how to live "pragmatically and immediately" (Dewey & Cohen, n.d.). In a critique of Dewey's work, Counts (1932) moved away from education and schools being about preparing individuals to live independently and towards them being prepared to live within society. These debates have continued into more recent times. Adler (1982) suggested that there were three main purposes for schools, and they were the acquisition of knowledge, development of skills and understanding of concepts and values. A pragmatic view was posited in the 1990s by De Marrais (1995), who believed there were four major purposes of schools that were: economic, social, intellectual and political.

The debate continues into the 21st century where over a decade of accountability measures, imposed through the government and enacted by OFSTED, have seen many schools shift towards a predominantly academic focus. However, some believe that they are places where young people learn how to form healthy relationships, and develop emotionally, behaviourally and cognitively, as well as gain their independence (Wilson, 2004; Cohen et al., 2009). An area worthy of further consideration, and which I will develop in chapter 4, is how power in school influences young people and their ability to take responsibility. It has long been argued that schools are responsible for socialising (Henslin, 1999) and promoting conformity of young people (Saldana, 2013). However, conformity should also be recognised as a power-laden tool which could be seen as an insidious practice and as a way in which society and school values are enforced (Saldana, 2013).

Over the past 40 years schools have changed. They have become businesses in their own right, leading to competition between institutions for the budgets that follow young people. As capitalism has taken hold, so the values in schools have changed. There is a greater emphasis on exam results, which has a twofold effect. It can put undue pressure on young people, some of whom develop mental health issues, whilst the support for them is often reduced with budget restrictions resulting in pastoral staff reductions. A perfect storm for increasing mental health issues.

Education in 2023 is at the centre of an ideological struggle between traditionalists who favour a purely academic focus and progressives who approve of more rounded and holistic approaches. This has partially been addressed by the latest OFSTED framework introduced in September 2019, which has a more socially-balanced focus. Inspectors are now asked to look at a school holistically when making their judgements. They have removed expected exam progress as the limiting measures (OFSTED, 2019). The way in which a school approaches key policy areas such as curriculum, behaviour, inclusion etc., will therefore directly impact upon school character. And so, this takes us full circle to the values of the school and how it is led. As educational policy changes and schools are put under increasing pressure, are the decisions they make ethical, and for the benefit of the school or young people? What a school leader believes and how they implement this belief will impact directly upon the staff, young people, the environment, and the relationships between all involved. This is the character of the school, something I will now go on to explore.
2.4 School character and how it may influence outcomes for young people

2.4.1 Overview

It is essential for schools to understand how the character of the school may impact students, their emotions and, therefore, their willingness to connect with the school and any available school support. This section explores the concept of school character and how it may impact upon the mental health of young people, including the measures put in place to support this. First, I consider what is meant by the term school character before investigating the theories that may influence it. The impact that school character has on students is likely to impact their outcomes. In order to enhance the theory around school character and draw it into the specific context of my research, I will supplement this section by reviewing appropriate literature and exploring exemplars from my teaching career. I will conclude by summarising the potential impact of school character on the mental health of young people.

2.4.2 What do I mean by the term 'school character'?

In section 2.2.4, I explore how changing education policy has impacted upon school values. The changes made in the last 40 years have meant that schools are in a competitive marketplace where performativity is the key measure through which they succeed or fail. This has resulted in changes to the ways in which schools operate that, in turn, have shifted their value base. Where there was once a focus on success for the sake of young people, the focus has now shifted to being about school success. I suggest that this subtle change has impacted the school experience for young people. School character is exactly how young people experience their education. It is linked to the feel of the school, what the climate or culture of the school is and how this impacts on young people. The importance of school character cannot be understated as how young people feel about their time in school is likely to impact upon wellbeing and performance (Lester & Cross, 2015). An exploration of the key components of school character will be the focus of the following section.

2.4.3 Values and climate as subsets of school character

The first acknowledgement of the concept of school climate was early in the 20th Century (Perry, 1908). The first research was in the 1960s (Halpin & Croft, 1963) and is deemed to be

somewhat oversimplistic as school climate is now accepted as being complex and multidimensional (Wang & Degol, 2016). There has been little consensus on a definition as to what school climate is (Rudasill et al., 2018); some, such as Freiberg & Stein, (1999) choose abstract definitions that use terminology suggesting school climate is the heart and soul, and the essence, of the school. Others have more concrete definitions. Aldridge & McChesney (2018) explain school climate as being about how the expectations and beliefs of a school create environments where young people feel emotionally, physically and socially safe. Terminology is also a problem as the term school climate is often interchanged synonymously with terms such as classroom climate, school belongingness, school connectedness and school culture (Rudasill et al., 2018). There are those who differentiate between school climate and culture, defining climate as "the character and quality of life within a school" and culture as "a set of beliefs and values" (Lester & Cross, 2015, p. 2). Furthermore, school climate could be described as a facilitator for school culture (Lester & Cross, 2015). There are clear links between the two descriptors and, although they are conceptually close, it is however important to differentiate between them. Culture is related to shared norms, whilst climate is about shared perceptions (MacNeil, Prater, & Busch, 2009).

Whilst it is tempting to adopt a definition of either school culture or school climate, I have adopted the term school character as a more inclusive term that includes all aspects of a school environment. In addition, school culture can be viewed from an anthropological perspective and, as such, there are ethnographic features that need to be drawn upon in this thesis (MacNeil et al., 2009). I, therefore, want to ensure that any cultural aspects are not overlooked. The definition of school character that I will therefore adopt for this thesis is:

School character is the interaction between cultural, physical, emotional, relational, and social aspects of school environment. The interplay between these individual facets of school character is set by school leaders and includes their values and behaviours. School character relates to how these varying aspects impact staff, young people, and their relationships, as well as the wider outcomes of the schools.

Wang & Degol (2016) introduced the Conceptualisation and categorization of school climate framework (Figure 1) which divides school climate into four broad categories: academic, community, safety, and institutional environment. This can be further broken down into sub-

categories that encompass the areas of school life that impact the lives of young people who attend them. By drawing on personal experience and focussing on each of the categories I can contextualise how this framework can contribute to my research.

The **safety** category in relation to school climate, and therefore character, refers to physical and emotional safety as well as order and discipline (Wang & Degol, 2016). Emotional safety is also defined by whether staff are caring and supportive, and whether the school provides services to support those who may be experiencing mental ill health (Kuperminc et al., 1997; Kuperminc et al., 2001; Swearer et al., 2010). Again, my reflective diary extract exemplifies the point:

"My experience as the school leader responsible for student health and safety allowed me great insight into this area. Young people who were bullied either physically or emotionally were unhappy, frightened and neither enjoyed school nor performed well there. On the occasions that classes were disorderly and discipline was lacking the emotional health of young people suffered."

As I explore all categories within the model, the importance of the **community** category will become self-evident. Positive relationships are key to respect, trust, support and caring, and as such are vital within the context of school character (Birch & Ladd, 1997; Pianta, 1999; Wang et al., 2012) and mental health. Equally important is the approach of staff to equality and diversity by valuing young people's opinions as well as their autonomy and interest (Weinstein et al., 2003). This is exemplified by another extract from my reflective diary:

"In my previous career as a teacher in an inclusive school, I can attest to the fact that character of the school was all the better for the educative approach to correction when young people made the mistake of not respecting minorities or diversity. When young people made the mistake of abusing others because of differences such as race, gender, or sexuality, they were required to attend one-to-one education sessions, in their own time, where the pertinent issues raised were discussed and explored."

The **academic** category focuses on leadership, teaching and learning (T&L) and the professional development of staff. Whilst each are key, a good leader will set the vision for all other areas of school life, including T&L and professional development. As such, a good

leader will be a good communicator, offering guidance to the staff (Leithwood & Riehl, 2003). The best school leaders ensure that there are open lines of communication between school staff and students (Waters et al., 2004; Kelley et al., 2005). T&L is about how teachers communicate understanding to young people. It encompasses areas such as instruction, academic challenge and feedback to young people, and is thus important in relation to school climate (Stefanou et al., 2004). The final aspect of this area is professional development for staff. From my own experience this area is crucial as a school develops its vision and ethos, as illustrated by my reflective diary extract below:

"For example, when leading on whole school behaviour, in my school career, I implemented a restorative approach to discipline. At this time my focus was to ensure that I educated staff in relation to both the restorative theory and school policy and so ensure that there was consistent implementation. Without this professional development, which was ongoing and took place over several years, the vision would not have been realised and school ethos would not be as I had wished it to be."

As will become clear in the findings section of this thesis, interpersonal relationships are pertinent to this.

The final category of **institutional environment** is related to the impact that the "sensory quality" of the school environment has on young people who inhabit it on a daily basis (Wang & Degol, 2016). The particular focus is on the condition of the buildings and whether it is a pleasant place in which to learn. Dawson and Parker (1998) found that the quality of instruction and teacher effectiveness improved with the quality of the environment. This in turn improved students' academic performance. Carpets, decoration, lighting etc. were regularly maintained to ensure that the environment was clean, welcoming, and a pleasant place in which to study. I reflected:

"Early in my career one of my first headteachers had an annual percentage of the everdecreasing school budget designated to modernise the school buildings and keep them in good decorative order. In this era of school marketisation, I have also seen the push to increase school numbers as a way of improving the school budget so that money could be spent on resources and the upkeep of the school. The whole purpose of such approaches was to ensure

that the environment was pleasant and welcoming for the school community but young people in particular".

The conceptualisation and categorisation of the school climate model is a comprehensive framework that complements my own definition of school character. I, therefore, intend to adopt the term school character as a reference tool throughout my thesis. Although the model is comprehensive as regards the breadth of topic areas, it cannot encompass all areas for all schools. I will therefore, where appropriate, contextualise, innovate, and adapt it to work for the school in which I am conducting the research.



Figure 1: Conceptualisation and categorization of school climate (Wang & Degol, 2016)

Sections 2.2 to 2.4 have given context from a historical educational standpoint that introduced the idea of a value shift in schools and how this has filtered through to impact the character of the school. It is also important to explore theoretical concepts related to young people and their development within both societal and educational structures. This will be the focus of the next section.

2.5 Theoretical influences

Research into child development, environmental impacts and the links between them demonstrate they may well influence school character. The area that can offer this work the greatest insight is Bronfenbrenner's work on ecological systems theory (Bronfenbrenner, 1977) and bio-ecological framework (Bronfenbrenner, 1994).

2.5.1 Bronfenbrenner's ecological systems theory (EST)

Bronfenbrenner's (1979) EST and bio-ecological framework suggest that development is the result of reciprocal interaction between a human (bio-psychological organism) and others within the environment. The EST focuses on the nested model that places the child at the centre of five interconnected ecological systems (see Error! Reference source not found.). The first level of this system, known as the "microsystem" is about the child's immediate environment, including the people around them both in and outside of the home. These are very often personal and intimate relationships that are bi-directional. Young people spend much of their childhood in schools, interacting with peers, school staff and the environment, and as such schools are therefore instrumental in a child's development. The levels of cooperation, conflict and collaboration between all parties, as well as academic expectations, all go to make up school climate and character (Haynes, Emmons, & Ben-Avie, 1997; Juvonen, 2007). It has therefore been proposed that the school is seen as the microsystem, as character is directly influenced by the combined perceptions of the school community (Rudasill et al., 2018). The implication for schools is profound. As demonstrated in the writing earlier in this chapter, school character is about the interplay between the human, physical, emotional, and social aspects of schools, as set by the values of its leadership. As with the EST, the young person sits at the centre and should be the focus of any improvement work. In section 2.2.3 I discussed the marketisation of education and how this has skewed many schools' aims. Instead of centring on young people, schools have been distracted and

now focus on satisfying external bodies such as OFSTED. The young person needs to return to the centre of everything that school is about and the focus of schools should be on ensuring their "microsystem" is the priority.

The "mesosystem" is the interaction between a child's microsystems. It is about the impact that the relationship between teacher and parent may have on them to influence effort and engagement in academic work (Galindo & Sheldon, 2012). However, there are also times when parents and schools have different values and give conflicting messages. This can often leave the young person trying to negotiate a path between the two (Spencer, 1999).

Bronfenbrenner also identified both formal and informal social structures not necessarily containing the child, but possibly impacting upon them. These "exosystems" have a distal influence on young people as it has an indirect impact on school character. It may be a government policy implementation, a fund-raising campaign by the parent association or an effective charity drive such as Red Nose Day. These experiences have a bearing on the life of the young person in the school (Marino, 2011).

The cultural elements that impact upon a child's development, such as poverty, ethnicity, religion etc., are known as "macrosystems". These are the effects that, through immersion, influence a child's beliefs and perceptions. As mentioned previously, a school's values and ethos are part of its character and, more specifically, its macrosystem, and something that, from my experience, will impact upon a child's beliefs and perceptions.

Finally, "chronosystems" are the major events over time that impact a child's development. Be it starting school, transitioning to secondary school, or a high-school shooting (Hong & Eamon, 2012), all such events may leave young people feeling insecure within their school environment.

It will often be difficult to identify exactly which aspect of school character influences an individual within the school setting. It could be the physical structures, systems, curriculum or young people/staff relationships, but the bio-ecological theory asserts any of these can influence student development (Way et al., 2007). Furthermore, the EST has been designed to include all influences upon the young person at the centre; they are the focus and, as such, the model suggests that school character should be responsive to factors beyond their control,

such as community dynamic or government policy. What happens in practice is still however subject to the values of school leaders. As we saw earlier in this chapter, ethics and ideology play a big part in this; one leader may respond to their school being in a deprived part of the country by enforcing a zero-tolerance behaviour policy, whilst another may respond by introducing a relationship-rich restorative one. Another aspect that should also be considered is how much of an influence a school can have on a young person, bearing in mind they are subject to numerous influences outside of school. Conflict may often arise when there is a lack of congruence between the school and these other areas. The most obvious example being a parent with a very different value base to the school refusing to support the school and coming into conflict with it.



Figure 2: Bronfenbrenner nested model of ecological systems (Guy-Evans, 2020)

Returning to exemplars from my teaching career, an extract from my reflective diary gives clarity to how there can be conflict around the school external factors influencing young people in school:

"From my own experience in schools, I can recollect numerous occasions where young people who I knew to be responsible outside school ended up in serious trouble within the school environment; the young farmer who took massive 'adult' responsibility at home running aspects of the farm, the young carer looking after the alcoholic mother being but two examples. The environment in which they lived enabled, allowed, or possibly forced them into positions of responsibility however when they were in school, they were required to follow rules which they often saw as irrelevant and rebelled against. Both in and out of school I would argue that the bio-ecological framework, with its multidimensional nature could possibly explain the differing behaviours of young people from different environments."

Rudasill et al. (2018) have developed a systems view of school climate (SVSC) (see Figure 3) that is broadly based on Bronfenbrenner's nested model. It develops the nested concept and, whilst it is based around the school, it also links with family and peers that interact with the school microsystem. The SVSC chose a new term, "nanosystem", to describe adaptations specific to schools that allow for the interaction of sub-groups within school microsystems. This places the student as separate from the school microsystem and impacted through more remote 'nanosystems' such as classrooms, peer groups and extra-curricular activities, including involvement in research such as this.



Figure 3: Systems view of school climate

The extract from my diary below shows how the introduction of nanosystems, as a subcategory of microsystems, presents opportunities to explore the potential influences on young people's experiences in school:

"From past involvement in schools, I have seen the impact, both negative and positive, of staff. The support staff with specific student welfare responsibility who over a period of years builds relationships with young people and their families. They build trust and improve student experience in school; the role allowing the development of relationships, has a direct impact on school character for young people. Similarly, there are staff with attitudes of 'zero tolerance' around behaviour as they have little understanding or interest in child welfare. They expect all young people to behave impeccably and have inflexible behaviour systems often resulting in serious conflict with students. This again impacts on the school character for young people who attend those classes. These differing adult behaviours have the potential to impact young people's mental health, something not to be underestimated."

2.6 Chapter summary

In this chapter I have laid out the context for my research starting with how historic education policy has played out in the past 40 years, and how there has been a shift in values due to the

marketisation of schools. I have explored how this shift in values has impacted the school experience for young people as the school character has also undergone a period of change. I have concluded the chapter by considering the research context through the theoretical lens of EST. EST holds a mirror to the changes seen in education in the past 40 years. It prioritises the young person, and appreciates everything in their lives that can impact upon them. The following chapter concentrates on aspects of mental health in relation to adolescence and schools so I can apply recent learning to this research.

Chapter 3: Young people, education and mental health

3.1 Chapter introduction

This chapter explores mental health in general terms, and more specifically in relation to this research. I begin by looking at how mental health has been viewed historically before focussing on education and mental health. The impact that the Covid-19 pandemic has had on young people's mental health is explored alongside a critical exploration of educational responses. I investigate adolescence through the lens of attachment theory before reviewing literature related to mental health research developments. This includes how magnetic resonance imaging (MRI) has transformed brain function research. I complete the chapter by investigating how MRI discoveries have contributed to understanding adolescent mental health. In particular, I focus on adolescent social cognition and self-concept, including social media's role in this.

3.2 What is mental health?

There is great ambiguity around the term *mental health* (American Psychological Association, 1959; Manwell et al., 2015). This includes the fact that it is sometimes used as a euphemism for mental illness (Cattan & Tilford 2006). Mental health has been defined as an absence of disease, as well as being associated with biopsychosocial factors that may contribute to an individual's ability to function within society (Manwell et al., 2015). There are other definitions that go further, including intellectual, emotional and spiritual development (Herrman et al., 2016), together with physical health and feelings around self-perception and self-worth (Bhugra, Till, & Sartorius, 2013; Caswell, 2021)

As highlighted by Manwell et al. (2015) discussions around mental health have been further informed by a debate exploring a redefinition of the term health (Huber et al., 2011). This debate has challenged the World Health Organisation's (WHO) definition of health as achieving "a state of complete well-being" (WHO, 1946/2022). The WHO definition was lauded as inventive and progressive when enacted in 1948, as it attempted to move away from a traditional and problematic view of health, towards a positive and holistic definition. Huber et al. (2011) argue it is a static definition that, due to its longevity and changing world

demography, is outdated. Furthermore, the term 'complete' is unachievable, and I contest most of the world's population would fail to reach the goal of 'complete health'!

However, in the 70 years since the original WHO definition, the understanding of mental health and mental illness has moved on. It is now seen as a desirable asset that enhances life experience for individuals and society (WHO, 2019a). The WHO were building upon the work of figures such as Huber et al. (2011), who looked at health definitions that included concepts of capacity for an individual to adjust and develop over time. As an inclusive statement, this gives power back to individuals; it is optimistic and positions health within a person's control as it "restores one's integrity, equilibrium and sense of wellbeing" (Huber et al., 2011, p.3). The balance between passing responsibility on to individuals and supporting them to take responsibility is very fine; related changes need to be supportive. The WHO revisited the concept of health in 1984, reframing its thinking towards a vision of health as a resource, and not an objective measure or goal. Within this definition the term 'change or cope' is used in a similar way to adjust as used by Huber et al. (2011). This is reinforced further as the definition goes on to state "Health is a resource for everyday life, not the objective of living; it is a positive concept, emphasising social and personal resources, as well as physical capacities" (WHO, 2019b). This realignment from a judgement (either subjective or objective) to an asset is a discrete and definite move away from a deficit model towards a positive and inclusive framework.

Manwell et al. (2015) build on the work of the WHO (Boxer, 2005) and Huber et al. (2011) in adopting the three domains of health: physical health, mental health and social health for their transdomain model of health. By integrating them they have been able to identify key and interlinked components for each domain, whilst also highlighting areas of common ground (see Figure 4 below).



Figure 4: Transdomain Model of Health (Manwell et al., 2015, Figure 4.)

As I have already discussed above, this is about individuals being able to draw upon their holistic resource base in order to thrive. Mental health can be affected by genes, brain chemistry and neurons, but it also has to do with an individual's interactions with the wider environment. A person's relationship with self and others, society, environment, and political and economic systems are among the components that all have an effect on mental health (Manwell et al., 2015).

3.3 Mental health in the context of this research

I embarked on my own professional mental health journey as I followed a pastoral career path in schools. My job was about building relationships with young people and their families while supporting them, linking them with external services and ensuring they were able to benefit from education and progress in life. In my early days of teaching, mental health was a term reserved for medical settings and rarely used in schools (RothÌ & Leavey, 2006). In my experience, it has only been in the past ten years or so that the term mental health has received widespread use in the education sector. It was initially mentioned in relation to certain behaviour problems, anxiety and bullying issues, and one of the issues was these types of problems seemed to be on the increase and schools were not in a position to deal with them. The issues related to the term mental health and its meaning in a school context continue to this day. This is something I will return to later in this section. My experience of the last 30 years has seen schools move from being places of education that were primarily run by teachers to quasi-welfare institutions with a wide range of multi-disciplinary staff within them (Simkins, Maxwell, & Aspinwall, 2009). The changes in the needs of young people have been gradual, and thus the profile and skills of staff in schools have needed to adapt. There were no rule books and changes were organic, and developing at a time of decreasing school resources. With the introduction of devolved school budgets, individual institutions prioritised their spending (Ball, 2021). Some schools, such as mine, ensured the pastoral dimension was well served, while others neglected it. Working with schools from across the country, I saw a patchwork of pastoral approaches, meaning an inconsistent provision in the care for young people.

Furthermore, school change has been accompanied by reductions in support services across many sectors of society. In 2018, the Sutton Trust estimated that up to 1000 Sure Start centres had been shut (Butler, 2018) since 2010. The YMCA reported that between 2010/11 and 2018/19 the reduction in funding for youth services saw the loss of over 750 youth clubs and over 4000 youth workers (Weale, 2020). In addition, austerity, the coalition government's stated aim, resulted in a stripped-back welfare system, as well as an NHS that is being funded at a far lower rate than previously (Stoye, 2018). The result of these changes, together with the changes to the experiences of adolescents outlined earlier, was an increasing problem with fewer resources. In schools and society, there are increasing adolescent mental health problems, alongside a reduction in vital support and medical services (Local Government Association, 2022).

There is an acceptance that the period of adolescence finds many vulnerable to mental health issues, due to several conflicting factors impacted by this sensitive period of brain development (Fuhrmann, Knoll, & Blakemore, 2015). Whilst most young people navigate this period without major mental health issues, the growth in such problems is truly worrying. In 2017 it was estimated that, in England, one in nine, equivalent to 850,000 children and

young people, had a diagnosable mental health disorder (DfE/DoH, 2017). Figures for anxiety in children (3.3%), depression (0.9%) and mental health disorders (10.8%) (DfE/DoH, 2017; NHS Lifestyles Team, 2020) are all concerning. However, one of the issues is there seems to be an upward trend in mental health disorders statistics. They show a steep increase between 2017 and 2020 to one in six, with an estimated increase of 425,000 to 1,275,000 (NHS Lifestyles Team, 2020). One of the contributions to this rising trend may be increasing acceptance of the self-reporting of mental health concerns, resulting from efforts to destigmatise it. Furthermore, some believe the recent focus on mental health may in itself be increases in reporting is further reinforced by figures that show that CAMHS receives 183 referrals every school day, with 56% of these coming from primary schools, an increase of one-third since 2015 (The Lancet, 2018). This has been further compounded by the COVID pandemic (NHS Lifestyles Team, 2020).

The reality is that young people who experience mental health issues are more likely to have problems in later life. Thus, as well as impacting the quality of an individual's life during this most impactful period, it can cause problems as young people mature. Young people with mental health issues are more likely to have a disrupted education, with apparent consequences (DfE/DoH, 2017):

- There are links between mental ill-health in childhood and poorer employment-related outcomes, including regularity of employment and earning power (DfE/DoH, 2017).
- Links between mental health issues and crime are well documented, with young offenders more likely to experience mental health difficulties (DfE/DoH, 2017).
- The cost to society cannot be ignored either, with over 11.4 million working days lost due to work-related stress, anxiety and depression (DfE/DoH, 2017).

My experience in schools was that there were generational mental health problems. Having taught across two, (and possibly 3) generations in the same school, I came across young people and their parents whose patterns of mental health followed a similar trajectory. The concerns above suggest it is of the utmost importance that society does everything it can to address the rise in adolescent mental ill-health. Whilst many of the challenges mentioned in this paragraph are societal problems, they are often positioned as issues that schools can address through policy (Ball, 2021). Whilst the Mental health support teams pilot

(Department of Health and Department of Education, 2017) and its subsequent fuller implementation (NHS England, 2021) is a recent initiative that aims to address some of the issues of mental health in schools, the reality is that schools have limited options to draw upon. In section 2.2 I raise issues around the marketisation of schools that have drawn on both economic resources and school values; both can be barriers to improving young people's mental health. An area that schools can use to support emotional difficulties is the SEND process, which includes an application for an Education and health care plan under the category of Social, emotional and mental health difficulties (Department for Education, 2015). Where the problems go beyond being just educational and impacting life outside school, there are options to apply for an Early help review from local authority partners (OFSTED, 2015). However, in my experience, rarely did individuals with mental health issues benefit from these measures.

3.4 Mental health and school approaches in 2021

3.4.1 Overview

The term *strange times* could not be more apt for this thesis. In the following short section, I present the mental health dichotomy society, as schools find themselves embroiled in this at the beginning of the 21st Century. A recurring theme throughout this thesis is growing calls by many for a focus on supporting children and young people's mental health needs. This has been exaggerated further by the COVID pandemic we have been experiencing. This has magnified societal inequalities, as health and financial concerns have created trauma for many, with young people being the hardest hit (Meredith, 2020). Issues around bereavement, anxiety and loneliness are among the key factors impacting young people (Cava, Buelga, & Tomás, 2021). It is also important to recognise that there is evidence that some young people experienced "improved mental wellbeing" (Soneson et al., 2022, p. 1) during the lockdown. Whilst this thesis is not bound by the pandemic, the research occurs at a unique historical point for young people and school mental health.

Research at this time has also linked young people's mental health and attainment in secondary education (Smith et al., 2021). Smith's report also found that young people with similar characteristics to the participant cohort were likely to be at the most significant risk of underachievement. However, another point of view needs to be considered, which is the rise

of what has been called 'therapeutic education', which is seen by some as doing more harm than good (Didau, 2016). As I will set out, the arguments against therapeutic education are further confused by the ideological debate in schools, which pits the knowledge curriculum against the skills curriculum (Gibb, 2021). At its simplest, this is about traditional versus progressive education, and is where emotional or therapeutic education comes under pressure. The UK government follows a traditional path (Gibb, 2021) and favours a knowledge curriculum. This combines with the pressures of assessment, discussed above, to create an environment where increasing numbers of young people struggle emotionally.

3.4.2 Therapeutic education and how it is viewed

The ideology mentioned in the section related to educational history and mental health goes deeper than just an education debate. After the 1997 election victory, New Labour championed a more progressive and child-centred approach, including the introduction of social and emotional attitudes to learning (SEAL). Ecclestone & Hayes (2008) suggest that New Labour's approach was about appealing to the public via a popular therapy culture. Since the re-election of a Conservative government in 2010, the focus has been on implementing a knowledge curriculum, as referenced by Gibb, (2021). These political decisions then divide opinion and add to the broader education debate that follows an ideological script. What seems to get lost is what the right thing to do for young people in our schools is.

Although some are demanding a greater focus on emotional health in schools (Anna Freud National Centre for Children and Families, 2022; Lioncare, 2022), there are also those who believe that, by its very nature, such an approach disempowers and undermines young people's self-esteem. An example is the work of Didau (2016), who supports the view that therapeutic interventions aimed at suppressing negative feelings can do more harm than good (Borton & Casey, 2006). Emotional curricula, such as SEAL, promote a belief that young people struggle emotionally, are vulnerable or are challenged by low self-esteem (Ecclestone & Hayes, 2008). The argument is that however well-intentioned this approach is, it lowers expectations and aspirations and replaces the goals of an academic curriculum (Didau, 2016). Whilst this may be true, I suggest most modern interventions are not about the suppression of feelings, but about understanding and coming to terms with them. During my school career, I saw examples where interventions enabled a deficit approach whereby young people, and

sometimes their families, were labelled as being in need and therefore invited intrusion into their lives.

Much of the argument against school interventions seems to be based on the polarisation between a therapeutic or knowledge-based curriculum. My time working in schools with young people and families tells me that it is not a black-and-white decision. Just as there should be a balance between skills and knowledge, there is a place for an emotional curriculum. The research evidence around the impact of COVID on young people demonstrates that many need support and help to deal with the trauma of what they have been through (Carpenter & Carpenter, 2020; Meredith, 2020; Waite, 2020). Furthermore, Smith et al. (2021) are clear that increasing mental health issues are impacting attainment, particularly for the cohorts of disadvantaged young people. If we ignore the problem, then it certainly will not get any better.

3.5 Understanding adolescence and links to mental health

The natural processes that ensure humans change biologically from childhood through puberty into adolescence, and then to adults, have been misunderstood and often demonised for centuries. Rather than seeing adolescence for what it is, it seems that through history a deficit model has been developed that still exists today.

In ancient Greece, both Socrates and Aristotle (469-322 BC), when talking about the youth of the day, wrote about young people having bad manners, showing disrespect and being fickle, things that would not be out of place in today's tabloid newspapers. (R. W. Hill, 2017; Hohnen, Gilmour & Murphy, 2019). In the 17th century, Shakespeare exemplified the worst of adolescent behaviour in The Winter's Tale (1623) when he wrote "…stealing, fighting— Hark you now! Would any but these boiled brains of nineteen and two-and-twenty…" (Shakespeare, 1978).

In the first decade of the 20th century, the first definition of adolescence was committed to academic text when Hall (1904) wrote about it as a time of 'storm and stress'. Again, the implications were that this was a difficult time for young people, where their behaviour was scrutinised and often below the standards that society expected. This pattern continues today, as young people are often labelled as abnormal because of the adolescent traits that, it could

be argued, result from the biological changes happening to them. A newspaper headline that exemplifies this, "Tories five-point plan to tackle yob culture" (Daily Mail, 2005), explains 'yob' behaviour on a mix of young people and cheap alcohol. The article discusses how the leader of the then opposition would 'fix' the problem by 'frightening' these 'yobs' with the police, whilst simultaneously adjusting licencing laws. The language used in the article assumes that the readers understand terms like 'yob' and 'thuggish', and plays into a societal fear of young people (Arnett & Hughes, 2012).

Adolescence is seen as a developmental stage that starts with puberty and ends when an individual maintains a stable, independent position in society (Mann & Blakemore, 2021). A generally accepted age range was from 10-19 years (WHO, 2014), however more recently an argument has been made that the age definition needs to be expanded to 10-24 years (Sawyer et al., 2018). Earlier onset of puberty and neuroscience developments, as well as societal changes, suggest that adolescence is now lasting longer, and adulthood is often starting later, particularly in many Western societies.

Typical behaviours highlighted above and associated with adolescence, such as conflict with parents, mood swings and risk-taking behaviours, are seen as negative and disruptive. However, these views are slowly being replaced by a transformative (Sawyer et al., 2018) point of view suggesting they are a necessity, enabling the growth from a child, (a being in their own right) to a becoming a responsible adult (Arnett, 1999; Corsaro, 2018). It is a unique period in a person's life where they exercise their agency to challenge, grow and develop. In so doing they go through a period of mastering enabling experiences that should be viewed as a normative process (Bandura, 2006). Western society still has a deficit approach to adolescence, but this developmental period should be seen through an adaptive and rational lens (Blakemore, 2018). It is now accepted that peer pressure plays a large part in the life of some adolescents; they engage in risk-taking behaviours rather than jeopardise their social standing within a group. Furthermore, what is far less accepted in Western society is that during this period in their lives many pro-social behaviours are contributing to their development (Blakemore, 2018).

My personal experience in schools certainly supports the opposing views of adolescence. As a teacher for many years, the overriding consensus from those with power in schools is that there are young people who choose to misbehave. Therefore, there are others who do not know how to behave and, whilst mitigating factors are sometimes considered, the consensus

is one of negativity towards certain sections of the student population. Having responsibility for the behaviour system for over 12 years put me at the centre of this dichotomy. Whilst we tried to develop a system based on relationships rather than behaviour, there were occasions when the intensity of the work, the drive by outside forces and pressure from targets became overwhelming, resulting in a deficit mindset around this group of young people. Over the years, there were many occasions where I would have to take stock and remind myself that the number of behaviour issues in school created pressure due to the interaction of a tiny proportion of the population. Whilst on occasions I might have to remind myself that I was in danger of lapsing into a deficit mindset, some other staff were squarely in the deficit camp. As with society, there were those who took on the view that we had a yob culture, and would treat young people accordingly.

Society's view of adolescence continues to be informed by research, and there is a growing body of evidence suggesting the attachment between a child in the early years of their life and a primary carer can contribute to the impact upon their behaviour during adolescence. I explore attachment theory below.

3.5.1 Attachment and how it can impact adolescence

Attachment theory is one of the critical foundations of school climate work as it emphasises the importance of secure child-adult attachment, as well as warm and consistent environments (Bowlby, 1988). At its most basic, attachment is about the bond between a child and a unique and particular adult (or adults) in their life. Bowlby (1988) initially investigated the role of the primary carer in providing a safe and secure base from which the child will be able to thrive and flourish. Adults, through attunement, ensure the emotions and the psychological state of the child are the centre of attention and the motivation to their work (Trevarthen, 2011). Attunement helps develop a child's emotional security and enhances their feelings in relation to themselves and others. Furthermore, it influences their thoughts, feelings and behaviours, helping them cope with stressful situations (Sroufe, 1995).

The relevance of attachment theory for secondary schools and adolescents comes as it has been discovered that interventions in later life can positively impact attachment issues (Smith et al., 2017). Furthermore, it has also been suggested that events in later life can affect relationships and cause attachment issues (Cummings & Davies, 1996). Whilst attachment was initially used as a therapeutic model in counselling and psychotherapy, it has more recently been linked to educational settings. The development of positive sustained relationships that promote attachment in schools is one of several requirements needed to ensure that a young person's developmental needs are fulfilled (Darling-Hammond et al., 2020). It has been established that attachment influences young people's relationships with their friends and peers, as well as school staff, something referred to as a microsystem by Bronfenbrenner (1979). It therefore stands to reason that, in schools prioritising exam success ahead of pastoral care, these types of relationships will be more difficult to develop.

Furthermore, it has been recognised that schools that develop environments with opportunities for stronger adult/young people relationships provide a more productive context for learning. Young people in such environments are more likely to have a better attachment, greater motivation, higher attendance, better behaviour and improved achievement (Hammond & Harvey, 2018). As a teacher, whilst I saw many young people develop attachment with staff through their time in school, others could not build these independent relationships through a lack of confidence. Schools can and should do more. In secondary schools with robust transition arrangements with their feeder schools, young people who find relationship building difficult will be known to the school. Schools then have a unique opportunity to create systems whereby these young people can build a meaningful relationship with a key adult before transfer. By employing a staff member whose role is to support and advocate for the young person, a start can be made to ensure healthy relationships develop; this then creates the greater possibility for attachment to be made.

Trusting and positive relationships between young people and school are one of the key areas when investigating the success of young people in school (Smyth, 2007; Martin & Dowson, 2009). Research has also demonstrated that young people can form bonds with adults outside the home, such as school staff (Kennedy & Kennedy, 2004; Riley, 2009). Once again, I draw upon my reflective diary to illustrate this point:

"In my school experience, I have witnessed how some staff have built trusting relationships with young people. In particular the staff member responsible for overseeing the school's pupil premium cohort produced remarkable results through building relationships with her charges. She built the trusting, parent-like relationship with the heroin addict's daughter or the young man who lived with grandma because mum and dad were alcoholics and could not

care for him. She helped, supported and advocated for them but also held them accountable for their behaviour. She had a parent-like bond allowing them to succeed at school."

Bergin & Bergin (2009) suggested that at least one-third of young people have insecure attachment issues that could impact their school performance. Unfortunately, the behaviours exhibited by such young people are often misinterpreted by school staff, whose reactions can make the situation worse. Attachment behaviours may include aggression, lack of concentration, demanding behaviours etc. My reflective diary extract once again illuminates how my school experience supports this theory:

"The problems that I experienced in school was the need to balance the good of the individual against the good of the rest of the school population. School budgets were never enough to support those with the greatest needs. Often fallback positions were punishments and sanctions rather than the support or specialist help these young people required. Whilst the school used restorative approaches to foster relationships, on occasions, young people were excluded from school. I also saw some schools focussing solely on student achievement, often with zero-tolerance behaviour policies. I would suggest that such policies have the potential to do more harm than good for young people with attachment issues."

I now explore more closely how the relationship between mothers and new-borns can also impact young people. During adolescence, young people are more likely to self-report lower life satisfaction than at any other period of their lives (Patalay & Fitzsimons, 2018). In section 3.7, I discuss the development of the social brain during adolescence. This supports the thinking that young people are likely to experience emotional instability and stress in a way they have never experienced before (Mónaco et al., 2019). This period also offers opportunities; as the social brain is developing, it has been discovered there may be flexibility within the brain, allowing the possibility for adolescents to learn how to manage their emotions (Mónaco et al., 2019).

Secure attachment, traditionally seen as being built on warm and caring parent-infant relationships, is developed through relationships exhibiting trust, communication and a lack of disaffection (Koehn & Kerns, 2018). The early connections between the primary caregiver and child are likely to impact the development of how an adolescent may view themselves and others as they grow up in their social environment (Bowlby, 1982). For a child who is

nurtured with trust and love by attentive parent(s), a secure attachment is something likely to endure into their later adolescent life; as a consequence, it is probable the child will develop a fundamental attitude of trust towards others (O'Connor et al., 2019). Furthermore, it is thought those who have benefited in this way will have more effective emotional and interpersonal skills, enabling them to develop positive relationships in adult life (Allen et al., 2018). The significance of this cannot be understated for my work, as my thesis specifically explores what support young people need to take responsibility for their own mental health. It is therefore pertinent to highlight that society needs to ensure that those who exhibit symptoms of insecure attachment, and have difficulties with their fluctuating emotions, should be supported to develop their emotional competencies. My research is designed in such a way that it promotes relationship to be beneficial to both parties. I believe that the potential bonds (Kennedy & Kennedy, 2004; Riley, 2009) that develop between the two groups of young people may help develop the emotional competencies that are so important to young people. I will explore this in more detail in the discussion chapter.

3.5.2 Neuroperson model

Emotional competencies are related to how an individual manages their emotions to function adequately in a given social setting (Petrides et al., 2016). This is about how one perceives, names, expresses and regulates their emotions (Mónaco et al., 2019). Also, these now being recognised as traits that may function as mediators between attachment and wellbeing (Sabri et al., 2015). Research suggests emotional competencies are seen as having a buffering effect against the negative influence of an individual having to maintain insecure attachment relationships. It is recognised that good emotional competency indicates subjective wellbeing (Di Fabio, 2016) and physical wellbeing, such as less somatic issues and lower levels of perceived stress (Matthews et al., 2017). Furthermore, it was found that those with secure attachments, based on trust and communication, were more able to talk about their emotions, meaning they understood how they felt and therefore had a better chance of being able to cope with this (Keaten et al., 2008). Balluerka et al. (2016) also suggest that introducing emotional competency training for adolescents may be a way of preventing the onset of wellbeing issues. It is interesting to note that there are also sociodemographic differences, as females seem able to perceive and express emotions more than males. In contrast, males can develop more effective strategies to regulate negative emotions than females (Martínez et al.,

2020). There is also a need for additional research about those identifying as non-binary or gender-fluid. The intention suggests, therefore, that we want young people to take responsibility for their mental health. If this is the case, we need to understand the existing knowledge available and use it to ensure that any targeted help is well-informed.

The Neuroperson approach is underpinned by the Three part socio-emotional mental skills as seen below (McNeil & Stuart, 2022).



Figure 5: Three Part socio-emotional mental skills

Error! Reference source not found. suggests behaviours are influenced by young people's mental skills and their neurobiological support systems. Schemas have a powerful influence on how young people feel and behave. They consist of non-verbal memories that, when triggered, can result in emotional responses. Schemas are deep-set and long-term memories which take time and perseverance to change (McNeil & Stuart, 2022). In contrast, beliefs can change relatively quickly and are therefore easier to impact upon. The power a person has to be aware of their thoughts and feelings, and consequently the ability to focus awareness at a specific moment whilst purposefully changing focus, is a crucial management skill (McNeil & Stuart, 2022). Schemas are directly related to attachment patterns; therefore, it is important to understand their influence on emotions and behaviours. In turn, as adults look to support young people, relationships between supporting adults and young people are essential. The emphasis in the Neuroperson model is on practices that engage young people's schema-developing awareness of feelings and challenge-seeking behaviour. These are known as broaden and build schemas, and they contrast narrow and constrain schemas.

McNeil & Stuart (2022) have developed an outcomes framework 2.1, which focuses on six domains of socio-emotional skills, described by them as life skills; once learnt, they are applied across the life course (Figure 6). The outcomes framework 2.1 is designed to inform informal youth provision; there may also be a place for it in schools that are seen as formal provision for young people. The aim is to assist practitioners as they, in turn, support young people in developing socio-emotional skills.

Emotion Management is the ability to be aware of and constructively handle both positive and challenging emotions;

Empathy is the ability to relate to others with acceptance, understanding, and sensitivity to their diverse perspectives and experiences;

Initiative is the ability to take action, sustain motivation, and persevere through challenge toward an identified goal;

Problem solving is the ability to plan, strategise, and implement complex tasks;

Teamwork is the ability to collaborate and coordinate action with others; and

Responsibility is the ability to reliably meet commitments and fulfil obligations of challenging roles.

Figure 6: Domains of socio-emotional skills and their contexts

The implications of the neuroperson model and the domains of socio-economic skills are therefore relevant to my research. Schools need to understand the neuroperson model so they can fully recognise how young people's emotional competencies can influence their behaviour. This is however a challenge on a number of levels. When SEAL was first introduced to schools, it was short-lived as other priorities squeezed emotional education out of the curriculum. As discussed in section 2.2.4, the neoliberal agenda prioritised exams, something that altered the school value structure. This also links to section 2.3 and the arguments about school purpose; what are schools for and who is responsible for young people's neurobiological functioning? I suggest that, like mental health, schools are not responsible for it but do have a part to play in it. It is also essential to recognise that dynamics between adults and young people in provision such as schools are key drivers in the socio-emotional learning (Peck, & Smith, 2020). Secondly the domains in Figure 6 need to be understood by adults in schools so they can be systematically incorporated into activities and, at times, the curriculum. Furthermore, the skills and competencies my research have exposed

the two groups of young people to align with the above-mentioned domains. The YRT quote supports this in section 5.10 as it is explained how, through the research process, uncertainty was replaced by confidence both in actions and feelings. This is something that I explore more fully in the discussion chapter.

3.6 Adolescence and brain development

3.6.1 Societal approaches to mental health: From biomedical to biopsychosocial

The introduction of the biopsychosocial model (Engel, 1977) as an alternative to the biomedical approach changed how mental health was understood. Engel argued that mental health should be looked at through three separate lenses: biological, psychological, and social, with each of the three different disciplines being examined collectively as the others impact each. This approach is seen as being inclusive of the patient as opposed to them being the passive victims of the biomedical approach. Again, this deficit approach considers the problem as the patient who will be put right by the medical profession.

The biopsychosocial approach is a good fit for diagnosing and addressing mental health (Babalola et al., 2018). Much research has linked how a person lives with the state of their mind. Marmot et al. (2020), in their review of health in the UK, discovered that health equity is a significant problem as deprivation is one of the greatest indicators relating to health; they bluntly stated, "The poorer the area, the worse the health" (Marmot et al., 2020 p.13). The report is clear in highlighting that throughout a child and young person's life they are at three times the risk of mental health issues if they live in poverty than if they are relatively well off. Inequalities at school are linked to poorer life-long outcomes, including income level, future employment quality and physical and mental health. Children excluded from school are ten times more likely to have mental health problems and four times more likely to live in poverty than their peers. Absence from work due to a young person's poor mental health is increasing; between 2009 and 2017 it grew from 7.2% to 9.6%. Marmot addresses other issues such as debt, food security, housing, heating poverty and poor community cohesion as contributors to an increase in young people suffering from anxiety and depression. The social part of biopsychosocial needs to be addressed by society if we are to support young people towards improved mental health outcomes. Unfortunately, we seem to be moving in the opposite direction in many ways. Since the universal credit introduction in 2010, 92% of

NHS trusts cited the new benefit arrangements as responsible for increased referrals for mental health issues (Marmot et al., 2020). By electing to work with the pupil premium cohort, I have attempted to address social aspects impacting them. In my experience, this group is disadvantaged by its circumstances and, as such, deserves every opportunity that can be given to help them improve their own lives and future.

3.6.2 Medical advances and their contribution to an understanding of mental health

The introduction of MRI has enabled the medical profession to get inside the brain (literally). It has allowed scientists to study how the different parts of the brain react under certain conditions; studies have also been carried out with other animals that supplement emerging evidence (Orben et al., 2020). Some argue that the brain undergoes continuous change throughout an individual's lifetime (Weedall, Wilson, & Wayte, 2019). However, it has also been recognised that there are specific life periods during which the brain undergoes substantial development, making it vulnerable to environmental impacts (Romer, Reyna, & Satterthwaite, 2017). The brain starts developing around the third gestational week and will continue to grow throughout the foetal period and beyond. Another important period for brain development is the post-natal pre-school period from birth to six (Weedall et al., 2019). This is a significant period in child development and a particularly sensitive time. It is now accepted that adolescence is also a period of brain maturation. Whilst this brings development opportunities, it is also a period of vulnerability for young people as their brains can be susceptible to mixed messages leading to erratic decision-making (Steinberg, 2014). From my school experience, I observed some young people who found it challenging to fit in. This is something that I explore in more detail in section 3.7.1.

The earlier quotes from Socrates and Aristotle are words which would not be out of place in schools today. As highlighted earlier, these negative attributes are accepted as part of the maturation cycle. However, what is different today is that, due to the advances in neuroscience and technology, we are more able to understand what is happening in this phase of life and, more importantly, how this can impact an individual's behaviour. Through this lens, I will explore what is happening and how this is relevant to my research.

Adolescence results from the onset of two major biological events; the release of pubertal hormones and the consequent change in both the structure and function of the brain

(Fuhrmann et al., 2015). As recognised above, this time of sensitivity within the brain can lead to some individuals behaving in a risky or challenging way. This can result in adults developing negative belief systems related to young people in general, or to specific individuals. Adolescence provides opportunities to enhance life experiences; therefore, we need an understanding of what influences this.

One of the striking differences between adolescence and childhood is the switch in focus away from family and towards friends (Larson & Richards, 1991; Blakemore, 2018). This push for independence is part of the transformative process, and enables a child to grow into an adult. This draw towards spending more time with friends is part of the transformation in adolescence; most young people will start to assert their authority about whom they want to be with and where they want to go. In turn, young people are often more likely to make choices influenced by their peers that may make them more vulnerable to risk. Furthermore, MRI has discovered that different parts of the brain control distinctive functions. There is a pleasure part of the brain sensitive to rewards (the nucleus accumbens) as well as a regulatory section of the brain (frontal cortex) (Crone, 2017); as we mature these parts of the brain also develop. However, the frontal cortex takes longer to develop, and many adolescents will behave in an unregulated manner. Their desire to please their peers and fit into the group means they may, on occasion, behave in a risky way even though they know the dangers; their need to fit in overrides the regulation from their immature frontal cortex(Steinberg et al., 2008).

3.7 Applying brain development research to this thesis

3.7.1 Adolescence and social cognition

Social cognition is the ability to make sense of the world through processing signals from others (Blakemore & Mills, 2014), and was previously thought to have developed and matured during childhood. However, more recent studies have shown that this continues through adolescence, with activity in the brain concentrated in a network that is sometimes known as the social brain (Blakemore, 2018). This is significant because peer relationships are now thought to influence social decision-making. Increasing self-awareness during adolescence is considered to have implications for integrating self-judgement and peer

evaluations (Blakemore & Mills, 2014). This links directly to a primary need of humans to be part of a social group.

As we have already seen, adolescence is a time for separation from family with a move towards peers. This is when adolescents spend most of their time with peers and learn the rules of the group; by trial and error, they go through a social process of learning how to interact with others and fit in (Crone, 2017). Rejection from one's peers around this period of brain sensitivity can be particularly painful and can result in depressive symptoms (Rigby, 2000). Rejection can also result in poor school attendance and all the associated issues. Furthermore, the use of MRI has found that the reaction to social rejection is most significant for those who are anxiously attached (DeWall et al., 2012), have low self-esteem (Onoda et al., 2010) or have a history of abusive behaviour (van Harmelen et al., 2014). Some adolescents are so sensitive to rejection that they encounter social anxiety to the extent of experiencing depressive feelings and psychosomatic issues, such as stomach aches and headaches. The result can be that they remove themselves from the equation by not attending school. In this way, it could be viewed that they are no longer able to function normally in society (Crone, 2017) by saving themselves from the hurt they feel when rejected. The implication of this research for schools looking to support young people and to developing meaningful mental health strategies is self-evident:

- Be aware of who these young people are.
- Schools need to consider developing targeted intervention strategies.
- Train staff to recognise the warning signs.

Just as rejection causes pain, acceptance can result in feelings of pleasure or reward. In experiments, it was found that the sense of fairness also caused feelings of pleasure (Tabibnia et al., 2008). Furthermore, it is suggested that cooperation is rewarding and results in positive emotions. Therefore, through this research, I must develop a process that encourages a teambuilding approach on multiple levels. This is something that I explore further in the findings/discussion chapter.

3.7.2 Adolescence and self-concept

The development of self is about the awareness of becoming an individual with an identity, and starts around 12 months (Crone, 2017). However, the real advancement in an individual's concept of self occurs during adolescence, as an ability to integrate the views of others into one's perspective of self starts to develop (Selman, 1980; Garber, Frankel, & Herrington, 2016). Self-concept comprises several components, such as the distinction between selfknowledge and self-esteem (Harter, 2012). Self-knowledge is about an individual's personality traits, how one judges one's ability to do something and how one sees others judge that ability. Self-esteem, however, is about how individuals value themselves in relation to self-knowledge. Chapter 4 focuses on how agency, identity, power and capital all impact upon the ability, or lack of ability, young people have to take responsibility for their mental health. How self-concept develops during adolescence is also crucial to this area, as it is linked directly to identity. As I discuss in section 4.10, the social position an individual finds themself in can impact self-esteem. In turn, numerous studies are showing low selfesteem is linked to anxiety (Beck et al., 2001; Muris et al., 2003), depression (Mann et al., 2004) and, in some cases, externalising behaviours such as aggression (Donnellan et al., 2005). This is pertinent as my research is designed to address some potential negative school influences on the young people involved, whilst simultaneously using the research to address issues associated with capital, power, identity, and agency.

Pubertal hormones are released during adolescence, which can lead to mood fluctuations in the same period of development in self-concept. Mood swings coincide with changes in self-esteem that peak during early adolescence. This is when young people worry about how others perceive them and are more likely to experience anxiety and tension, something they can remain sensitive to throughout their lives (Steinberg, 2008; Blakemore, 2018). Conversely, those young people who have developed a more complex self-concept take a rational perspective on aspects of their life beneficial to their wellbeing (Crone, 2017). Furthermore, it has been found that having a more developed self-concept enables adolescents to differentiate between their true self and the self they fear; studies have often found that youngsters in the criminal justice system are abundantly clear about whom they do not want to become (criminals) but have no clear sense of whom they wish to be (Oyserman & Markus, 1995; Brewer, 2017). More recent research has linked self-concept with other constructs such as mastery, defined as how well an individual is under control of their life chances (Turner et al., 2017). Mastery has been associated with concepts such as; locus of control, personal control, self-efficacy and fatalism, as these all incorporate the idea of

individual agency (Turner et al., 2017). Whilst there is a possibility that poor self-concept may be linked to a lack of aspiration for some young people, there is also evidence that the discourse of aspiration is contextual rather than individual (Harrison & Waller, 2018).

3.7.3 Social media and the adolescent mental health debate

As discussed above, a crucial area for adolescents lies around whether they feel they are an accepted part of a group. Acceptance or rejection can make all the difference to them. It could be the difference between them withdrawing from society, potentially developing mental health problems, and leading a relatively normal life (Guyer et al., 2014). Whilst beyond the scope of my thesis, social media has a strong influence on young people and I have, therefore, decided to explore this further. We all live in a media-saturated world; however, it is the most recent generations who have grown up knowing little else (Crone & Konijn, 2018). As any parent or teacher can attest, young people communicate via social media platforms. It also allows young people to share ideas and opinions (Crone & Konijn, 2018). This is where young people stay connected and check out their status with their peers; they monitor the feedback they receive from their friends. Our understanding to date suggests it is likely this plays a role in adolescent development (Wartella et al., 2016; Donelle et al., 2021). Society's approach to social media is complex; on the one hand, it is used as an advertising and marketing tool (Siddiqui, & Singh, 2016) that targets individuals and encourages use. On the other, society is also quick to criticise young people for being obsessed with their phones as they are seen to be addictive (Ignat & Galatiu, 2022). To add to the complications, they are also seen as a valuable asset in targeting support issues such as mental health (Ridout & Campbell, 2018). This debate is carried over into schools where the use of phones divides opinion (Gajdics & Jagodics, 2021).

Earlier in this chapter, I highlighted how MRI has transformed neuroscience development and how the social brain undergoes significant change during the sensitive brain period of adolescence (Crone & Konijn, 2018). How young people use social media is therefore significant in this regard as it amplifies social influence. The evidence suggests a similarity between online and offline social experience and the adolescent brain development (Crone & Konijn, 2018). What is concerning is that the habits of young people, who have between 6-9 hrs per day of screen time (excluding school work) (Rideout & Robb, 2019), are developing to the extent that their online social experience is a more intense experience than ever it was;

their exposure to rejection and the negative consequences associated with this is also increased (Crone & Konijn, 2018). Social acceptance online also produces very similar results as face-to-face acceptance, with activity seen in the pleasure or reward centres of the brain (Crone & Konijn, 2018); research also suggests this type of social acceptance may be responsible for the mitigation of depressive symptoms (Nesi & Prinstein, 2015).

The influence of peers online is also seen to be significant, and it is found that both adults and young people will adjust their input to fit in with their friends. However, this is more significant for adolescents, due to their brains going through a sensitive developmental period (Crone & Konijn, 2018). Interestingly, young people were also found to be susceptible to peer influence regarding body image, suggesting social media can influence how adolescents view themselves and others (Crone & Konijn, 2018).

Social media is part of the day-to-day lives of most adolescents in Western society and beyond. It impacts how they live, and the negatives of this are often highlighted at the expense of any positives. As with many topics, the position on adolescent use of social media and mobile devices is one that divides opinion as to whether it is a benefit to them or not (Donelle et al., 2021). For example, numerous schools throughout the country have taken a deficit approach to mobile phones, and banned their use on the premises as they are seen as a distraction and encouraging of cyberbullying and sexting. However, I take an asset approach and believe using social media and mobile devices in schools present opportunities. This modern technology is here to stay, and its use will only increase. I suggest that we seize the opportunity to educate young people about how we get the best from this union with mobile technology. Schools have cohorts of young people with a powerful computer in their pockets that could support their education and learning. Beyond this, I suspect there are opportunities to harness their power positively, and enable a contribution to young people's well-being. I suggest engaging with and educating young people about social media, which should be seen as an imperative for school.

3.8 Chapter summary

This chapter has explored the key areas of mental health, adolescence and the relevant theory that links to these topic areas. I have framed them in this research's context and viewed them from both societal and educational perspectives. The pandemic, which included periods of

lockdown, anxiety and uncertainty, brought a greater focus on mental health for young people. This has come at a time when young people's mental health services are under increasing pressure, and schools are encouraged to take a more significant role in supporting students. Medical advances in brain research mean that society's understanding of mental health is developing at a greater rate now than ever before. This thesis intends to learn from the latest research and apply it within the context of mental health support in schools.

Chapter 4: The impact of agency, power, identity and capital for young people taking responsibility

4.1 Introduction

This chapter looks at young people's challenges as we explore how power is central to them and their ability to take responsibility. The mental health strategy I designed at William Howard School was about how young people could take responsibility for their mental health. It recognised a need to provide a supportive school environment where young people would feel safe exploring mental health issues (McPartlan, 2019a). I believe that many of the problems around agency (Houlders, Bortolotti, & Broome, 2021), power (Durkheim, 1982), identity (Cooley, 1902; Onu et al., 2016), and capital (Bourdieu, 1977, 1986, 2011) that exist in wider society are replicated in UK schools. Therefore, there is a need to try and understand how these issues may impact young people and their ability to take responsibility. By understanding the potential problems young people may be experiencing, it should be easier to address them. Presuming we can know what young people need to take responsibility for their mental health, it is unrealistic to expect all of them to engage with these concepts. It is, therefore, incumbent upon me as an educator to explore and understand the causes of such issues; only then will I stand a chance of reducing or mediating the impact.

Whilst society and schools may want young people to have agency in personal mental health, the concept is hindered by other factors. Identity plays an ever-increasing role in young people's lives today. It is entwined with the adult's perceptions of young people, which may, in turn, impact the competencies of these young people. The inequalities in society today highlight that, socio-economically, young people cannot be categorised as a single homogenous group; there are rich and poor, advantaged and disadvantaged. Power is also an area worthy of exploration. However, we live in an adult world, which influences how young people are treated, and this impacts their behaviour. How these disparities affect different groups of young people is an important area to investigate and, in this chapter, I explore how some young people are less likely to be able to engage than others. This is a complex area and is wrapped up in the debate around the structures and agency (Alderson & Yoshida, 2016), something that I touch on below. Finally, I take a closer look at class and education through the eyes of Bourdieu (1973, 1974,1977, 1986, 1988, 1990, 2011) and his
how marginalised young people behave in schools, and also help provide solutions as we endeavour to improve their school experience.

Having experienced schools, either as a student or teacher, for the vast majority of my life, I can attest that they are institutions dominated by power. As a young person in school, I experienced power through violence via corporal punishment. As a teacher, I saw hierarchical power influence staff and young people in various ways. Foucault claims power is an essential element in the development of individuals as they are constituted through social interactions, all of which are power-laden (Saldana, 2013). Their interactions in school, therefore, shape a young person's identity. Foucault explores the role power plays in forming an individual's agency that develops through interactions within the family and school (Saldana, 2013). Furthermore, young people are subject to dangerous strategic power relations within social institutions, including schools, such as sexism, class oppression, and adultism. Power is at the core of much of Bourdieu's work (Power, 1999), which has valuable contributions to make as I explore power in schools and how it impacts young people.

This chapter will centre on the role of power in schools, how it has impacted identity, agency and capital, and how it has contributed to my research.



Figure 7: How power influences young people's agency, identity and capital

4.2 Power, young people and schools

Power relationships between adults and young people shape many lives within schools and wider society. This section explores the power relationship both from a school and societal perspective. In writing below, I address the power issues between myself and young people involved in school research. This writing explores more general relationships between young people and adults (including staff working in schools), and how this impacts young people to take responsibility.

In the 19th Century, sociologists were altering perspectives of society by suggesting individuals were created by their environment and society (Durkheim, 1982). Described as functionalism or structural functionalism, Durkheim highlighted power as being critical to how society functions within its unique structures. Positioning power centrally in society enabled systematic thinking about how power and society interact. However, these views had limitations, as the state was overemphasised at the cost of the population, who were seen as objects. Durkheim argued from a functionalist perspective that society creates roles for individuals, and the interplay of these roles enables a society to function, with the noncompliance seen on the part of deviants. An opposing view was developed by Parsons (1937). Whilst he agreed with Durkheim that structures were central to societal processes, he also argued that the status quo within society was maintained by individuals accepting their prescribed roles. He believed that as actors we choose our societal roles, and the role of the elite was to uphold society, as the lower classes were not intelligent enough to do this. Whilst only seven per cent of the British population attend private schools, the vast majority of top jobs come from this seven per cent: 71% of barristers, 71% of British army officers, 61% of senior doctors and 64% of the government cabinet are all privately educated (Mohamed, 2020). As demonstrated, the class problems that fuelled our past are still with us today. This is exemplified in what we have seen from two of our last five prime ministers, David Cameron and Boris Johnson. They are products of a privileged upbringing that included a private education at Eton, somewhere that has provided this country with 20 of our prime ministers (Overton, 2021). I reflect in an extract from my diary below:

"There is also an uncomfortable reality for me, a privileged, well-educated, middle-class white man in a position of power in the school. Having left teaching, I am now researching school mental health, particularly about young people from poorer sections of society. This

area is linked to relational ethics, something I explore in detail in chapter 7. Within this context, it could be argued that whilst my intentions may be laudable, I am as guilty as those I castigate above as I am from a privileged section of society trying to further myself by 'using' others. My response is that I understand and accept my failings, which is why I have ensured my research is conducted from a young person's point of view and is founded upon values and respect (Vervliet et al., 2015). As discussed in my methodology chapter (5), I have developed this as YPAR to ensure that it is seen through the eyes of young people and not from the perspective of a privileged, white, middle-class man."

Parsons's (1937) view that people accept or choose their roles and ensure the smooth running of society is somewhat idealistic. For many young people, the choices they can exercise may be narrowed by life circumstances. The example of Steph (a pseudonym) from my school pastoral work illustrates why Parson's assumption that all in society are in a position to make a choice does not stand up to scrutiny.

"Steph was a student of mine who ended up permanently excluded from school. Since then, she has become addicted to heroin, been the victim of domestic abuse and hospitalised on several occasions. Her daughter is on the child protection register is now at school. She has witnessed fighting at home and the death of her mum's friend from an overdose. She has been arrested for shoplifting and has started experimenting with drugs."

Steph's daughter has seriously restricted choice regarding her future direction. As it is, her life is structured around disadvantage and chaos. These are, Durkheim might say, the social facts of this young person's life. In a larger sense, structures both develop a society's culture and are created by society; they are factors controlling individuals, including norms, rules, laws, and discourses; they are what is accepted as the way things are (Maynard & Stuart, 2017).

As I demonstrate later in the chapter, Bourdieu's (1977) theories of habitus and capital are about social inequalities and how power can help reproduce inequality. All staff in school can be described as power-laden. Compared with the study participants, most staff are from a relatively privileged background gaining valuable social, cultural and economic capital from their advanced education. It is, therefore, more likely that young people will perceive them as the ones who dominate; consequently, some young people may also take on the role as the ones who are dominated. Some see power as a negative force and controlling factor, while Foucault (1991) regards it also as enabling within society (Gaventa, 2003). Foucault (1991), believing that language around power needed to change and arguing this should not always be expressed as 'excludes', 'represses', 'censors', 'masks', or 'conceals'. He also believed power transcends politics and is embedded within society. This has been my experience in schools where power is also embedded. Whilst some young people see only a school's negative power dynamic, others choose to become empowered within such structures. I observed the following when I was on the school's leadership team:

"As a member of the school's leadership team, one of my roles included running the student council. Every year we asked for new volunteers to stand, and we had elections for young people who wanted to stand. However, it was predominantly the same young people every year. They were motivated young people from privileged backgrounds. We tried expanding the pool of young people to include a more representative and diverse collection of young people but with little success."

Gramsci (1971), and Bourdieu (2000) had similar views about the ruling classes and their manipulation of the masses. Gramsci introduced cultural hegemony to explain how the powerful ruling classes imposed on society to protect the status quo and thus retain power. He believed a struggle is created when revolutionaries advance alternatives under the banner of social justice. My intention with the research is to develop an alternative to the cultural hegemony created in the school. Young people involved in the research, both the YRT and the participants, have been enabled to contribute to how the school is run. The aim is, therefore, to shift the power base in favour of young people, empower them, improve their sense of self, and cultural and social capital, and help develop their agency to act.

This research is an exercise that supports an inclusive approach for some of the most vulnerable young people, many of whom do not fit in to the middle-class hegemony of the school. Teacher-student relationships are often seen as interpersonal (Frymier & Houser, 2000), and research has shown teachers use social influence to persuade young people to be obedient, comply (Richmond & McCroskey, 1984) and learn (Richmond, 1990). However, there is evidence that young people also apply social influence, thus developing the power to sway teachers (French, & Raven, 1959; Golish, 1999; Golish & Olson, 2000). The somewhat complex boundaries between social influence and power can get blurred; teachers may use

both, but young people generally lack legitimate power and rely on social influence to get their way. The question then arises as to how this impacts the power dynamics between young people and teachers that influence relationships.

4.3 How power impacts relationships

There is a broader question about relationships between young people and adults across different settings. This is exemplified by the relationships that visiting youth workers developed with young people, compared with those between teachers and young people. The power relationships available from youth workers and school staff differ significantly; as such, the response from young people reflected this. Young people are required by law to attend school; it is compulsory and, in my experience, some young people resent this. Also, school staff are the ones who often have to enforce this law. In contrast, the relationship between a young person and a youth worker is voluntary. Young people cannot choose whether they have a relationship with school staff because of the compulsory nature of school. By engaging with a youth worker, young people are making a positive choice that will likely benefit the relationship. This further supports my rationale for ensuring this work was directly influenced by young people and completed as YPAR. In this way, I ensured the research was robust and from a young person's perspective. French and Raven (1959) suggested five relational bases how teachers (and I would imagine other adults in society) may exert influence through power that can then be used for pro-social and anti-social effects. Pro-social power includes reward, and expert and referent power, and anti-social power includes coercive and legitimate authority. Relating this theory to my research is important as the dynamic it created contradicts much of the above. In relation to pro-social power, the rewards young people get are both intrinsic and extrinsic. However, in discussion with the YRT, their motivation for volunteering was primarily based on their interest in mental health and helping other young people. They were, therefore, likely to benefit through their own experience rather than purely extrinsic factors. The research was designed as a collaborative exercise. I deliberately broke down barriers by positioning young people as experts in their own lives (McPartlan, 2021). Referent power was virtually inconsequential as young people's actions were primarily motivated by a belief in what they were doing, rather than what an outsider told them to do. I believe that French and Raven's (1959) anti-social power was largely irrelevant as my research was designed in a way that disempowered me

and empowered young people; I have continually encouraged young people to do what they believe to be the right thing (McPartlan et al., 2021).

I have seen shifts in society's attitudes to power that mirror the discipline structures in schools. This ranges from being a pupil who received corporal punishment (possibly the ultimate exertion of school power) to being a senior leader in a progressive, pro-social leaning school. As attitudes have changed, anti-social use of power has been discouraged, as alternatives to corporal punishment have been suggested (Maurer, 1984). Schools moved towards supporting young people, as opposed to punishing them. Initiatives such as social and emotional learning (Department for Children Schools and Families (DCSF, 2007)) focussing on child wellbeing were introduced as an alternative to, or alongside, sanction-led approaches. The introduction of restorative approaches also helped young people to understand how their behaviour impacts others (Hopkins, 2002). William Howard School adopted this approach to shift the school's power relationship dynamics. It introduced an element of fairness to disciplinary procedures by ensuring incidents and their impact were understood by all. Crucially, it also gave young people a voice and shifted the teacherdominated power dynamic. This approach was also about both parties taking responsibility for their part in the incident, something that individual young people and staff found difficult to do on occasion. It would be naive to imagine these changes were appreciated by all young people, as some still rejected the rules and sanctions imposed.

The restorative approaches also acknowledged that teachers and young people share power and cooperate (Devine, 2003). Whilst a teacher's power over young people is often implicit and invisible (Bernstein, 1977), it can often shift and change; power-sharing and conflict are not mutually exclusive as teachers and young people continually create and modify their relationships (Woods, 1980; 2012).

So far in this writing, I have deliberately focused on negative young people/teacher power relationships in order to highlight the largely negative connotations of power in school. However, it is also important to acknowledge that students expect to have rules (Kim, 1998; Thornberg, 2008) and they have confidence both in the rules and in the teachers (Cullingford, 1988; Thornberg, 2008). Furthermore, research suggests young people judge staff in terms of honesty and are mainly only critical of unfair treatment (Gorard, 2012); in my experience, the majority of young people/teacher relationships in schools are positive and supportive.

4.4 Agency and how it is impacted by power relationships

Broadly speaking, agency concerns the ability of an individual to shape his/her own life by making meaningful interventions to bring about change (Houlders et al., 2021). Being an agent relates to intentionally making things happen so that, over time, an individual can adapt, develop and renew (Bandura, 2001; 2012). The core elements of agency include intentionality, forethought, self-reactiveness and self-reflectiveness (Bandura, 2001; 2012). My concern arose as I tried to apply the principles of agency to this research as I was less than confident that all young people had the required agentic competencies to be aware of their mental health, let alone seek support for it. The concept of agency, and particularly that of young people's agency, is a complex one as there are those who believe that it is not something that individuals possess (Oswell, 2013; Horgan et al., 2017). Oswell (2013) suggests that a young person's agency only exists in the context of their relationships with others. Earlier in this chapter, I focused on power and relationships relating to adults and young people in schools. In the context of agency as a relational concept, the link between the two is clear, young people in schools who are subject to non-pro-social power by adults are likely to have a diminished sense of agency. Agency is something that is created through a number of different factors, including family circumstances and upbringing, as well as interactions within social institutions such as schools. Young people do not choose their family (Archer, 1995), moreover "they are context-dependant relational beings" (Horgan et al., 2017 p.276) who are impacted by the environments in which they live. As I will demonstrate, the circumstances in which children and young people live differ and, in turn, so too will their agency to act. The importance of supporting children and young people to develop their sense of agency cannot be underestimated (Maynard & Stuart, 2017).

Before looking specifically at agency and young people, it is prudent to take a step back and explore a number of theoretical aspects relevant to my work. The role that social structures take in a young person's life is impactful. The family unit, the schools, the local community, and involvement with local sports clubs or local churches all have a bearing on how an individual develops. For many years academics, including Parsons (1937), Weber (1978), Durkheim (1982), Bourdieu (1990), King (2010) and Lamsal (2012), have debated social structure, and more latterly the relationship that social structure has with how individuals act, thus informing their sense of agency.

Giddens (1979) developed his theory of structuration describing agency and structures as two sides of the same coin (Marsh, 2010). Simply put this posits a duality whereby individuals within a group act and, in so doing, create the rules and structures for that group. However, the other side of the coin is how the rules and structures of the group also impact individuals, empowering and/or limiting their actions (Lamsal, 2012). Archer (1995) proposed morphogenesis as an alternative view to such relationships. She suggests individuals and social structures are a dualism rather than a duality, and are independent of each other; meaning they are intertwined but distinct. Whilst critiques have highlighted disagreements regarding the detail in their work, there is a fundamental agreement that human agency is interlinked with societal structures (King, 2010). Whilst the debate around the similarities and differences between the two approaches is interesting philosophically, the pertinent issues for my work suggest agency and structure are linked. Schools, as social structures, are powerladen institutions and the staff/young person relationship is at the centre of this. It is therefore important to recognise the impact school values may have on school character and the agency that young people display. I, therefore, suggest where schools adopt a non-pro-social powerbased approach, some young people are less likely to be able to develop their sense of agency as the staff/young person relationships will be based on coercive or legitimate power. This is an area beyond the scope of this thesis which requires further investigation.

Schools, as social structures, play a major part in a young person's life, impact the agency they have and, ultimately, how they live their lives. Morphogenesis also helps explain how differing school value sets can impact upon an individual's sense of agency, as Archer (1995) takes a sociocultural view where context is all important. This can be seen within society and schools. As an example, schools with a zero-tolerance policy on behaviour will elicit a different response than schools with restorative-based behaviour policies (Nassem, 2019). Young people's agency will be impacted by the different structures. However, it is not just young people. Staff, parents, and governors will all also respond in different ways depending on the school's value base. Young people not only have to navigate a more or less harsh set of rules, but those rules/structures send a subliminal message to the school community which in turn impacts upon agency and behaviour. This subsequently, be it acceptable or not, will then exert influence on the structures.

4.5 Sense of self

Agency, and particularly epistemic agency, is linked with a sense of self (Houlders et al., 2021). Epistemic agency is the ability and motivation to refine and alter one's belief-forming methods and practices. To fully explore the complexity of this phenomenon is well beyond the bounds of this thesis. However, an overview will help situate young people's ability to control their own behaviours. The following elements are some of those that make up a sense of self and include: memory; relationships; bodily awareness and affect (Houlders et al., 2021). As someone who has spent his life working with young people, I have seen first-hand how a sense of self can be enhanced or diminished due to circumstances that have been thrust upon them. Relationships with primary caregivers or friends, personal events and traumatic life events are just a few of the examples of how young people's sense of self can be influenced by outside agents (Houlders et al., 2021). In relation to this research, school experiences may also be included within this list of events for some young people. In addition, there is also consensus in research that autobiographical narratives are linked to feelings of selfhood (Bruner, 1991; Dennett, 1992; Huttunen & Kakkori, 2002; Stenberg, 2011). Throughout my teaching career, I saw how individuals from different backgrounds experienced life differently, impacting their sense of self. This was often communicated through a personal narrative that was also affected by sociological structures, such as schools and interpersonal relationships, and these events and interactions often framed their lives (Houlders et al., 2021). Those young people who live in stable family units, who have enough money coming in for food, clothes, heating, and leisure activities, are likely to have a more stable sense of self than less advantaged young people. What is more, there are those that believe experiences in these formative teenage years are likely to shape individual identity (Addis & Tippett, 2010).

The equalities literacy framework developed by Stuart et al. (2019) exemplifies this. It draws together five elements: pre-existing context, lived experience, positioning of others, technologies of oppression and positioning of self. It also suggests these all contribute to the sixth element, impact and trajectory. Using this as a lens through which to observe young people's lives allows us to see their lives in context. An individual's self-narration can be seen in a general sense; it can be via oral or written communication, and does not need to be part of a formal process (Gallagher, 2007). The feelings attached to some of our self-narration are significant in relation to one's sense of self as they can directly shape how we see

ourselves (Houlders et al., 2021). Furthermore, it is also important to recognise the potentially significant role others play in our sense of self (Gallagher, 2007). Parents, siblings, and teachers are among the people who can often influence and shape how young people see themselves.

As a school leader, I often witnessed reactive policies that took little of a young person's history into account. The boy from the chaotic home, whose first responsibility was to feed his little sister before taking her to school, but was given a detention for not bringing the correct equipment, is one of many examples I could cite. Once again, schools could benefit from using the equalities literacy framework to support all young people. The tension between the greater good of the school and the individual is enduring. This could not be exemplified in my career better than my experience with different headteachers. At one extreme, there were the headteachers who were determined to reduce the number of young people excluded from school; they invested in support services and would see exclusion as a last resort as they wanted to avoid the potential harm to an individual. The other extreme was a new headteacher who insisted that we exclude someone as soon as possible in the new year to send a message to the other young people. As such, it was the same school run by headteachers with differing values. The equalities literacy framework aims to position an individual and contextualise their life so that there is awareness of the limiters in their lives. Stuart et al. (2019) suggest schools need to be aware of young people's backgrounds, and professionals needs to be cognisant of how young people's lives have been shaped and how this may impact behaviours. This allows staff to use this information to benefit young people. However, as discussed earlier, in section 2.2.3, issues remain around the marketisation of schools, with very few resources to spend on this sort of work; schools are increasingly challenged as they attempt to support all young people. Hattie (2015) argues that this is an overly pessimistic view, and that there will always be issues beyond our control.

There is an ongoing struggle between traditionalists who advocate hard-line zero-tolerance behaviour policies and progressives who favour more educative ones that include restorative approaches. This is down to how society sees young people. If we recognise young people as complete human beings with the facility to participate fully within society, then we need to treat them in this way (Freeman, 1996; Spyrou, 2020). The danger is that if punitive sanctions are solely relied upon, lessons will not be learnt, and further damage may be done (Ttofi & Farrington, 2008). A young person's sense of self will impact their behaviours, so any policy

should consider this. The link between power, relationships and sense of self is played out in schools throughout the country. This requires consideration if we have expectations of young people being able to take responsibility.

4.6 The influence of power on epistemic agency in schools

The methods and practices that result from epistemic agency can be understood as doxastic disposition or belief-forming abilities that can be seen as one's proclivity to form true or false beliefs (Olson, 2015). The context of how a young person matures within society plays a significant part in the growth of their epistemic agency. However, schools also have a role to play in this epistemic development. School structures communicate different expectations regarding staff and young people's role in deciding which or whose knowledge is valuable, and how that knowledge should be constructed (Ko & Krist, 2019). A school's power structures characteristically ascribe epistemic agency and authority to staff, rather than young people, with little opportunity for young people to co-construct knowledge (Ko & Krist, 2019). A young person may therefore have the dual impacts of a less advantageous upbringing alongside attendance at a school where their epistemic agency development is at best discouraged, and at worst ignored. Epistemic agency should not be seen as a binary concept that one either has or does not have. It is a complex, ever-changing and relational construct that occurs through interaction with others (Ko & Krist, 2019). If there is an expectation that young people should take responsibility for their behaviour (including their mental health), schools need to understand their role in developing young people's epistemic agency.

It has also been suggested that one's autobiographical narrative can directly impact one's sense of agency (Ratcliffe & Broome, 2012; Ratcliffe, 2016; Tate, 2019). One's agency to act is directly linked to what an individual thinks or feels is possible (Houlders et al., 2021), and includes Bandura's (2001) core elements of agency. As a teacher, I observed young people having had their lives shattered by years of neglect at home, and withdrawing into themselves. Their autobiographical narrative possibly cast them as quiet and unassuming, thus meaning they find it challenging to fit in. These responses then impact the individual's ability to exercise epistemic agency or controlling actions, ultimately impacting events in their wider world (Haggard & Chambon, 2012).

To have a sense of agency is to believe that you are in control of your faculties; you initiate your actions and influence the direction in which your life goes (Tapal et al., 2017; Christensen et al., 2019). My experience of young people similar to Steph (pg. 62) is that they rarely feel they are in control, and often feel helpless. I suggest some young people in schools are in survival mode, as their lives are in turmoil due to circumstances beyond their control. It is doubtful they will have a complete sense of epistemic agency as they lack one or more of the areas of competency, authority, or credibility to comprehend their own experiences (Houlders et al., 2021). Therefore, this has profound implications for my research; I am asking how young people can take responsibility for their mental health in school or, more generally, within society. If, as Houlders et al. (2021) suggest, some need to gain the skills to comprehend their own experiences, then there is little chance they will be able to take responsibility for their mental health. If this is the case, schools must be aware of these assumptions to mitigate them. This reinforces the Stuart et al. (2019) equalities literacy framework for use in schools to get a clearer picture of those young people who would benefit from extra support.

Furthermore, and linked to this, we can learn from Maslow (1943) and his hierarchy of needs. A significant number of young people from disadvantaged backgrounds are surviving. Schools must be aware of these assumptions to mitigate them if this is the case at the lowest level of this hierarchy. In the cost-of-living crisis of 2022/23, many struggle for adequate diet, warmth, and even reasonable housing. Many young people do not have their psychological needs fulfilled, yet the set expectations and targets from many schools are based on self-actualisation; all young people are expected to achieve their full potential, whatever their circumstances (Maslow, 1943; McLeod, 2018).

Society and schools need to recognise that many young people are not in a position to take direct control of their lives because they lack epistemic agency; their life circumstances prevent this. It is, therefore, unlikely they will be able to take responsibility for their mental health. As such, they are subject to epistemic injustice, something I will explore in more detail in the following section.

4.7 Epistemic injustice

As I contextualise the theory for the benefit of my thesis, I am drawn towards the notion of epistemic injustice being the harm done to young people as epistemic subjects (who know, reason or question). Young people are harmed as their capacity to make sense of their own experiences or give knowledge to others is undermined (Fricker, 2007; Crichton et al., 2017); I see this happening in society and schools as they facilitate the building of structures that enable and limit this epistemic injustice.

Epistemic injustice can be divided into two substantive parts; distributive and discriminatory. Distributive injustice is about inequalities in the distribution of epistemic goods, such as information and education (Crichton et al., 2017). Reay (2017) argues that inequalities throughout our society are mirrored and reinforced by the UK's education system. The economically advantaged generally benefit from high-quality primary and secondary education, enabling their progression to the best-funded universities. The financially disadvantaged often struggle either in inadequate schools or in the lower sets of better schools. Their eventual educational outcomes are often defined early in their schooling when testing occurs as they start primary school. Those from economically advantaged groups start school at an advantage over those from economically-disadvantaged groups; the education gap is visible when children start school (Hattie, 2015). This often condemns these children to lower sets and a second-class education, often for the entirety of their school lives (Reay, 2017). Discriminatory injustice can be further divided into the more commonly recognised testimonial and hermeneutical injustice (Crichton et al., 2017). Testimonial injustice is the wronging of someone in their capacity as a speaker or knower (Byskov, 2021). It occurs when a person from a marginalised group suffers prejudice, diminishing their credibility.

In schools and broader society there are numerous examples of young people being on the end of such treatment. The term snowflake generation, now widely used in Western societies, is associated with an attitude that sees young people as emotionally weak. This attitude is further subsumed into society through the introduction of popular television characters who legitimise and possibly encourage the vilification of minorities. Vikki Pollard from *Little Britain* (Lipsey, Lowney, Posner, & Lucas, n.d.) and Lauren Cooper from *The Catherine Tate Show* (Anderson, & Gernon, n.d.) permit us to laugh at young people from poorer backgrounds, as described by Tyler (2013) in her work on social abjection theory. In schools, I saw numerous occasions where adult staff would dismiss information from students because they believed them to be untrustworthy, incompetent, or irrational; young people were

stereotyped by adults. I think most, if not all, young people could claim to be the victims of testimonial injustice. This is also linked to epistemic agency and power, as the adults in school were only in a position to discriminate because of their ability-laden status. It is reasonable, therefore, to assume that a young person subject to epistemic injustice through discrimination is unlikely to have trust in those who discriminate against them. I suggest that, where there is a lack of trust, young people are unlikely to take responsibility whether they can or not.

Hermeneutical injustice is a structural phenomenon that again happens to hermeneutically marginalised groups such as young people (Dieleman, 2012). As the term hermeneutical suggests, it is related to interpretation. In this case, the opportunities people have to interpret the social world and, in particular, their social experience (Dieleman, 2012). However, as explored above, some young people's life experiences do not allow them to develop these self-interpretative skills. There will therefore be social experiences where individuals do not have the interpretive skill that enables them to explain to others or even comprehend it themselves (Dieleman, 2012). Some young people will find this challenging when schools are tasked with supporting young people to take responsibility for their mental health. Considering poverty is a significant factor in young people's mental health (Wickham et al., 2017), we must acknowledge this as an issue to find solutions.

4.8 Overcoming epistemic injustice

Distributive epistemic injustice is a result of the epistemic structure of society and, as such, is a societal problem (Bai, 2020). Thus, is it necessary to tackle it at an international and national level. This research will play a role in raising awareness of such issues. However, the problems are far broader than what happens in one school. Distributive inequalities are a societal issue and something that is at the heart of this country's education system. It is, therefore, beyond the scope of this thesis. We can address discriminatory injustice through a mixture of education and professional training. Society as a whole, but schools in particular, need to be made aware that young people, as a sociological group, are discriminated against both hermeneutically and in terms of testimonial hearing. Fricker (in Dieleman 2012) writes about virtuous hearing. She describes how, in relation to testimonial injustice, individuals need to be made aware of any potential prejudice so they can neutralise its influence. This is

about neutralising the influence of possible prejudice that one may have about a marginalised group (Dieleman, 2012).

Stuart et al. (2019) suggest working with young people in need to improve their own equalities literacy and, therefore, start to break the cycle of social justice inequalities. This is something schools are in a position to be able to do by developing awareness training for staff related to hermeneutical injustice. Staff need to be aware of the characteristics of someone suffering hermeneutical injustice. Individuals may struggle to communicate their feelings as they may not understand what is happening to them. Fricker suggests sensitivity and care are required, which may include gentle open-ended questioning to help the young person explore their feelings so they can understand what is happening (Dieleman, 2012). I suggest a restorative questioning framework could be used to frame virtuous hearing. A person's ability to communicate with others is essential to an individual's social life and agential action (Crichton et al., 2017). It is, therefore, paramount that, as educators, we are not only aware of this, but are also fully trained to respond to it.

4.9 Identity, what it is and how it impacts upon young people and their ability to take responsibility

As someone investigating how young people can take responsibility for their own mental health, I needed to critically evaluate what was enabling or preventing this. The school participants in this research are volunteers from the cohort who qualify for pupil premium funding. In my experience, many from this cohort found school a challenge; this was often seen as either rebellious behaviours or in how they physically present, something that Bourdieu (2000) described as "bodily emotion". As a professional who spent many years supporting many from this cohort, I suggest their difficulties were a complex mixture of issues related to power and trust, sense of self, identity, and agency. By drawing from several ideas and theories related to identity, I explored what enables some, but not others, to engage. Within this, I also acknowledge that schools are the most important public institution in developing young people's identities (Reay, 2009).

In an attempt to understand selfhood and identity, numerous academic disciplines have conducted studies in this area. Psychoanalysis, through Freud (1856-1939), Erikson (1902-

1994) and Marcia (1864-2014), have explored adolescent development, each building on the former's work. Similarly, in sociology, Cooley (1864-1929), Parsons (1902-1979), Durkheim (1858-1917) and Mead (1863-1931) have all made contributions to theories related to self and identity (Adams & Marshall, 1996; Geldard, Geldard, & Yin Foo, 2020). Socialisation relates to how an individual successfully integrates within society and is also seen as essential to both an individual and society as a whole, (Damon, 1983; Blos, 1985; Hauser et al., 1991; Grotevant & Cooper 2009). The processes of development and socialisation are based on the conflicting association between the opposing elements of agency (related to uniqueness, individuation and separateness) and communion (being about connectedness and belonging) (Adams & Marshall, 1996; Geldard, Geldard, & Yin Foo, 2020). Both are required to create a positive self-image of someone who is significant and makes a difference (Rosenberg, 1985). Individuals require both belonging and distinctiveness, and must be able to develop positive relationships with others (Damon, 1983; Geldard, K, Geldard, D. & Yin Foo, 2020). Young people struggling to develop an affirmative self are more likely to be in danger of harm as they feel less secure about their place in society and are, therefore, more likely to engage in risk-taking behaviours (Silverberg, 2020). Furthermore, where self-identity is a problem, the development of meaningful relationships is likely to be hampered. These factors are directly linked, as young people who have life challenges and who may not fit in may also be the ones to find it difficult to seek help. Identity, therefore, requires further investigation.

Adams & Marshall (1996) propose identity as an ongoing process punctuated by sensitive points, often seen as life stages (adolescence, mid-life etc). In addition, they also suggest that in most Western societies, an individual selects a set of psychological and interpersonal goals. These are based upon feelings of uniqueness, agency, approval from others, belonging, equity, social responsibility and caring for others. This is articulated within society in different ways, depending upon one's circumstances. When young people are exposed to consistent adult values and expectations, they develop a positive sense of self with a purpose, direction, commitment, and identity, with perceptions of power and mastery giving them feelings of independence and control (Silverberg, 2020). In addition, they develop values consistent with individual and social responsibility for themselves and others.

In contrast, the outcomes are very different when young people find themselves in poverty, where conflict may be rife, and there are conflicting goals or expectations. What results is confusion, cynicism, a lack of direction and a poor sense of self (Ianni, 1989). Taken within

the context of my research, it is no surprise that the participants may find taking responsibility a challenge, as many of them live in poverty. If these young people have a weak sense of self and are challenged by identity issues, it is little wonder they lack confidence in a formal school setting. I believe this is a significant reason why many of them do not have the awareness or self-confidence to be cognisant of their own mental health, let alone take responsibility for it.

4.10 Social and learner identity in education

Social identity is another aspect to consider if we want to fully understand how some young people can engage whilst others cannot. Social identification is a cerebral process where social categorisation is internalised to become part of an individual's self-concept (Turner, 1982; Reynolds, 2009). Much of the work around social identity looks at how individuals define their identities in relation to their place within social groups (Islam, 2014). Social identity theory builds on these concepts to explore social categorisation and social comparison, and the impact that these may have on self-esteem, stereotypes, and discrimination.

Theories related to social groups can be used as a lens through which to filter the research. Of the eight social identifiers (race, gender, ethnicity, sexual orientation, faith traditions, social class, ability and age), because of the demographic of the participants, the most significant for this research are age and social class (Onu et al., 2016). Whilst social categorisation will usually commence in the home, there is also a belief that children begin to form self-concept and build their social identity through interactions with other groups in more formal settings, such as schools (Albarello, Crocetti, & Rubini, 2018). Jenkins (2004) suggests that as young people reach adolescence, their identity develops further as they are drawn towards their peers.

Schools are where young people mix, meet others from various backgrounds, and engage with differences. This makes schools important places in identity development (Reay, 2009). It is further complicated in schools by the intersection of social and learner identity, as young people construct themselves and others as a learner in relation to their peers and school staff (Reay, 2009). If I return to chapter 2, where I explore how school values have been developed, other lessons can also be learnt from this work. The overemphasis on school

assessment has resulted in the valorisation of ability as measured by tests and exams. This leads to young people internalising the messages given to them by the school system (Reay, 2009). Bernstein (2000) develops this further, claiming that there is also deliberate policy by the state to develop young people's identity through education. In section 2.2.4, I discuss how Thatcher (1995) intentionally replaced what she called the 'dependency culture' with the value of 'self'. This education policy direction was about changing a mindset. Society was seeing a reduction in state welfare, and education became about preparing young people for a culture of the market (Bernstein, 2000). This was an identity founded upon self-support, self-reliance and individualism, and is something that governments have continued to promote for the past 30 years or more (Reay, 2009). The education system was now at the centre of identity formation (Reay, 2009).

Cooley's (1902) looking glass theory helps explain how young people from these lower socioeconomic groups may self-identify. Cooley proposed that individuals imagine how they appear to others, what judgements others make, and how others feel about them. However, individuals are active within this role, and our self-image is shaped by others but only through the filter of our minds (Squirrell, 2017). Our perceptions can often be inaccurate as we cannot know what someone feels about us or how they might judge us. We are also selective in what we process; we care more about some things (people and their judgements) than others, and we use our looking-glass self to regulate and inform our response to others. Being constantly aware that others are observing us alters our behaviours and projects what we think others may want to see (Squirrell, 2017).

Many participants' identity and self-concept are a complex mix of these theories. Social identity highlights categorisation, comparison, and self-esteem, which are issues for many young people engaged in this study. As a teacher, I saw examples of their low self-esteem as they categorised themselves in extremely negative ways; academically weak, troublemakers or generally unworthy. I now understand what I saw as a lack of aspiration is a far more complex construct. These young people are only partially in control of their destinies. Yes, they can make choices, but these are often limited. I therefore suggest some young people have less capacity to take responsibility than others.

4.10.1 Viewing young people in schools through a lens of habitus and culture capital

Within the following section, I draw heavily on Bourdieu and Reay using their language of social class for clarity. However, I know that some believe this to be contested. One of the problems with education in the UK in 2022/23 is that it is a relic from the past, producing an antiquated hierarchical system that still benefits the most privileged in society (Reay, 2017). The existing inequalities are a constituent part of the processes within the system. As I exemplify in this section, all aspects of society contribute to the continuation and success of this grossly unfair status quo. The mostly upper-class young people privileged to go to private schools have generational support with an unwavering belief in their right to be at the top of the academic tree (Reay, 2017).

Bourdieu's work allows us to frame the agenda in terms of transformation rather than reproduction. By understanding what habitus means for the poorer in society, and by aiming to create school opportunities for them to develop cultural and possibly social capital, my work creates openings to alter the field in which they live (Mills, 2008b).

4.10.2 Habitus

Habitus develops within an individual from childhood as they grow into their family and surroundings. The repetitive nature of everyday life ensures these young people take on the beliefs, values and behaviours that impact their speech, dress and manners as a subconscious social class compass that directs them towards the acceptable norms within their group (Mills, 2008a). Therefore, the dispositions they exhibit reflect their upbringing. It is expected that someone from a working-class background may behave differently to someone from the middle or upper classes (Bourdieu, 1990; Thompson, 1991). Furthermore, some young people will attune to the school's values by the virtue of their upbringing. This is related to how someone may think and, more generally, to their cultural disposition (Grenfell, & James, 1998; Purdy et al., 2021). In relation to my own experience, whilst teaching, many individuals did not understand how to behave in and around the school when they first arrived. Some young people found the school experience extremely challenging; if they did not rebel and get excluded, they would often merge into the background, keep their heads down and get by with the minimum effort, often underachieving. Bourdieu & Passeron (1964; translated by and cited in Grenfell & James, 1998) also posited that those who fit their environments were more likely to succeed in school. This has implications for my research, as those young people, mainly from the lower socioeconomic groups and who find schools a

problematic place to be, are less likely to engage and trust adults, accept assistance, or go looking for help, as they are not comfortable in the school environment.

Within critiques of Bourdieu's work there is a discussion about whether his theory of habitus is about reproduction or transformation. Kenway & McLeod (2004) ask whether there is room within his theory for agentic improvisation or whether it is simply deterministic. Whilst some, such as Jenkins (2002) and Nash (1990), argue that his work is too deterministic. Whilst his work shows little evidence of transformation, there are others, such as (Reay, 2004), who argue that in his works such as The Weight of the World (1999), Bourdieu and his co-authors are searching for ways of transforming the lives of the vulnerable. Reay, David, & Ball (2005) suggest habitus can be both reproductive and transformative. This area cannot be divorced from agency and identity and, as such, habitus can shape but does not have to determine choices young people make in their lives (Mills, 2008b). Being responsible for behaviour and welfare in schools resulted in me observing many adverse, deterministic outcomes. Sometimes, I needed to remind myself that these negatives were a hazard of the job that temporarily blinded me to the many positive stories happening in school. I, therefore, also had to be conscious of another group of young people who could have followed similar negative routes. However, by being in a position to accept support, they employed their agency to make positive life choices. The habitus of one individual may be more transformative than that of another, and this may well enable one to take an opportunity whilst another cannot.

Furthermore, young people may see possibilities for action in one situation, and simultaneously prevent themselves from recognising other situations, as habitus merely sets boundaries in which agents are free to act (Codd, 1990; Mills, 2008b). This research exemplifies this, as both the participants and the YRT were volunteers. Two separate, and seemingly homogenous groups of young people, were given opportunities to become involved with this research. Whilst some came forward and volunteered, others decided they did not want to be involved; both groups were agents free to act and they all did, in either one way or another.

Another question is whether we, as professionals, should strive to transform or change young people. This is beyond our remit as, by doing so, we will be interfering with cultural issues beyond our understanding, something that could do more harm than good (Mills, 2008b).

Some would suggest that attempting to impact the working classes in this way is to risk pathologising them (Lawler, 1999; Friedman, 2014). Whilst it is difficult to argue against offering young people transformative opportunities, schools should be more concerned with providing educational experiences that create life-changing possibilities. This is particularly true for the marginalised, excluded or disadvantaged (Lingard et al., 2003). By offering the participants a position within this research I offered them an educational experience that would create possibilities for them to change their lives, even if only in a small way.

4.10.3 Culture capital

My experiences working in state schools leads me to agree with Bourdieu's premise that schools are places that reproduce both social and cultural capital (Mills, 2008b; Burger & Walk, 2016). Bourdieu's (1973, 1974) assertions were that it was the culture of the dominant classes that was embodied in schools. This is something that would also give those young people who could identify with this accepted culture an educational advantage (Mills, 2008b). Others, however, would need to adjust their behaviours to fit into the academic structure or stand up and fight against the system (hooks, 2017). The problem, however, is somewhat more entrenched than even this implies, as performance at school is now misrecognised as being about an individual's talent (measured by test scores), whilst social class is ignored. As demonstrated earlier in this chapter, there is a gap in the educational starting point for those from different backgrounds. This profoundly impacts both educational provision and outcomes (Mills, 2008b; Croizet et al., 2017). Bourdieu, and Passeron's (1990) cultural capital describes the codes, rules and ways of behaving that individuals take on from the lives they lead. However, Reay's (1998) work identified substantial differences between the cultural capital middle-class mothers generated for their children and the difficulties immigrant parents had in helping their children within our education system. The difficulties were down to the lack of cultural capital the incomers had, or, to be more accurate, how the cultural capital these immigrants did have was no longer valid within this their new environment (Fresnoza-Flot & Shinozaki, 2017). As such, educational consequences exist for those who possess cultural capital in the wrong currency (Gewirtz, Ball, & Bowe, 1995; Raffo et al., 2009). Our system, often including the schools, teachers and parents, unwittingly supports this social segregation that sees unfair educational advantage given to the privileged in society (Bourdieu & Passeron, 1990; Reay, 2017).

We are left with a system that enables young people to succeed based on class privilege rather than merit. Using my research as a tool to broaden the cultural capital we use in school, this work can be transformational (Mills, 2008b). By working and co-producing with young people as researchers in school, I am taking a more critical perspective and using the benefits of contemporary society to give them access to both cultural and social capital (Comber & Hill, 2000; Raffo et al., 2009). Those young people who have yet to accumulate the various forms of capital are likely to be the ones whose habitus does not enable them to fit in at school as well as others might. This suggests that capital plays a significant part in a young person's development. Therefore, I argue this adds to the complex factors influencing why some young people find it difficult to take responsibility for their mental health.

4.11 Chapter summary

This chapter situates the position of power in schools and highlights its importance within this research. The literature has demonstrated how it can impact young people's agency, identity, and capital. This is particularly true for those from the lower socio-economic groups, which includes the participants in this study. The links between power, agency, identity and capital are complex and, at times, being able to separate the impact of each one is a problematic process beyond the bounds of this thesis.

Chapter 5: Research methodology

5.1 Introduction

This chapter presents my philosophical position and rationale for the study. I investigate the social constructivist foundations for the research and how this has influenced my decision to adopt an approach synthesising action research (AR) and critical communicative methodologies (CCM). I explore how the context of the research, particularly collaborating with young people as co-researchers, has also been influenced by my social constructivist beliefs. I delve into the reasons behind my decision to incorporate young people in the research, and how I recruited both the YRT and the participants, before looking at how my partnership with the YRT influenced the research cycles and the methods used in the study. My positionality is briefly examined before I consider using both reflectivity and reflexivity in this research. The chapter is concluded with a section on how the analysis developed through the process of crystallisation as we collected data.

This research collaborated with young people in a school setting and explored the efficacy of the whole school mental health strategy to find improvements. I intended for this research to directly benefit the two groups of young people involved, the YRT and the participants. It was also about re-balancing social inequalities in school. My qualitative research is about understanding young people's subjective experiences in a school setting and investigating their unique perspectives (Stringer & Ortiz Aragon, 2021). Therefore, this research was not about the objective measurement of a school process for universal implementation, but about constructing a critical pedagogical approach to achieve knowledge democracy (Stuart, 2020).

Later in this chapter (section 5.2), I argue that this research is taken from a social constructivist standpoint; it is important to contextualise this within the broader social constructionism family. Confusion arises as the terms 'constructivism' and 'constructionism' are sometimes interchanged (Burr, 2015). However, constructivism has also been theorised as a branch of constructionism, where individuals see the world differently through their perspective, creating meaning from events (Burr, 2015). From this perspective, I approached the research by choosing an AR methodology as I sought to collaborate with young people. The collaboration was through a combination of CCM and YPAR, which I discuss later in this chapter.

As an ex-teacher, my researcher position felt vulnerable. I was returning to a school where I held power and was concerned this could influence both the young people and the authenticity of the findings. I, therefore, adapted learning from CCM that put us on an equal epistemological level and re-balanced the power dynamics, further enhancing knowledge democracy. The final part of my methodological philosophy, which I conceptualise in Figure 8 below, is the approach I took to data collection. The mixed-method data were from a variety of sources, but at its core were information collected by the YRT in their meetings with the participants. The YRT was given the freedom to develop their data collection strategies and were equal partners in the analysis of the research. I made a deliberate decision to further enhance knowledge democracy within the process; this area is explained in more detail in section 5.7 and the data collection philosophy is summarised in Figure 9.



Figure 8: Concentric methodology model

5.2 Social constructivism

My ontological position as a social constructivist aligns with my decision to conduct youth participative action research (YPAR) (to become youth participative dialogic action research (YPDAR)). By committing to empower and enable young people to work with me as corresearchers, I was morally invested in their wellbeing and welfare, something I saw as being enhanced by the research process (Hipolito-Delgado et al., 2022).

Social Constructivism owes its foundations to the work of many including, Piaget (1970, 1971), Dewey (n.d., 1938, 2007 [1916]) and Gergen (1985). It focuses on social interactions' role in developing reality, knowledge and learning. Some believe that for social constructivism to be active, only one pre-requisite is required: a social action must occur between two individuals (Teague, 2000; Safari, 2020). A transaction (sharing of prior knowledge) occurs between them, resulting in both participants leaving with knowledge gained. Culture is seen as important within social constructivism as individuals can influence transactions with their values and beliefs formed from the social background or environment they inhabit (Whiting, 2007).

Social constructivism is a value-laden philosophy as the knowledge that each individual creates is, by its very nature, influenced by the learning process through which it has travelled. An individual's life experience will therefore impact how they go through the learning process. Thus, my own life experience influenced the philosophy of this research. As a teacher for over 35 years, I was concerned with helping young people improve their lives. Through this research, I came to see that from a social constructivist standpoint, young people in school had different starting points due to the circumstances in which they were born. It was, therefore, important to me that I treated this as an exercise in social justice. I created a participatory learning environment that helped young people develop their agency, and thus improve their life chances (Brubaker et al., 2010).

Crethar et al. (2008) identified *equity, participation, harmony*, and *access* as four essential components or aspirations that they believed underpin socially-just qualitative research. My research was inclusive as it encompassed *equity* via the participants coming from the pupil premium cohort. This was also achieved as a *participative* project, as it included young people as co-researchers. The focus of the project was to ensure *access*, as it was about working with young people to identify the correct questions to make a difference and come

up with pertinent resolutions. *Harmony* was achieved by ensuring that the community benefited from the project as a whole (Lyons et al., 2013).

From my perspective, returning to my previous school, a social constructivist framework was the best fit to explore mental health strategy and the impact this was having on young people within the institution. My history within the school and my relationship with staff and students meant that the constructs that informed the development were central to my role in the research. As I set out in the prologue, the school where the research was conducted was a comprehensive academy with a mixed rural/urban catchment. All young people come with unique perspectives and, as such, have different social constructs, which schools need to address to provide equal opportunities for all. Taking a social constructivist approach enables an understanding of how the individuals within such organisations function because schools are fundamentally social structures; they are living communities and, as such, they are socially constructed.

Reality is developed through human activity; individuals within society fashion the properties of the world (Kim, 2000; 2006) and, from this perspective, the same can be said for schools. Individuals and groups from different backgrounds, cultures, and communities inhabit them. The basis of interactions of the individuals (both within and between the groups) is based upon a shared understanding of their environment, which influences their assumptions and should also lead to cooperative learning within the school (Lee, 2020). This intersubjectivity is fundamental to the processes that take place in the development of knowledge, which is constructed socially and culturally by the interactions between individuals and their environment.

Therefore, it was essential to design my research using these principles as a guide to ensure the findings were authentic, as my collaboration with young people creates them. We worked together to design the research; they worked with participants to collect data and took a leading role in analysing and disseminating the findings and recommendations. Young people involved in the research were engaged in social activities, ensuring the learning process they were involved in was meaningful (Kim, 2000; Hein, 2000).

The knowledge created was through a democratic process that placed young people, the school environment, society, and culture at the centre of the work and was thus indeed socially constructed.

Young people in school were a source of knowledge housed within their community (Wood, 2016); I was determined to ensure the creation of new knowledge came from this foundation. The term authentic is crucial to my research. Throughout this thesis, I use it to signal to the reader that any findings are from the insight of young people rather than seen through the researcher's lens. In section 5.8 I explore my position within the research and, as such, accept it is impossible to remove my lens from this research. However, I took on a reflexive responsibility where I continuously scrutinised my assumptions and conclusions, as well as checked them with the YRT (Dávila, 2014).

The formation of reality, knowledge and learning will alter within and between the specific groups that are part of this complex school community. Within schools, it is accepted that young people face numerous risk factors that may result in mental health difficulties (Bostwick & Glazzard, 2018). The risk factors vary; they include influences from within schools, such as high-risk testing, and wider society, such as socio-economic disadvantage (Bostwick & Glazzard, 2018). Any school mental health strategy needs to consider this; just as young people experience mental health in different ways, they will have different experiences of school strategies. The school's role in mental health is also pertinent at this point. It is accepted that pressure from assessments can create mental health issues for young people (Păduraru, 2019). Other problems, such as approaches to behaviour and the general approach to inclusivity, may all have a bearing on young people's mental health. Therefore, schools must have a good understanding of how they may be influencing mental health as they prepare a strategy to combat it. My methodological approach to the research was influenced by social constructivist thinking, and it is important to present my research principles based on this theory. They are not a tick list but served to guide me and the YRT as we developed the process together.

- Ensure that the processes of the research are founded upon a sharing of experience.
- All parties involved in the research are to be viewed as co-creators of knowledge.
- The researcher/YRT relationship is to be based upon support and assistance rather than direction and instruction.

- Tasks within the process to be collaborative and ensure inclusivity.
- The research outcomes and processes are to be viewed as equally important (Adams, 2006).

With the above principles laid out, I now intend to explore the place of CCM, AR, and YPAR within my research.

5.3 Critical communicative methodology (CCM)

After Jurgen Habermas gave a lecture addressing 'The intercultural discourse on human rights' during a 2001 conference in Barcelona, he was challenged by a lay non-academic about women's labour rights. Whilst some in the audience were contemptuous of the questioner, Habermas addressed the issue, calling it a brilliant and critical question. He applied his proposal of universal dialogue that states that all can offer arguments in whatever way they wish (Gómez et al., 2011). This dialogic approach is the basis of CCM, that at its core believes everyone has the critical competence to analyse their own lived world; this analysis is achieved through dialogue between people from differing backgrounds to ourselves (Gómez et al., 2011). This dialogic turn in society is directly relevant to the critical communicative approach I intended to take (Puigverta, Christoub, & Holfordc, 2012). Habermas' thinking influenced CCM, and the key to his thinking were his seven postulates, three of which are fundamental to my work. He believed that everyone could interact and communicate; he called this the 'universality of language and action'. The second postulate of relevance was the 'absence of interpretative hierarchy' that stated all interpretations coming from the research process are equally valid, regardless of the position of the person putting them forward. The final postulate is where the researcher and researched work on an 'equal epistemological level', each an expert in their area, be that academic or lived experience (Puigverta et al., 2012). These are areas I will return to in the discussion chapter.

Habermas (1987) suggests that we have moved into an age of dialogue as there has been a shift from instrumental rationality to communicative rationality, where people use their knowledge gained from lived experience, which is something that, by its very nature, is socially constructed. Such an approach aims to achieve accord rather than allowing power to be the leading force for change. As with the rest of society, which has seen this dialogic transformation, qualitative research has shifted away from traditional hierarchical research

relationships. A redressing of power imbalances within some research areas has been achieved (Råheim et al., 2016). As a result, scientific knowledge about our social world has increasingly come about through egalitarian dialogue (Gómez et al., 2011) and has produced a more democratic, socially useful and politically responsible knowledge (Denzin & Lincoln, 1998; 2000; 2017) that also has the potential to be transformative (Gómez et al., 2011; Tracy et al., 2015). Importantly, CCM states that social interactions build social situations and, as such, reality does not exist autonomously from the subject; only where researchers have intersubjective relationships with social actors, can objectivity be reached (Gómez et al., 2011). The premise within CCM is that everyone can contribute to knowledge construction, which is further enhanced where there is dialogue between people with differing cultural intelligence. The relationship between the researched and the researcher is vital, and the dialogue between the two enables a path to empirical truth (Gómez et al., 2011). This process, therefore, gives the researchers a deep insight into the lived experience of those they are collaborating with, and offers opportunities to transform people's lives. By developing a dialogic methodology, I followed the principles of CCM and this was instrumental in fostering a process with the potential to enable young people to be part of transformative research and, thus, change their own and others' lives. By accessing the participants' social constructions of reality, the YRT will then be able to construct their realities in school. The participants came from a cohort of young people who were some of the most economically disadvantaged; the YRT were a self-selecting group in a comprehensive school. Both groups shared their lived experience with a researcher through various forms of dialogue. I believed that by giving these young people the opportunity for rich learning experiences, I would enable them to improve their capabilities for life-long learning by facilitating their agency to choose and develop the things they value (Hi Kim, 2017).

The CCM dialogic approach aligns with my arguments in section 4.10.2. I agree with Reay, David, & Ball (2005), who suggest that whilst habitus can be reproductive, it can also be transformative. Freire (1970) distinguished between those who engage in dialogue and those who are anti-dialogic and want to impose their will on others. Habermas (1984) positioned the fight as between those with validity and power claims. Young people who live in poverty have been set as individuals who are deficient in an ability to choose and achieve (Bok, 2010). Furthermore, young people might adapt to their challenging circumstances through selfrestriction, limiting their ability to open up opportunities for them (Teschl & Comim, 2005). This can lead to young people lacking inspiration and aspirations for enriching life experiences (Hi Kim, 2017). By placing young people at the centre of my CCM research methodology, I have attempted to counter the deterministic hegemony I have highlighted previously. Social constructivism is about how human interactions enable knowledge creation, which I aimed to achieve by developing this research synthesising CCM with YPAR.

Following a CCM is about demonopolising the knowledge of experts (Beck, 1992) and is also attempting to ensure that the knowledge creation, whilst necessarily more complex, is more inclusive; it is likely to impact the lived reality of the social actors as they have had a dialogic collaboration with academics (Schütz & Luckmann, 1974). Young people have been described as "social actors and experts in their own lives" with perspectives that are important in their own right (Cowie & Khoo 2017 p. 234). Working in this way, with egalitarian dialogue between myself, the researcher, and these social actors, we aimed to better understand the complex nature of the inequalities that have impacted them.

5.4 Action research

5.4.1 Introduction

The action research method is congruent with the socially constructivist paradigm in that it is value-laden; it has social intent and a social or "sound moral purpose" (McNiff, & Whitehead, 2012; Groundwater-Smith et al., 2015, p. 142). My values were integral to this study's purpose and outcome, particularly as it was subjective in exploring people's everyday life experiences. Crotty (1998) suggests we need to view the research from the participant's point of view, and minimise the risk of imposing my assumptions on it. Furthermore, Crotty suggests that to protect against researcher biases, the participants needed to be involved in the research process to ensure interpretations are made from their social construction and not from that of the researcher. In the context of this research, there was the opportunity, therefore, to develop an action research project that was both participative and potentially transformative by its nature. The issue of shared ownership within the research was addressed through widening participation between myself, the researcher, and the researched; as the

process developed and control was devolved through the team, my role evolved from designer to facilitator (Ennew & Plateau, 2004). As PAR has the potential to be emancipatory and is most closely aligned with social constructivism (Langhout & Thomas, 2010), I decided to apply it to this school study. There was the exciting prospect of developing a research team that would include me as the researcher/facilitator, volunteer sixth formers as members of a YRT, and a cohort of young people as participants.

5.4.2 What is action research, participative action research, and youth participative action research?

Action research assumes that those closest to a given issue are experts in understanding the root of the problem, and are in the best position to help find solutions to such issues (Stringer & Ortiz Aragon, 2021). It addresses real-life issues that impact people's lives through a systematic cyclical investigation that incorporates observation, reflection and action (Stringer & Ortiz Aragon, 2021). PAR is a collaborative approach to AR where the research team includes community members with lived experience of the research topic. The aim is the reconstruction of knowledge through understanding and empowerment. PAR is often carried out with marginalised groups who rarely have their voices heard (Bergold & Thomas, 2012) and is seen as a way of ensuring social change is informed by the voice of such groups. YPAR is where youth are the participants. There is evidence that collaborative projects enabling young people to contribute towards research design can be challenging, but have also been found to be productive (Bowen et al., 2013). I cite examples from my 35-year teaching career to counter this. Young people rarely get the opportunity to have a say in the running of their school. Some schools run student councils, but my experience suggests they often have a staff-led agenda with little real impact on young people or the institution. I, therefore, suggest young people should be viewed as a marginalised group within a school setting.

There have been a variety of action research cycles developed in recent years. Some, like Hendricks (2017), conceptualise the cycle as a two-dimensional iterative process of reflection, action and evaluation. Others such as Kemmis, McTaggart and Nixon (2014) suggest a three-dimensional spiral process of plan, act, observe and reflect. I agree with Stringer & Ortiz Aragon (2021), who highlight that the key functions within any action research cycle are linked to looking and thinking. The model they suggest includes mini-

cycles of looking, thinking, and acting within a broader cyclical process of planning, implementing and evaluating. The model that I developed was based on the group's preferred way of working, which I will explain below.

5.4.3 The development of iterative action research

My research created two separate action research cycle processes. The first was a mini cycle between me, the YRT, and the participants, and the second was a less frequent process of me stepping back from the day-to-day work to reflect upon, consider and further plan the research cycle.

Over the data collection period, weekly mini-cycles were taking place, and these can be simplified into four stages:

Stage 1. I met with the YRT team, where we identified a research issue, discussed how we would investigate it and then planned the YRT/participant meetings.

Stage 2. The YRT members held separate meetings with their participants.

Stage 3. I held meetings with the YRT members immediately after their participant meetings. These meetings fulfilled several purposes as they allowed me to capture and transcribe the information, but also enabled us to discuss the research meetings and explore how they worked. This was a time for reflection that considered the process and the research material.
Stage 4. I then met with the YRT to discuss how the meetings and data collection had progressed. I developed a process that ensured that we learnt from each other to improve our practice. This was also the time that we planned the following week's meetings.

As a social constructivist, and as mentioned at the beginning of section 5.2, I was morally invested in young people's wellbeing and welfare. It was incumbent upon me to support and look after them (Hipolito-Delgado et al., 2022). I, therefore, decided to prioritise three specific areas relating to both the development of the YRT and the good of the research. These were:

- Active support and care for young people.
- Empower and give them as much autonomy as possible.
- Ensure research integrity and authenticity through an open and honest approach.



Figure 9: Action Research Cycle

In order to understand fully how the above process worked, it is important to exemplify how stage two of the research cycle, above, worked. As discussed in section 6.3, I encouraged the YRT to creatively develop their own research sessions with the participants. While many of the sessions differed, they all had the same aim. This was to create a process that enabled the young researchers and the participants to develop an understanding of the given area under discussion. As the data collection commenced, I deliberately chose to give the YRT the freedom to discover the most appropriate method with which to engage the participants. All YRT members went into their initial meetings with a question-and-answer approach, which was unsuccessful. The participants were difficult to engage in conversation, they gave very simplistic answers, and little meaningful data was gathered in the first few meetings. We gradually evaluated this approach, and many of the YRT team decided to employ more active, creative and fun approaches to engage the participants. In order to get a flavour of

how this worked, I will use one such session, run by Charlotte, a YRT member, to exemplify the typicality of these sessions.

Charlotte worked with two participants, both of whom were in year seven. In one of her sessions with the participants, the topic of 'stress' was discussed, and Charlotte decided to explore this further. She researched this topic and found a resource she describes as a 'stress wheel' that divides into eight areas of life that can impact an individual's stress levels. Each area is scaled from one to ten, indicating the differing stress impact for each area (**Error! Reference source not found.**).



Figure 10: Stress wheel data collection tool

Charlotte then describes her approach:

"I briefly talked about what potentially does stress them in school. I introduced this stress wheel that categorises different aspects of life and allows them to scale their stress in different areas. After explaining how it works, and the scale of 1 to 10 for each category they worked on their own wheel."

The two participants were then given their own stress wheel and asked to fill in each category to develop their personal 'stress profile'. This then acted as a discussion point for the three of them. Here are some of Charlotte's thoughts relating to one of the participant's wheel.

"Happiness and family had high stress scores, and this surprised me, so I tried to dig into those. Participant 1 explained that she was worried about her parents because her dad wasn't very healthy so she was scared he was gonna die.

She also worried about her friends liking her and she worried that if she doesn't do well in school she won't get a job and she explained how she couldn't help but look forward, and worry all the time. She said that her dad gave her 150 pounds worth of Amazon voucher as a present and she's already got all of her friends presents from it. And I was like oh you should get something for yourself. It seems to me that she uses presents to her friends as a safeguard against losing them"

Charlotte used this simple tool in a number of ways. It provided her with a simple mechanism to engage the participants and enable them to open up about their experiences in and out of school. It allowed her to record what they were saying on a third 'wheel', which she completed during their discussions (**Error! Reference source not found.**). This activity also enabled a relationship-building process to develop week upon week.

This activity took two weeks to complete; in that time, I met with Charlotte at a debriefing between the two of us, where we discussed the approach, what worked and what she needed to do next. Our debrief meeting also gave her time to reflect on her data collection, and the stress wheel exercise as well as plan her next steps. Charlotte also shared this approach with the other YRT members at the weekly team meeting. This created a lively discussion where others helped evaluate the process and findings. Again, reflection on an individual and group level enhanced this process. This approach to the research helped the team as we supported and learnt from each other. The collaboration within the team also enabled creativity as other members were inspired by the ideas brought forward and demonstrated by individuals.

This activity serves to demonstrate the approach we took at stage two in the research cycle see Figure 9.

5.4.4 Why I chose youth participative action research

As this research was being carried out within a social constructivist framework, the aim was to work with young people as part of a research team and involve them in all processes. This was about valuing them as individuals in their own right, and not just using them as a data collection tool. By working in this way, young people were at the centre of the resulting transformational changes, and this meant that such changes were something that they will be able to take credit for (Cook-Sather, 2020). I saw that we all made sense of our worlds in our ways, and everyone's experience was valid and worthy of respect (Crotty, 1998). However, I also saw that we were shaped by our background, the space we lived in and the people we lived around. Social constructivism played a part in how we saw the world and in our knowledge production (Nyandarai & Egbuonu, 2018; Crotty, 1998). This framed the importance of having young people from the school working alongside me as co-researchers. They were the social actors with lived experience and, as such, the real experts in our field of investigation (Reid, Flowers, & Larkin, 2005; Cowie & Khoo, 2017). Therefore, I needed to work with them to integrate their thoughts, feelings and reflections as part of the process. Somekh (2002) goes one step further when she suggests that knowledge produced independent of active practitioner participation is only partial knowledge. Therefore any knowledge created about young people without the involvement of young people lacks integrity as it has not been grounded in their worlds (Jones, 2004). I was duty-bound to ensure the YRT were fully involved in the research. Having recognised that young people can be the co-producers of new knowledge, I needed to explore their roles and responsibilities within the research.

As discussed in Chapter 2, education has been impacted by the neoliberal agenda, with young people being used as commodities so schools can improve themselves. It is important to distinguish between economic theories of human capital and theories of human capital (Sen, 1999). Capital accumulation should move towards being seen as a process in which humans
are intrinsically involved in the production of their own futures; as a society, we need to build on individuals' capabilities (Sen, 1999; Patton et al., 2016). YPAR would therefore promote the growth of students, as participants and co-researchers, into becoming part of their reality so that they would be integrated within a process creating new theory, new knowledge and new practice.

Working in a social constructivist framework informed my decision to adopt a PPI as a preresearch approach. This is something that I touch on again in section 7.7 when I discuss the involvement of young people in more detail. PPI is a process embraced within the health service that aims to improve patient outcomes by involving them in the design of the research questions, and methods of collection and analysis, as well as decisions to do with the dissemination of findings (Hoddinott et al., 2018). This is why the approach is philosophically aligned with YPAR and CCM and, as such, fits well with this research.

The processes involved in my action research are inherently related to interactions between people. My relationship with the YRT and the relationship between the YRT and participants were part of AR cycles that created new knowledge and altered our individual realities. Furthermore, this research explored the support required to enable young people's agency about their ability to take responsibility for their mental health. Therefore, the intention was to ensure that, through this research, we developed theories that were first tested and then applied to help others. In this way, I agree with McNiff and Whitehead (2012 p.34) when they say, "The idea of agency is that people can, and should, take an active part in decisions about how they and others should live". By taking a participatory approach, I intended to be dialogic, which meant more than just listening to the voices of young people. This approach was based on relationships oriented towards young people's self-understanding and agency. It is suggested that engaging in this way ensures change grounded in respect (Graham & Fitzgerald, 2010). This leads to questions about what my influence as a researcher may bring to the project. Unlike in traditional research, where the researcher should not influence or be influenced by the research, in AR the researcher takes on the responsibility for influencing the process as a primary instrument (Stringer & Ortiz Aragon, 2021). Within this AR, I took control by using my agency to identify a need for change and investigated social situations that benefitted from improvement. At the same time, my learning process was also necessary; as the researcher, I needed to ask questions about my independence and decision-making ability. This is something that Etherington (2004, pp. 31-32) describes researchers as having

the capacity to "acknowledge how their own experiences and contexts ... inform the process and outcomes of inquiry". Therefore, I saw AR as being about honesty, integrity, and authenticity, and whether I could be critical of the process and myself. Consequently, the process was heavily value-laden.

Running concurrently with this is the concept that AR is about social improvement and the development of new theories explained through one's own and others' learning. This is why I have developed theories that I have subjected to critique and am publishing for further scrutiny (McNiff & Whitehead, 2012). Having been the person who created the whole school strategy, I believe that I was ideally positioned to influence this work. However, I was also mindful that in having this initial investment in the strategy, I needed to take a reflexive approach to ensure that my bias did not interfere with research findings, one of the reasons the YRT were engaged. When Ofsted (Ofsted, 2019) inspected the school, they made favourable judgements about the mental health provision within the school. However, the brevity of the visit (two days) meant that their judgments were generic; I argue that the Ofsted findings were based on anecdotal evidence that did not meet research standards. Therefore, a robust, in-depth investigation was required to explore whether young people in the school benefited from the strategy. As a social enterprise project, I intended to introduce new practices that came about through theory and knowledge development from dialogic action research with young people.

The business of action researchers is to develop innovative practices that improve aspects of society, academia and the intellectual world; it is about being a theorist and a practitioner (McNiff & Whitehead, 2012). Whilst only a few choose this path, all humans have a story to tell. Some feel that action research needs to be heard; what is the purpose of it if it is not being used to improve society at whatever scale? Steele (2010) explores Foucault's understanding of parrhesia and free speech. Learning from this, I deduced there was a responsibility on those involved to tell their truths from their perspective, something my research was intent upon doing. Some believe researchers in social science could and often should go further. Stuart (2020) argues that the scholar activist contributes to society by using their research as a vehicle for change. Within the context of this research, the scholar-activist approach is somewhat appropriate as I aimed to discover new theories to influence social change and, through dialogue, "become jointly responsible for a process in which we all grow" (Freire, 1970, p. 53). In relation to the YRT, it is through this that, as scholar-activists,

they can help "transform that structure so that they become beings for themselves" (Freire, 1970, p. 47).

5.5 The barriers to young people's participation

The narrative around how young people are seen by society has been changing for a number of years. Since the 1990s, the UK government have been officially promoting the voice of young people as an important asset in service review and delivery (Davidson, 2000). This agenda was further enhanced with legislation such as Every child matters (Thornton, 2007), which centred young people in service delivery. Young people are now seen as social actors with the right to participate in decisions that impact their own lives (Hill et al., 2004; Kalliala, 2014; Lawrence 2022).

Unfortunately, for many years, the role of young people in research was limited to them being used as a data source. They were often seen as incompetent and unable to understand research processes (Christensen & Prout, 2002). Kitzinger (1997) argues that images contrived through the word incompetent disempower young people by denying them access to knowledge, which leads them to become more vulnerable. More recently, research has made a conscious effort to include young people in research and, importantly, to ensure that they are involved in shaping it. This then ensures the research is focused through young people's worldview, rather than the worldview of the adult researcher (O'Brien & Moules, 2007). It has also been suggested that working with young people to investigate the services provided for them has the potential to develop new knowledge and more democratic communities (Aubrey & Dahl, 2006).

Furthermore, Laws (1998) concluded that it was important to recognise the capacity of young people to evaluate service provision and give reasoned opinions. More recently, it has been recognised that young people can participate in critical inquiry in schools and make a difference to themselves, their peers and their school environment (Mitra & Serriere, 2015). Without young people's perceptions, it is difficult to know what they need for systems to be improved (Hart & Chesson, 1998).

The level of participation within a given project is important as it is inevitably linked to power relationships and control. Arnstein (1969) developed his ladder model, one of the first

widely used participation models. More recently, it has been developed by scholars such as Hart (1992) with reference to children and young people (Figure 11).



The Ladder of Participation

Figure 11: Ladder of children's participation (Hart, 1992)

Whilst these attempts to categorise participation levels of citizens, particularly children, were initially welcomed, they also have limitations in that levels of participation are rarely found to be sequential in the manner depicted as a ladder. Although Hart clearly states that the model is not intended to be used in this way, using the metaphor of a ladder gives the impression that participation at level eight has greater validity than at level five. There are

instances when it is more appropriate to consult with young people than to have a childinitiated decision-making process developed. Further development came from (Treseder, 1997; Driskell, 2002), who reconceptualised the ladder away from a sequential image towards more bespoke units that depicted the roles of both children and adults within the process (Figure 12)



Figure 12: In Empowering Children and Young People: Promoting Involvement in Decision-Making (Treseder, 1997)

As we have seen, participation is so much more than demonstrating that children or young people are involved in a decision-making process: it is about ensuring that when decisions are to be made, we listen to what they have to say, so their views can be taken into account. The debate then moves from one about what could happen to one about children's rights and what should or must happen.

5.6 Young people's rights

The United Nations Convention on the Rights of the Child (UNCRC) was written in the 1980s, adopted in 1990 and came into force in the UK in 1992. It is a legally-binding human rights treaty that focuses on the rights of the child. Whilst the document is designed to be read and enacted as a whole, Article 12 is clear that children have the right to:

- Express their views in all matters relating to them
- Their views being given due weight, taking account of age and maturity
- Be heard in administrative proceedings (UNCRC, 2019).

The significance of this is that, for the first time, children were recognised as full human beings with the facility to participate within society fully (Freeman, 1996). Moreover, by signing up for this, the UK government is legally bound to apply it without compromise. The Committee on the Rights of the Child oversees compliance with the treaty and has, on a number of occasions, criticised the government for failing to apply aspects of it fully. In particular, in 2002, it recommended they needed to do more in society as a whole, but specifically in schools, to ensure there was effective and meaningful participation of children (Lundy et al., 2020).

Arguments about the participation of young people in the school decision-making process go beyond legalistic ones. There is evidence that, at the most basic level, young people want to be heard; they want a voice and want adults to value it (Kilkelly et al., 2004). This is also about dignity and respect (Morrow, 1999) and some argue their rights should be written into public decision making (Byrne & Lundy, 2019). Furthermore, from experience, young people in schools are more content when they feel listened to and happier when they feel respected by staff. This view is supported by the recognition that where children are consulted in schools, there have been improvements in the teaching and learning (Flutter & Rudduck, 2004). The experience of many school staff is that they do not believe young people have the capacity to contribute towards decisions or the rights to do so; such arguments are deeply flawed. Ironically the ignorance of some staff towards this legislation strengthens the case for framing it as a rights issue. This is not a gift for adults to give or take away (Lundy et al., 2020).

One of the difficulties in schools is perceptions of what constitutes participation. Again, my experience suggests many schools have a student council as a decorative participative feature

to satisfy OFSTED. However, this is a dangerous approach as there is evidence that having a tokenistic school council is illegal and, more importantly, counterproductive (Pleasance, 2016). It has been found this sort of council has a greater negative effect than having no school council at all (Alderson, 2000; Lafferty-Jenkins, 2017). Kilkelly et al. (2004) reported students complaining their school council meetings were heavily staff-led with trivial agendas that usually include topics such as canteens. Students are unhappy with school forums where nothing ever changes (Alderson, 2000); they see them as a waste of time. It is, therefore, no surprise that the existence of school councils, parliament or senate is no guarantee of children's rights (Wyse, 2001).

The work that Lundy (2020) has done in developing the Lundy model of participation is primarily about simplifying how young people can be given an active role in the decisionmaking process regarding aspects of their lives. Lundy's space, voice, audience and influence principles align perfectly with my thesis' research design. I aimed to give young people voice and space as co-researchers and participants, as well as an audience through the school's leadership team. The influence was to come with the commitment the school has made in promising to listen and give due regard to the findings and proposals from the research. They intend to implement change to develop the school's mental health strategy.

5.7 Knowledge – accessing it, creating it or co-producing it?

Plato wrote "Those who know and do not act and those who act and do not know" (n.d). From the perspective of a researcher working on a participative action research study with young people in a school setting, this statement challenges me. I reflect on my years of teaching, and it encourages me to question whether I was someone who knew but did not act. The answer to this question may well be for others to provide, but will depend upon the subjective views of those involved with me during that time. From a personal perspective, I know that I worked incredibly hard to improve the lives of thousands of young people. However, this sits counter to what I am learning now as I return as a researcher. My role as a teacher was constricted by a complex mixture of social, political, and institutional factors that limited what I could do. Although I often made a conscious effort to give young people a right to reply or a say in the running of the school, my position often limited rather than enhanced the voice of young people. I was at the centre of what I would call a traditional school model where the teacher holds the knowledge and transmits this to the student. In this respect, I would be seen by Plato as one of those who "know and do not act". As a researcher, I see potential in young people, that I rarely saw when I was a teacher. From the outset of the study, I have worked hard to switch the emphasis, both in terms of power and knowledge, from **ME**, the adult and ex-teacher who has the knowledge, to **US**, the researcher and the YRT, who will co-produce the knowledge. Over time, as our relationship developed and their confidence grew, they have shown themselves to be perceptive, thoughtful, and knowledgeable individuals with the capacity to contribute much more than previously allowed. I have realised that young people such as these can develop into independent researchers with limited input from adults. I was therefore duty-bound to ensure participants and the YRT were as fully involved in the research as possible. Having recognised that young people can be the co-producers of new knowledge, I needed to explore their roles and responsibilities within the research.

5.8 School context

My aim from the beginning of the research was to involve young people within the school setting as fully as possible; I had concerns about the extent to which this research would be participatory. Being aware that there are different levels of young people's participation (Mercer, 2002; Cook-Sather, 2020) in educational research allowed me to explore how my time as a teacher may impact my work as a researcher. As mentioned earlier in this chapter, my school role would often impact my behaviour; personal values were occasionally compromised as I was required to make uncomfortable decisions. This is one of the reasons that I was drawn towards participatory action research. I wanted to help young people by improving their access to support in schools. The partnerships I developed with them during this research led me to conclude that they can become change agents in schools with guidance.

Being aware that young people are often **used** as objects of research within a school setting (Erickson & Christman, 1996; Wöhrer & Höcher, 2012), I intended to ensure this research was conducted from the young person's perspective, something not commonly adopted (Noffke & Somekh, 2008). As I discuss in section 7.7, I visited the school and conducted a PPI exercise. The findings from this exercise concurred with Moules & Kirwan (2005), who found young people were more likely to open up to their peers than they were to adults. If I

intended getting a critical view of the whole school mental health strategy from a young person's perspective, and they were more comfortable talking to someone closer to their own age, they were then more likely to give insightful and authentic answers in this setting.

5.8.1 My positionality within the research

Whilst the methods used will be explored in detail later in the following chapter, my position within the work was an important aspect to consider. Before looking at the theory underpinning this, it is important to understand the context of my history in the research school. I was the Assistant Headteacher with a pastoral responsibility, and my potential influence cannot be underestimated. I was responsible for attendance, behaviour and welfare issues, and found myself in conflict with young people, often every week. One of my tasks was administering all exclusions from school, a contentious area that again proved to be a challenge regarding my relationships with some young people and sometimes parents. By its nature, my position was seen by young people as powerful, as overseeing behaviour, I often had to make decisions regarding young people's sanctions. As a researcher returning to my previous school, it was clear that my lived history with the institution and its population could not be ignored. Having explored the literature concerning power in detail in chapter 4, it is pertinent to reinforce the idea that adults are seen to have more power than children (Cooke & Kothari, 2001). In accepting this, it is important to ensure this is not replicated in my research. Therefore, I needed to ensure power issues were addressed ethically (Edwards & Mauther, 2002), something that I explore in detail in Chapter 9.

Bearing in mind my history outlined above and returning to the school as a researcher, I was in a somewhat unique position as an outsider who was an insider, and now probably sit somewhere between the two. As I went into the research, I was aware of the insider/outsider debate, and I followed learning from Corbin, Dwyer & Buckle (2018) to surmise that the space in-between was the place I was going to occupy. However, my position was complex and did not fit neatly into the insider/outsider debate. The research structure saw me as an exteacher returning to the school with which I had an intimate relationship. I understood how it functioned, I had worked with many of the staff and, as such, felt like an insider. Within the research, there was also an indirect triangular relationship between me, the YRT and the participants. After the recruitment of the participants, I had no contact with them, leaving the YRT to work with them. However, although this neutralised my influence on them, they

would still see me as an outsider. I constantly deliberated over my relationship with the YRT and quickly recognised that as an ex-teacher, I would be seen as an outsider. However, I wanted to work collaboratively with them, make joint decisions and empower them to act; something needing trust. This would require my position to shift from an outsider towards being an insider.

The consideration of my positionality was about how "tuned-in" (Maykut & Morehouse, 1994) I was to the experiences of all the groups involved, the school and staff, the YRT and the participants. I was cognisant of the fact that, having worked at the school, I had subconscious biases at play and did not want these to influence the research process with any of the groups above. Furthermore, I was also aware that as a qualitative researcher, "*There is no neutrality. There is only greater or less awareness of one's biases*" (Rose, 1985, p.77). I worked hard with the YRT to reinforce my neutrality through conversations with them individually, as a group, and on occasions when we were involved in meetings with school staff. At times I deliberately needed the YRT to see me as part of their team rather than as an ex-teacher. This ongoing process made me aware that my position was constantly shifting.

This process enabled me to embrace my positionality. I became aware that I was not separate from the research but a central cog, part of all aspects and essential to it (Corbin Dwyer & Buckle, 2018). Further to this was my need to separate out the bias my past life may bring to the research. As I discuss in section 5.8.3 this was achieved through a constant reflective process, which enabled me to recognise that the most important aspect was an awareness of both my own feelings and the feelings of others. One cannot have true self-understanding without other-understanding (Fay, 1996). From this research perspective, it is impossible for me to position myself in one camp or another; the complexity of schools and their populations makes the task too difficult. Rather than attempting to position myself within the project, exploring how to remain true to myself and ensure honesty and integrity was more important. This thinking process is an area I will explore in the following section. The tensions that my positioning created throughout the research are something I will return to in the discussion chapter.

5.8.2 How is robust thinking ensured within the process of action research?

In section 5.4.3 I highlight the AR cycle and how it has been conceptualised in different ways; at its core are the twin processes of looking and thinking. Hendricks (2017) and Kemmis, McTaggart and Nixon (2014) use the terms 'reflection' and 'reflect' as the thinking part of the AR cycle. Reflection in AR is an explicit and active role that needs to be directly linked to further action. It is not about a passive reflective approach (McNiff, & Whitehead, 2012). The data produced within this research had a complex foundation coming from the three-way dialogues between me, the YRT and the participants. This research was about giving voice to young people, who often go unheard (Rothman, 2007). I agree with Francis & Hester (2012), who propose that the daily lives of the researchers should be part of the society explored, and their experiences and action used as data. The methodology I designed for this research required a process through which I and the YRT could turn our own experiences, actions and emotions into data for analysis (Roy & Uekusa, 2020). Furthermore, as researchers, we were part of the school's social world. Therefore a starting point of this social inquiry could be our own experiences and activities, and our reflections upon them (Francis & Hester, 2012).

The data collection period saw me adapt rigorous engagement with reflectivity as a tool through which I could enhance the research process. My daily meetings with individual YRT members and my weekly YRT team meetings were functioning reflective sessions. My task after that, having decided to develop a reflective methodological thread through the research, was to engage the YRT to take a more active reflective role. Enabling the YRT to develop reflective approaches also enhanced the study's validity by giving them deeper insight and a more rigorous understanding of their sociocultural experiences (Roy & Uekusa, 2020).

5.8.3 A process of collaborative reflection to enhance the research process

The reflective process I developed was an incremental one. In the early days of the data collection, as the YRT attended weekly meetings with their participants, I ensured that I met them after each meeting to discuss the process, record the conversation as data, and build the YRT's self-confidence. This allowed close one-to-one conversations where we discussed what was working and what was not, enabling us to develop creative solutions to obstacles arising. As the process moved forward, I became aware of the need to draw the young researchers together more closely so they could share their experiences and contribute to group problem-solving. The needs of the YRT varied. Some were confident, enjoyed working

individually and chose to continue this way; this was particularly true of young people in Year 13. The other YRT members decided to work in pairs or small groups; they experimented to find the best option for themselves and the participants. However, I also initiated weekly meetings so we could explore the details of what we were doing and how we were doing it. This work was not merely about data collection but about developing relationships to improve people's lives. The younger Year 12s had a different dynamic; whilst one or two of the group worked similarly to Year 13, the majority preferred group work; four of them paired up and met with two groups of younger people. As with Year 13, we also had weekly group meetings to ensure we could focus on our all-important goals and values.

As our confidence grew and we started to understand each other and develop a deeper trustbased relationship, the group meetings began taking on a greater reflective significance. I recognised that I had an opportunity to establish this process more formally by using the discussions as prompts which would spark reflection via a written process. As we embarked on a joint paper, the writing that came back from the YRT was not only high quality but gave me an insight into the challenges and successes of the process from their viewpoint. Reflection is about facilitating mental processes to enable a better understanding of confused or unstructured ideas based on reprocessing our thoughts and emotions (Moon, 2005). In exploring the theory around reflection, numerous models offer frameworks to structure this type of process, such as those by Kolb and Moon. Whilst these seem to concentrate upon individual reflection, I was searching for something which could support us as a research team to explore our individual and collaborative experiences. To this end, I developed the collaborative self-reflection tool (CSRT) (Figure 13) to support each YRT member to reflect and learn. The group then met to collaborate through a discussion of individual learning, enabling collective synthesis and planning as the research moved forward. This tool came from a collaborative process between the YRT and me. We worked individually, contemplating an aspect of the research, before meeting as a group to collect our thoughts and plan for the following work.



Figure 13: Collaborative self-reflection tool

Collaborative self-reflection tool descriptors:

Step 1. Individuals reflect and describe

Each team member explores their thoughts and feelings during specific aspects of the research. What did that aspect tell them about the research, the young people they were doing the research with or the research process?

Step 2. Individuals evaluate and learn

Each team member identified what worked and what did not. What has been learnt from the process, and what needs to change going forward?

Step 3. Collectively connect and collaborate

The group of researchers come together to connect and share their learning. The team develops an empathetic appreciation of the work through this collaborative process.

Step 4. Synthesise and plan

Learning is synthesised from an appreciative position of common understanding to revisit the previous work or move on to the next aspect of the research.

Since completing this section of the research process, I realised that there was a further step in this reflection. This saw me, in the researcher role, sit outside the whole collaborative process and view it from an autonomous perspective. I realise that the word autonomous could be challenged, but I think it is important for me to distinguish the part I play in the research process. In Figure 13, I actively work with the YRT as we collectively reflect on what is happening to learn and plan for the next cycle. However, I am suggesting that as the researcher responsible for this work, I have an autonomous role, which means I need to take responsibility for the outcomes. I have therefore included an updated model, Figure 14, titled Researcher summative reflection. For this, I have included a fifth step:

Step 5. Researcher's summative reflections

In addition to being within the research process, the researcher also takes on an outsider role. This research is his responsibility, and he needs to be prepared to make decisions after viewing the process from an autonomous perspective.

Again, this could be challenged, as anyone involved in steps one to four would find it very difficult to divorce themselves from what has taken place and make independent decisions. However, once the researcher accepts this premise, there is a valuable role to be played by them as they conduct an overview of this part of the research.



Figure 14: Researcher summative reflections

I intended that the benefits of pursuing a reflective process with the YRT would bring substantial gains. From academic writing to using SWOT analysis charts, or one-to-one and group discussions through to the use of the CSRT, a steady process of repetitive reflective engagement has supported the development of individuals, the team and the research process.

Developing this further thinking links my research with that from the University of Chicago Consortium of School Research (CCSR) (Figure 15) and their model to show how various factors interact and contribute towards young people's development, including how they interact with the world around them (McNeil et al., 2019). The model highlights how young people can grow through an action and reflection cycle. This process builds their skills and increases their experiences of agency. It is enhanced by a method of learning by doing, assisting young people in developing their sense of self and perspectives on others. This feeds into the growth process and enriches their journey towards intentional life choices (McNeil et al., 2019).



Figure 15: The UChicago framework. Taken from 'A Framework of Outcomes 2.0' (McNeil, et al., 2019)²

5.9 Reflexive approaches

The approaches described explain how the work was constructed between myself and the YRT through a reflective approach. However, this only partially covers how I worked as the person leading the research. I found the answer in the term 'reflexivity', which is defined in the Cambridge Dictionary in the following way: "the fact of someone being able to examine his or her own feelings, reactions, and motives and how these influence what he or she does or thinks in a situation". Cunliffe (2003 p.985) meanwhile stated: "...that we are constantly constructing meaning and social realities as we interact with others and talk about our experience."

I used reflexivity to examine my feelings, reactions, and motives within the research process. In particular, attention was focused on the knowledge and reality constructed through the social interactions between myself and the YRT in the research. This was an opportunity-

² The original source from the University of Chicago could not be found

laden yet critical process that contributed original knowledge about procedures, relationships, and personal learning of the YRT, the participants and myself, the researcher.

Through dialogue (Reason, 2006), this approach focussed on ensuring all participants had the opportunity to contribute and communicate in a way that was accessible to the individual regardless of age, ability or status. It has been suggested that dialogue is about the processes, suppositions and reality that make up everyday life (Issacs, 1993; Chiva et al., 2014); this is the approach I took by tapping into the lived experience of young people in the school.

Throughout the research, I 'stood back' and enabled young people; they were experts in their own lives (Cowie & Khoo, 2017). My aim was for the ones with lived experience to be reflexive about my role to ensure that I did not dilute the findings of the young people. Through dialogue, I encouraged the YRT to think for themselves and decide what was important to them, what they should include and what they should reject; they needed to be the decision-makers. By being reflexive, I challenged myself to ensure the findings were from their perspective, not that of an older white male. The iterative work that I did with the YRT, in examining the data in conjunction with my reflexive diary, and continual reading of literature, helped us make new decisions about the next steps, but also ensured that I was aware of both myself and the research processes (Robertson, 2000). My recognition that I could not stand outside this research was evidence that my values would permeate the inquiry, something the YRT needed to be aware of. In this context, reflexivity required my commitment to the value of this awareness (Robertson, 2000). Equally important was my academic input. Making decisions to introduce ideas, concepts and thoughts from other projects needed to enhance the decision-making process for young people. There was a continual balancing act between allowing the YRT to get on with the work and recognising that there may be a need for abstract thought, ideas and input from my academic work. By working this way, I aimed to safeguard the research and align with CCM.

My position within this research, the space in-between, has been highlighted above, as has my need to be critically reflexive. This thesis is based upon a philosophy of social constructivism; using autobiographical techniques has enabled me to develop a reflective process to derive meaning from experience (Saldana, 2020). This is not conducted in isolation but takes account of context and social interaction, and therefore supports my social constructivist philosophy (Saldana, 2020). I have taken the opportunity to use critical

reflexivity by drawing on my autobiographical writing to develop and explore my learning and understanding. I agree with Roberts (2019), who suggests that using an autobiographical tool improves critical thinking. This, in turn, assisted me in addressing potential imbalances with the YRT. It was important that, as an ethical researcher, I paid attention to and addressed, rather than ignored, power positions (Edwards & Mauther, 2002).

5.10 How the reflective and reflexive approaches helped me navigate the mess

My wish to enable the YRT as co-researchers required me to relinquish a certain level of control to them. My loss of control, particularly around processes, has been like learning to ski again, travelling downhill with little ability to control the descent, having frequent crashes and showing little, if any, style. I arrive at my destination, but how is somewhat of a mystery! The likeness to skiing continues beyond this, as the process as a novice skier and novice researcher can lead to a scrambled brain, confusion about the direction, and a lack of confidence about any tangible outcomes. In relation to research, Cook (2009) describes this as mess, something she sees as essential for progress in action research. Furthermore, she argues that it should be acknowledged within any research process. Firstly, because it is there, secondly because denying its existence would undermine other researchers and thirdly, and in my opinion most importantly, because there is a reason for its existence.

Barthes (1982) writes about 'punctum', the concept of knowledge that we have that is just beneath the surface but difficult for us to articulate. We are aware of it, but it is hidden from us and, as researchers, we will often shy away from it because of the difficulties we have engaging with it and making sense of it. This is the mess I found myself working in. As I worked with the YRT, wrestling to make sense of our collective thoughts and deliberations, we were engaged in a collaborative process of co-labouring that involved hard work, often leaving me feeling distinctly uncomfortable (Sumara & Luce-Kapler, 1993). This uncomfortable place was the space for creativity and the development of new thought (Cook, 2009). It should be a reason for celebrating that, on occasions, we did not know what we were doing (Atkinson & Claxton, 2000). This punctum, or mess, is where there was an innovative collaboration between those who know and those who act to develop the new knowledge required to advance the research; it is where the work with the YRT was situated. As one YRT member reflects: "At the beginning of the project, I found it difficult to communicate and express my ideas and thoughts on the different topics we discussed as I was unsure if what I was saying was applicable. Though as the project grew more into a collaborative process, I felt it was easier to share how I felt about matters — especially during our weekly meetings between the young research team and Dave. We started to analyse our findings as a group during these meetings, meaning there were varying perspectives on how our plans came together and what could be improved, leaving us plenty of opportunities to get the most out of the project we possibly could. We concluded that the more creative the meetings were, the more open, engaged, and expressive the participants were. I have noticed the confidence of everyone in the group has significantly improved — not only by creating a rapport with the participants, but also with creative thinking/ generating ideas, problem-solving, and working together as a team."

Taking a constructivist approach opened up the opportunity to reap the benefits of working with people, guided by their perspectives and developing new knowledge on their grounds. This was about developing the research from the viewpoint of the young people I was working with, and not about introducing theoretical frameworks imposed upon them. This leads to authentic research based on appropriate, well-informed choices by the co-research process (Manning, 1997). Validity and authenticity was secured further by returning to those who act, the YRT, to check that what had been discovered made sense to them (Lather, 1986). Engaging with the mess was, therefore, part of the collaborative process and about exploring what we did not know and what we nearly knew, thus resulting in a more rigorous research process (Cook, 2009). A crucial aspect of getting to grips with the mess was the ability to stand back, find some space, and reflect upon observations and the process. As I have explained, this needed to be done individually and as a group to make sense of it.

5.11 Research analysis

One of the central themes of this research was to collaborate with the YRT to ensure the findings were authentic, having been created from our three-way dialogue. I had to ensure the analysis came from the subjective experiences of everyday life, and this was about how young people's world was understood rather than an objective reality of it (Boyland, 2019). Social constructivism enables a relational reality to be constructed by individuals working together. Biosocial interpretation develops through biological cognition evolving via social interaction as consensus is reached (Cottone, 2001). In this research analysis, the biosocial interpretation came from the dialogue between the three parties, people from differing backgrounds collaborating on an equal epistemological level (Gómez et al., 2011). This research was about enabling young people to bring their reconstructions together around the consensus (Boyland, 2019).

As I have highlighted in the introduction, this was practitioner-based research. As such, the research process was not straightforward; the challenge came about as we aimed to conduct data collection followed by a process of data analysis. As I explain below, this was an assumption on my part. At the outset, I intended to apply an inductive thematic analysis approach; the YRT would meet with participants, and through transcripts of my debrief meetings with them, we would codify the data. I expected to conduct a text-book thematic analysis to systematically identify, organise, and gain insight into patterns of meaning within the data (Braun & Clarke, 2012). Whilst I used the six-phase approach to the thematic analysis (Braun & Clarke, 2006) as a guide, what transpired, due to the nature of the research design and the make-up of the research team, was a hybrid dialogic version. My adaptation saw codes and themes develop through discussion and experimentation, as well as scrutiny of our texts, made from the transcripts of YRT/participant meetings.

The importance of the ongoing dialogue between all research parties at each step of the process cannot be underestimated, as it enabled the team to scrutinise, adjust, re-test and confirm threads in the data. Whilst ongoing dialogue was about the data, it was also coupled with the sharing and reinforcing of my values. This was achieved by being fully attentive to the team by listening, discussing, and collaborating with group members as we navigated a complex and circuitous route through the research and the analysis.

This was true for my meetings with individual YRT members, our YRT meetings and the YRT/participant meetings. The weekly AR cycles became more than a data collection exercise as they involved an ongoing analysis process. This process is explained in detail in the following chapter.

5.12 Chapter summary

In this chapter, I have detailed the underpinning philosophy of this research and how this has informed the research processes. As a qualitative study that has followed a social constructivist path, I have adapted an AR cyclical approach alongside CCM, enabling me to collaborate closely with the YRT. This approach was initially conducted as YPAR, which has since evolved into YPDAR. To ensure the integrity and authenticity of the research, I have adopted both reflexive and reflective ways of working. These approaches, including the CSRT, helped me support the YRT and helped us think through the issues individually and as a group with a biosocial approach. In addition, this has enabled me also to distance myself from the collaborative reflection and view it as the author of this research. This chapter has also contextualised the data collection and analysis approaches, which I will develop further in the next part of this thesis.

Part 2. From theory to action

Part two of this thesis is concerned with the practicalities of the research. There are 2 chapters designed to take readers from the philosophical and theoretical context of the research to what I did and how I did it. I start with a chapter highlighting the research methods, from data collection to analysis. The following chapter then explores my ethical approach to the research as it reflects my journey from procedural ethics at the commencement of the research to the practicalities of the relational ethics I developed.

Chapter 6: Research methods

6.1 Chapter introduction

This chapter explores the participative principles of the research, and how these principles of equality, inclusion and dialogue impacted the methods used to collect data. This is also reflected in how young people were recruited for the research. The cycle was also influenced by my understanding of the terms vivencia, praxis and conscientisation, and I have developed my thoughts on these concepts in relation to this research. I then look in detail at how the relationship between myself, the YRT and the participants influenced the action research cycle we created. This includes explaining how the data collection and analysis became part of the same process. I share examples of how this came about and how the YRT, in particular, was instrumental in this. The developing relationships between myself and the YRT were also fundamental to the success of the research.

As discussed earlier in the methodology, introducing a YRT as collaborative partners is one of the ways I intended to tackle the concerns around my previous role in a school. Whilst I hoped this approach would start to redress the balance, I needed to take care not to imagine that all issues around power would be resolved (Groundwater-Smith, Dockett & Bottrell, 2015). It was, therefore, important in delivering authentic work; I needed to reflect on these developing relationships and the processes involved in this PAR.

6.2 The recruitment of young people for the research

An important aspect of this research was how young people would be selected to participate in the study. In the introduction, I explained my rationale for working with the pupil premium cohort as participants and the 6th form as a YRT. I worked with staff members who supported both cohorts of young people to develop strategies to recruit volunteers. I will explain the process for each below.

6.2.1 Participant young research team and recruitment

I arranged with the staff member responsible for the pupil premium cohort to meet with small groups in order to explain the research and their potential role in it should they choose to volunteer. For those interested, I prepared letters and consent forms for them to take to their

parents. These were then returned to the staff member by those interested. During the recruitment process, I met approximately 100 young people, 40 of whom took letters, of which 38 were returned. During the same period, I attended assemblies for the 6th form and promoted the research as something in which they could contribute to positive change in the school, and something that would enhance their own skill set; this resulted in 20 young people volunteering.

The initial plan involved pairing YRT members and participants for meetings. However, the pandemic interrupted the process, as the case study school was shut just as the meetings were due to start. After consultation with school staff, we attempted to complete meetings via an online video platform. Of the initial groups, six participants agreed to participate, and only three of the YRT wanted to do it this way. Despite numerous attempts to run these meetings, only two were held before this approach was abandoned. When we returned to school, the two groups were approached to start the meetings. We experienced dropout from both groups; ten participants and seven of the YRT decided to withdraw from the research.

6.3 Young people as collaborators in research design

To develop a research project that is a sustained collective inquiry, the collaborative process must include its design (Eynon, Torok, & Gambino, 2013). I had already identified the project's scope; however, the YRT needed input into the most appropriate methods. There was an expectation that in an attempt to collect data from a wide range of participants, multiple methods would be required. At the beginning of the research, I had thoughts about how the YRT would collect data, but it was clear that the YRT themselves should lead this area. I, therefore, developed a methodological tool (Figure 16) that I introduced to the YRT. This aim was twofold. It was an educative instrument as it helped me explain to them the broader purpose of the research in relation to knowledge creation. It also enabled me to introduce a variety of potential methods the YRT could use with participants.



Figure 16: Methodological tool

Initial conversations based on the tool centred around more traditional data collection methods such as questionnaires, one-to-one interviews, focus groups, blogs, and diaries. We also discussed the possibility of using more creative methods, such as drawing, story-telling through narratives, visual sociology and semiotics, and mapping or play-based activities. However, I was also conscious that the YRT may need to experience conversations with the participants to make decisions about how they collected data. The scope of this thesis does not allow a detailed analysis of each method mentioned. However, research has shown (Groundwater-Smith, Dockett & Bottrell, 2015) each has its advantages and challenges. Table 2 below summarises the advantages and disadvantages that the YRT expressed about the methods they chose to engage with the participants.

Table 2: Advantages and disadvantages of YRT selected methods

Methods chosen by YRT to	Advantages	Disadvantages
collect data		
One-to-one conversations	Opportunity to get an	Sometimes difficult to
	individual's thoughts	motivate a response,
		particularly at beginning of
		research
Group conversations	The collective can motivate	Without careful
	others to speak and engage.	management one individual
		can dominate.
		An individual's comments
		can sometimes lead others to
		follow.
Photo elicitation	Engaging and active way of	Didn't lend itself to the
	collecting data.	specific task that.
	Helped develop relationship	
	between participants and	
	YRT.	
Art based activities	Engaging way in which to	Some participants lacked in
	develop dialogue.	confidence and were self-
	Active and fun.	conscious about drawing etc.
	Helped develop relationship	
	between participants and	
	YRT.	
Concept maps	Engaging way in which to	None identified
	develop dialogue.	
	Active and fun.	
	Helped develop relationship	
	between participants and	
	YRT.	
	A good visual tool for	
Y	connecting ideas.	None identified
Interactive tasks based on	Engaging way in which to	None identified
YRT developed activities	develop dialogue.	
	Active and fun.	
	Helped develop relationship	
	between participants and	
	YRT.	

I was also acutely aware that the key to the data collection would be the three-way relationship between myself, the YRT and the participants. The initial focus for the YRT/participant meetings focused on getting to know the participants. On reflection, my meetings with the YRT also followed a similar pattern: the more we met, the more open we became and the better we worked together. What very quickly became apparent was that, to engage the participants and to develop the trust required, the YRT needed to introduce a variety of different activities with a common thread. Meetings were required to be both fun and active. The YRT started experimenting with potential data collection methods by adapting a sorting exercise I had used with them. Once they saw how successful this was in engaging the participants, they used their imagination to design their own activities. These included using outline figures, hexagons, photo-elicitation and poster making. These methods helped the participants find their voice as it enhanced their engagement and, ultimately, the relationships between themselves and the YRT, which is critical to this research (Broussine,

2008). I regularly reminded the YRT that these qualitative methods aimed to draw out a dialogue between themselves and participants in a search for rich data. I was becoming aware that dialogue was a central tenant to this research, being much more than a method. As I will explore later in this chapter, it also became crucial to the analysis process.

6.4 The influence of vivencia, praxis and conscientisation on the methods

The concepts behind this work were straightforward; they were about participation, action, reflection and research. However, the processes that enabled the realisation of these concepts were more complex (Glassman & Erdem, 2014). In section 6.5, I will explain how the data collection process, through numerous action research cycles, merged with the analysis. However, it is important to understand the theory underpinning the process before exploring it. By choosing to follow a participative route with a CCM influence, one of the most critical aspects of the research was the collaboration between myself and young people. The lived experience of the YRT was vital to the success of the research. This is an example of vivencia, defined as the full experience of events lived through their participation (Glassman & Erdem, 2014). The power of vivencia is that it cannot be observed; it can only be experienced, felt and lived (Glassman & Erdem, 2014). As someone who has not experienced the mental health strategy from a young people. Thus, this is not something I can draw on unless I can partner with young people. Thus, this is something I did and which I outline in section 7.5.

The transformation of vivencia in this research, bringing about change in school, is achieved through praxis (Glassman & Erdem, 2014). The process I expand upon below is an example of praxis, as it was an ongoing form of authentic action and reflection to bring about change (Crotty, 1998). Freire (1970) insisted the elements of praxis could not be divided; they were one and the same, and as such were a creative force (Crotty, 1998). By working in this way, I ensured the YRT's critical self-insertion into their reality, enabling them as re-creators and not just spectators (Freire, 1970). The process required a constant cycle of dialogue, reflection and action between myself and the YRT, the YRT and participants, and amongst the YRT when we met as a group; this was something other than what could be planned (Glassman & Erdem, 2014). Crotty (1998, p.153) describes it as "Action/reflection in fellowship and solidarity". This was what Freire (1970) called critical reflection and links to the process of conscientisation or critical consciousness, which is the awakening of people to

their capacity to change their own lives (Glassman & Erdem, 2014). Only dialogue can create critical thinking, but it cannot exist without it, meaning that dialogue is central to conscientisation (Crotty, 1998).

Conscientisation is therefore closely linked to vivencia, and praxis, as those marginalised within communities become aware of new perspectives, empowerment and a possibility of change. Through the research methods described below, the YRT took the opportunity to think and reflect critically, before taking action as part of a praxial process (Glassman & Erdem, 2014). Our research was not based on an abstract hypothesis but on observation of human relationships within a complex social structure. We aimed to gain insight into the working of the whole school mental health strategy to improve it, and so improve the lives of young people. The research process was a cycle of ongoing exploration and understanding, "action as praxis, research as conscientization and reflection leading to transformation of praxis" (Glassman & Erdem, 2014, p. 214). As young people are marginalised in schools, this research was also about rebalancing power in this setting; young people were given an opportunity to contribute to running an aspect of the school. It was also important that the research methodology and methods respected young people as partners in the research and not as subjects of it; hence our adaptation of CCM and its equal epistemological philosophy. This was a collective problem-solving process in a shared world which was challenging and problematic (Borda, 2006). This was about benefiting the community by creating a setting for young people to share their vivencia with adults in their world through processes of reflection and change (Cammarota & Fine, 2008). Section 6.5 details how the research unfolded into a five-step process incorporating data collection and analysis.

6.5 From data collection to analysis

6.5.1 Detailed research analysis

In the section below, I build a step-by-step picture of how my analytical method developed. As I worked through the AR cycles and the steps below, what struck me was the overlap between methodological processes and the methods of analysis. The steps below are an indication of the sequential and iterative nature of the work. What is significant is that whilst all steps were integral to the process, Step 3 was the one where time was spent discussing, deliberating and reflecting within an intense dialogic process. This is where we came together for collaborative reflection after individual reflection from individual meetings. This was the time we made decisions, and the critical step within this dialogic process.

Step 1. YRT/participant meetings

The YRT met with their participants every week for four months. These took on various different forms depending upon the YRT members. Some worked on a one-to-one basis, and others chose to work in pairs. On other occasions, the YRT decided that all the participants should meet together, so large whole group sessions were also held at times. My desire to give autonomy to the YRT meant that when challenges arose and solutions came through a process of dialogue within the research team, my default position was to encourage the YRT to take control and make the final decision themselves. This process' success was based on the relationships that developed between the YRT and their own participants; I learnt to trust the YRT members' judgements as they were more than capable of making the right call on the working of the YRT/participant groups.

Step 2. Debrief between the researcher and individual YRT members following YRT/participant meeting

After each YRT/participant meeting, debrief meetings were held between myself and the YRT member(s) where we would explore what had taken place; this was recorded, and a transcription was made that both myself and the YRT members reflected upon and scrutinised. During the meetings, we would look at what had been successful and what had been less so. These transcripts were used for the thematic analysis process as we developed an inductive, bottom-up data-driven approach (Braun & Clarke, 2012)³. This is where we started to discover the codes and themes as our process of insightful invention supported us in discovering new knowledge (van Manen, 1990).

Step 3. Weekly research team meetings

The research team meetings process was built upon the foundation of collaborative dialogue based on epistemological equity. Reflective conversations drew initial observations from individual meetings and allowed us to compare and contrast the participants' thoughts. It also facilitated in-depth dialogue within the group that enabled individual and group

³ Whilst I am aware of Braun and Clarke's 2021 edition of 'Thematic Analysis' I still refer to their 2012 edition in this thesis.

interpretations. As a research team, we discussed the findings from the previous week, and our deliberations then led us to a focus for the following week. To divorce these discussions from the analytical process is an almost impossible task, as the research process enabled us to develop threads that the YRT would then go and explore with the participants; it was part of the research and analysis methods. The acts of dialogue between myself, the YRT and the participants were part of communicative action as they were based upon validity, rather than power claims (Flecha, 2009). This stage of the work was crucial if I was to develop findings based on the voice of the YRT. I prioritised my own individual reflexive process, where I regularly challenged myself about whether findings were from the YRT or if I had filtered and distorted them.

Throughout this process, the YRT were given opportunities to summarise the work and draw conclusions. There were challenges to this work around the YRT's time availability and their varying personal commitments. I supported and led them; I suggested tasks for them to complete and could not (and did not want to) force work on them. The process through which the analysis grew and developed was based on the meeting patterns described above. On one occasion, to help us reflect on our progress, I conducted a hexagon exercise with the YRT. We explored the causes of mental health problems related to school character and grouped and linked them to help us create codes and themes. The results can be seen below in Figure 17.



Figure 17: School Climate Hexagon Summary Exercise

To develop this exercise further, and after reviewing school character literature (chapter 2) concerning mental health, I developed a tool to use as a discussion starter with the YRT,

Figure 18 below. Whilst YRT voice was my ultimate aim, I was responsible for guiding and prompting them by introducing academic theory to the process. The objective was to link theory on school character to young people's lived experience to support us all in developing new knowledge.



*Figure 18: School climate and mental health tool*⁴

From this, we discussed their findings, and they then completed the hexagon exercise (Figure 17). In discussion, we decided that a productive way forward would be for them to develop the hexagons into a framework model linked to the influences that impact young people's mental health in schools; unfortunately, they had little time to do this because of their academic commitments and so we all agreed I would produce a version. This, however, would be my interpretation of our joint hexagon exercise. The result was Figure 19 (below).

⁴ At this point in the research process, I had not formulated my views on school character. The tool therefore uses the term School Climate.



Figure 19: Framework developed by the researcher taken from YRT hexagon exercise

I circulated this to the YRT for feedback, asking for their thoughts. What I got back surprised me initially, as two of them went away and changed what I had presented. I had produced something from their own work, but as my diary entry from the time recognised, the results should not have shocked me.

"...my diagram was their information filtered through my eyes, the eyes of a 60-year-old white man! It isn't, therefore, a great surprise that they wanted to change it so that it is their work seen through their eyes."



Figure 20: YRT framework model 1 adapted from researcher model fig. 19

"... it is interesting to reflect on the differences between Aimee's model and my own. She has kept the basic components the same as mine however she has reflected teachers as key people whereas I neglected them. She has also developed her explanations more than I did. Therefore, I suggest that by getting the YRT members to develop these ideas gives a greater subjective element from the YRT point of view. Something I could not do".



Figure 21: YRT framework model 2 adapted from researcher model fig 19

".... Susannah focused on stress and pressure categorising it into a number of different areas. School was the biggest direct impact but followed by future prospects which links to 'expectations' in my first model. As with Aimee's model, this again showed the importance of enabling the YRT to develop their own ideas independently".

Whilst the two models, Figures 20 and 21, they developed were different I was encouraged for several reasons. Two of them had taken the time to challenge me, which was significant in terms of the research process and my relationship with young people. This also heartened me as it exemplified that I had enabled the voice of the YRT to come to the fore. This was a collaborative project; working together, we developed a process enabling young people to
present their findings. An important aspect of this phase of the work was that out of the 13 YRT members, only two had committed to creating their own model. This is something I return to in the discussion chapter. The two models also demonstrated to me the importance of the YRT developing their own work as their contributions were different to mine.

Step 4. Supporting the YRT to contribute to the analysis

Enabling the individual members of the YRT to develop their own narrative relating to the findings was important as this reinforced their autonomy within the research and built their confidence towards the analysis of the data. It was a continuation of my journey to ensure that the voice of the YRT was at the core of the process. To support the YRT's involvement in the analysis, I took four significant steps.

- 1) Introducing them to Quirkos, the tool I had identified to help with the data analysis
- 2) Running tutorials on qualitative analysis that included a help sheet (Figure 23)
- 3) Supporting reflection through the use of my Collaborative self-reflection tool (CSRT)
- 4) Inspiring them to develop reflective writing

6.5.1.1 Quirkos

Together we used the transcripts from our meetings to develop the analysis within this programme. It allowed us to develop codes and themes that indicated areas that the participants saw as being significant to them taking responsibility in school. The final one we developed is seen below (Figure 22) and shows how it visually simplifies the text into the themes and codes as an intuitive model. I imported all the transcripts into the programme, and together we codified it and developed the themes. In order to create this pattern, we referred to the framework models, Figures 20 and 21 above, and two previous Quirkos exercises (causes of mental health and school support for mental health), as well as notes from our meetings.



Figure 22: Final Quirkos analysis diagram

6.5.1.2 Qualitative research support

I produced a help sheet, Figure 23, for the YRT to support them in their analysis. I then ran tutorials discussing how we might go about developing our findings and recommendations. Again, the aim was to enable the YRT to develop their own thoughts, analyse their data and draw their conclusions from the research.

Qualitative Data Analysis and how to get started!

Analysis of qualitative data is about looking for patterns in the writing. Trying to identify themes see how those themes are broken down (or made up) and identify links between them.

"Think of this in the same way that you think about doing a jigsaw."

Step 1

Identify the key questions and try to break them down; write down what you think the question(s) is about. This can be done individually, in pairs or in a group. This will help you to focus on exactly what you need to be looking for.

Time to do your jigsaw!

Step 2

Just as when you do a jigsaw you will need to organise a strategy. Where do you start? Which bits come next? And so on!

- <u>Start with the 4 corners of the jigsaw</u>. In the analysis see this as looking for the main themes. When you read through the different data what stands out? At WHS we have developed a frame work so it may be that everything we read will be assigned in to one of these categories. Be careful, it may be that we have missed some things out and they may need adding
- <u>Separate out all the edge pieces.</u> Within each theme can you subdivide in to further categories? Look for themes within themes
- <u>Are there pictures on the jigsaw that help you link bits together?</u> How do we link themes or sub-themes together? Are there areas that overlap and need to be included in more than one theme?
- <u>Are there bits of the jigsaw that don't fit?</u> In the same way as a jigsaw there may be evidence in the data that sits by itself or is even contradictory. Make a note of it, put it to one side but keep trying to find a solution; you may get it to fit or you may come across a reason why it doesn't fit.
- <u>Are there bits of the jigsaw missing?</u> Is there anything that is obviously missing from the data? Keep asking yourself the question, "am I missing anything here?"
- <u>Always do the sky last!</u> I always keep the hard, uniform bits until the end. In the same way in the analysis, you might come across something that you find difficult. Put it to one side and move on, go back to it at the end or with someone else and try and work it out.

Step 3

- The above process may take you 3 or 4 attempts reading through the data. Once you are happy with it try and pull it all together.
- Write a paragraph about each theme. Within each theme write about each sub-theme.
 When you are writing try to quote examples from the text to back up what you are putting forward.
- Write about what doesn't fit and why you think that is.
- What do you see as the links or threads that run through the analysis?

Have a Happy Jigsaw!

Figure 23: Qualitative analysis help sheet

Step 5 Reflective dialogue to develop findings and recommendations

The final step in the analysis process was to draw all of our information together and conduct a dialogic review process between myself and the YRT, where we revisited our thematic analysis via Quirkos, our reflexive writing and our discussion notes as captured in my reflective diary. My invitation to the YRT to write up their own findings resulted in three reports written by five of the team. Through a lengthy process of collaborative deliberation, discussion, and creativity, we crafted an understanding of the key findings. Due to the work commitment of the YRT, I drafted these before a concluding discussion with the team to finalise the findings and recommendations for the school.

The process I described above was about how dialogue became central to the research and how the principles of CCM enabled a collaborative process that encouraged egalitarian relationships. This, in turn, countered hegemonic school power relationships. The dialogic process between the three parties involved ensured robust data collection and analysis. It then enabled us to produce findings and recommendations which were presented to the headteacher, with whom we had further dialogue before agreeing to implement. This is summarised below in Figure 24.



Figure 24: Young research team participative action research cycle

One of the difficulties throughout the data collection phase was the logistics of working with the YRT. My ideal was to have weekly group meetings with all YRT members attending so that we could agree on plans together. Because of school commitments and a lack of free time together, this was not possible; instead, I was left acting as a go-between, trying to negotiate an agreement between the two groups. Whilst, on the surface, not a difficult task, I always had tensions. Throughout the process, I questioned whether my negotiator position meant that I influenced decisions too much. I continuously reflected and evaluated the decision-making process, asking myself whether I was really allowing the YRT to decide on the direction of the research. In reality, there were many joint decisions as the YRT constantly impacted my thinking.

6.6 Triangulation or crystallisation?

As described above, the project's design enabled both groups of young people to make meaningful contributions to this research through their own lived experiences and interactions with each other. It was about them developing their relationships to build an interpretative position as social constructivists (Charmaz, 2000), and was not about taking an objectivist standpoint (Ellingson, 2008). My dilemma, however, was that when I first engaged in this research, I aimed to use conventional forms of qualitative analysis so that findings could be presented to improve practice in the school (Charmaz, 2000). What transpired was a thematic analysis approach that was informed by the dialogic processes between the three parties within the research. My search for authenticity was less about triangulation and its objective overtones, and more about the intuitive flexibility of crystallisation, which enhances and deepens my thoroughly partial understanding of this given area. This approach brought a greater depth of understanding whilst also highlighting that there is so much more for us to discover (Richardson, 2000).

My decision to use crystallisation as a guiding philosophy for the analysis was reinforced by my commitment to understanding the social world through subjective interpretation within the research team (van Lieshoult & Cardiff, 2011). As the research developed over time, the relationships matured. As I will demonstrate in the discussion chapter, trust and understanding built between myself and the YRT, as it did between them and the participants. This enhanced our interpretive mechanisms as we started to believe in one another's capabilities and intentions. Through the repetitive reading and discussion of the transcripts, the collective development of new ideas became cyclical and generative as the group and research developed. By ensuring the perspectives of the research parties were interwoven, I was evaluating the quality of the work based on criteria of convergence rather than criteria of objectivity (Ricoeur, 1976); although this was more about being comprehensive in approach rather than about convergence as a goal (Varpio et al., 2017). This, I believe, is crucial for the analysis of the research as, by fusing multiple interpretations of reality, I am guarding against and, as far as possible, removing "authorial intent" (Ricoeur, 1976). The clarification process through repetitive speaking and listening tests boundaries and redefines understanding, enhancing trustworthiness (van Lieshoult & Cardiff, 2011). It is this that can then give us a fully involved understanding of this topic (Varpio et al., 2017).

6.7 Who is the analyst?

Whilst I have ownership of this research and the thesis is written in my voice, I am also acutely aware of the tensions existing within the process of analysis around the critical question of 'who should analyse the work?'. This is a question I will now seek to explore and answer. From the outset, this research has been founded upon my deeply-set values of honesty, integrity, authenticity, and inclusion. By developing this research with young people and for young people, I aim to help improve their lives both now and in the future. To be involved in this work is their right and, as I quote in section 5.6, this right is not one for "adults to give or take away" (Lundy et al., 2020). My intention was that involving young people in the design and data collection would result in data from their perspective, and would therefore be authentic. This also holds for the analysis. Were I to collect the data from young people and analyse it myself, I would be undermining my values, and I also believe I would be letting the young people down. This is research from their perspective, and the analysis also needs to be done this way.

I worked with the YRT team to design the research, and they then worked with participants to collect the data. Collaborating with them to conduct the analysis proved to be challenging in a number of ways. Firstly, as intimated above, this was my research and my qualification which I needed to own; I had opinions about the research, dialogue, and findings, and I wanted to be able to voice them. Initially, I considered two choices; do I present my opinions to the group, impose my power to force these on the YRT (researcher analysis), or do I step back and allow the group to decide (participatory analysis)? There was a compromise

137

whereby I offered my opinion and suggested it as just one voice within the group, and we analysed the work together. I found this a challenging time and this is reflected in an extract from my reflexive diary.

"The tension I have is that I do not want to impose myself or my opinion, I also question the validity of what the YRT may decide. The echoes of adultism and teacherism are reverberating in my head as I ponder the correct decision for myself, the YRT and the research outcomes. If I am to be true to my values, then the YRT needed autonomy to make their own findings within a supportive structure, something I need to develop. Just because the YRT are made up of young people does not give me the right to disempower them and impose my will. I believe that by taking the 'middle road', I will offer opinions without imposing my will."

The development of our relationship was crucial for this as trust had been developed between us. At every meeting, I would actively encourage them to challenge me; at times, I would challenge myself by putting counterarguments to points I made. I would also take time to reflect on conversations and return to the group and revisit discussions where appropriate.

I was aiming for intersubjective consensus (van Lieshoult & Cardiff, 2011) or mutual adequacy (Boog, 2007), a critical validity check in PAR. Reciprocal trust was developed through our relationship, which also helped deepen truthfulness, morality and authenticity (van Lieshoult & Cardiff, 2011). This was multi-voiced research and required multi-voice analysis. Ultimately, I am responsible for telling a story and reporting the findings as I see them. I worked extremely hard to give young people the support, direction, and autonomy to enable them to develop new knowledge using their experience and research. I had impact, but how much cannot be quantified. I have set myself a high standard of self-scrutiny throughout the research. I have questioned, discussed, and deliberated on decisions to ensure my primary goal of inclusion for young people was met.

6.8 Summary

This chapter has outlined how this research developed between the three parties as we searched for new knowledge through dialogue. I have explored how the methods of data collection were chosen. I clarified the five-step part of this process, highlighting how it

enabled me to ensure the YRT were central to decision-making. I also gave examples of how the collaboration developed between the YRT, the participants and me. Section 6.6 has explored how the analysis and data collection merged into a crystallisation process. The following chapter examines how ethical procedures became fundamental to this way of working.

Chapter 7: Ethical considerations

7.1 Chapter introduction

At the conception of my research, ethics was an area which had not been fully considered. However, what transpired was a journey from the necessity of protectionist ethics via the university board to an in-depth analysis of ethics in the context of youth participative action research in schools. I did not intend to commit to a lengthy exploration of ethics until I realised its centrality in my thesis. Chapter 4 explores how young people are impacted by power and how this then impacts their agency, identity and capital, areas which are at the heart of what ethics is. A detailed exploration of ethical issues related to young researchers and schools follows. I decided to concentrate my ethics writing here in one chapter, presenting the challenge of what to include and omit. I refer to this in section 5.10 as mess; the reason why the reader may interpret parts of this chapter as findings.

Therefore, this chapter charts my ethical progression through this research journey. I start by reflecting on the procedures demanded of me to gain approval. I then explain my approach before the research, involving a consultation, PPI process; this informed two critical elements of this research. The importance of relationships to my research cannot be underestimated. Therefore, I explore how my relational research processes aligned with my social constructivist standpoint, which led me to ensure the YRT were involved in ethical decision-making. This, in turn, steered me towards relational ethics as a way of working and ensuring young people were at the centre of the research as equal partners. I investigate the key elements of relation ethics and how they influenced my research, before I conclude by exploring the importance of dialogue to my relational ethics approach.

7.2 Seeking ethical approval

My introduction to research ethics was via the university ethics review panel, as a requirement to receive ethical approval was a prerequisite for conducting research with young people. Whilst I was aware of how important the relationship-building process would be in the research, I did not deliberate on the relevance of relationships as I conducted the university's ethics procedures. Although aware that the relationship status between all parties would be important, I did not know then that I would be working on an equal epistemological

level with the YRT. As my research developed, I came to understand how, together with the YRT, we would be co-producing dialogic knowledge. As part of this, we would need to be critically reflective and grounded within what we were doing (Banks & Brydon-Miller, 2019).

Similarly, my understanding of critical reflection developed, and I became cognisant that it would allow us to interact with the research, the researcher/co-researchers and the results, enabling us to interpret our lives through the experience (Martin et al., 2003). Were I to complete the process again, I would want to involve young people in a discussion about ethics and explore the potential advantages and disadvantages of their involvement in this type of research. Whilst the focus of such discussions would be on the relational aspects of the research, I would also find a space for a discussion around research ethics as protection, something I discuss in detail below.

7.2.1 Research ethics as protection

One of the challenges within ethics for qualitative researchers is that the origin of ethical guidelines comes from a scientific and bio-medical background (Artal & Rubenfeld, 2017). The initial impetus came after the atrocities of the unethical research practices by the Nazi regime during the second World War that led to the Nuremberg Code in 1947. The Declaration of Helsinki updated this in 1964 and has done so regularly since, with the latest version published in 2013 (World Medical Association, 2013). The Declaration of Taipei (World Medical Association, 2016) continued this ethical modernisation process. However, whilst ethics guidance has taken great strides forward, disturbing cases have continued, and as recently as the 1970s dehumanising research was still taking place that resulted in yet further legislation. In the United States of America, the Belmont Report (The National Commission for the Protection of Human Subjects of Biomedical and Behavioral Research, 1979) was implemented in response to the Tuskegee syphilis study. Since the basis of ethics development has come from a natural science or biomedical standpoint, issues of significance to qualitative research, such as the importance of relationship development, are largely overlooked. There is a belief that there has been too little focus on ethics within the behavioural and social sciences (Schrag, 2010). Applying ethics within a qualitative research setting still requires consideration and development.

The background I have given to the development of research ethics suggests that it was seen as a purely protective measure for many years. Whilst some academics started to address ethical dilemmas, such as research relationships and ownership of findings (Burgess, 1989), there was (and is) still an overemphasis on protection from harm (Banks & Brydon-Miller, 2019). There is a debate across society about the benefit of this approach as it can be seen as a deficit one searching for what an individual needs (McCashen, 2014). Instead, society should take an alternative and asset-based approach (Stuart & Perris, 2017) and concentrate on promoting good to support people's strengths, helping them take responsibility in their lives. This area is explored more fully in section 9.4.2, where I look at relational ethics as an approach within YPDAR. This asset-based approach builds on the positive relationship between the parties involved as they are created through equal epistemological power relations. However, I feel compelled to state that guidelines designed to protect young people from harm must be understood and followed, although they should not be used solely to exclude other ethical considerations.

7.2.2 The process of completing ethical procedures

I completed the university ethics documents systematically, but with reservations due to the procedural nature of what was required. This was the most straightforward aspect of the ethical procedure as it dealt with tasks such as the safety of individuals, data collection and storage. A set of robust procedures were developed following the university's guidance. It was a relatively uncomplicated process whereby I had to document how I would keep young people and researchers safe, as well as collect, store and then safely dispose of all data. As I mentioned above, what I now realise was missing was any input from young people. I am uncomfortable with this approach as, on reflection, it feels as if young people were being treated as research objects. My belief is counter to this, as they are thinking beings who are socially constructed. There is a danger that this approach could undermine young people's agency (Hilppö, Chimirri, & Rajala, 2019).

Cutting & Peacock (2021, p. 2) describe "navigating our institution's ill-fitting ethical procedures alongside the realities of conducting participatory research". The principles applied by review panels may be necessary to ensure that guidelines are being followed, but they are also about power and control (Banks & Brydon-Miller, 2019). Cutting & Peacock (2021, p. 2) also describe the gap between ethical procedures and ethics in practice as

"dilemmas" or "moments", sometimes described as "ethical slippage". This is when the reality of everyday life is ignored and comes about when working with other human beings, as the unexpected will often happen. New ethical dilemmas continuously present themselves and need to be considered, understood, and acted upon. At this point in time, the signing of the forms will ultimately be irrelevant and could also be seen to absolve a researcher of the responsibility to reconsider issues (Hoonaard, 2011). My concerns about the university's ethical procedures were many. They included the suspicion of a top-down approach which was as much about the university abdicating responsibility as young people's safety. In addition, it disempowered young people I was collaborating with, as they had no say in the same rules being applied to them.

The ethics approval process began before I engaged with the co-researchers or participants. I thought through the process and discussed potential issues with my supervisors and school staff. This was about following a rules-based traditional approach and was located around my deliberations with other adults. Crucially, other than the PPI activity (section 7.7), I did not engage with young people themselves (Hilppö, Chimirri, & Rajala, 2019). The ethical guidelines I was working towards in preparation for my research should be seen as rules of thumb rather than absolute, and used as part of a suite of measures in any ethical decision-making process (Brinkmann & Kvale, 2005). Once ethics was granted, I moved on to the data collection process. However, this was about accepting and embracing a generalised other rather than a concrete other in relation to ethics (Pollard, 2015). Whilst the process I had put in place protected all parties, I now question how I did it. The approach I took was functional and, as I will explore in the following section, dissonant with my preferred relational way of working.

During the research, I recognised that an interpersonal bond was being developed within the research team through our cyclical conversations and dialogue; actions were from the heart and mind (Slattery & Rapp, 2003). The positioning of the term relationship was important as it was about all parties understanding each other and each other's lives to ensure that everyone bought into the concept; these relationships were a primary consideration (Hilppö, Chimirri, & Rajala, 2019). As humans, we have a constitutive relation with others, which implies one already has an ethical responsibility to the world and others in it. This was not required as part of my ethics application for this research, but it was an essential part of the research process.

7.3 The significance of ethical considerations to this study

Ethics are closely linked to morals and related to how we embrace such issues when working with others (Gregory, 2003). In a research context, ethics is about values and responsibilities; complexity arises, however, as it needs to be seen from various standpoints, including harm prevention (Fry, Treloar, & Maher, 2005), democracy (Marrero Castro & García, 2021), and human rights (Reis Monteiro, 2014). More recently, its meaning has been broadened to include animal welfare and environmental issues (Chatfield & Morton, 2020).

As described in the context chapter, this research is about working with young people within a school setting. Two groups of young people were involved in the research, with each group requiring both conjoined and separate ethical considerations. The two groups included the YRT, a group of 16–18-year-olds who acted as co-researchers with myself and undertook regular meetings with participants, and the participant group, which included groups of young people aged between 11 and 15 years old from the pupil premium cohort of the school. From the outset, there were a multitude of interconnected issues that required care and consideration regarding ethical concerns that will be explored throughout this chapter.

Working as a social constructivist is about discovering meaning and understanding through my active involvement in constructing that meaning (Kim, 2014). Within this paradigm, it is therefore important to consider the social interactions with young people involved (Kim, 2014). At the outset, I was aware that the relationship development between myself and the YRT would be vital to successful outcomes. I had yet to consider the implications of these relationships on the ethics within this work. However, I am subscribed to a way of working that, from my perspective, demands a research partnership with young people on an equal epistemological level. During the data collection period, there was a gradual transformation in my own ethical development as I realised the intricacies involved when conducting research of this type. Having failed to involve young people in the procedural ethics at the outset, I was wakening up to the realisation that the ethical integrity of this research was at stake. To be true to my values, I needed to develop a way of working that had relationships at its core. This research is about working with young people as partners, using a process of collaborative dialogue. It is, therefore, vital that the relationships between parties in the research are also true to the values and intent of the research, including care, fairness,

144

openness, and inclusion. The foundation of this research is based on social constructivist principles (see section 5.2), and it is important to explore how this has influenced my ethical approach (Cottone, 2004).

7.4 Ethical consideration of working in a social constructivist paradigm

Following a social constructivist paradigm, my research was built upon the pre-requisite that social acts occur between individuals (Teague, 2000); any transaction results in knowledge gained for all parties. The processes that guide the research design heavily rely upon the values held by those within the research relationships, something significant within my research. The research I have undertaken has ensured that the knowledge created has come through the engagement of individuals in social activity (Kim, 2000). This section will explore how working in a social constructivist paradigm has implications for the ethical approach I need to take to remain faithful to my values.

We all inhabit the same relational world, and important questions need to be explored to do research and live in this world together (Hilppö et al., 2019). Furthermore, research ethics is under constant revision, and much has been written about how we ensure it can be adjusted and redefined as research situations change and develop (e.g., Baumrind, 1964; Etzel & Watson, 2006; Harlow & Oswald, 2016; Campbell & Morris, 2017). One of the central themes of this writing is the claim that human ways of being and becoming are essentially relational, and who we are and whom we become is through our interaction with others (Hilppö, Chimirri, & Rajala, 2019). In research where one aims to understand the lives of others, essential consideration is given to the interactions between the researched and the researcher, and how these impact how both parties live their lives. In other words, we have a moral responsibility and ethical commitment to ensure that we look after each other (Hilppö, Chimirri, & Rajala, 2019). Collaborating with the YRT and working on an equal epistemological level meant I brought my academic knowledge while they brought their lived experience and life knowledge. This did not necessarily bring equity of power to the relationship. Still, it enabled both young people and me to bring our unique and different power to the partnership (Pollard, 2015). In addition, there was a need for a deep and trusting relationship, particularly between the YRT and me, as well as between the YRT and the participants. In my view, it would be difficult for work to be conducted on an equal epistemological level without a strong element of trust between all the parties. Within a

145

school setting, this empowered the YRT to be part of the ethical decision-making process. In conjunction with me, they were actively involved in ethical discussions and decisions they would otherwise not be allowed to contribute towards. These decisions were constructed within the context of the research but, more importantly, the context of the relationships (Pollard, 2015). This interdependency between young people and me was crucial to the research's success and depended upon the quality of our relationships. I agree with Cottone (2004) who argues ethical decision-making as a social constructivist is a joint decision, as decisions are made in a context of social interaction. This is described as consensualising, which is "acting according to what is 'known' through social interactions" (Cottone, 2004 p. 8). These were the social interactions taking place between us.

As I suggest in this writing, young people in schools are often subjected to teacherism, which is systemic across UK education; but it is important to set out precisely what I mean by this term. Building upon Shier's (2015) definition of adultism, teacherism can be understood as a subset of the orientation of cultural norms that reinforce the superior position of adults in relation to young people (Corney et al., 2021). This position is further exaggerated within a school setting as the historical power structures in schools reinforce the position of both adults and young people. The definition I put forward is as follows:

Teacherism is a dysconscious act carried out by adult teachers over young people in a school setting. Schools, as institutions, are adult dominant power structures that reinforces societal belief systems that adults are superior to young people. Schools place teachers in positions of power reinforcing young people as 'becoming' adults, something that in turn disempowers them as beings in their own right. Teachers are described as educated and professionals, terms which are loaded positively and that engender respect. In contrast young people are often described as pupils, students, or adolescents, terms defining young people as 'becoming' and which can also have negative association connected to them.

Figure 25: Teacherism definition

Working on a YPAR project in school enables young people to rebalance and shift the accepted school power hegemony legitimately, as well as contribute to school improvement (Cook-Sather, 2020). This research enables young researchers to contribute to ethical decisions and benefit from moral and social development. There needs to be a move away from an ethics of intention to an ethics of responsibility (Searle, 2004) and, in doing this, we

commit to the consequences of interaction (Gomez et al. 2011). YPAR is a way of working from a values base recognising the rights of young people to contribute to research directly impacting them, and ensuring they play a central role in this research. In particular, the methodological innovation around the fusion of YPAR and CCM requires reframing the traditional protectionist ethical approach to a relational ethics approach. It, therefore, follows that relationships within the research context require further scrutiny.

7.5 The importance of the research relationships

As discussed throughout this chapter, the caring, respectful, and trusting relationships between the three parties were crucial to the project's success. Below I will explore each element in turn.

7.5.1 Participant/researcher relationship

I had a fundamental decision regarding my relationships with the two groups of young people, the participants and the YRT. My relationship with the YRT was central to the research outcomes; the YRT were the link between the participants and me. If we did not trust and respect each other, the process of data collection and analysis (explored in chapter 6.5) would have been close to impossible. However, my relationship with the participants was somewhat different; it did not need to be as close as they met regularly with the YRT. From my perspective, the participants were in the background as it was the YRT, and not me, who would have weekly contact with them. I had concerns regarding how my historical relationship with the research participants may influence their approach to the research. This is best illustrated by a diary extract reflecting on an incident during the recruitment of the participants:

"As I sat waiting in a room for a small number of potential participants to arrive, a young girl came in, I reassured her she was in the correct location, but her nervousness was obvious. I explained what the meeting was about, and she breathed a big sigh of relief and said, "*Oh, thank goodness for that. I thought I was in trouble with you*". I was extremely surprised; I had never spoken to the girl before and had left the school 12 months previously. However, she remembered me from my past role as a powerful member of staff who would often deal with discipline issues. The power issues I discussed in chapter 3 immediately struck a chord, and the PPI consultation explained in section 7.7 proved

147

pertinent; I was definitely not the most suitable person for the participants to be talking to. At this point in time, I also made the conscious decision that I needed to leave the contact to the YRT, trust them to build relationships with the younger participants and deliberately remove myself from the day-to-day contact with them."

The following comments from members of the YRT indicate that this process was successful, as they all highlighted trust in the research development. Josh commented:

"In my opinion I've moved away from being a strange year 12 that talks to them about topics which are difficult to speak with friends about, never mind complete strangers. I am someone who my participants can talk to about the school's mental health support system, with enough trust and confidence between us that they know they can critique the system without being reprimanded by staff due to what they've said in our meetings. This confidence and trust have also helped me develop more as a person"

Susannah commented:

"Though as the project grew more into a collaborative process, I felt it was easier to share how I felt about matters especially during our weekly meetings between the young research team and the researcher."

Poppy and Katie commented:

"As well as my confidence growing, the participant's confidence also started to grow as he began to open up more and became more trusting of me."

7.5.2 Young research team/participant relationships

Whilst I had overall responsibility to ensure the safety of all young people within the research, the YRT were the ones building relationships with the participants meant I needed to ensure systems and lines of communication between all parties were clear. The YRT were in a position to recognise distress or have a concern about one of their participants. However, their responsibility was to pass any concerns on to either myself or a school staff member, ensuring the participants were safeguarded; to my mind, it would be totally unfair and ethically wrong to place the burden of care at the door of the YRT. Therefore, I worked with

the school and developed a safeguarding system to allow the YRT to pass any concerns on to a staff member if I was unavailable. I also conducted safeguarding, child protection and confidentiality training with the YRT.

Furthermore, I reinforced this whenever appropriate to ensure they were reminded of their responsibilities and to ensure they were comfortable bringing things to my or the school's attention. This all took place before the research started, and whilst I took feedback from the YRT it was not planned in collaboration with them. I was applying my procedural ethics as advised by the university. On reflection, working as a partner, I should have built in a consultation phase with YRT rather than imposing something on them.

Whilst guided by procedures, there were no specifics from the university to guide me on what I have just described. It was down to me to work through all the issues, and this was something I did through conversations with my supervision team and staff in school. Whilst completing the university's ethics application process, an area I did not highlight but that came to light was the pupil premium status of the participants. I decided to discuss the issue with the YRT. This was an ethical dilemma as I was giving the YRT access to confidential information about the status of the participants, something students would not usually have. However, I was also cognisant that without this information, the YRT were unlikely to be able to conduct their work either effectively or empathetically.

By entrusting the participant conversations to the YRT, I knew that the relationships between these two groups would become an important aspect of the research. The safeguards I put in place aimed to ensure the safety of the participants and YRT. However, I discovered that one of the most critical aspects of this area of the research was the direct and indirect support I gave to the YRT to ensure they developed caring and productive relationships with the participants. In the section below, I explore the development of my relationship with the YRT. However, it is important to highlight that as part of the action research cycle (Figure 9), I intended to model, to the YRT, the way in which you develop a trusting and productive relationship. Through discussion at individual and group YRT weekly meetings, I could draw on the research's values; why we were doing it, how the participants may benefit, and what we were learning about ourselves. This demonstrated to the YRT what a caring/trusting relationship required so they could get to know and support their participants both in the research and in the broader school community. In the following section I quote examples from the YRT's writing to demonstrate that confidence and trust between the two parties grew as their relationships matured as we moved through the research. In addition, members of the YRT developed positive relationships with participants they had worked with over this period of time.

7.5.3 Young research team/researcher relationship

In chapter 4, I explored power issues between young people and adults in schools. As such, I was aware my past position in school would impact in some way upon the relationships between myself and the YRT. I, therefore, undertook a continuous weekly process of discussing with them how they could shape the research by taking control of aspects of it. It was also important that this was not just a discussion; I demonstrated to them that I was also allowing them to take control. This was a delicate path to tread as this was my research, but as part of it I wanted to empower the YRT to be architects of their self-improvement. However, I was surprised by the extent of this impact on members of the YRT; it was not just about empowering them to take control, but also about developing their confidence and new skills and recognising this development in themselves. As YRT members reflected:

Katie:

"Empowerment has become a big part of this project for me especially as I have learned many skills which have allowed me to provide better support to the participants, also seeing the young person's confidence and knowledge around the subject of mental health in school develop and increase has ultimately improved my confidence and trust in myself." Poppy:

"As well as my confidence growing, the young person's confidence also started to grow as he began to open up more and became more trusting of me, this was very empowering for me." "Over my time doing the mental health project I feel my confidence has grown and developed in a variety of ways." Joanne:

"The success of this meeting was mostly down to my increased confidence in what I was

doing and also helped by the relationship we'd built"

The action research cycle (Figure 9) enabled me to build relationships with individual YRT members at our weekly debrief meetings and develop a relationship with the group at our weekly group meetings. The intensive process that developed through the mini-action research cycles enabled the relationship-building process to mature. As I discuss in section 5.8.3, I deliberately introduced a collaborative self-reflection process involving individual, followed by group, reflection. This, along with the process of the writing our academic paper (McPartlan et al., 2021) together, helped cement trusting relationships between the team and me. I would also suggest our relationships were further enhanced as I prioritised my relationship with them over my relationship with the school staff and the school itself. Again, this was often a delicate balancing act, but it was essential that the YRT could trust me. They knew I was an ex-teacher, still in touch and friends with the staff at the school. Therefore, it was essential that I actively showed the YRT I would put both the research and the YRT above these other historical relationships, something I had to do on several occasions.

This section has highlighted the role and importance of relationships within this research. The part that relationships played in ensuring young people's ethical voice was heard is an area that I will now explore further.

7.6 Ethical voice in this research

7.6.1 What do I mean by voice and ethical voice?

As an ex-teacher, I was aware that within the school system it was adults who made decisions and had the overwhelming voice within the institution. I wanted this research to be different; I wanted young people to have a voice where they could participate, be listened to and contribute to change in their school. I tried to enable them to express their beliefs and preferences (Sharma, 2009). This research was about empowering young people to influence critical decisions. It was, however, also about who should have a voice about the ethics, something I describe as ethical voice.

7.6.2 Who has an ethical voice in this research?

The concern for me was around the questions who within this research has or should have an ethical voice, who is listened to and whose voice is heard? As the leader of this research, it is clear I have an ethical voice, and through knowledge of school systems, and including school staff, I have enabled the institution of the school to have a voice. The YRT also need to be heard when it comes to ethical issues. However, in terms of ethics, in my experience, young people's views in schools are rarely considered. I sought for them to be involved and ensured we had conversations to discuss the ongoing research; I listened to them and took on board suggestions, altering some of the procedures as we implemented them (Hipolito-Delgado et al., 2022). Young people had a stronger ethical voice through the research cycle, as issues arose requiring us to make decisions to develop a strategy. Over the months, our relationship developed, so our discussions had greater significance and depth. In one of our weekly group meetings, discussing how to get more effective mental health teaching into the school, I proposed a non-confrontational strategy for an obstructive staff member with a traditional outlook when it came to young people and their place in school. I was taken aback, slightly shocked, but also delighted when a reticent member of the YRT basically called me out. She said:

"We have done the research; this is what we have found out and we have Mr X who is not happy. Will we allow a member of staff to continue working in the same way and disadvantage hundreds of young people? That is not fair on young people".

She made a valid point. We then spent considerable time discussing the issue and trying to resolve it. This incident reminded me of the importance of active listening within this dialogic process, and being particularly attentive to less forthright partners.

Developing a YPDAR project in schools is challenging as young people have to cope with adultism (Ginwright & Cammarota, 2002; Delgado, 2008; Shier, 2015; Corney et al., 2021) and, more acutely, teacherism (see 5.4.4 introduction to YPAR). Again, there is an ethical

aspect to this that was not covered through the traditional university process. My strong values-based approach to the research means that, as the researcher, I have a responsibility to ensure the YRT are not let down by the school or the process within the school. There were a number of occasions, as demonstrated above and also discussed as part of my relationshipbuilding process with the YRT (section 7.5.3), when individual staff took exception to what was happening or being suggested. On one such occasion, a member of staff argued with members of the YRT that the more creative, activity-based classroom approaches they were suggesting were unrealistic as "none of the children wants to be in the lesson" and because "they did not know how to behave when sitting behind a desk let alone doing activities". As an ex-teacher, I was shocked as these were issues that were professional and not to do with the research. I had to decide my ethical position. As a teacher, it would be very unusual to take the side of a young person (in my teaching career, this did occasionally happen but was rare). However, as a researcher, I had an ethical responsibility to the YRT. The headteacher and I had promised them that these findings would be implemented. However, it seemed there were staff who were not aligned with the school's wishes. I then spent time discussing the teacher's approach with the YRT, particularly how unprofessional it was and how we could move the project forward. My past teacher persona needed to be suppressed as I felt distinctly uncomfortable talking about a staff member, particularly with a negative opinion. This experience enabled me to reflect on my position as someone occupying the space inbetween, something I explore in section 5.8.1 (Corbin Dwyer & Buckle, 2018). Whilst I ensured that the YRT were clear that this was a confidential conversation and I trusted them not to gossip about what I was saying, I still felt uncomfortable. This was an example of my working with the YRT and using embodied knowledge (Craig, Jeong Ae, & Zou, 2016). We used our judgement based on perceptions, which I discuss in more detail in section 7.8.1.3. (Nussbaum, 1990). Reflecting on this has allowed me to clarify exactly my position within this work. I am a researcher with a responsibility to my co-research team and participants. This is captured in a short extract from my reflective diary:

"What a weird experience that was! First the feedback to staff where one of them was unprofessional and the YRT were the professional ones; a role reversal. Then debriefing with the YRT, I was unnerved by the fact that I had to carefully negotiate my way between an ex-colleague (someone I like and got on with) and my 'new' colleagues the YRT. As the first meeting was taking place, I could feel myself being torn and very quickly had to clarify in my head which side I was on. By the time we got to the debrief my head was clear and I

153

knew that my responsibility was to the YRT. Whilst I know I did not let the YRT down; I also know that I was having to process my feelings of discomfort at criticising my excolleague......wow."

This research and the approaches I have taken have enabled the YRT to have their ethical voice heard. As co-researcher, they have bought into the value base of fairness and justice; they expect that the research to be implemented where possible to ensure it creates better systems. Furthermore, and equally important, they have also found themselves in a unique position as this work, possibly for the first time, has given them a legitimate and transformative voice in the school (Hipolito-Delgado et al., 2022).

I will summarise by addressing the less-than-straightforward question of whether the participants had an ethical voice in this research. My experience of working with young people has helped me form an opinion that young people are often ignored, patronised and silenced within a school setting. Being true to my values, I wanted to use my research to change this status quo, and if I were to do that the participants needed a voice and to be heard. The youth participation in schools model (see Figure 35) that I introduce in the discussion chapter is designed to ensure there is a process whereby this happens. I disengaged myself from the participants by enabling the YRT to take responsibility for the data collection. This had the negative effect of reducing their access to me. We had a remote relationship, so the chances of them directly impacting ethical decisions were reduced. We set up systems enabling them to speak to the specialist pastoral worker, and they also met their YRT members every week. Whilst the YRT developed close and mutually beneficial relationships with the participants, there was no systematic provision for the discussion of ethical issues. The YRT made me aware that they occasionally had conversations that could be placed within this category. However, the concern of how participants may be able to have a greater ethical voice is an area that requires further exploration. Whilst the significance of whose ethical voice can be heard is important, it was also critical that young people influenced the shape of the research. This is something that I discuss in the following section.

7.7 Using a patient and public involvement (PPI) approach to enable young people to shape the research

Throughout the first few months of my PhD, I was challenged about my position within the research. In chapter 5, I explored my use of the term authentic research and how this related to my position as an ex-teacher and my desire for this not to compromise my findings. As my research shape started to develop, my first ethical deliberations began. A fundamental question I needed to address was whether young people who had known me as assistant headteacher of the school a year earlier would accept and trust me in my new role as a researcher. I intended that this research needed to reflect the thoughts and understanding of young people in the school; my knowledge of the school and its systems was different to young people who experienced it. To put it bluntly I made the rules, and they were expected to follow them. I wanted to ensure that any findings were not my interpretation of what young people said, but their own interpretation. To be able to do this would require the development of trusting relationships (see section 7.5), and the experiences of my previous school life could potentially hinder that.

I needed to start as I meant to go on and talk to young people to determine their views and preferences about how they would like me to approach the research. Mindful of the power imbalance due to my previous role, I aimed to attempt to deconstruct any existing issues. To do this, I arranged a series of meetings with groups of young people in school using a PPI process often used in health settings to improve the relevance and quality of the research (Hoddinott et al., 2018). PPI is defined as research being conducted either with or by patients and members of the public (Hoddinott et al., 2018). It is a flexible approach with the general aim of improving patient outcomes by involving them in the design of the research questions, methods of collection and analysis, and also decisions to do with the dissemination of findings (Hoddinott et al., 2018). Fundamentally, this is a research process ensuring outcomes matter to patients as they are relevant to those who will benefit from the work. Here was an opportunity for young people to contribute with knowledge gained from their own lived experience, making the research more equitable and ethical. Using this process, it became clear that their feelings of worth would be boosted through being included from the outset (Hoddinott et al., 2018).

Through this PPI process, the most important questions I asked were about communication. All young people involved in this research needed to be comfortable with the arrangements we made for the data collection, including whom they were expected to talk to. I had four one-hour meetings, two with young people aged 13 and 14, and two with sixth-formers aged 16 and 17. Over the four hours, we focussed on two key questions:

- How would those involved in the research want to be addressed?
- Who would young people in the research feel most comfortable talking to?

When I was teaching, I often felt uncomfortable when young people addressed me as 'sir'; I was far happier when young people used my name, 'Mr McPartlan' and, on rare occasions, 'Dave'. I would always address young people by their first names as a mark of respect for them. However, within the context of this research, I thought it important that the young people involved had the opportunity to define themselves. Were they students, pupils, teenagers, or adolescents? The 16- and 17-year-olds did not want to be referred to as 'teenagers or adolescents' as, in their minds, these terms had negative connotations. Many young people were uncomfortable with the terms 'student and pupil' as this defined them through their education status. They, therefore, suggested the term 'young person', something up until that point I had not considered. In my session with the 13- and 14-year-olds, I added this to the activities, and they agreed with the sixth formers that the term 'young person' should be adopted. This is noteworthy as one of the aims when initiating this research was to ensure young people were at the centre, with a significant voice within it. There was a substantial jeopardy to my integrity as a researcher if I did not talk to young people about such issues. It would have been unethical of me to assume how young people wanted to be addressed; I would have chosen one of the other terms, leading to resentment among some young people. In turn, this was likely to have impacted the relationship between young people and me, and possibly could have impacted the authenticity of the research findings.

The second question was more straightforward. I was open with the young people in the PPI meetings and explained my trust and power issues concerns. I asked them, "Who should conduct the research interviews?" Again, starting with the 16- and 17-year-olds, we worked together to develop three options. I could do the empirical data collection myself; I could get an outside researcher to collect data for me, or I could work with older young people from the school to collect the data. The younger cohort was clear that they would not be keen to talk to an ex-teacher like me, as there may be trust issues, and they would not be comfortable talking to an adult researcher who was new to them. There was unanimous agreement from all young people involved in the PPI work that the best model would be for sixth-formers to work with

me; they would be the ones to collect the primary data from the participants. The younger group believed their peers would be more likely to be open and honest talking to young people closer to their own age.

Reflecting on the PPI, an extra round of discussions with young people may have benefitted the process. As I suggested above, an exploration of ethical issues, including relationships, prior to the research's instigation would benefit the research's start. It would allow the researcher to set the tone with young people and commence the crucial relationship development between the parties. Whilst this consultation with young people helped shape the research, it was completed from a position outside the research. As the working relationships developed through the research process, I realised that an alternative ethical approach needed to be considered. In the following section, I will explore the applicability of relational ethics as an approach for this research.

7.8 Relational ethics

The premise of relational ethics is that ethical decisions and actions are made within a context of a relationship (Pollard, 2015). Much of the work around relational ethics has come from healthcare (Cook, 2012), although there are lessons that education can learn from this. Studies have explored how some relationships are better than others for providing healthcare (Bergum & Dossetor, 2005), in particular ones that follow the principle of a concrete other as opposed to a generalised other. That is where relationships are developed, and decisions are informed by learning from the relationship rather than by often oversimplified and universal knowledge. Bergum and Dossetor (2005) also identified a number of key interdependent principles (discussed below) underpinning relational ethics in healthcare, which YPDAR in a school setting can build upon. There has been a shift in healthcare from traditional thinking, positioning the nurse and patient as independent agents, towards a more integrated approach (Pollard, 2015). It is now accepted that ethical commitment, agency and responsibility for oneself and others develop through interactions involving relations between individuals (Pollard, 2015).

Similarly, as I discuss in section 5.4.4, YPAR sees occasions where young people are used in a tokenistic way (Alderson, 2000; Charteris & Smardon, 2019), or are seen as objects of research; procedural ethics are likely to dominate in such instances. YPDAR ensures that the

157

researcher and young people are interdependent and connected, so that ethical decisions are constructed through a collaborative partnership. This ethical commitment develops from relationships built on social situations and involving individuals with agency. They take responsibility for themselves and others within the research relationship (Pollard, 2015).

7.8.1 Fundamental principles of relational ethics

7.8.1.1 Mutual respect

Mutual respect is generated through responsibility for one another and has a role in rebalancing power dynamics (Pollard, 2015). Within this research, the YRT and I had different power; I had the power associated with being an ex-teacher and researcher, whilst they had the power gained from their lived experience as young people who, amongst other things, had attended the school for six years. The YRT also had the power to disengage from the process and pay lip service to the study, or even subvert it. The different power we had was what complimented, rather than excluded, within the relationship (Behabib, 1987). As the premise of these relationships was from a value base of equity and complimentary reciprocity (Pollard, 2015), we needed mutual respect. This developed through a common and non-opposition standpoint that acknowledged our phenomenological experiences.

7.8.1.2 Engagement

Building a relationship, a central pillar of YPDAR and relational ethics, was how I engaged with the YRT. This was about me being able to move my position towards the YRT (and the YRT towards the participants) (Bergum, 2012) so that we were alongside each other and engaged with the other (Olthuis, 2001). The engagement resulted from this real 'researcher-self' being present only as a result of our relationship (Pollard, 2015). A significant aspect was gaining an understanding of others' perspectives, complexities and vulnerabilities. As the researcher, if I had tried to imagine myself in their place, I would have risked discounting their phenomenological experience. Instead, by working alongside them, I identified their unique needs, limitations, talents and capacities (Pollard, 2015), which enabled me to develop positive relationships and work on an equal epistemological level with them to support them to improve their own and others' lives.

7.8.1.3 Embodied knowledge

Embodied knowledge concerns the distinction between an intellectual decision-making exercise and decisions made through cognitive, affective and emotional experiences gained through our lived experiences (Pollard, 2015). This was the desired way of working for the YRT and me. It was about making judgements based on perception, as it was impossible to apply systematic universal rules to each and every situation within the research (Nussbaum, 1990). As the research developed and our relationships matured, how we interacted and made ethical decisions changed. Earlier in this chapter, I discussed how, on occasions, I had to decide how to respond to inappropriate comments by staff. As a teacher, I would not have commented to the YRT about it, as this would have been seen as unprofessional. However, from a relational ethics approach, I referred to embodied knowledge and opened a conversation with my research partners about these inappropriate comments (Tomaselli et al., 2020).

7.8.1.4 Environment

As social beings, and in relation to our environment, we are connected to others and to wider society; who we are, our perceptions, values and concepts are, therefore, a product of our environment (Sherwin, 1998). However, within relational ethics, there is a subtle difference in that we are the environment (Bergum, 2012); we cannot be separated from others, and this mutuality defines this as the key aspect of the environment. This mutuality was how the power differentials were redistributed. This aspect of relational ethics was further complicated by three constituents within its relationships: the YRT, the research participants and me, the researcher. Mutuality existed between me and the YRT, as well as between the participants and the YRT. Further research will be required to look at whether mutuality was in any way transferred between the participants and me via the YRT.

7.8.1.5 Uncertainty

As discussed throughout this chapter, values are at the centre of this research. However, these values often brought dilemmas, difficult decisions and uncertainty (McPherson et al., 2004). With uncertainty came the realisation that there may well be no perfect answer to a given problem; instead, a best fit was sought within the context of unique situations (McPherson et

al., 2004). Rather than power, information and ideology, what was required was humility, understanding and relationships (Bergum & Dossetor, 2005). The knowledge constructed through context when we worked as a team in this way was often incomplete. We found the process challenging as we worked to draw our findings together and make recommendations to the school. Therefore, we used a self-reflective process including various methods such as discussion, writing, my collaborative self-reflection tool (Figure 13), and negotiation to address the tension regarding what needed to be done (Pollard, 2015). Whilst the key principles above are fundamental to relational ethics, I believe within the research context time and the nature of the dialogue are also highly significant.

7.8.2 Time and contextual experience

In addition to the fundamental principles Bergum and Dossetor (2005) identified, a further element yet to be discussed is time and contextual experience. As discussed in chapter 4, power is important when considering school research. Young people are subject to teacherism and may be regarded as oppressed (Cammarota, 2002; Delgado, 2008; Ginwright & Shier, 2015; Corney et al., 2021). It is unrealistic to imagine a teacher/researcher from within the school, or a researcher from outside, can turn up and expect young people to engage and trust them immediately. The complexity compounds the problem in relation to the power differential between the adult researcher and young people in the school (Schäfer & Yarwood, 2013). Relationships take time and effort to develop. Whilst relationships are never static, and constantly change in one way or another, this nascent relationship could and probably should take many months to develop. My experience was working for many months with the same group of young people. Some of the conversations we had and decisions we made in the latter part of our research would not have been possible earlier. Whilst the time, in relation to the number of interactions, discussions, arguments, and decisions, cannot be quantified, the contextual experience of the relationship is all important. The researcher held the key to this by living out the values in everything done within the research. In this way, young people were in no doubt of my intentions and, crucially, their own place within the research.

7.8.3 The nature of dialogue within relational ethics

Whilst dialogue is part of YPAR, as I synthesised it with CMM, I realised it was a fundamental and essential constituent of our way of working. It was the element that ensured young people were taken seriously as collaborative partners, rather than as tokenistic objects, as sometimes can happen in YPAR (Alderson, 2000; Charteris & Smardon, 2019); dialogue is also fundamental to the success of relational ethics. Relational ethics are not about following rules or guidelines, nor are they solely about solving problems on the ground (Hilppö, Chimirri, & Rajala, 2019). They are about being attentive and responsive to others within the relationship (Austin, 2007). To do this, the researcher must commit to being selfreflexive in relation to the assumptions and predispositions of others within the relationship. They also require self-awareness on behalf of the researcher so there is an appreciation of the relationship as defined by those whose lives are being studied; this, in turn, defines the researcher (Hilppö, Chimirri, & Rajala, 2019). This means relational ethics are about being open and available to others. They are about responsivity and preparedness to listen to others and take their concerns seriously. There is a need for sensitivity and inclusivity through open dialogue, but this does not mean that anything goes; it is about being in this together (Austin, 2007), as the research builds through honest dialogue, enabling all involved to bring their reflections to the table. Openness is established through the development of the research relationship building during and beyond the research project; it impacts the life of the researcher and the researched (Ellis, 2007).

Relational ethics within research involves walking alongside your research partners as their supporter and their advocate; it engages them, as a participant, *with them* rather than as a spectator *of them*; it is about the research, but it is also about the everyday life of those partnering in the research (Maffesoli, 1989). As researchers are not only *in* the world of their research partners but also *of* the world (Barad, 2007), and as social relationships have implications relating to contextualised power balances (Erickson, 2006), relational ethics move beyond preventing harm. Instead, the focus should be developing a partnership that will improve the lives of those being researched and possibly that of the researcher. It is thus a symbiotic relationship.

7.9 Chapter summary

In this chapter, I have set out the ethical journey this research has taken me on, from my introduction via the university's ethical procedures ensuring the safety of all involved to my realisation of the central role relationships has played within the research. My position as a social constructivist has enabled me to view ethics through a collaborative mindset, supporting my decision to promote the position of the YRT. I have explored how the principles of relational ethics have influenced my research and have positioned dialogue within my ethical approach.

Part 3. What did we find out and what does it mean?

In parts one and two, I have laid out the context of the research and have both described and explained how and why the research was carried out. Part three of this thesis will show what I learned in my research, and what I hope this means for future practice. It also looks at the limitations of this study and how further research is required to develop ideas raised in this thesis.

Chapter 8: Findings

8.1 Chapter introduction

This chapter introduces the findings and is divided into three sections. Firstly, I present my findings concerning the whole school mental health strategy. I then present two sets of findings on my new methodology, YPDAR; its impact on young people and on school character. This is then followed by a chapter containing detailed discussion as well as limitations to the study.

8.1.1 Development of my discoveries in relation to the whole school mental health strategy

My findings about the whole school mental health strategy resulted from the five-part collaborative process between myself, the YRT and the participants, which are highlighted in chapter 6. Our work was conducted within the context of praxis; learning from action, reflecting and building forward. In addition to this process, I also conducted more formal reflective exercises, including developing frameworks to categorise and help apply the findings. This was carried out with the YRT, who mirrored our reflective meetings with the participants to sense-check our findings. After completing frameworks built through praxis, this ongoing cycle of action and reflection enabled us to develop our findings, which I have summarised below.

8.2 Findings

I will write the following section in a neutral voice, and will not use any of the texts outlined on page 10 of my introduction. The findings related to the school mental health strategy were created collaboratively between myself and the YRT, whilst the methodological findings were created by myself.

Throughout this research, and as I highlight in 5.8.1, I wrestled with my positionality within it. I was always conscious that I had developed a methodology, YPDAR, which empowered young people and subsequently took some of the control away from me, the researcher. In exploring the ethical considerations in Chapter 7, I look at the importance of the relationships within the research process. Section 7.5.3 explores what was involved in handing power to

the YRT and how this, in turn, reinforced the development of trust and confidence in the methodological process. I was developing what I call the methodological voice of the YRT, see section 7.6.2. This entailed taking on responsibility to ensure they were not let down in the research process. I set them up as partners in the research working on an equal epistemological level, and I had to ensure this principle was adhered to. We worked on the analysis of the research together and collaborated to develop the findings. When we settled on the findings, I asked the team to write them up and present them to me. They presented them in bullet point format, which created a problem for me. I was aware that traditionally findings chapters were often substantial works rarely presented in bullet point format. I, therefore, had to choose to either abandon the work of the YRT and re-write the findings in my own way or accept their work as it was. To be true to my inclusive values and the YPDAR methodology, I had little choice but to use the findings as they had been presented; I had to have the courage of my convictions. This research was built around a collaborative process where we worked as co-researchers. I, therefore, adopted the YRT recommendations and used their bullet point format. Having made this decision, it then made sense to use this same bullet point format for my own methodological findings also.

Whole school mental health strategy findings

The focus of the findings related to the whole school strategy concentrated on three areas, two of which were interlinked. Firstly, young people reported that whilst the school had set up referral systems for them, this entailed approaching staff for help. This was problematic for some young people due to trust issues. This links with the second issue around trust. The school strategy relied upon school surveys as a self-awareness tool for young people and a gauge for the school. However, young people did not trust the survey process. Finally, young people also reported dissatisfaction with how they were educated about mental health in school.

Oppressive school structures encourage teacherism, which has a detrimental effect on staff/young person's relationships.

- *i)* Good trusting staff/student relationships are of the utmost importance because:
 - Young people requiring support will only go to those staff they trust.
 - Young people requiring support prefer to go to staff they know and trust.

- Young people requiring support will not trust staff they feel are judgemental and display teacherism.
- *ii)* Pastoral staff are often the preferred choice of staff they will choose to go to because:
 - For many young people, these staff have dealt with personal/family issues previously, and many pastoral staff have followed cohorts of young people through the school and are likely to have a more trusting relationship with an individual.
 - Young people do not have a perception that pastoral staff have the same conflict of roles that class teachers may have.
 - These staff give more opportunity for confidentiality than teaching staff do.
- *iii)* The member of staff responsible for the pupil premium young people has built one-to-one relationships with many from this cohort and was the most popular choice for someone to confide in because:
 - Young people trust her.
 - She does not judge young people.
 - Young people are comfortable talking to her as they have built a relationship with her.
 - She is consistent, kind and understanding.

Trust

School surveys are not trusted by young people because:

- They don't know who will see the information and what those people will do with it.
- When completed in classrooms on computers, they feel others can see what they are entering.

Curriculum

 Personal development lessons should be used more effectively to educate young people about mental health and reduce the stigma around mental health by:

- Focussing on the recognition of the signs of poor mental health, self-help strategies and whom to go to for help if needed.
- Making mental health a frequent and regular topic in lessons.
- Making lessons positive, fun and, where possible, personal to the individual.
- ii) Other subjects don't contribute enough to the education around mental health and reducing stigma because:
 - They don't explain the part they have to play in supporting mental health (e.g., PE, drama and art).
 - They don't reference mental health during routine lesson time.

By synthesising YPAR and CCM, I developed my new methodology, YPDAR. There were two primary beneficiaries of YPDAR, young people with whom I collaborated and the school within which the research was carried out. My methodological findings are presented below.

Methodology development and its impact on young people

The development of a new methodology, YPDAR, has enabled young researchers to develop their agency and build their confidence, trust, and empowerment.

- i) Developing a school-based research team empowers young people and improves their agency to act by:
 - Giving them support to develop and lead research projects across the school.
 - Mentoring them as they choose appropriate research methods.
 - Demonstrating the use of reflexive practice within the research process.
 - Supporting their development as critical thinkers.
 - Developing their confidence and trust in school staff and structures.
 - Creating a mutually-constitutive duality as the agency young people develop impacts upon the school structure that, in turn, puts their trust in what young people suggest. This further develops the trust, agency and empowerment of young people.
 - Redistributing power dynamics.
- ii) Developing a school-based research team builds researcher confidence by giving them the opportunity to:
 - Connect and engage other young people.
 - Collaborate with school staff, working on an equal footing with them.
 - Prepare resources and run research sessions.
 - Develop their communication skills.
 - Realise their own potential and further develop their sense of self.
 - Develop knowledge in key areas (school mental health).
 - Participate in academic forums through paper writing, conference presentations and wider research group membership, thus developing new forms of social and cultural capital.
- iii) Developing a school-based research process builds researcher trust with the school by:
 - Putting a system in place that enables young people to bring forward their own ideas for school improvement.
 - Giving young people the time, support, and resources to build meaningful relationships with school staff and structures.
 - Engaging in conversations regarding their ideas about school improvement.
 - Implementing findings from their research.
- iv) YPDAR can assist the development of young people's social and emotional skills and capabilities by:
 - Creating opportunities for young people to rehearse emotional management, empathy, initiative, problem-solving, teamwork and responsibility in real-life situations.
 - Supporting young people to develop a broaden-and-build schema through exposure to situations that require the domains of socio-emotional skills.
- Working collaboratively with a young research team to develop research furnishes researchers with the skills to understand and develop appropriate research methods by:
 - Demonstrating different methods and supporting them to experiment with their own ideas.

- Encouraging them to use creative techniques.
- Demonstrating, supporting, and encouraging them to be reflexive in their approach to the research work.
- vi) Through research projects, 6th formers have a key role as mentors to younger students in school because:
 - Researching issues that are important to young people develops real-world situations that enable meaningful dialogue between groups of young people.
 - They are looked up to by younger students and can inspire confidence in their mentees.
 - Working with the younger participants on meaningful school research can shift the narrative around them and contribute to improvements in their sense of self and self-identity.
 - They can develop attachment-like relationships that make a positive contribution to how younger students feel about themselves and the school. This, in turn, improves their experience of and engagement with the school in addition to improving their general confidence.
 - They can provide a trusting outlet for younger students.

Methodology development and its impact upon school character

The implementation of my new research methodology, YPDAR, can enable schools to improve school character.

- i) By adopting YRTs in school to collaborate with staff and other groups of young people, the school benefits from:
 - The opportunity for young people to be involved in counter-cultural research.
 - A more balanced power dynamic between staff and young people with the potential to improve staff/young people relationships and improve young people's agency.
 - Greater trust between young people and school structures.
 - A potential reduction in young people's perception of teacherism.
 - The opportunity to support young people as critical thinkers and so engage in policy with the potential to transform the school from one supporting banking education to one with a problem-posing focus.

The development of YPDAR frameworks safeguards the integrity of both the research as well as the parties within it.

- i) The YPDAR framework sets the parameters for this research in schools by:
 - Making the principles and values explicit.
 - Centring dialogue and equity of relationships within the research.
 - Ensuring multi-voice analysis through a process of crystallisation.
 - Positioning relational ethics at the heart of the process.

Chapter 9: Discussion, limitations and challenges

9.1 Introduction

My research has chartered the dichotomy that is our schools. Schools are for young people; however, they do not necessarily cater for all of them. This discussion chapter will therefore concentrate on how my research can contribute to developing ways schools can further support young people's mental health. My emotional mental health framework and school mental health model are specific contributions to knowledge in this area.

To do this, this chapter will explore what my research means for future practice and how it has contributed to new knowledge. This will include several interconnected areas, all of which are linked by one common denominator, young people. My initial discussion will investigate the process through which the research was created, developed and completed. This is divided into two parts which I set as aenigmas. The first focuses on the literature around power issues and their impact on young people's agency, identity and social and cultural capital. The second part, linked to the first, is about the English education system, how it has developed through time and how socio-political influences have shaped it. I then reflect upon how this impacts young people. My research confirmed my view that schools' expectations of young people are as passive (Freire, 1970) recipients of education. The system in which young people are forced to fit in ensures they have little active voice, and therefore have little chance of real growth and development.

Following discussion related to the two aenigmas, I explore how the synthesis of CCM and YPAR created YPDAR, a methodology for use in schools, which I position as a solution to the two aenigmas above. YPDAR centres young people as transformative activists and change agents as they contribute to improving their school community and lives. I present YPDAR as an alternative to Freire's (1970) banking educational system. Rather than seeing young people as objects used within an oppressive system, this research positions them as capable young people who can use their subjective knowledge to overhaul a broken system. I will therefore explore how young people's participation and rights need to be included when considering consultation, collaboration and research with young people in schools. I conclude the chapter by applying this learning to my youth participation in schools model, a further contribution to knowledge.

9.2 The process

Each of my aenigmas is explored through a revisiting of the findings in light of the literature reviewed, something I describe as a praxis. The literature review in chapter 4 focuses on the role power plays in the lives of young people in school. As I discuss below, power structures dominate education, but power's negative impact on some young people is often unseen in this area. In particular, it reduces their epistemic agency as well as their agency to act in school. In addition to this, the participants, coming from the PP cohort, are more likely to have a poor sense of self and may be challenged by concerns around their identity. The final piece in this complex jigsaw, is how this same cohort may find school challenging as they may not have access to the social and cultural capital required to succeed. I, therefore, developed Figure 26 as a lens through which to view these young people in school today. This suggests a deficit perspective to young people and schools. However, having completed the empirical data collection, and having seen the benefits of YPDAR on the young people, I then returned to explore these relationships through an asset-based lens, which resulted in Figure 27, something I discuss later in this chapter. These common threads run through the above elements of the aenigmas. Within the context of the state school system, this research (section 4.9) has reflected on how some young people lack meaningful power and agency, and their identity is often being set by others. When this happens in a deficit way, their sense of self can be challenged, and a negative script can be set. Their capacity to generate meaningful social and cultural capital is hindered, and they are challenged both in life and school.

Similarly, our education system is a legacy of this country's class-driven past, which is reflected in how it imposes upon our young people. From my teaching experience, I suggest schools are largely anti-dialogic organisations that take a tokenistic approach to listening to young people's views. This can result in aspects of some young people's personal development being curbed. However, this research has also demonstrated how schools can empower young people and help develop their sense of agency which, in turn, helps improve young people's sense of self. It also plays a part in challenging the often-negative script ascribed to them. Chapter 2 touches on how neoliberal policy and other external influences have resulted in schools being fixated on assessments at the expense of broader cultural and social experiences. As the school curriculum is often narrowed and constricted by the focus

on assessment, young people are less able to benefit from the social and cultural capital that could help them. The asset approach that this research takes can be a counter to this.

9.2.1 A deficit approach

In section 7.4 I highlighted that young people are oppressed through adultism in society and teacherism in schools. This literature is supported in my research by the example I cite in chapter 7, when a teacher acts to undermine the legitimacy of the YRT when they report their findings to her. Chapter 4 highlighted some young people's difficulties in engaging with and taking responsibility for their mental health. This exemplifies how imbalances of power, between adults and young people in schools, can lead to young people being overlooked, dismissed, or even dominated by adults, something I review in the literature in chapter 4. Therefore, I suggest Figure 26 as a lens to explore how this can impact my research.

Connected to this are two further areas for consideration relating to adultism and structures of oppression. As discussed in section 4.4, whether structuration or morphogenesis, agency is interlinked with societal structures such as schools. My original assertion was that these two theories agreed that structure and agency were linked. However, reflecting on my findings, it is now clear that this research was an example of structuration as they are a "mutually constitutive duality" (Jones & Karsten, 2008, p. 129). In the findings chapter, I posit that this research developed an approach whereby the school started to trust the YRT, and they, in return, began to develop trust in the school.

I had created a situation whereby the research process empowered the YRT and developed their agency. The trust young people had previously not had in the school started developing as they completed the research and put recommendations forward. Trust development was a symbiotic relationship from a school's perspective as they were engaging with and acting upon suggestions made by young people about an area of school improvement. In this instance, it is hard to see how trust between the school and young people could have come about in isolation as each side required the other. Furthermore, I suggest that research of this nature, if taken on systematically by the school, can positively impact the school's character. When developed further, I posit that a school with an emerging culture of dialogue with young people is likely to afford these young people greater agency in their school lives. In this way, a mutually constitutive duality has been created as agency and structure rely on each other to survive.

As I exemplify in chapter 2, the marketisation of schools since the 1980s has seen a shift in school values that, in turn, has impacted how teachers operate. Examples of such teacher behaviour surfaced during the research in two ways. Some teachers refused participants permission to miss a part of a lesson, citing work pressure as the reason; the participants were at least a year away from any exam classes. This was despite explicit instructions from the headteacher to the contrary. Similarly, early in the research, a number of the YRT withdrew from the project as they were being pressured by teachers who suggested they consider whether they had time for the research.

Young people were pressured into following the teacher's advice, which was given without due care relating to young people's wellbeing. It is how the school controls young people by imposing its will on them. These are the behaviours of schools striving to ensure their exam results meet expectations. In my experience, schools and staff who work in this way undermine many young people they are aimed at helping. While the focus of Figure 26 is the four key areas discussed in chapter 4, it is important to note how societal and school structures frame the theoretical principles. The power structures at play in society and schools are about how adults perceive and treat young people, something linked to hermeneutical injustice. I argued in chapter 2 that England's education system is an unfair and antiquated hierarchical structure that benefits the more affluent. It employs a deficit approach to working with young people. As demonstrated in Figure 26 below, both examples are related to power and its related structures, including teacherism. They were denying young people an opportunity to develop many skills, but also to develop their agency and their sense of self.

My findings identified what was being gained by young people was a trust and connection with the school, the development of potential and a sense of self, opportunities to develop skills, and social and cultural capital. I support my findings further with observations I made over the research period. In the first few meetings, the YRT wanted to be spoon-fed; they kept asking, "how do we do that?" or "what do we do next?" By the end of the research, they were challenging me, making critical reflections about our practice and presenting their findings at conferences. The participants and their YRT members were both critical and resentful towards the teachers who limited their involvement in the project. I suggest that they would be unlikely to trust these staff moving forward, and thus reflecting the findings in this chapter related to young people/staff relationships. There are lessons for schools (and researchers) to learn about communication with staff. All staff knew of the research and the expectations on them; however, I am less sure that they had a clear understanding of its aims and the potential benefits it could bring both young people and the school.

As I reflect on my journey of becoming a researcher, I have recognised how the processes of learning take time. The findings of this study have demonstrated that when young people are given time to complete research, such as the YRT and participants, they can flourish and problem-posing education results (Freire, 1970). Unfortunately, teachers are being forced into fixing young people by cramming them with tests and extra classes, and trying to force their development. In contrast, I see a banking system in education (Freire, 1970). I suggest that my research has lessons for policymakers, teacher trainers and educators alike.



Figure 26: Inability to take responsibility: a deficit-based tool

9.2.2 Barriers to research success in schools

As discussed previously, the English education system is a legacy of the class-based society that has evolved in this country. As I progressed on this research journey, I became aware of potential barriers to its success. One of the most significant barriers to effective and progressive research with young people was adultism. Which promoted the deficit-focused view of children and young people (Ginwright & Cammarota, 2002; Delgado, 2008). Adultism, which includes a lack of positive social norms related to young people and laws delegitimising young people, is further compounded by adult values that are internalised as negative beliefs by young people themselves (Flasher, 1979; Kennedy, 2018). In the ethics chapter, I suggest that it is possible to sub-categorise adultism further in relation to a school setting. As evidenced in the findings, I argue that a specific type of adultism in school should be described as teacherism, where teachers use their position and power to subjugate young people. I explore this in more detail and define it in section 7.4. Many young people feel totally disempowered and delegitimised within a school setting as there is a compulsion for them to attend an institution in which they often feel distinctly uncomfortable. Experiences of adultism are magnified through teachers shaping and supporting oppressive school structures. Teacherism has developed historically alongside adultism and is now a dysconscious act for many teachers; they behave towards young people as society, and the school's management structures, expect them to. The term dsyconscious refers to an uncritical habit of the mind that includes perceptions, attitudes, assumptions, and beliefs, and justifies inequity and exploitation by accepting the existing order of things as a given (King, 1991).

I should have prepared the teachers to be involved in the meetings; in hindsight, that was a mistake. In chapter 7, when I explore ethics within the research, I refer to an incident when a teacher became extremely defensive as the findings were explained. As I now explore with the Academy Trust how the next stage of this research will work, I need to be cognisant of this as a potential issue. One of the challenges moving forward will be how schools and their staff can be prepared for such situations. As I discuss in the limitations, all in the school community need to be aware of such research projects and their wider aims. The attitude of the member of staff suggested that the contributions of young people were incompetent and untrustworthy, and therefore the adult/teacher would impose what they thought was right; and young people were ignored (Schelbe et al., 2015). On reflection, my understanding of the importance of trust has developed since the end of the project. I was determined to encourage

the development of relationships between the YRT and participants, as well as between myself and the YRT. I also aimed for all young people involved to develop confidence and trust that the school would take their work seriously. However, as I alluded to above, I should have acknowledged the importance of the school trusting the YRT. The incident above, where I believe the teacher was unprofessional, was partially down to the fact that I neglected to prioritise trust development, from the school and staff in the young people. I have therefore concluded that, for this research to be successful, trust must be a two-way process; schools also need to play their part.

Schools must be committed to the research and cognisant of the potential for such studies to be culture-changing. To ensure research of this nature is more than tokenistic, schools must commit to prioritising the work as a valued school improvement project. This will need to include continued professional development for all staff and the regular celebration of what has been achieved by the young researchers to their peers. To facilitate this, and as the potential research facilitator and the teacher-researchers supervisor, I have offered to share the theory and background philosophy with all school staff. I have suggested regular updates be made available to staff from the teacher-researchers and myself. This is about slowly changing the school's character through a process that drip-feeds information, celebrates success and demonstrates the strengths and abilities of young people. I hope that ensuring staff are well-informed, interest and conversations between staff, young people and researchers working in schools need to be aware that unless schools are fully committed to working in this way, and all staff adequately prepared, trust is liable to break down (from either direction), meaning the likelihood of success will also be limited.

Teacherism is a more extreme example of adultism because the structures of the school support teachers. In particular, a school's policy will reinforce how the teachers can impose their will on young people. One such policy implemented in English schools after perceived successes within the criminal justice system is zero tolerance for what schools deem inappropriate behaviour. Despite the fact that research from the United States of America (USA) (Skiba, 2008) found that the suggested successes of zero tolerance in schools were, in fact, ill-founded, it seems that many secondary schools in England have gone down a similar road of zero tolerance behaviour policies. However, research has shown that young people in schools who use these policies feel victimised, and believe that once identified by teachers as

someone who does not follow the rules, they will be subject to a vicious cycle of sanctions (Nassem, 2019). These policies further alienate already-disillusioned young people who often find it a challenge to fit into schools. This is important in the context of this research as my findings show that some young people already lack trust in the staff. They find the school systems that create or further promote teacherism work counter to the positive school character that promotes openness. Nassem's (2019) research goes further. It suggests that the policy, and the attitude it creates in the school, allow staff to ostracise and humiliate young people, particularly those with disabilities and learning difficulties. To compound things further, the research findings also highlighted that disadvantaged young people, often from working-class backgrounds, were disproportionately impacted by such policies.

The example above strengthens my argument that teacherism is a consequence of adults having their power base strengthened by the policies and structure in which they work. These are the very young people who, from my experience, can lack the cultural capital to fit in, enjoy and fully benefit from attending school. As discussed in chapter 3, neuroscience has shown that adolescence is a period whereby the brain develops and that risk-taking is more likely. The needs of young people at this time are more likely to benefit from adult support, identity negotiations and academic self-efficacy (Skiba, 2008). I have been involved as a teacher in numerous schools throughout my career and have observed elements of teacherism in them all. My research highlights the importance of creating other ways of knowing. I suggest that a zero-tolerance behaviour policy is a "structure of oppression" and supports a banking concept of education (Freire, 1970, p. 47). In contrast, the type of research that I have been conducting gives young people problem-posing education that has the potential to transform these structures of oppression and, in the process, enable the development of new knowledge (Freire, 1970).

I used Gramsci (1971) to demonstrate how power plays out across society, alongside his ideas of cultural hegemony. The same could be said for struggles in schools. As I demonstrate below, the findings support this view in that I set this research as counter-cultural. What has emerged during this investigation is a form of research that has also been able to counter the oppressive structure young people find themselves in today. Young people have little power in society or school, and their role in this research has empowered them and given them the opportunity to create change within the organisation. By offering young people this chance to get involved in the research, I played a part in repositioning them as

epistemic agents by a soft redistribution of the hegemonic school power (Ko & Krist, 2019). This was done by intentionally opening up dialogic spaces, and thus allowing young people to become co-collaborators in knowledge (Ko & Krist, 2019). I deliberately use the term soft in relation to the redistribution of school power, and suggest that any power shift needed to be through evolution rather than revolution. In my experience, schools are not revolutionary places, and whilst I have observed numerous school leaders promoting concepts linked to critical pedagogy, few seem to support their implementation actively. This may be due to critical pedagogy being seen as a revolutionary concept.

Shifting power can create uncomfortable positions in schools built around implicitly designed systems (Ko & Krist, 2019). To change systems directly would have required an enormous values shift. However, collaborative research, such as this, gives opportunities for a soft shift in power that redistributes epistemic agency (Ko & Krist, 2019). My findings support the approach encouraged by the literature. The participants and YRT reported increased confidence, empowerment, and a sense of agency as they developed their trust in the school. I suggest this was the beginning of a redistribution of the hegemonic school power. This is reflected in comments made by a YRT member in section 7.5.3. I see the soft power shift, through student-led school improvement research, as a contribution to new knowledge. There have previously been young researchers in schools, but my research has positioned it as research that, if done systematically, can change the hegemonic character of the school.

The findings in chapter 8 also highlight the success of the research in this respect. By collaborating with the YRT and supporting them, they have become more able to use a reflective approach and develop their critical thinking. The ability to be a self-reflective critical thinker is a skill that helps prepare researchers to recognise and help transform oppressive structures (Clark, 2010). The YRT, through this research, were at the beginning of a transformative process of changing school structures. Maybe, more importantly, it was also about transforming themselves. This has resulted in young people who have grown in confidence as researchers. We jointly published research findings as an academic paper and have co-presented these at a conference.

I ensured the YRT had the space and voice to contribute, and the audience and influence to make a difference (Lundy et al., 2020). This was about developing a new authority or hegemony with young people at the core, with the intention of contributing to dismantling the

adult-dominated hegemony that has limited the powers of these young people (Cheney, 2019). The YRT recognised their development and acknowledged the role the research has played in their self-confidence, as reflected in their comments in section 7.5.3. I recognise that the YRT were presenting at an adult-led conference, and I realise the limitations of this. As a YPAR special interest group (SIG) member in the Action research network of the Americas (ARNA), I am engaged with academics from the USA, South Africa, France, Italy, Portugal and Greece. We are developing forums for young researchers from all over the world to meet (initially online) and learn from each other by sharing knowledge from their own participative research. The ultimate aim is to bring these groups together for their own conference. In this way, our ultimate aim is to establish a mechanism enabling young people to collaborate with minimum interference/influence from adults.

To summarise, I return to the inequalities of our class-based system and reflect on how it deprives some young people of the social and cultural capital often required to succeed in society today. The status quo is protected by societal and school structures that privilege the middle and upper classes and deny many young people the opportunity to thrive in society. As I discussed in chapter 4, these may well include schools that I have described as technologies of oppression (Stuart et al., 2019). I believe that the school and societal approaches to some marginalised young people hinder young people's ability to take responsibility. This deficit approach tool is summarised in Figure 26, that is based upon this discussion. In chapter 4, I discuss in detail the impact power differentials and the opportunities young people have for social and cultural capital can have on their identity and their agency to act. This research has given both the participants and, more particularly, the members of the YRT an opportunity to develop different forms of social and cultural capital. The writing of an academic paper gives these young people an opportunity they are unlikely to have had. It broadens their educational and cultural scope and maybe the extra ingredient that helps prise open a door of opportunity for them later in life. Likewise, presenting at a conference gave them a new experience; they rubbed shoulders with academics and professionals from different worlds. It gave them new experiences away from their home and school environment; opportunities to develop social and cultural capital away from the norm.

9.2.3 Power, structures and agency

In section 4.4 I explore how agency intersects with power relationships. Oswell (2013) suggests that the concept of agency should be viewed through a relational lens; it exists not within an individual but in the context of relationships with others. Horgan et al. (2017) go one step further, suggesting that young people are relational beings impacted by their environment. The views of Oswell (2013) and Horgan et al. (2017) are not mutually exclusive. Indeed, my findings show that young people involved in this research improved their trust and confidence in the school, and through this their agency to act. This came about as the power dynamics shifted from school and staff towards young people.

Below I highlight the key findings in relation to the importance of young people/staff relations and, as such, I agree with Oswell that young people will have an increased agency to act if they trust a member of staff. Horgan et al's. (2017) view that choosing to participate (agency to act) is supported by my suggestions in the previous section that schools that develop an environment of open dialogue with young people are more likely to support young people's agency to act. In particular, the relevance of societal structures and schools takes on great significance in this respect. In the previous section, I discussed schools as structures of oppression and how teacherism reinforces this concept. It, therefore, follows that some young people in schools are impacted by the oppression of both the structure and the school staff, many of whom will display characteristics of teacherism. This is about the imposition of power by teachers on young people and, as discussed in section 4.4, some young people are less likely to be able develop their sense of agency within this school context when challenged by a non-pro-social power-based approach. This links to my findings in Chapter 8: Findings suggesting the importance of young person/staff relationships in relation to young people being prepared to confide in staff should they need support for mental health issues. Young people's agency is paramount if we want them to come forward and ask for help. If they do not have a trusting relationship with a school staff member, it is unlikely that they will seek help. This new insight gives educators an alternative lens through which to view a school mental health strategy, as I found that young people were unwilling or unable to engage with the staff they did not trust. If some young people do not trust adults in schools and therefore will not come forward, we need to find an alternative to support them. We need to focus more on relationship development in schools and ask young people about

their preferred option before then supporting them to get it. This area requires further research.

As I emphasise in the deficit tool, Figure 26 above, antiquated social structures reinforce the disparities between sections of society. Also, as I have highlighted, this is seen in English schools. They combine to undermine young people as their self-identity is weakened, and they need more power, agency, and capital assets to take responsibility in areas of their life both in and out of school. However, this research has demonstrated that it does not need to be this way. As I will demonstrate below, my research took an asset-based approach to collaborate with young people.

9.2.4 An asset approach

One of the challenges to ensuring young people's development, is society's (and school's) deficit approach. We live in a culture that wants to identify those in need, what is needed and what provision is required (McCashen, 2014). This approach is risky as it has the danger of developing a deficit discourse that further marginalises, oppresses and disadvantages those it is trying to support, and often blames them for problems in society (Pitzer, 2013). This discourse is ably reinforced by the media, who take great delight in reporting young people as the problem (Tyler, 2014). The media messages are starkly portraying young people as delinquent and criminals with the need to be controlled (Stuart & Perris, 2017). This, in turn, then supports the deficit agenda that young people need fixing. What is required instead is an asset-driven approach.

YPDAR was designed to ensure that the YRTs' involvement enabled them to participate fully in this research as co-researchers and thus ensure their rights as contributors to their own futures. As the research process developed and the findings emerged, I became increasingly aware of the research's positive impact on the young people involved. I, therefore, returned to my original deficit-based tool (**Error! Reference source not found.**) to develop an assetbased one (Figure 27). The findings, (see figure 8.2), were clear that developing a schoolbased research team empowered them and therefore started to rebalance power in school. This, in turn, enabled young people to develop their agency to act. The findings showed that the research process boosted young people's confidence through the development of their socio-emotional skills. It also improved young people's trust in the school. These processes

combined in a way that enhanced their sense of self and grew their self-identity. The following sections will further explain and explore the evidence and theory underpinning the asset-based tool (Figure 27).

An asset or strength-based approach focuses on what people can do, rather than what they cannot do (Stuart & Perris, 2017). As discussed in chapter 4, the theory of capital and its unequal distribution (Bourdieu, 1977) is closely linked to privilege and disadvantage. Furthermore, capital has also been connected to one's possession of assets (Mathie & Cunningham, 2005). Therefore, developing an asset-based approach to supporting young people can be seen as a way of championing the cause of the marginalised and, as such, is socially-just work (Stuart & Perris, 2017). I returned to the literature in chapter 4 to develop an asset-based tool to explore the potential of this research, Figure 27 below. My findings focus on the potential for an asset-based approach to the school. I have reached this conclusion because I found during my years as a teacher, and subsequently as a researcher, that much activity in school sees young people as being a glass half empty rather than a glass half full. This builds upon the work of (Foot & Hopkins, 2010) and demonstrates how my research, with young people as co-researchers, can address some of the highlighted issues.

The evidence from my YPDAR research has demonstrated how the young research team have grown and developed throughout the process. My aim at the beginning of the research was to ensure the YRT had as much input and control as possible. Whilst my research topic was set, they all volunteered, and thus bought into the subject area; we worked as a team to set the research agenda and methodology, continually discussing and evaluating what we did. My philosophy and methodology focused on developing the team as capable individuals who would contribute with their lived experience as their guiding light; this was an asset-based approach. What I learnt from this research was that the process was not a simple or linear one. As we approached each section of the research, new challenges were met, and these needed to be addressed. This took time, as we gently assessed the YRT strengths and weaknesses, so we could identify the skills gaps and address them as a team. Also, we needed to give whatever time was required to build relationships between myself and the YRT, as well as between the YRT and participants.

In section 7.5 I discuss how this research helped develop young people in various ways. These young people were demonstrating aspects of a problem-posing education system rather than a banking education system (Freire, 1970). They very quickly started to realise that they had the potential to contribute and make a difference in their community.



Figure 27: Taking responsibility: an asset-based tool

Freire, (1970, pp.46-48) wrote, "The more students work at storing deposits entrusted to them, the less they develop the critical consciousness which would result from their intervention in the world as transformers of the world." This research was, in effect, about being able to "...transform that structure [school] so that they can become "beings for themselves."" This transformation required trust, which this research identified as lacking within this school's systems. As demonstrated in the findings, the participants were clear that there were elements of these systems that they would not engage in because they lacked trust in both the school structures and the adult staff. Earlier in this chapter, I give an example of how the YRT were subject to teachers during this research. As the project became established, trust developed, primarily between the young researchers and myself. They also started to trust the process, meaning they trusted that the school took the research seriously. A subtle switch in the power base became apparent. This aligns with what Freire (1970, p. 47) calls becoming "fully human" as the research acknowledged the role of both participants but more notably the YRT. By working as a research team, we engaged in critical thinking and were on a quest for "mutual humanisation". I had worked extremely hard to develop this critical ethic within the team by being a "partner(s) of the students"; I had "profound trust in people and their creative powers" (Freire, 1970, pp.46-48).

The school was committed to the research and, despite resistance from one teacher, most of the findings have been implemented. The school have made it clear that they want to listen to young people and allow them, where possible, to contribute to school improvement. The YRT, therefore, became empowered and transferred this to the participants, who also saw themselves as change agents.

Therefore, the processes we have developed in this research have positively impacted the school, young people and the structures that influence them. Whilst the impact so far has been small scale, there is potential for this to grow. The school has shown that they are prepared to engage with young people and listen. The message that the school is sending is vitally important. As long as the school follows through on its promises and commits to further such work, I am confident it can be transformative for all involved. The improvements

in confidence, trust, and relationships have helped develop young people's sense of agency; they have discovered a way to voice both their opinion and the opinion of their peers. Whilst small, there has been a shift in the power base. Previous to the research, the school had no real student voice relating to school improvement and the subsequent changes represent, a process of evolution rather than revolution. The research has empowered the YRT and participants, which has challenged the school's traditional power base. The research process, to date, has seen young people involved in the research, develop new skills, work with new people, and contribute in different ways. They have presented at meetings and conferences; some have contributed to an academic paper, and they have developed research skills. They have experienced something different from their peers, and mixed with professionals from various backgrounds new to them. Their experiences have been broadened and thus improved. Their social and cultural capital has been enhanced and will help them as they move forward. Overall, this work has impacted the identity of young people who now see themselves as confident and capable individuals with an important role to play in improving their own lives and those of their community.

In this way, an asset-based approach to working with young people has been developed and aligns with other research in this area. This includes Positive youth development (PYD), which focuses on evolving young people's competencies in the context of healthy relationships and communities (Teixeira et al., 2021). My research is also similar to Transformative student voice (TSV). TSV is about working with "adult allies" (Hipolito-Delgado et al., 2022, p. 2) in schools and developing marginalised young people's critical reflection skills through open dialogue. It aims for young people to become change agents (Hipolito-Delgado et al., 2022). This research has given a new perspective linked to young people researching in schools and is a significant contribution to knowledge.

By enabling young people to achieve through this research, I also believe that we are addressing epistemic injustice, something I discuss in detail in chapter 4. Houlders et al.'s (2021) arguments associated with this and how it links to self-identity and an individual's feelings of trust and confidence are also explored in detail in this chapter. I suggest that this is further compounded by schools displaying teacherism. As educational structures, I believe they may undermine young people's competency, authority, and credibility to comprehend their own experiences. Both distributive and testimonial injustice can, however, be addressed by schools.

In contrast to my reflections in chapter 4, where I discuss how some teachers denigrate and belittle young people, this research has the potential to promote young people's strengths as equal research partners. The literature I reviewed in chapter 4 related to hermeneutical injustice suggests that this is about individuals developing self-interpretive skills through social experience. It is also about the problems some young people from poorer backgrounds have in developing it (Dieleman, 2012). As demonstrated in the findings, we discovered that by involving young people from the pupil premium cohort in this research, they have been able to develop relationships with older young people which were reported as extremely positive in a number of ways. They have had the opportunity to speak their mind about school experience, debate and argue openly about the school, and develop their interpersonal skills through this. They are contributing to school improvement and developing their sense of worth. Possibly in a small way, this research provided a social base that enabled the development of hermeneutical agency.

After presenting the initial findings of the research to the school, the headteacher recognised the benefits and agreed to work further to develop a student-led, research-based school improvement model; a follow-up project investigating young people/staff relationships and how they impact the willingness of young people to ask for help in school has been completed. I am in discussion with the Academy Trust's CEO, looking at developing this model across a number of the Trust schools. The YRT have been given the guarantee of space, voice, audience and influence. In the introduction to this section, I referenced how schools as structures are part of the oppression of young people. Introducing young people from the pupil premium cohort to this research enables some of the poorest in society to develop self-confidence and feel valued as contributors to school improvement.

As I evidence in the findings above, this contributes to their sense of self and self-identity by changing the narrative around them. This research and the commitment of the school/Trust to enable its development is a step towards addressing structural issues in English schools. The asset tool above exemplifies the structural changes it has started to shift. The power structures have shifted, so young people's hermeneutical injustice is being addressed. This has come about as the research, based on young people's assets, is an enabling rather than disabling structure. In section 4.2, whilst exploring the literature related to power, young people and schools, I discuss how structures both develop society's culture and are created by society.

This is how I view my research, as it has started to develop the school's culture but is also being created in collaboration with the school. Maynard & Stuart (2017) write about the factors controlling individuals, including norms, rules, laws, and discourses, as being what is accepted as the way things are. This is the direction in which my research is moving as it aims to develop practice in schools that become an accepted way of working. This work also takes strides towards developing schools as truly democratic structures. This research has given a new perspective on how research collaboratives between young people and adults can positively impact young people and aspects of school character. Whilst, this work is unlikely to affect government economic policy, it has started to impact this school, which is a significant social structure in the lives of the young people in its community. Whilst I do not think this is enough to solve all related problems, I believe that it contributes to small changes that, with careful management, can be grown further within the institution, initially other schools in its Academy Trust and eventually throughout other schools in the country.

I have reviewed some of the literature regarding power in relation to society and schools in chapter 4. However, my focus on power throughout this thesis is also deliberate, and as expressed above I believe that it is an essential factor in the negative way in which some of our young people experience school; from a functionalist perspective (Durkheim, 1982) many young people fit into the role that they believe to be theirs. However, following Foucault's (1991) thinking around power as an enabling force, I believe the research has demonstrated precisely this. The participants from the pupil premium cohort and the YRT have blossomed through this research period. I have observed both groups of young people develop and grow. Many were shy young people lacking in confidence, but some have gone on to present their own experiences to academics, teachers and headteachers, whilst others have presented at conferences. As I have demonstrated, they have reported feeling empowered by the process as they contribute to school improvement and the mental health strategy.

9.2.5 School strategies to support young people's mental health

My initial inspiration for this research was around young people's mental health, particularly how schools could support it and how young people could contribute towards developing this support. The original school strategy was introduced as part of an external programme that produced a framework for developing an individualised school strategy. On reflection, whilst developing and implementing a strategy for the school, I was unaware of the implications of research linked to young people's development and the impact different environments may have on them. Furthermore, as I report in section 9.2.3, some young people in schools are reluctant to come forward and seek support from staff due to a lack of trust. By returning to literature in chapter 2, I draw upon Rudasill et al. (2018) and their system view of school climate (SVSC), in which they introduce the term nanosystem. They suggested nanosystems include peers, sports teams and academic school settings. To this, I would add a school mental health strategy. Nanosystems are seen as a bridge between the school and individuals. If schools view the strategy this way, recognising that some young people find relationships with adults difficult, they are then cognisant that this issue needs resolving if the strategy is to be successful. Schools must prioritise staff/young people relationships if they want young people to come forward for support. Applying my research to this literature, I offer new insight by suggesting that a school mental health strategy should be viewed as a nanosystem; this new perspective is a further contribution to knowledge.

A school implementing a mental health strategy will implicitly signify that, as an institution, they care about young people's mental health and want to support them in improving their mental health. However, my findings suggest that this is more complex. One of the strategy's aims was to encourage young people to self-refer to school staff if they felt they needed support. However, as stated earlier in this section, young people were reluctant to self-refer to school staff they did not trust, who were judgemental or who displayed aspects of teacherism. Also, school surveys, implemented as part of the strategy, were disliked by young people for two reasons. Firstly, they did not know who would see the information (and therefore whether they were someone who could be trusted) and also what would happen to the information. Secondly, young people perceived the environment where the surveys took place as lacking confidentiality, as surveys were completed on a computer screen in a classroom.

The school strategy, designed to help young people, was therefore also seen in a negative light. The issues of trust were once again at the forefront, and acted as a barrier to the strategy which was therefore not always observed as being a positive and supportive school act. Whilst some in the school community saw the concept of the strategy as being positive, others would not fully engage in it because it required the ability to trust the adults in school. Therefore, teacherism has resulted in young people feeling that they are not able to engage. This also supports the view in section 4.4, expressed by Horgan et al. (2017), that agency is a

relational concept. If young people were able to trust the staff in school, they would be more likely to come forward and ask for support, and also more likely to complete the school surveys. This reinforces the need for young people to be involved in the design of school policies, especially those that have the potential to impact their wellbeing. This also supports my desire to ensure young people's rights are recognised in schools, and this links directly to the Lundy model of youth participation, which I discuss in more detail later in this chapter.

My research has built upon Rudasill et al.'s (2018) work and contributed original thinking in relation to my interpretation of the school mental health strategy as a nanosystem, something that facilitates relationships between young people and the school. However, to ensure that this contribution is meaningful, I need to apply what I have learnt and explore how this can shape the thinking of educators and educator trainers in relation to school mental health strategies. Therefore, I intend to draw together learning from this research and address the issues challenging schools in this area. What follows is my suggested framework to support schools in helping young people to look after their wellbeing and mental health.

9.2.6 A framework for schools

Whilst completing the literature reviews and in preparation for the empirical data collection, I became aware of the potential this background work could have in relation to young people's wellbeing and mental health. As I discovered the work of Bronfenbrenner, I reflected on my school experiences and the context reading from chapter 2. The historical changes I highlight and the impact these have had on school values led me to develop a conceptual framework for emotional wellbeing in school. As an ex-teacher, it occurred to me that schools would benefit from a deeper understanding of how young people and their wellbeing outcomes fit into the broader education picture. The framework I consequently developed (Figure 28) also drills down to give a perspective of young people, mental health and schools including all internal and external influencing factors. Young people's outcomes are influenced by their own world and the world around them, including the world at large (UNICEF, 2020). Any model or framework for school mental health needs to place young people at the centre and demonstrate how factors in their lives impact them. This can be achieved by adapting Bronfenbrenner's (1994) bio-ecological model as a framework through which schools can explore young people's emotional wellbeing. It will benefit staff to understand the determinants of those factors that directly and indirectly influence young people. In turn, the

outcomes they display will depend on the circumstance in which they live and learn. This is not about a pretence that schools are bubbles separate from society, but it is about recognising that they have a unique place in the lives of young people, and that they can have an enormous impact on them. This is about a school context.

Like Bronfenbrenner's model (Figure 2) the school's framework (Figure 28) puts young people and their outcomes at the centre. Learning from Bronfenbrenner, I believe that young people are impacted by their biological makeup and immediate environment, both in and out of school. Following the concept of concentric circles, I have chosen to focus on elements impacting young people. However, unlike Bronfenbrenner, I only explore how young people impact their school environment. This is deliberate, as my thesis investigates young people's mental health in relation to schools. Whilst there is evidence that young people impact their wider environments (Silvestri et al., 2009), this is beyond the scope of my study as I concentrate on influences that impact young people's wellbeing outcomes.

Determinates of YP wellbeing

- Government policy
- Local priorities
- Local investment
- Community engagement
- Local youth facilities
- Areas socio-economic status
- School's historic legacy
- School Values
- Leadership
- Investment priorities
- Student support systems
- YPDAR projects running in school
- Impact on YP wellbeing

- Distal society Regional influence Wider community Culture/ethnicity/religion Security Local community Family Youth Clubs & groups
- Neighbourhood Opportunities Friends <u>School</u> Happy in school Academic progress Relationship with adults and peers Specific wellbeing support from designated staff Extra-curricular involvement School character

Young person's wellbeing outcomes Positive physical, social & mental health Contributing to school improvement & feeling productive Social & emotional skills High self-esteem Feeling valued Connected

Figure 28: A conceptual framework for emotional wellbeing in schools

The outer circle and distal influences on young people's wellbeing are linked to the wider community and society. The broader determinates are related to past and present political choices. At the community layer, I chose to concentrate on local influence; the community supporting the young people, including family, youth centres, and their neighbourhood. The final influence on young people's wellbeing is the school itself. The determinates are associated with the nature of the school, from its history to its character, and how the leadership interprets their responsibilities to young people. As I have already highlighted, my findings suggest that well-being outcomes are likely to improve where school values focus on young people, rather than school success. The lack of trust that we discovered in this research suggests that a new approach is required if schools are to develop as institutions to be trusted by young people. Within the determinates for schools, I have therefore included the YPDAR approach used in this research, as I believe it can impact upon school character and also improve outcomes for young people's wellbeing. This is something that I explore in greater detail later in this section. This also enables me to take a fresh view of Bronfenbrenner's model, as I suggest that in the 1980s and 1990s, whilst his focus was on the development of young people, their position was set in society as becoming adults who had little say in matters relating to them. Whilst the UNCRC (2019) was written in the 1980s, it only became law in 1992. As I discuss in section 5.6, the fact that the law changed to give young people a right to a say in their own lives does not mean it will happen. Children's rights to a voice are now more widely accepted than in the 1980s and 1990s. My research findings demonstrate the benefits to young people of research I carried out. I, therefore, suggest that adding youth/child participation as part of the microsystem would add value to Bronfenbrenner's model and contribute extra knowledge and new insight from within the perspective of his EST.

In the following section, I set out how schools can benefit from adopting the framework and using it as a lens through which to critique their approach to young people within a broader context of national and local issues and policies. It is a tool for school leaders and pastoral staff, and may be used as a reference and discussion framework to improve understanding of young people in their care. Schools must appreciate the framework's purpose; therefore, I have developed considerations they need to explore before adopting it.

A conceptual framework for wellbeing in school's considerations:

- School leaders could use this conceptual framework to explore how the specific circumstances relating to young people impact wellbeing of young people within the school.
- There is scope for schools to adapt this framework to their local situation.
- The framework (Figure 28) raises the awareness of school staff regarding the issues of young people's mental health within the context of wellbeing outcomes.
- The term wellbeing is used in multiple ways depending on the context, and it must be seen from an education-specific perspective; that is, in how it impacts the education a young person receives.
- The framework is designed to be used independently or alongside subjective selfassessment criteria for young people, such as those developed by Burton et al. (2014), below.
- This is not intended for school staff to use as an objective assessment tool, but could be implemented as part of a professional development package. There is also an opportunity for initial teacher training programmes to adopt the framework as a reflective tool for graduates working in schools.
- This is a reference point and guide for school staff working with young people.

Subjective self-assessment (Burton et al., 2014) criteria may include:

- A capacity to develop and sustain mutually satisfying personal relationships.
- Age and ability-appropriate psychological development.
- Age and ability-appropriate development to play and learn.
- Developing a moral sense of right and wrong.
- Psychological distress and maladaptive behaviour fall within normal age and context limits for the young person.

As presented in chapter 3, there is great ambiguity around the term mental health, and it can be used in multiple ways to convey varying conceptions. Some believe mental health covers such broad areas that it is almost impossible to develop a useful definition (Manwell et al., 2015). I find this argument difficult to counter, as the term is used in various ways depending on professional setting, be it education, health, or social care etc. However, it is my experience that most young people today are aware of the term mental health; they are encouraged to think and talk about it and, as such, it is incumbent upon us as educators, to ensure that they have a clear understanding of the different definitions. Young people need to know what mental health, mental illness and emotional well-being are, and the difference between them. The imprecise terminology may well lead to a lack of clarity in an area where we are trying to upskill young people. If we do not support them in understanding the difference between these terms, how can we expect them to appreciate how they may vary in the impact on themselves and their families?

A multi-faceted approach is required to develop an appropriate school mental health definition encompassing the three areas of mental health, mental illness, and emotional wellbeing. As the model, Figure 29 below, demonstrates, there is a relationship between all three, and schools have a crucial role to play in facilitating the support for young people in each of them.

Whilst in my role as assistant headteacher and introducing the whole school mental health strategy, I discussed terminology with mental health professionals from various backgrounds. There was a strong feeling that if society was to reduce the stigma associated with mental health, the issue needed to be tackled head-on. The advice was to use the term mental health, and explain that it was different from mental illness and that the school was integral to ensuring that individuals in need were supported either in-house or signposted externally. For this work, I am clear with the distinction between a mental illness that is medically diagnosable and mental health that is an emotional state all young people experience. With education and support, young people can learn to understand their feelings, adapt, adjust and modify their behaviours to experience a fruitful education and prepare for a fulfilling life beyond school. Glazzard & Stones (2021) write about conceptualising mental health as a state along a continuum from being mentally healthy to being mentally ill. I advance this further by suggesting the School mental health model (Figure 29) be used by schools in conjunction with the mental health definitions to simplify the issue. Both educators and educator trainers can use the model to support staff and young people in this complex area.



For this research, I intend have adopted the following definitions:

Mental health

Mental health is an individual's psychological condition and, like physical health, is something that all young people experience and which can impact upon how they feel and behave. It can be influenced by what is going on in their day-to-day lives and it affects them in different ways and to differing degrees. Good personal mental health is important as it enables young people to be the best version of themselves. (Bronfenbrenner's 1994; Manwell et al., 2015; WHO Mental Health and Substance Use Team, 2022).

Mental illness/Mental health disorder

Mental illnesses are medically classified signs and symptoms that are often complex and multi-faceted. They require medical diagnosis and treatment to enable young people to live productively both in school and in the wider community. (Bentall, 2003; WHO Mental Health and Substance Use Team, 2022)

Emotional wellbeing

Emotional wellbeing is a positive mental state where an individual's basic needs are met to the point where they have a sense of purpose and can achieve personal goals and participate within the school community. Emotional wellbeing is further enhanced by an inclusive school climate where schools prioritise healthy relationships and support for young people experiencing difficulties. (Huber et al., 2011)

School mental health

This is an aspirational concept focusing on ensuring that all young people are in a position to achieve their potential. It encompasses education, support for young people and their families, and signposting to external services where appropriate (Bronfenbrenner's 1994; Huber et al., 2011; Manwell et al., 2015).

I give a new perspective by defining the term school mental health. It is an alternative that schools can use to frame mental health within their specific context, and give clarity to their role in supporting mental health for young people and their families.

Figure 29 above demonstrates how all four concepts interact within a school context. Furthermore, it shows how schools can work to ensure that young people are supported to achieve good emotional wellbeing, and therefore enjoy a productive education in preparation for life in the future. My previous teaching experience suggests that schools have a role to play in both mental illness and mental health. Dedicated pastoral staff can address some of these issues through in-school monitoring. These staff require training that will give them the skills to recognise the warning signs for potential mental ill health. This will then enable schools to signpost young people and their families to appropriate services that will be able to diagnose and treat them. I favour this educative approach that, if implemented alongside monitoring, can enable schools to signpost young people and families to diagnostic services when and where appropriate. The introduction of the Mental health support teams (NHS

England, 2021) will support this process; however, as a deficit model, more is needed to address the ever-increasing emotional needs of young people.

Mental health can be supported through whole-school strategies that require a multi-faceted approach (McPartlan, 2019b, 2019c). The aims of an effective strategy are:

- To educate young people about all aspects of their emotional wellbeing.
- To support young people to be able to take responsibility for their mental health.
- To develop school character that is inclusive and supportive.
- To develop a school curriculum that supports rather than harms young people's mental health.
- To develop a school curriculum that strategically normalises mental health through an approach that requires all curriculum areas to include the topic in subject syllabi.
- A broader community strategy that supports mental health education in the school.

By taking this approach, the mental health and emotional well-being of young people will be prioritised by schools, which will be central to their values and will ensure that they are in the business of developing well-rounded and well-adjusted young people.

9.3 Methodological innovation

9.3.1 Introduction

In chapters 5, 6 and 7, I develop the premise of YPDAR and how it is built around the notions of equity, inclusion and authenticity when researching with young people in school. To fully understand the development and creation of this synthesised methodology, I return to Freire's classic Pedagogy of the Oppressed (Freire, 1970). I have learned from Freire's concepts of banking and problem-posing education and agree that reality can only occur through communication rather than isolation (Freire, 1970, p.50). I draw on French and Raven's (1959) work, which influenced my thinking on relationship dynamics, hegemonic power (Arnot & Reay, 2007) and teacherism in school. I discovered that relationships in schools are lived out through a power dynamic, which is an inequitable dynamic in seeing young people as lesser beings who benefit (or do not) from teachers' knowledge, wisdom,

and experience. This was an area that required consideration within the sphere of the research as well as the development of YPDAR. In an attempt to dismantle the power dynamic in the research, I took learning from CCM, which celebrated a dialogic approach (see section 5.3). By developing the definition below, and with dialogue at the centre of my work, I empowered the young people I was collaborating with. I built on the work of Habermas (1984) by ensuring the findings of this research came from the YRT and were validity claims rather than power claims. I adapted learning from Freire (1970), who suggests research such as this is dialogic; I developed YPDAR to influence an anti-dialogic school which I saw as wanting to impose their will on young people.

9.3.2 Dialogue and critical communicative methodology

Dialogue and my understanding of this term is central to all aspects of my research. Therefore, it is crucial to understand what I mean by the term and why it is important. Freire's problem-posing concept suggests dialogue enables reciprocity between teacher and student in their symbiotic pedagogic relationship. This was central to my development of YPDAR. As I explained above, the research was designed so that both parties took responsibility for their contributions to the process; myself offering academic knowledge and the YRT offering knowledge from their lived experience. New knowledge was created from our dialogic interactions. Just as the students in Freire's problem-posing model are no longer passive learners but analytical co-investigators, so were the YRT. Both the younger participants and YRT were invested in a process through dialogue with their researcher. This is the dialogue that creates the impetus for the unveiling of reality and the emergence of critical intervention. This dialogic process is what enabled the YRT and me, the researcher, to collaborate and understand our reality, and grow our critical thinking. In this way, we developed the power to critically explore our relationships, not only with each other but also with the school and the world in which we live (Freire, 1970).

Freire argues that "something which is the essence of dialogue itself: the word" (p.60) is instrumental in our approach to life. Our understanding of the word praxis, and its dual dimensions of reflection and action, are vital to this work, as without either of them, a word becomes empty. Where reflection takes place without action, one is left with verbalism; similarly, where action takes place without reflection, what remains is activism (Freire, 1970). As the research team worked through the weekly mini-action research cycle (Figure 9) this understanding of praxis and its links to the term dialogue became our way of working. As explained in section 5.4, we discussed and reflected on previous actions, how these actions connected or not, and how our collective learning would inform our next step forward. Throughout this process, the values of honesty and integrity became fundamental to its success, reinforcing the development of relationships between all involved, which was a particularly essential element. Freire posits that unless we are prepared to collaborate in this collective spirit, we cannot be truly transformational, and the key to this way of working is dialogue. Our research design was what Freire described as the "encounter between men, mediated by the world, in order to name the world" (Freire, 1970, p.61).

I believe that the meaning of the word dialogue is fundamental to this work and, as such, I therefore suggest the following description to be used within YPDAR:

The dialogic approach in YPDAR is communicative activity that emanates from life reflections between two or more individuals. Creative activities are encouraged to stimulate inquiry that empowers individuals to self-critique their own life-circumstances as a transformative endeavour. In this context dialogue is egalitarian. Positive relationship development is essential to the process and is enhanced by equitable but different power interactions between the researcher and the researched.

Figure 30: Dialogue description

Whilst the position of dialogue within YPDAR is fundamental to its success, so is the alignment between the researchers and the participants. Therefore, I return to look at how Habermas and CCM have impacted my work.

9.3.3 Habermas' theory of communicative action

Habermas' thinking was influential in developing CCM and, therefore, also in my research. I have realised the importance of his theory of communicative action, which I explore in more detail below. The key to this theory was his seven postulates, three of which are fundamental to this research. These are 'universality of language and action', 'absence of interpretative hierarchy', and working on an 'equal epistemological level'. I will briefly investigate the relevance of each to my approach. I will then explore how they influenced my methods and analysis in greater detail.

9.3.3.1 The universality of language and action relates to Habermas' thinking that everyone can interact and communicate

This is one of the reasons we developed the research process as we did; layers of research between three parties focussing on dialogue to encourage consensus and mutual understanding. I believed that young people in the school were the only ones who had the answers to questions relating to their experiences of the mental health strategy. In section 6.5, I write in detail about the development of the analysis and how the YRT contributed to it. Within the process, I gave them as much autonomy as possible to conduct the research with the participants; I encouraged them to take chances and be creative. To do this, I had to develop checks to my power-laden position by engaging in reflexive activities and developing reflexivity in our group meetings. I explore this in more detail below.

9.3.3.2 Absence of interpretative hierarchy links to Habermas' belief that all within the research process had an equal part to play when interpreting the data

Throughout this thesis, I have discussed how my life as a teacher impacted this research. As the researcher, I had to contribute to the process but ensure my position did not enable me to overrule or impose my will on young people. It was elemental to the research. The findings came directly from young people and not via the interpretation of an ex-teacher. This is supported by the thoughts of Susannah, a YRT member, in her reflections in section 7.5.1, where she demonstrated how, as the research developed, her trust in me grew, and she could contribute how she 'felt about matters'. The importance of these meetings as vessels through which we created the findings together is also exemplified. I always intended to ensure an absence of interpretative hierarchy. This did not necessarily come easily; it took time for trust and relationships to develop between the parties involved. This leads me to believe the time-limited nature of my research was likely to hinder its outcomes. It also reinforces the importance of allowing time for relationship development in qualitative research such as this.

9.3.3.3 Equal epistemological level. Habermas believed all parties bring their unique expertise to the research

My positioning of the research did just that; I set my academic experience on the same level as young people's lived experience, and whilst our experience was different, it was of equal importance. This research would only work with young people's input alongside mine; one without the other would not provide the results we were aiming for. Whilst this is discussed in greater depth in chapter 5, I also believe that the implications of this philosophical approach have the potential to have a positive impact on schools. This is something I explore in more detail later in this chapter.

These postulates were about equity between the researcher and the YRT. However, to ensure their adoption was not tokenistic, they needed to be built into the research's foundation. To do this, I needed to support the YRT to enable them to interact, communicate, and collaborate if they were to contribute fully to the process. Therefore, I decided to develop strategies to improve their reflexive and reflective skills.

9.3.4 Impact of a reflexive and reflective approach with the YRT

The research was an action-orientated approach to inquiry, using small-scale theorising to investigate specific problems and situations (Denzin & Lincoln, 2017). As discussed in chapter 6, I aimed to work with young people in school as the knowledge was housed within their community (Wood, 2016). This was to ensure the research came from the YRT and from the school community. I saw the YRT develop skills and acquire knowledge, enabling them to sustain and build on the research (Schratz & Walker, 1995). The processes involved were about social change (Schratz & Walker, 1995) and democracy at its most basic level. Young people were being supported to develop a reflexive and reflective process that grew their critical thinking skills and brought about positive change in their school.

As I have already highlighted, the YRT were not in a position to deliver on the academic aspects of the research. Therefore, I was responsible for introducing academic principles to the group. I built in a reflexive process to ensure the YRT fully knew they were the vessels through which the research would flow (Borg et al., 2012). As part of a constructivist approach, critical reflexivity helped situate our experiences as data (Denzin & Lincoln, 2017). In addition, the reflexive process that we developed brought us closer together as a

team; it enabled us to build cohesion, trust (Simmons et al., 2021) and a collaborative mentality. Much of this can be exemplified through quotations from the YRT members in sections 5.10 and 6.5, as they talk about growth in their confidence and trust, and the collaborative process. In section 5.10 the comments also suggest that our emphasis on reflexivity improved creative thinking, problem-solving and generating ideas.

The reflexive approach was crucial, but the research also benefitted from my development of a systematic collaborative self-reflection tool in Figure 13 and **Error! Reference source not found.** 14. As I suggested in chapter 5, the tool enabled me to investigate the daily challenges of the research process with the YRT team. It was about collaborating with them and stimulating them to explore what they were doing, and how they were doing it. I took on two roles, firstly as a team member within the research and supporting the YRT. Secondly, I removed myself from the process and became the researcher overseeing the process. The collaborative self-reflection tool I developed is a valuable contribution to knowledge as it creates a framework that researchers and young people in schools can use. In addition to enabling a systematic reflective process, the tool also helps develop young people's critical thinking skills.

It was important that, as part of both the reflexive and reflective processes, the YRT could challenge both myself and my position; this is something I encouraged. The combination of the work we carried out in the mini-action cycles and this reflexive work enabled us to develop the findings in relation to the mental health strategy and the emergent YPDAR. The dialogic interactions within the research team suggested there was a more effective way of having an inclusive approach to involving young people in school research, that being YPDAR. This created a model for impactful involvement and resulted in authentic findings from a mixture of young people's lived experiences and my academic input.

My role, however, went further than just developing the YRT skills and facilitating the research. The relationships within the process were more than merely functional. Freire (1970) suggests that love is an essential ingredient of dialogue, and that true dialogue cannot exist without it. It is both the foundation of dialogue and dialogue itself. Dialogue cannot be about oppression or domination, but needs to have liberation as a focus. Therefore, I sat alongside the YRT and they sat alongside the participants. On occasions, we acted as advocates for our respective partners. We supported them and took an interest in them. I offered myself to the YRT in whatever way they wanted.
I wrote references, helped them with applications, lent them reading books and generally took an interest in their lives, both in and out of school. Similarly, the YRT learnt from my example in that they would stop and talk to their participants in school, offer advice when approached and, on one occasion, support a shy young person to access an out of school young farmers club. Freire (1970, p.143) calls this "communion" when describing how Che Guevara interacted with peasants. Investing in people in this way, be they Brazilian peasants or young people from a school in Cumbria, demonstrates a commitment and love for them and enables a genuine dialogic process. As I reflect, by the culmination of the research, the YRT and I were in communion. We were comfortable in the other's company, had mutual respect and trust for each other and went out of our way to support and help where appropriate. However, as the findings show, some young people in this research told us they did not trust school surveys (or staff) and would not take them seriously. As something designed to engage young people, it was obviously failing. This type of student consultation is tokenistic and not about communion with them, but instead supports Freire's (1974) view that schools are anti-dialogic. The benefits this research has created are many. However, they can be divided into two major categories. Firstly, I explore those benefits that impact the school as a social structure, and I then look at the benefit of young people as researchers working with their school peers.

9.3.5 Research impact on school

As a researcher in a school, working closely with sixth formers, I have adjusted my approach so that the work is non-hierarchical. The school saw the success of this research and employed me to develop it further. I returned to the school to work with another sixth-form YRT on similar research. After the success of this second project, I am in discussion with the Academy Trust to lead a team of teachers as researchers who will use YPDAR. My role within this will be to supervise the teacher/researchers and oversee the research project in several schools. I will also be interested in observing how the teacher/researcher relationship with young people changes from when I was the researcher. Whilst I was an ex-teacher and concerned about how this may inhibit the research relationships, I was, nevertheless, able to distance myself from the direct influence of school structures. This is something the teacher/researchers may find more challenging as they are part of the school structure, and integral in its direction of growth. As this further research progresses, work will need to be done to explore how a teacher/researcher's position will impact both research relationships and outcomes.

Another challenge for the school staff will be whether they can cultivate trusting and enduring relationships that build young people's confidence so that power relations between them can be redistributed. This will enable them to work on an equal epistemological level with young people. I agree with Darder (1997) when she describes this as democratic education which is only possible when educators see it as an act of social justice that enables young people to discover their own transformative power and potential. This is an important step away from Freire's banking model and towards his problem-posing one. I discuss school values, climate and school character in section 2.4, and this is where the challenge lies. Unless the school is committed to working with young people in this way, I suspect individual teachers may struggle to deliver. The findings of this research have already highlighted the benefits of increased confidence, trust and empowerment for young people. However, as I reflect below, the advantages gained from young researchers collaborating with younger participants also have the potential to impact the lives of those young people positively.

9.3.6 Young people researching with young people

In chapter 3, I review the literature on attachment theory (Bowlby, 1988). This requires further exploration as it is pertinent to this discussion. As discussed in section 3.5.1, Hammond & Harvey (2018) suggest that schools developing environments with opportunities for stronger adult/young relationships have a more productive context for learning, resulting in better attachment, motivation and behaviours, and higher attendance and achievement. Bronfenbrenner (1979) suggests that the microsystem of such schools has been enhanced, as productive relationships with staff may be more easily achieved. One of the outcomes of this research is the realisation regarding the lack of trust some young people have in school staff. This leads to a reluctance to seek support for any mental health issues. The design of this research has seen a group of young people, the YRT, collaborating with other young people, the participants. I touch on a new perspective in chapter 6: the potential transformational relationships between these two groups during the data collection period. On one level, the YRT worked with the participants to investigate issues around the school's mental health strategy. However, their work was more significant than just collecting empirical data. The YRT also developed a bond and team ethos with either individuals or small groups of young

people from the cohort of participants. As I suggest in the findings above, the participants also reported feeling empowered, more confident, and more trusting of the school. It is important to remember that, other than during the recruitment process, I had no interaction with the participants. Therefore, all the reported feelings resulted from the participant's interactions with the YRT. This suggests that the relationships developed were more than just researchers working objectively to collect data. Furthermore, at its conclusion and when asked to assess the project, the overwhelming majority of participants cited relationships with the sixth form YRT as the most important factor regarding their enjoyment. The social interactions between the two groups have resulted in reward feelings as they have been included and accepted, to the extent that they felt part of a meaningful and purposeful project.

My research has demonstrated that attachment-type bonds can be built between young people and adults outside the home (Kennedy & Kennedy, 2004; Riley, 2009). The growing relationships I observed between the two sets of young people fit with how Koehn & Kerns (2018) see secure attachment development. It was based on trust, communication and lack of disaffection. Allen et al. (2018) suggest this further enables young people to benefit from more effective emotional and interpersonal skills, enabling positive relationship development in later life.

The work between the YRT and participants developed relationships that were beneficial to both groups. The findings draw together evidence from my teaching experience and this research to suggest that the relationship could also be mentor-like. I have concluded that participants admired and looked up to the YRT; they inspired them. Trust developed between the two groups, enabling positive and productive discussions between them. During the PPI exercise I conducted, I was told participants were unlikely to trust me during the research process.

What I did not know was the potential for developing transformational relationships between the YRT and younger participants. Over many years as a teacher, I observed numerous failed attempts to foster mentoring relationships between groups of young people. Most, if not all, had failed. On reflection, I believe this was due to a lack of purpose within the projects. Whilst schemes were put in place, neither group of young people ever had a choice in the topic area for this relationship. The mentor schemes I came across were prescribed and delivered via a help sheet or guidebook. The relationships that developed within this research

were founded upon shared common goals. Both groups were interested in young people's mental health and how schools could support it. These young people wanted to be involved in improving the school and were prepared to engage with young people from a slightly different age group. In section 4.4, I explored how a sense of epistemic agency was boosted when young people felt they have the competency, authority and credibility to understand their own experiences and master concepts relating to their situation (Houlders et al., 2021). Therefore, I suggest this process has benefited the YRT in this way. Whilst not a specific focus of this research, my observations of the benefits of these relationships suggest this type of research has identified additional knowledge to the area of young people as researchers in schools. It is, however, an area requiring further investigation.

To take this learning further, I return to Chapter 3 and the exploration of the neuroperson model and the domains of socio-emotional skills. This section explained the theory underpinning how an individual perceives, regulates, manages, expresses and names their emotions (Mónaco et al., 2019). Also, it is important to understand the impact of emotional competencies on wellbeing, and an individual's ability to understand their feelings and come forward to talk about them. This is linked to secure attachment and the capacity to trust and communicate. My research, focusing on the school mental health strategy, suggests that some young people are less likely to be able to come forward due to their lack of agency. The neuroperson model indicates that some young people have narrow and constrain schemas that do not support the development of awareness of feelings or challenge-seeking behaviour. The domains of socio-emotional skills have been suggested to help these young people to develop a broaden-and-build schema.

My research has demonstrated that both sets of young people have benefited from their involvement in this research. They now have improved confidence and trust, and are more empowered by what they have done. Above, I suggest that the relationship between the YRT and the participants is attachment-like, as a bond was built during a mentoring-type relationship. The evidence would also suggest that this research could be viewed as developing young people's socio-emotional skills and competencies. I will take the domains and explain how each resonates with the development of young people, particularly the YRT, through this research (McNeil & Stuart, 2022).

- Emotional management YRT members demonstrated how they could be constructive in challenging situations at various points in the research process. In chapter 7, I recount the difficult meeting with a staff member. As I reflect, the YRT were able to stand back and see the bigger picture; they recognised it was only one staff member under pressure and being irrational.
- Empathy I draw upon the same incident of the member of staff's interaction
 with the YRT to illustrate that they displayed empathy. There were also occasions
 during the action research cycle meetings when I was impressed with the levels of
 understanding and sensitivity of the YRT to the participants. YRT members took
 time to help participants who were finding situations difficult.
- Initiative The empirical research process took over a year to complete, and throughout the YRT and participants remained steadfast in their commitment to the research. They persevered to complete the work despite setbacks, including three months of COVID disruption.
- Problem-solving This was at the fore throughout the research. However, the longer the process went on, the better the YRT became at it. The YRT quote in section 5.10 sums this up perfectly.
- Teamwork Once again, I refer to quotes in sections 5.10 and 6.5, where the YRT talk about the collaborative process developing over time, and how it helped them share their feelings and grow.
- Responsibility As I have mentioned previously, the YRT were volunteers
 within this research; they were fitting work in alongside their studies. However,
 they remained tenacious and determined throughout. They were loyal to the
 research, me and the participants, whilst also being understanding and taking their
 responsibilities seriously. They also committed to extra work by taking on and
 completing the academic paper we published together.

Another lens through which to understand the impact this research had upon the YRT is the Uchicago model Figure 15 that I briefly explore in chapter 5. My experiences of working closely with the YRT were mirrored by what the CCSR suggested. As I have explored above, this research process has helped them grow as individuals as they engage in various experiences. Importantly, I allowed them to reflect on the purpose and processes at play with this work. This action and reflection cycle has been vital to both what we have achieved and how we have achieved it. The YRT have been learning by doing, and have thus been able to

cultivate their own sense of self. This has been in addition to getting a perspective on other young people and school staff they have been forced to work with. Just as with the CCSR, YPDAR has built the YRT's skills and increased their experiences of agency, which I suggest will feed into the growth process and enrich their journey towards intentional life choices (McNeil et al., 2019).

The YRT relationship with the participants can benefit both parties, and I suggest YPDAR has also benefitted them. The processes within YPDAR bring new understanding to collaborative research with and between young people in schools. YPDAR can be used not only for school improvement research, but also to develop the socio-emotional capabilities and skills of young people.

I have highlighted how the theory from Habermas and Freire has contributed to my understanding of dialogue and its influence on the research. I have also explored how this research impacted the school and young people involved. I now analyse my new methodology to consider the development of YPDAR frameworks for use in schools.

9.4 A YPDAR framework for schools

To fully consider and engage with YPDAR, it needs to be dismantled into its primary constituent parts. The YPDAR framework (Figure 31) does just that by conceptualising it but also enabling an exploration of the ethics and analysis required for its success. The principles of YPDAR are based on the assumption that the researcher and YRT are working on an equal epistemological level (Gómez et al., 2011). The collaborative and dialogic act between the two parties (three if we include the relationship between the YRT and the participants) ensures that any claims made are from young people's lived experiences and perspectives. By approaching the research in this way, I have taken steps to safeguard its authenticity and ensure that findings have not come about through coercive power.

The research values and approach taken are reflected through the cards on the table in Figure 31; I ensured a caring, fair and inclusive approach to the research process, enabling the values of trust, respect and integrity to permeate throughout. The foundations of the research were from a social constructivist standpoint; it was conducted through action research with youth participation at the fore; egalitarian dialogic learning (Gómez et al., 2011) took place that

created emancipatory knowledge (Langhout & Thomas, 2010). This transformative process introduced critical inquiry to the school and contributed to the deconstructive process connected to hegemonic school power. As highlighted in the findings, young people's trust in the school grew, as did their self-confidence. Feeling empowered, they took on a responsibility to act to improve and change their school. By developing YPDAR, I have created a new perspective on young people as researchers in schools and the impact it can have on school character.



Figure 31: YPDAR framework for schools

YPDAR and the framework above are well-founded in both theory and practice. There are two additional elements needed that will further enhance it and also ensure the principles behind it are adhered to; these elements include (1) how analysis within YPDAR is conducted and (2) why relational ethics is an essential concept underpinning it.

9.4.1 YPDAR analysis

The processes I explored in the method chapter and summarised in Figure 9 were central to the research analysis. The research was a collective act where dialogue was used to understand, explain and learn from a social reality (Gómez et al., 2011). Habermas (1987) suggested this was a consensus perspective on truth created by dialogue within a community of competent inquirers. This was not about me interpreting their information but supporting the YRT to interpret their own information. The nuance within this process is also important to consider. Whilst my focus was on the YRT and their interpretations, it would be impossible for me not to have conscious thoughts about what was and needed to happen. This was a challenge for me as once I had thoughts, I could not remove them from my consciousness; they were with me, and I needed to embrace them positively. I decided to record anything noteworthy, and in the planning for the following group meeting, I would consider their appropriateness; thus, utilising reflexive deliberation. What was also important during this time was my alignment with Habermas' (1987) postulates, as explored in section 5.3. To fully understand the theory behind these and appreciate the importance of my approach, they need further exploration. By synthesising CCM with YPAR, I adapted theory from Habermas, taking a new perspective and creating unique insight on how we may collaborate with young people in schools and enable them to have a more significant say in an aspect of school improvement.

As I describe in section 6.5, the analysis process was challenging due to several tensions. In chapter 6, I explain how my initial intentions were to complete the analysis myself. This was my PhD and, therefore, my responsibility to draw conclusions. I wrestled with this dilemma and concluded that the YRT had as much right as I to write the analysis, recommendations and findings. One of the practicalities, however, was that these were young people studying for their A levels and had limited time for the research. I was careful not to pressure them, which I continually questioned myself about. The thought that they had committed to this research and therefore needed to deliver crossed my mind occasionally. I sometimes reflected that I had been too easy on them and did not push them enough. On a number of occasions, I asked if any of them had time to develop a model, write up our collective thoughts or summarise a diagram. When no one stepped forward, I took it upon myself to complete the

task. My concern then was whether this was my interpretation of the work or whether it was theirs.

An example of this is my framework (**Error! Reference source not found.**), which looks at the determinants of mental health in schools. My safety net to protect against my potential bias was the reflexive process highlighted above, where I encouraged the YRT to challenge me. On this occasion, I shared my framework with the YRT, who were encouraged to evaluate and adapt it. They did this individually before we came together to discuss it. This resulted in Figures 20 and 21.

Whilst I sometimes question my approach to the YRT, I also see another side to it. After our initial setbacks related to COVID and school closures, we were a happy, stable team. 13 YRT members committed to the research project, and they stayed throughout. In many ways, they went above and beyond what I expected of them. This included us writing an academic paper and attending conferences to present findings. I have written about the relationship development between myself and the YRT, particularly how trust developed and how we reached communion (Freire, 1970). I, therefore, have to consider whether, if I had pushed them harder, they would have been as engaged. Therefore, it is impossible to judge whether one way or the other was correct or misguided. Instead, it is important to recognise this as a tension, and when developing the research with others this tension needs to be addressed early within the research process. The decision around who should be involved in the analysis was organic; it emerged as we worked through the action research cycle. This was also the case in how we completed the analysis, which I explore below.

My initial thoughts about how to complete the analysis were taken straight from the literature and I identified a thematic approach as being appropriate (Braun & Clarke, 2012). Whilst we started in this direction and used thematic principles, our approach was more of crystallisation than triangulation (Richardson, 2000). We were developing praxis by looking for themes, talking about what was becoming apparent and revisiting the themes through our action research cycle. Every week, our action research cycle would deliver new and different knowledge we would take forward and act upon. In both my YRT meetings and the meetings where the participants and YRT met, we would discuss and test ideas before bringing new thoughts back to the team. Having worked this way for several weeks we wrote the analysis and recommendations together, before presenting them to the headteacher. This is illustrated in Figure 24.

9.4.1.1 YPDAR analytical framework

The aspects of how analysis fits as part of YPDAR are essential if this methodology is to be followed and maximised. I have therefore adapted the framework above to help clarify the importance of analysis in YPDAR. The intention is to emphasise the value base of YPDAR, but also include the essential elements that enable the analysis to take place through this process. The theory and practice I have discussed above have evolved into Figure 32 below. The framework is based on the principle of validity claims (as opposed to power claims) through collaborative dialogue. Still, as I discuss in chapter 6, the analysis focuses on crystallisation which comes through both individual and collective reflexivity and leads to multi-voice analysis. This is about subjective interpretation by young people and is, therefore, egalitarian. Using this analysis framework alongside the YPDAR framework will enable researchers and schools to understand the principles of this research process, and the philosophy underpinning it. A substantial outcome of this research is how, interpreting theory from Freire and Habermas for use in school research with young people, I have created a form of analysis that is true to the values of YPDAR and contributes to young people's development and school character. The final piece of this jigsaw is how relational ethics is also an essential aspect of YPDAR.



Figure 32: YPDAR analysis framework

9.4.2 Relational ethics in YPDAR

Chapter 7 explored how my approach to ethics changed from protectionism to relational. As my research grew, I started to appreciate the significance of my ethical positioning within it. As the dialogic approach became a focus, YPDAR emerged and I became aware of the importance of relationships. As I explored in chapter 7, how I was shaping connections between myself, the YRT and participants, meant a shift towards a relational approach was almost inevitable. This was about being attentive and responsive to others and showing empathy with and for the YRT and participants. As the researcher, I had a responsibility to be open with the YRT, as it was essential I was available for them and responsive to their reflections and needs; this was what Freire (1970) called 'love'. As I commented in chapter 7, it was about being in this together (Austin, 2007). I suggest this required me as the researcher to walk alongside them as a partner, supporter, and advocate. This had moved from a

protectionist position to a relational one. To protect the integrity of YPDAR, a relational ethics framework is required, something I discuss below.

9.4.2.1 A framework for relational ethics within a school YPDAR project

My aim in developing this framework was to ensure relational ethics can be understood in the context of school-based YPDAR. The principal thread that runs through this research is the essential positioning of young people at the centre of this work, and how their relationship with the researcher enables and safeguards this. The framework (Figure 33), below, acts as a guide for researchers and young people, and supports their ethical deliberations before and during the research process. It ensures that the ethics align with the research and the everyday life of the collaborators. As research relationships developed, I became aware that, just as I showed care and attention for the YRT, they too showed similar care for the participants. Teo (2009) calls this conscious act virtuous living, and differentiates it from following ambiguous utilitarian beliefs.

YPDAR, in any circumstance, but particularly within a school environment, is subject to a unique context. This framework (Figure 33) is again based on the original YPDAR framework, but its foundations (the table front) have been adapted to prioritise a sequence-ensuring relational ethics approach. Whilst the values and principles are similar throughout all three frameworks, the relational aspect of this model requires fundamental acceptance and commitment from any schools adopting it. As I discussed previously, power issues at play in schools require them to engage with young people as active partners in research. This would require a shift in school ethos for many.

As the findings demonstrate, developing the trust and confidence of the YRT was crucial to the success of the research. It empowered them to take responsibility for aspects of the work, something that would previously not have happened. Therefore, engaging young people in school-based research requires a rebalancing of young people/staff power relationships. Mutual respect comes not from instruction but through a demonstration of open and honest dialogue, and as such it comes via "acts of cognition not transferrals of information" (Freire, 1970, p.52), and is a step away from teacherism. This links directly to the discussion earlier in this section, where I explored my hopes for the teacher-led project and, in time, one led by young people. Although I imagine the YPDAR ethics framework may need adjusting further

as teachers take on the role of the researcher, the dynamic teachers have with young people in school are likely to differ from that of a researcher and young people. I aim to explore this as the Trust schools' research develops.

If this framework is to be effective, there must be engagement with it prior to the start of the research, in addition to its being used as a reference point during the research. It will thus ensure all involved are clear about the value base of the project and are fully invested in the moral purpose of the work. The initial engagement and exploration of the framework, conducted by the researcher and the researched, enables the relationship development process to commence. It has been suggested that marginalised groups are more likely to doubt whether the researcher has their best interests at heart (O'Doherty & Burgess, 2019). Trust issues may not be resolved regardless of the informed consent process; the power imbalance between adults and young people in schools is likely to accentuate this further. Using the framework as a starting point provides an opportunity to initiate and develop relationships engendering honesty, integrity, and trust between the researcher and the researched. What better way to start to develop trust than to engage in discussions about the values and purpose of the research? The opportunity to explore and disentangle a school's intent and position in relation to researching with young people is designed as an exercise in equality; it lays bare school structures, opening them up to critique whilst simultaneously empowering young people. In this respect, and as explored in section 5.7, I agree with Ko & Krist (2019) that school structures send variable messages about the role staff and young people play in deciding the value of different knowledges, including how the knowledges should be constructed. One of the differences between this framework and the previous YPDAR ones is that the intent of how YPDAR will work is a sequential series of commitments and actions. It starts with a commitment to young people as partners in school improvement. This is achieved by the school setting out their vision of collaborating with young people to be researchers in school. Developing an ethos of respect where young people and teachers learn from each other to counter teacherism should be discussed early in the process, further empowering young people. From the outset, I ensured young people understood how the research would work and how the research may be challenging for the school. The school was not used to young people collaborating with staff to make decisions; therefore, the research was likely to uncover tensions as some staff pushed back against it. This is, in fact what happened on occasion.



Figure 33: A relational ethics framework for YPDAR in schools

9.5 Applying YPDAR for use in schools

Having laid out the benefits derived from this research, and set out a vision for young people's involvement in school improvement via research, it is important to develop a working model for how this can be established in schools. I address this below with my model for Young people's participation in schools.

Returning to the models of participation in chapter 5 can help define where this research sits in relation to participation. Whilst the limitations of the Hart (1992) model bring little to this process, the Treseder (1997) model can offer insight. Over the data collection period, the YRT involvement clearly shifted from being consulted and informed to an adult-initiated, shared decisions with children process, something that is closer to my initial aim. However, my research has demonstrated the potential for a school model of participation that goes one step further. Taking learning from Treseder (1997), I would adapt the model to create a sixth spoke for use in schools. Young people's participation level will be equal to school staff; there will be collaboration between them. Projects will be young person-initiated and directed with school staff acting as collaborators; staff as co-researchers will have the responsibility to develop academic knowledge in the given area of investigation. Also, Habermas' three postulates, which I synthesised into the creation of the YPDAR methodology, 'universality of language and action', 'absence of interpretative hierarchy' and working on an 'equal epistemological level' will underpin the research.

Young person initiated. Shared decision making in a collaboration with school staff.

Young people in school decide on an aspect of school improvement they wish to investigate. In discussion with school leaders' agreement is reached regarding which groups of young people will be involved. Young researchers then work with participants to undergo school improvement research, staff act as academic experts. The young researchers write up findings and recommendations which are presented to the school leadership team and disseminated to all young people in school.

Figure 34: Young person level of participation in school research (after Treseder, 1997)

Whilst the various models of participation can be used to categorise work in school, there is a clear need to develop a school-based model. A model, which encompasses that of Lundy et al., (2020), is required to ensure the views of young people are given due weight, and they comply with young people's rights. As I discuss in chapter 5, schools have a role to play in ensuring that young people's rights are upheld. It is important that schools are at the forefront of this work as they need to be leading by example and making explicit to young people their rights; by running my proposed model, they will be going out of their way to demonstrate this. Taking this approach will enable the views of young people are also important. There will be a gentle power shift as schools move to a more collaborative model where young people start contributing to the decision-making process. My proposed model below, Figure 35, has been developed to ensure this research is examined through the lens of Article 12 from the UNCRC. The model is designed to make explicit the contrast between how schools have operated as zones of control and how YPDAR can develop zones of empowerment. By giving young people their full rights to participate in decisions affecting them, schools will be

confronted with issues related to power and control. This returns me to how school values have changed over the past 40 years, and how I see YPDAR as a way of redressing the balance so schools can once again focus on the welfare of young people explicitly, rather than on the welfare of the institution. When young people are given the opportunity to instigate and lead research projects that are endorsed by the school and feed into the school improvement process, young people are likely to feel more engaged and empowered (Das et al., 2020). However, where schools only use surveys or dysfunctional school councils as tokenistic gestures to appease students or provide evidence for OFSTED, my experience suggests young people are likely to feel disengaged and disconnected from their institution.

9.5.1 A model for young people's participation in school research

The young people's participation in school model below (Figure 35) suggests schools need to move from a control-based approach to empowering young people. By encouraging them to participate in decisions leading to school improvement, schools will be investing in a transformative process which has the potential to improve outcomes for young people and school character. This is not about tokenistic involvement, but about ensuring young people's rights to make decisions about issues affecting them. To do this, it is important that a safe space is created where children are "able to express their views without fear of rebuke or reprisal" (Lundy, 2007, p. 935), which is why I have incorporated Lundy's model of participation within my model. As I have already demonstrated in this chapter that the benefits for the school and young people, be they participants or YRT members, are extremely positive. The Youth participation in school model incorporates a new perspective on work by Habermas; I adopt the three postulates mentioned above as underpinning and core theory for the participation model. In this way, it develops additional knowledge on school power through young people's research. To ensure this, I have also incorporated the principles from the Lundy model of participation, again developing new insight from an already well-accepted scholar.



Figure 35: Young people's participation in school model (incorporating aspects of Lundy's Model of participation (Lundy et al., 2020))

As previously suggested, there will need to be a seismic shift from those in power to accept and value that young people not only have the right to, but also can, contribute to the way in which their schools are run. Furthermore, should this shift happen and such methods are accepted as the way forward, there will still be significant work to do. From my own experience, I argue that many well-meaning teachers, who possibly see themselves as childcentred, would initially find it challenging to work in the way I have described, as their behaviour is a dysconscious act seen as adultism (Ginwright & Cammarota, 2002; Delgado, 2008) as well as teacherism in schools. Furthermore, there are also a significant minority of other teachers who could not and possibly would not work in this way. The challenge, therefore, is to ensure work such as this is shared and celebrated to ensure the benefits are clear to headteachers, teachers, professional associations, and policymakers alike.

9.6 Limitations and challenges

9.6.1 Limitations

As I set out to conduct research into a whole school strategy, I was aware of many of the potential pitfalls relating to the authenticity of this research, including researcher bias and potential trust issues between young people and the ex-teacher turned researcher. Whilst this particular context may be seen as a limitation, I believe that by developing this as a participative study and incorporating a reflexive process into it, I have been able to counter such claims. I am confident that the checks and balances I have introduced have ensured that the findings from this research can be held up as authentic and make a valuable contribution to both young people's mental health education and the field of youth participative action research.

This small-scale study was conducted in a rural comprehensive school in northeast Cumbria and was never intended to be representative of all UK schools. The design enabled an indepth study by having only a small number of participants with a limited age range. If the participant numbers had been greater, the research would not have benefited from the detail I was able to draw from this approach. Although the school's ethnic diversity was minimal, it was a truly comprehensive school that drew from a socio-economically diverse catchment. I argue that it is typical of many other schools in the country. Other schools wishing to refer to this research need to be aware of the specific research context. Still, they can take learning from the findings in relation to the whole school strategy and the potential benefits of YPDAR as a research tool in school. The strengths of this approach are that whilst it may be small scale, the detail we were able to focus upon has given a real insight to the thoughts and feelings of the young people involved.

My approach to this research was focussed on the young people as I wanted their input on the whole school mental health strategy. Whilst I stand by this, I have also realised that the school staff do have a part to play as they are the ones who hold the power and without their cooperation, change in schools will not happen. I therefore believe that to develop this work further, I need to ensure that staff are included in regular feedback and involved in the process. In this way they will be fully informed, feel included and have a level of investment which then enables them to support changes suggested by the young people and research.

9.6.2 Challenges

As suggested above, an ex-teacher returning to his previous school can be potentially problematic. My relationship with the YRT was a potentially challenging one. This was a collaborative project; as we were partners in research, co-researchers and working on an equal epistemological level. I wanted a horizontal structure with us working side by side. Over the research period, we made substantial progress in this direction as we learnt to trust each other. By employing reflexive processes, I challenged the YRT to take more control of the research and, in turn, challenge me. It was not perfect, and I am not sure that it could ever be perfect; much of the behaviour we exhibit is unconscious and therefore there was a power differential between us. This is something that I managed and reduced as far as I possibly could.

As the research progressed and we drew towards the analysis, findings and recommendations stages, these trials continued. I had to continue to challenge myself both in terms of whose findings these were and who should make the decisions in this respect. I was forever telling the research team that this was joint research and these were joint findings. I would check and re-check with the YRT that what was being put forward was what they wanted. Over the months in which I worked closely with them, we grew together, becoming more cohesive as a team; we developed a collaborative mentality within the group; this was down to the reflexive approach that I encouraged (Simmons et al., 2021).

As the research progressed, it became clear to me that this research may have benefitted from an understanding of two further linked areas. The first was student voice; I became conscious that there was a body of literature related to my concerns about young people being listened to in schools. Unfortunately, I needed more time to investigate either this area or another linked area, that being the impact of peer mentoring in schools. Whilst both bodies of work may have given me further insight, I decided to focus on young people as researchers and the potential impacts this could have on young people and schools.

The most significant practical challenge occurred at the start of the data collection process as it coincided with the COVID lockdown of spring 2021. Face-to-face meetings were impossible as the school was forced to close. I worked with school staff to set up safe online

meetings and attempted to collect data in this way. Unfortunately, the YRT and the participants were reluctant to engage, and only three of the YRT and five participants opted to partake. It proved unworkable, and after a month of trying, we abandoned it and awaited the re-opening of the school. On reflection, young people lacked the willingness or on-screen confidence to participate. At the commencement of the data collection, I reflect that relationships between the two groups of young people needed more time to develop sufficiently to allow trust between the parties. Also, new to the research, the YRT likely lacked confidence in the process reflected. Once school reopened and even though young people were working in bubbles, the weekly process of meetings engaged all parties, enabled relationship building and yielded rich data collection.

Chapter 10: Future research, recommendations and conclusion 10.1 Future research, recommendations and impact

My discussion chapter explores how the findings relate to the school's mental health strategy. As part of the discussion, and in an attempt to apply my research, I develop several models for use in schools. Due to time constraints, I have not been able to evaluate the models and frameworks fully, and this is now required. The Conceptual framework for emotional wellbeing and the School mental health model would benefit from in-school evaluation.

One of the key findings concerning the school mental health strategy is how poor relationships can hinder young people from asking for help. Many mental health strategies are focussed on educating young people to come forward for mental health support. Schools are positioned as places which can offer such support. However, should my research findings be applicable across all schools, relationship development would significantly contribute to the school mental health support debate. I, therefore, suggest that the area of young people/staff relationships and how they impact a young person's ability to ask for help requires further research.

My thesis suggests my research has been taking place within a school, which is a structure of oppression. I discuss teacherism and the impact this has on relationships, and introduce the positivity created as I collaborated with the YRT. As I attempt to scale up my work and position it as school improvement research led by young people, I see school staff taking on the role of lead researcher, the part I played. What needs further investigation, however, is whether school staff, as researchers, can break through the shackles the school structures impose to create beneficial relationships similar to those made through my work with the YRT. Further work is also required about the autonomy of young people in YPDAR and whether they can develop research independently of the school staff.

By using YPDAR and developing participative dialogic action research communities, we can have an opportunity to bring about change by working with those in power. Relationships are at the centre of YPDAR, and further research is required to identify the benefits that were gained by all the parties involved. There were clear benefits to both the YRT and participants, as highlighted in the findings. I have shown how this way of working has improved the confidence and trust of young people, empowered them, and improved their socio-emotional skills. My PhD was time-limited research; therefore, a longer-term investigation is required to test the longevity of my initial results. Does this, as I suggest, improve trust between the school and young people over a longer period, and is this trust enduring? My Youth participation in schools model also requires evaluation. Further questions need to be answered regarding how YPDAR impacts school staff.

In my thesis, I suggest that YPDAR could be transformational in relation to school character. This will require several years to develop. Over time, as young people become empowered and start to influence school policy, so relationships and trust should develop. This will likely take five years or more to impact school character. Therefore, a longitudinal study is required to track YPDAR in a school(s) over a prolonged period. Another area of benefit could well be the use of YPDAR in other youth settings, such as youth clubs or young person's advisory boards. Alongside such projects, complete evaluations could add further value to this methodology.

In my thesis, I highlight a number of areas where the process has benefitted young people. The relationship between the YRT and their participants was beneficial to both parties. However, I posit that the participants benefitted from an attachment-like bond that developed between them. This is something that requires more focused attention and further research. The development of research as a way of improving the socio-emotional skills of young people is another area for future research.

10.2 Conclusion

The aim of this research was threefold. Firstly, I wanted to explore how the whole school strategy could be improved. Secondly, it was important to me that young people were at the heart of this research, and they informed any investigation. My final aim was to work with them to develop a process to counter my potential unconscious bias. My chosen methodology of YPAR was fundamental to this, and this is what grew into YPDAR. Therefore, the two primary outcomes of this research were my new methodology and findings related to whole-school mental health strategies.

In addition, learning from Freire, Habermas, Flecha, Gomez, and Lundy has enabled me to use the research and reflect on the underlying state of this UK state school, one I suggest is typical of many in the country. I have discovered that many young people today find school difficult as they attempt to fit into systems created by neoliberal society. These systems pressure their communities by using exams as the currency of success at the expense of their broader education; social and cultural capital opportunities are often overlooked. Schools are anti-dialogic, and relationships, something required for young people to feel accepted in school, are sometimes neglected or ignored. Young people can have a poor sense of identity and lack confidence and agency, resulting in disempowerment. The systems set in place by the school to support mental health rely on trust in both staff and systems. However, some young people lack trust in teachers, meaning support systems are less effective than they should be.

By developing YPDAR, however, this research has the potential to be part of the solution. It allows schools to initiate a change process by collaborating with young people as research partners. By committing to YPDAR's dialogic process, opportunities open up to build trust and confidence with young people, and empower them to act and develop them as critical thinkers. YPDAR is about relationships, building them, restoring them, and ensuring that they are at the heart of school values and culture. This is about transforming a deficit-based institution into one that has asset-based approaches.

10.3 Impact

The initial aim of this research was to explore the efficacy of a whole school mental health strategy, but the impact of the research goes far beyond this. By taking an inclusive approach through the use of PAR, I have included young people in transformative work that can positively impact their and others' lives. In discovering the strategy's flaws relating to trust, I have uncovered a valuable insight that, once disseminated, has the potential to impact school leaders' thinking.

This work has also directly impacted those young people involved in the research. The YRT have grown through the research. Their confidence was boosted through the process where they built relationships and supported their younger peers, developed research skills, co-wrote an academic paper and presented at conferences. The research has also allowed them to

see their school in a different light; a trust-building process has commenced. They have become empowered to act as agents for school improvement. In addition, their effort as coresearchers has developed their socio-emotional skills as they worked with their younger peers. The participants benefited from the research, and their confidence and sense of agency grew, recognising that the school valued their work. Furthermore, their relationships with the YRT members have positively impacted them.

YPDAR has been developed as a methodology that ensures young people are at the centre of decisions impacting them in school; it is a research model for young people's school improvement. The potential impact YPDAR may have on school character is untested, but the potential for transformative change is there.

Whilst my PhD may be complete, my research is not. In the spirit of action research, the work continues through praxis. As I develop the research, I do so by acting and reflecting. Dialogue is central to all of this work, as school staff, academics, and young people are encouraged to learn from one another in this ongoing, long-term reflective process.

Epilogue

I refer to myself as 'becoming a researcher' in my thesis. As I conclude the research process, I reflect on this three-year journey which has seen me grow and change from a task-driven teacher into a creative and process-driven researcher. As the pastoral lead in a large school, my role required me to be reactive. No two days were the same, and I regularly failed to complete any of my planned tasks as day-to-day events took over. Everything was frenetic and had to be done straightaway, leaving a long list of things to do with little time to do them in.

As I started my research, it took me many months to recognise and appreciate the different mindsets required to become a researcher. It was all about time. As a teacher, I had little time to do the many things required. However, as a researcher, I started realising that a prerequisite was the ability to stand back, observe and question, listen and think before further action. This involved me building reflective skills, such as reading, deliberating, and questioning. As I became more and more aware of the necessity of this approach, I also realised that young people with whom I was collaborating may also benefit from working in this way. If my experience as a schoolteacher is anything to go by, then I expect the same to be true for young people in schools. They have little time to reflect, learn and process. This certainly aligns with my thoughts about the changing value base of schools.

In the prologue, I wrote about how my values of inclusivity, integrity, honesty and trust are important to me and how I believe in relationship building. I now realise that, as a teacher, the structure of oppression in which I was working created barriers. On occasions, this meant it was difficult for me to express my values as I might have wished to. The PhD has given me the opportunity to live out my values from within the research. As a researcher, I have put my belief and trust in young people into practice. My work with them has been value-driven as I have completed substantial research by collaborating with them on an equal level. I believe productive relationships are essential in an altruistic society, and my work partnering with young people has exemplified this, as we problem solved and often clarified complex issues.

One of the joys of this research has been working alongside young people to find solutions to a common cause. By developing this research together, we have created a new methodology with the potential to transform a societal structure, meaning this corpus of work has been about social justice and young people as activists. We have also made valuable contributions to the debate around young people, mental health and schools.

This research has been a life-changing journey of discovery for me. My belief in relationships, young people and the ability to draw on my educational experience has enabled me to grow and mature as a researcher. I am excited about the next steps on this journey. I aim to work with colleagues both in the UK and abroad to champion young people as researchers and to support them to become transformative activists as they impact their own and others' lives.

Bibliography

- Acker, S. (2000). In/out/side: Positioning the researcher in feminist qualitative research. *Resources for Feminist Research*, 28(1/2), 189–208. Retrieved from https://www.proquest.com/docview/194882377/791A6A125EAA4BF2PQ/2
- Adams, G. R., & Marshall, S. K. (1996). A developmental social psychology of identity: Understanding the person-in-context. *Journal of Adolescence*, Vol. 19, pp. 429–442. https://doi.org/10.1006/jado.1996.0041
- Adams, P. (2006). Exploring social constructivism: Theories and practicalities. *Education 3-13*, *34*(3), 243–257. https://doi.org/10.1080/03004270600898893
- Addis, D. & Tippett, L. (2010). The Contributions of Autobiographical Memory to the Content and Continuity of Identity. In F. Sani (Ed.), *Self-Continuity: Individual and Collective Perspectives*. New York: Psychology Press.
- Adkins, A. (2017). Neoliberalism, Moralism, and Critique in Salman Rushdie's The Satanic Verses. *Critique - Studies in Contemporary Fiction*, 58(1), 12–27. https://doi.org/10.1080/00111619.2016.1144556
- Adler, M. J. (1982). *The Paideia Proposal: An Educational Manifesto*. New York: MacMillan Publising Company, Inc.
- Albarello, F., Crocetti, E., & Rubini, M. (2018). I and Us: A Longitudinal Study on the Interplay of Personal and Social Identity in Adolescence. *Journal of Youth and Adolescence*, 47(4), 689–702. https://doi.org/10.1007/s10964-017-0791-4
- Alderson, P. (2000). School students' views on school councils and daily life at school. *Children & Society*, 14, 121–134. Retrieved from https://discovery.ucl.ac.uk/id/eprint/1493893/1/Alderson COUNCILS.pdf
- Alderson, P., & Yoshida, T. (2016). Meanings of children's agency: When and where does agency begin and end? *Reconceptualising Agency and Childhood: New Perspectives in Childhood Studies*, (January), 75–88. https://doi.org/10.4324/9781315722245
- Aldridge, J. M., & McChesney, K. (2018). The relationships between school climate and adolescent mental health and wellbeing: A systematic literature review. *International Journal of Educational Research*, 88(September 2017), 121–145. https://doi.org/10.1016/j.ijer.2018.01.012
- Allen, J., Grande, L., Tan, J., & L. E. (2018). Parent and Peer Predictors of Attachment Security From Adolescence To Adulthood. *Child Development*, 89(4), 1120–1132. https://doi.org/doi:10.1111/cdev.12840

- American Psychological Association. (1959). Ad hoc Planning Group on the Role of the APA in Mental Health Programs and Research. *American Psychological Association*, (14), 820–825.
- Anderson, G. & Gernon, C. (n.d.). *The Catherine Tate Show*. Retrieved from https://www.imdb.com/title/tt0441051/fullcredits?ref =tt ov st sm
- Archer, M. (1995). *Realist Social Theory; the Morphogenetic Approach. Cambridge:* Cambridge: Cambridge University Press.
- Arnett, J. & Hughes, M. (2012). *Adolescence and Emerging Adulthood; A Cultural Approach*. Harlow: Pearson Edfucation Limited.
- Arnett, J. J. (1999). Adolescent storm and stress, reconsidered. *American Psychologist*, 54(5), 317–326. https://doi.org/10.1037/0003-066X.54.5.317
- Arnot, M., & Reay, D. (2007). A Sociology of Pedagogic Voice: Power, inequality and pupil consultation. *Discourse*, 28(3), 311–325. https://doi.org/10.1080/01596300701458814
- Arnstein, S. R. (1969). A Ladder Of Citizen Participation. Journal of the American Planning Association, 35(4), 216–224. https://doi.org/10.1080/01944366908977225
- Artal, R., & Rubenfeld, S. (2017). Ethical issues in research. Best Practice and Research: Clinical Obstetrics and Gynaecology, 43, 107–114. https://doi.org/10.1016/j.bpobgyn.2016.12.006
- Atkinson, T. and Claxton, G. (2000). The Intuitive Practitioner: On The Value of Not Always Knowing What One Is Doing. (G. Atkinson, T. and Claxton, Ed.). Bristol: Open University Press.
- Aubrey, C., & Dahl, S. (2006). Children's voices: The views of vulnerable children on their service providers and the relevance of services they receive. *British Journal of Social Work*, 36(1), 21–39. https://doi.org/10.1093/bjsw/bch249
- Austin, W. (2007). Relational ethics. In L. Give (Ed.), *The SAGE encyclopedia of qualitative research methods*. London: Sage.
- Babalola, Emmanuel, Noel, P., & White, R. (2018). Prioritizing rural and community mental health in India. *Indian Journal of Social Psychiatry*, 34, 285–288. https://doi.org/10.4103/ijsp.ijsp
- Bai, H. (2020). Epistemic injustice and scientific knowledge distribution. *Filosofija*, *Sociologija*, 31(3), 217–224. https://doi.org/10.6001/fil-soc.v31i3.4269
- Ball, S. J. (1981). Beachside Comprehensive: A Case-study of Secondary Schooling.Cambridge: Cambridge University Press.
- Ball, S. J. (2006). Education Policy and Social Class (1st ed.). Abingdon: Routledge.

Ball, S. J. (2021). the Education debate (4th ed.). Bristol: Policy Press.

- Balluerka N, Gorostiaga A, Alonso-Arbiol I, A. A. (2016). Peer attachment and class emotional intelligence as predictors of adolescents' psychological well-being: A multilevel approach. *Journal of Adolescence*, 53, 1–9. https://doi.org/10.1016/j.adolescence.2016.08.009
- Bandura, A. (2012). Social cognitive theory. Handbook of theories of social psychology. In
 E. Van Lange, P., Kruglanski, A., & Higgins (Ed.), *Handbook of theories of social* psychology (pp. 349–373). Sage Publications Ltd.
- Bandura, Albert. (2001). SOCIAL COGNITIVE THEORY : An Agentic Perspective. Annual Review of Psychology, 52, 1–26. Retrieved from https://www.annualreviews.org/doi/10.1146/annurev.psych.52.1.1
- Bandura, Albert. (2006). Adolescent development from an agentic perspective. *Self-Efficacy Beliefs of Adolescents*, 1–44.
- Banks, S. & Brydon-Miller, M. (2019). Ethics in Participatory Research. In M. Banks, S. & Brydon-Miller (Ed.), *Ethics in Participatory Research for Health and Well-being* (1st ed.). Abingdon: Routledge.
- Barad, K. (2007). *Meeting the universe half way: quantum physics and the entanglement of matter and meaning*. Durham: Duke University Press.
- Barber, M. (2014). Rab Butler's 1944 act brings free secondary education for all. Retrieved January 8, 2021, from BBC website: https://www.bbc.co.uk/schoolreport/25751787
- Barthes, R. (1982). Camera Lucida. London: Jonathan Cape Ltd.
- Baumrind, D. (1964). SOME THOUGHTS ON ETHICS OF RESEARCH: AFTER READING MILGRAM'S "BEHAVIORAL STUDY OF OBEDIENCE"." American Psychologist, 19(6).
- Beck, A. T., Brown, G. K., Steer, R. A., Kuyken, W., & Grisham, J. (2001). Psychometric properties of the Beck Self-Esteem Scales. *Behaviour Research and Therapy*, 39(1), 115–124. https://doi.org/10.1016/S0005-7967(00)00028-0
- Beck, U. (1992). Risk society: Towards a new modernity. Newbury Park, CA: Sage.
- Behabib, S. (1987). The generalised and concrete other: The Kohlberg-Gilligan controversy and feminist theory. In & D. C. Benhabib, S. (Ed.), *Feminism as critique: Essays on the politics of gender in late- capitalist societies*. Cambridge: Polity Press.
- Bergin, C., & Bergin, D. (2009). Attachment in the classroom. *Educational Psychology Review*, 21, 141–170.
- Bergold, J., & Thomas, S. (2012). Participatory research methods: A methodological

approach in motion. *Historical Social Research*, *37*(4), 191–222. https://doi.org/10.17169/fqs-13.1.1801

- Bergum, V., & Dossetor, J. (2005). *Relational ethics. The full meaning of respect.*Hagerstown, MD: University Publishing Group.
- Bergum, V. (2012). Relational ethics for nursing. In J. Storch, P. Rodney & R. Starzomski (Eds.), Toward a moral horizon: Nursing ethics for leadership and practice (pp.127-142). Toronto, Canada: Pearson Education Canada. In P. R. & R. S. J. Storch (Ed.), *Toward a moral horizon: Nursing ethics for leadership and practice*. Toronto, Canada: Pearson Education Canada.
- Bernstein, B. (1977). *Class, codes and control. Vol. 3. Towards a theory of educational transmissions.* London: Routledge and Kegan Paul.
- Bernstein, B. (2000). *Pedagogy, Symbolic Control and Identity. Theory, Research, Critique.* (Revised). Oxford: Rowman and Littlefield.
- Bernstein, Basil. (2000). *Pedagogy, Symbolic Control and Identity (Critical Perspectives)* (1st, Ed.). Retrieved from http://www.amazon.co.uk/Pedagogy-Symbolic-Identity-Critical-Perspectives/dp/084769576X
- Bhugra, D., Till, A., & Sartorius, N. (2013). What is mental health? *International Journal of Social Psychiatry*, 59(1), 3–4. https://doi.org/10.1177/0020764012463315
- Birch, S. H., & Ladd, G. W. (1997). The teacher-child relationship and children's early school adjustment. *Journal of School Psychology*, 35, 61–79. Retrieved from https://www.sciencedirect.com/science/article/pii/S0022440596000295
- Blakemore, S. J. (2018). Avoiding Social Risk in Adolescence. *Current Directions in Psychological Science*, 27(2), 116–122. https://doi.org/10.1177/0963721417738144
- Blakemore, S. J., & Mills, K. L. (2014). Is adolescence a sensitive period for sociocultural processing? *Annual Review of Psychology*, 65(August 2013), 187–207. https://doi.org/10.1146/annurev-psych-010213-115202
- Blos, P. (1985). Son And Father : Before And Beyond The Oedipus Complex. New York: The Free Press.
- Bok, J. (2010). The capacity to aspire to higher education: "It's like making them do a play without a script." *Critical Studies in Education*, 51(2), 163–178. https://doi.org/10.1080/17508481003731042
- Boog, B. (2007). Quality of Action Research:Reciprocal understanding of (scientific)researchers and participating researchers. In *The quality of paractitioner research:Reflections on the position of the researcher and the researched*. Rotterdam: Sense.

- Borda, O. F. (2006). The North-South convergence: A 30-year first-person assessment of PAR. *Action Research*, 4(3), 351–358. https://doi.org/10.1177/1476750306066806
- Borg, M; Karlsson, B; Kim; Hesook, S & McCormack, B. (2012). View of Opening up for Many Voices in Knowledge Construction. *Forum: Qualitative Social Research*, 13(1). Retrieved from http://nbn-resolving.de/urn:nbn:de:0114-fqs120117
- Borton, J. L. S., & Casey, E. C. (2006). Suppression of negative self-referent thoughts: A field study. *Self and Identity*, *5*(3), 230–246. https://doi.org/10.1080/15298860600654749
- Bostwick, R & Glazzard, J. (2018). *Positive Mental Health: A whole school approach* (1st ed.). St. Albans: Critical Publishing.
- Bourdieu, P., & Passeron, J. (1964). Les Héritiers . Paris: Les Éditions de Minuit.
- Bourdieu, P., Accardo, A., Balazs, G., Beaud, S., Bonvin, F., Bourdieu, E., Bourgois, P.,
 Broccolichi, S., Champagne, P., Christin, R. (1999). (1999). *The weight of the world: Social suffering in contemporary society*. Cambridge: Polity Press.
- Bourdieu, P. & Passeron, J.-C. (1990). *Reproduction in education, society and culture*. London: Sage.
- Bourdieu, P. (1973). Cultural reproduction and social reproduction. In R. Brown (Ed.), *Knowledge, education and cultural change* (pp. 71–112). London: Tavistock.
- Bourdieu, P. (1974). The school as a conservative force: Scholastic and cultural inequalities.In J. Eggleston (Ed.), *Contemporary research in the sociology of education* (pp. 32–46).London: Methuen.
- Bourdieu, P. (1977). Outline of a theory of practice. Cambridge: Cambridge University Press.
- Bourdieu, P. (1986). Forms of capital. In J. Richardson (Ed.), *Handbook of theory and research for the sociology of education* (pp. 241–258). New York: Greenwood Press.
- Bourdieu, P. (1990). The Logic of Practice. Translated by R. Nice.. Cambridge: Polity Press.
- Bourdieu, P. (2011). On television. Cambridge: Polity Press.
- Bowen, S., Sustar, H., Wolstenholme, D., & Dearden, A. (2013). Engaging teenagers productively in service design. *International Journal of Child-Computer Interaction*, 1(3–4), 71–81. https://doi.org/10.1016/j.ijcci.2014.02.001
- Bowlby, J. (1982). Attachment and loss: Retrospect and prospect https://doi.org/10.1111/j.1939-0025.1982.tb01456.x. American Journal of Orthopsychiatry, 52(4), 664–678. https://doi.org/https://doi.org/10.1111/j.1939-0025.1982.tb01456.x

Bowlby J. (1988). A secure base: Parent-child attachment and healthy human development.

London: Routledge.

- Boxer, J. (2005). Promoting mental health. *The Handbook of Community Mental Health Nursing*, 149–163. https://doi.org/10.5840/ncbq201616462
- Boyland, J. R. (2019). A social constructivist approach to the gathering of empirical data. *Australian Counselling Research Journal*, 13(2), 30–34. Retrieved from http://www.acrjournal.com.au/resources/assets/journals/Volume-13-Issue-2-2019/Manuscript5 - A Social Constructivist Approach.pdf
- Brady, L. (2017). RHETORIC TO REALITY : AN INQUIRY INTO EMBEDDING YOUNG PEOPLE 'S PARTICIPATION IN HEALTH SERVICES AND RESEARCH A thesis submitted in partial fulfilment of the requirements of the University of the West. (March). Retrieved from https://core.ac.uk/download/pdf/323893755.pdf
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2), 77–101. https://doi.org/10.1191/1478088706qp063oa
- Braun, V., & Clarke, V. (2012). Thematic analysis. APA Handbook of Research Methods in Psychology, Vol 2: Research Designs: Quantitative, Qualitative, Neuropsychological, and Biological., 2, 57–71. https://doi.org/10.1037/13620-004
- Brewer, K. (2017). Possible Selves on Probation: The Role of Future-oriented Identity Beliefs in Promoting Successful Outcomes for Adolescents on Probation. Columbia.
- Brinkmann, S., & Kvale, S. (2005). Confronting the ethics of qualitative research. Journal of Constructivist Psychology, 18(2). Retrieved from chromeextension://dagcmkpagjlhakfdhnbomgmjdpkdklff/enhancedreader.html?openApp&pdf=https%3A%2F%2Fwww.tandfonline.com%2Fdoi%2Fpdf% 2F10.1080%2F10720530590914789%3Fcasa_token%3DOFBFULHNu2gAAAAA%3A 4x5u32xcvp7mIL0pwtF0ZP8iewJFvvrGB85PWaGP7KuA9tgdPkc2v1FIjo9X
- Bronfenbrenner, U. (1977). Toward an experimental ecology of human development. *American Psychologist*, 32(7), 513.
- Bronfenbrenner, U. (1994). Ecological Models of Human Development. In *International Encyclopaedia of Education,* (2nd ed., Vol. 3). Oxford: Elsevier.
- Broussine, M. (2008). Creative Methods in Organizational Research (1st ed.). London: Sage.
- Brubaker, M., Puig, A., Reese, R., & Young, J. (2010). Integrating social justice into counseling theories pedagogy: A case example. *Counselor Education and Supervision*, Vol. 50, pp. 88–102. https://doi.org/10.1002/j.1556-6978.2010.tb00111.x
- Bruner, J. (1991). The Narrative Construction of Reality. *Critical Enquiry*, 18(1). Retrieved from https://www.academia.edu/9503047/The_Narrative_Construction_of_Reality

- Burger, K., & Walk, M. (2016). Can children break the cycle of disadvantage? Structure and agency in the transmission of education across generations. *Social Psychology of Education*, 19(4), 695–713. https://doi.org/10.1007/s11218-016-9361-y
- Burgess, R. (1989). The Ethics of Educational Research (R. Burgess, Ed.). The Falmer Press.
- Burr, V. (2015). Socia Constructionism (3rd ed.). Retrieved from https://www.taylorfrancis.com/books/mono/10.4324/9781315715421/socialconstructionism-vivien-burr
- Burton, M., Pavord, E., & Williams, B. (2014). An Introduction to Child and Adolescent Mental Health. Retrieved from https://uk.sagepub.com/sites/default/files/upmbinaries/61583_Burton_An_Introduction_to_Child_and_Adolescent_Mental_Health.pdf
- Butler, P. (2018). 1,000 Sure Start children's centres may have shut since 2010. Guardian Newspaper. Retrieved from https://www.theguardian.com/society/2018/apr/05/1000sure-start-childrens-centres-may-have-shut-since-2010
- Byrne, B., & Lundy, L. (2019). Children's Rights and Policy-Making: a 6 P framework. Children's Rights and Policy-Making:, 1–2. Retrieved from https://doi.org/10.1080/13642987.2018.1558977
- Byskov, M. F. (2021). What Makes Epistemic Injustice an "Injustice"? *Journal of Social Philosophy*, 52(1), 114–131. https://doi.org/10.1111/josp.12348
- Cammarota, J., & Fine, M. (2008). Youth Participatory Action Research: A Pedagogy for Transformational Resistance. In M. Cammarota, J., & Fine (Ed.), *Revolutionizing Education* (pp. 1–13). New York: Routledge.
- Campbell, R. & Morris, M. (2017). The stories we tell: Introduction to the special issue on ethical challenges in community psychology research and practice. *American Journal of Community Psychology*, 60(3–4).
- Carpenter, B & Carpenter, M. (2020). Think Piece. https://doi.org/10.1109/MAHC.2002.988588
- Cattan M, T. S. (2006). *Mental Health Promotion: A lifespan approach*. New York: McGraw-Hill International.
- Cava, M. J., Buelga, S., & Tomás, I. (2021). Peer Victimization and Dating Violence Victimization: The Mediating Role of Loneliness, Depressed Mood, and Life Satisfaction. *Journal of Interpersonal Violence*, *36*(5–6), 2677–2702. https://doi.org/10.1177/0886260518760013
- Charmaz, K. (2000). Grounded theory: Objectivist and contructivist methods. Thousand Oaks, CA: Sage Publications. In N. K. Denzin and Y. Lincoln. (Ed.), *The Handbook of*

Qualitative Research. Thousand Oaks, CA: : Sage Publications Inc.

- Charteris, J., & Smardon, D. (2019). 'Student voice in learning: instrumentalism and tokenism or opportunity for altering the status and positioning of students?' *Pedagogy*, *Culture and Society*, 27(2), 305–323. https://doi.org/10.1080/14681366.2018.1489887
- Chatfield, K,. & Morton, D. (2020). Ethics Dumping Case Studies from North- South Research (D. Schroeder, J. Cook, F. Hirsch, & S. Fenet, Eds.). Retrieved from https://library.oapen.org/bitstream/handle/20.500.12657/27812/1002193.pdf?sequence= 1#page=14
- Cheney, K. (2019). Decolonizing Childhood Studies: Overcoming Patriachy and Prejudice In Child-related RTesearch and Practice. In D. Spyrou, S; Rosen, R; Cook (Ed.), *Reimagining Childhood Studies*. London: Bloomsbury Academic.
- Chiva, R., Ghauri, P., & Alegre, J. (2014). Organizational Learning, Innovation and Internationalization: A Complex System Model. *British Journal of Management*, 25(4), 687–705. https://doi.org/10.1111/1467-8551.12026
- Christensen, P. Prout, A. (2002). Working with ethical symmetry in social research with children. *Childhood*, 8(4), 477–497.
- Christensen, J. F., Di Costa, S., Beck, B., & Haggard, P. (2019). I just lost it! Fear and anger reduce the sense of agency: a study using intentional binding. *Experimental Brain Research*, 237(5), 1205–1212. https://doi.org/10.1007/s00221-018-5461-6
- Clark, A. (2010). Young children as protagonists and the role of participatory, visual methods in engaging multiple perspectives. *American Journal of Community Psychology*, 46(1), 115–123. https://doi.org/10.1007/s10464-010-9332-y
- Codd, J. (1990). Making distinctions: The eye of the beholder. In C. Harker, R & Mahar, C & Wilkes (Ed.), *An introduction to the work of Pierre Bourdieu: The practice of theory* (pp. 132–159). London: Macmillan.

Comber, B., & Hill, S. (2000). Socio-economic disadvantage, literacy and social justice: Learning from longitudinal case study research. *Australian Educational Researcher*, 27(3 79-97.), 79–97. Retrieved from https://www.researchgate.net/publication/257275156_Socioeconomic_disadvantage_literacy_and_social_justice_Learning_from_longitudinal_case_ study_research

Cook-Sather, A. (2020). Student voice across contexts: Fostering student agency in today's schools. *Theory into Practice*, Vol. 59, pp. 182–191. https://doi.org/10.1080/00405841.2019.1705091

- Cook, T. (2009). The purpose of mess in action research: Building rigour though a messy turn. *Educational Action Research*, 17(2), 277–291. https://doi.org/10.1080/09650790902914241
- Cook, T. (2012). FORUM : QUALITATIVE SOCIAL RESEARCH Where Participatory Approaches Meet Pragmatism in Funded (Health) Research : The Challenge of Finding Meaningful Spaces 2 . Participatory Research : Terminology and Illusory Consensus. *Forum Qualitative Sozialforschung*, 13(1). Retrieved from https://www.qualitativeresearch.net/index.php/fqs/article/view/1783
- Cooke, B & Kothari, U. (2001). *Participation: The new tyranny* (U. Cooke, B & Kothari, Ed.). London: Zed Books.
- Cooley. C. (1902). Human Nature and the Social Order. New York: Scribner.
- Corbin Dwyer, S., & Buckle, J. L. (2018). The Space Between: On Being an Insider-Outsider in Qualitative Research. *International Journal of Qualitative Methods*, 17(1), 54–63. https://doi.org/10.1177/1609406918788176
- Corney, T., Cooper, T., Shier, H., & Williamson, H. (2021). Youth participation: Adultism, human rights and professional youth work. *Children and Society*, (November), 1–14. https://doi.org/10.1111/chso.12526
- Corsaro, W. (2018). The Sociology of Childhood (5th ed.). Bloomington: Sage Publications.
- Cottone, R. Rocco. (2001). A Social Constructivism Model of Ethical Decision Making in Counseling. *Journal of Counseling and Development*, 79(1), 39–45. https://doi.org/10.1002/j.1556-6676.2001.tb01941.x
- Cottone, R R. (2004). Displacing the Psychology of the Individual in Ethical Decision-Making: The Social Constructivism Model. *Canadian Journal of Counselling*, *38*, 5–13. Retrieved from

http://search.ebscohost.com/login.aspx?direct=true&db=eric&AN=EJ719896&site=ehos t-live&scope=site

- Counts, G. S. (1932). *Dare the School build a new social order*. Retrieved from https://www.curezone.org/upload/PDF/Counts_George_Dare_the_School_Build_a_New _Social_Order_1932_.pdf
- Cowie, B., & Khoo, E. (2017). Accountability through access, authenticity and advocacy when researching with young children. *International Journal of Inclusive Education*, 21(3), 234–247. https://doi.org/10.1080/13603116.2016.1260821

Cox, C. B. (1981). Education: The next Decade. London: Conservative Political Centre.

Craig, C. J., JeongAe, Y., & Zou, Y. (2016). The narrative nature of embodied knowledge: A
- Crichton, P.; Carel, H.; Kidd, I. (2017). Epistemic Injustice in psychiatry. *BJPsych BulletinPsych*, *41*(2), 65–70. Retrieved from https://pubmed.ncbi.nlm.nih.gov/28400962/
- Croizet, J. C., Goudeau, S., Marot, M., & Millet, M. (2017). How do educational contexts contribute to the social class achievement gap: documenting symbolic violence from a social psychological point of view. *Current Opinion in Psychology*, 18, 105–110. https://doi.org/10.1016/j.copsyc.2017.08.025
- Crone, E. (2017). *The Adolescent Brain: Changes in Learning, Decision-Making and Social Realations*. Abingdon: Routledge.
- Crone, E. A., & Konijn, E. A. (2018). Media use and brain development during adolescence. *Nature Communications*, Vol. 9. https://doi.org/10.1038/s41467-018-03126-x

Crotty, M. (1998). The Foundations of Social Research. London: Sage Publications.

- Cullingford, C. (1988). Children's views about working together. *Education 3-13*, *16*(1), 29–34. https://doi.org/10.1080/03004278885200071
- Cummings, E. M., & Davies, P. (1996). Emotional security as a regulatory process in normal development and the development of psychopathology. *Development and Psychopathology*, 8(1), 123–139. https://doi.org/10.1017/s0954579400007008
- Cutting, K., & Peacock, S. (2021). Making sense of 'slippages': re-evaluating ethics for digital research with children and young people. *Children's Geographies*, 0(0), 1–13. https://doi.org/10.1080/14733285.2021.1906404
- Daily Mail. (2005, April 19). "Tories" five-point plan to tackle yob culture. *Daily Mail*. Retrieved from https://www.dailymail.co.uk/news/article-345463/Tories-point-plantackle-yob-culture.html
- Damon, W. (1983). Social and Personality Development: Infancy Through Adolescence. New York: W.W.Norton and Company.
- Darder, A. (1997). Creating the conditions for cultural democracy in the classroom. In H. Darder, A.; Torres, R. & Guiterrez (Ed.), *Latinos and Education* (pp. 331–350). New York: Routledge.

- Darling-Hammond, L., Flook, L., Cook-Harvey, C., Barron, B., & Osher, D. (2020).
 Implications for educational practice of the science of learning and development.
 Applied Developmental Science, 24(2), 97–140.
 https://doi.org/10.1080/10888691.2018.1537791
- Davidson, D. (2000). The National Service Framework for Mental Health. *Journal of Integrated Care*, 8(1), 5–10. https://doi.org/10.1108/1476901820000002
- Dawson, C., & Parker, J. R. (1998). New Orleans: paper presented at the Mid-South Educational Research Association. A Descriptive Analysis of the Perspective of Neville High School Teachers Regarding the School Renovation. New Orleans.
- Delgado, M. (2008). Designs and methods for youth-led research. Thousand Oaks, CA: Sage.
- DeMarrais, K. B. (1995). *The way schools work : a sociological analysis of education* (2nd ed.; Longman, Ed.). New York.
- Dennett, D. C. (1992). The self as a centre of narrative gravity. In D. Kessel, F; Cole, P; Johnson (Ed.), Self and consciousness: Multiple perspectives (pp. 67–83). https://doi.org/10.5209/rev-ASEM.2013.v46.42862
- Denzin, N., & Lincoln, Y. (2017). The SAGE handbook of qualitative research (5th ed.; Y. Denzin, N., & Lincoln, Ed.). Sage.
- Denzin, N., & Lincoln, Y. (1998). *Collecting and interpreting qualitative materials*. Thousand Oaks, CA: Sage.
- Denzin, N. & Lincoln, Y. (2000). *Handbook of qualitative research* (2nd ed.). Thousands Oaks, CA: Sage.
- Department for Education. (2014). *Mental health and behaviour in schools*. (March). Retrieved from
 - https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment _data/file/1069687/Mental_health_and_behaviour_in_schools.pdf
- Department for Education. (2015). Schools: guide to the 0 to 25 SEND code of practice. Department for Education, p. 35. Retrieved from

https://www.gov.uk/government/publications/send-guide-for-schools-and-alternative-provision-settings

- Department of Health and Department of Education. (2017). Transforming Children and Young People's Mental Health Provision: a Green Paper. In *Journal of Immunology* (Vol. 110).
- Department of Health and Social Care, & Department for Education. (2017). Government Response to the Consultation on Transforming Children and Young People 's Mental

Health Provision : a Green Paper and Next Steps. Retrieved from

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment _data/file/728892/government-response-to-consultation-on-transforming-children-andyoung-peoples-mental-health.pdf

- DES. (1965). *Circular 10/65: THE ORGANISATION OF SECONDARY EDUCATION*. Retrieved from http://www.educationengland.org.uk/documents/des/circular10-65.html
- DeWall, C., Masten, C., Powell, C., Combs, D., Schurtz, D., Eisenberger, N. (2012). Do neural responses to rejection dependon attachment style? An fMRI study.pdf. Social Cognitive Affect Neuroscience, 7(2), 184–192. Retrieved from chromeextension://dagcmkpagjlhakfdhnbomgmjdpkdklff/enhancedreader.html?openApp&pdf=https%3A%2F%2Fwatermark.silverchair.com%2Fnsq107.p df%3Ftoken%3DAQECAHi208BE49Ooan9kkhW_Ercy7Dm3ZL_9Cf3qfKAc485ysgA AAsAwggK8BgkqhkiG9w0BBwagggKtMIICqQIBADCCAqIGCSqGSIb3DQE

Dewey, J. (1938). Logic: The Theory of Inquiry. New York: Holt, Rinehart and Winston.

Dewey, J. (2007). Democracy and Education,. Teddington: Echo Library.

Dewey, John, & Cohen, M. R. (n.d.). EXPERIENCE & EDUCATION.

- DfE/DoH. (2017). *Transforming children and young people's mental health provision*. Retrieved from https://assets.publishing.service.gov.uk/government/ uploads/system/uploads/attachment_data/!le/664855/Transforming_children_and_ young_people_s_mental_health_provision.pdf
- Di Fabio A, K. M. (2016). Promoting Well-Being: The Contribution of Emotional Intelligence. *Frontiers in Psychology*, 7. https://doi.org/10.3389/fpsyg.2016.01182
- Didau, D. (2016). The Dangerous Rise of Therapeutic Education. Retrieved from https://learningspy.co.uk/featured/therapeutic-vs-academic-education-new-dichotomy/
- Dieleman, S. (2012). An interview with Miranda Fricker. *Social Epistemology*, *26*(2), 253–261. https://doi.org/10.1080/02691728.2011.652216
- Donelle L, Facca D, Burke S, Hiebert B, Bender E, L. S. (2021). Exploring Canadian Children's Social Media Use, Digital Literacy, and Quality of Life: Pilot Cross-sectional Survey Study. *Journal of Medical Internet Research*, 5(5). Retrieved from url: https://formative.jmir.org/2021/5/e18771
- Donnellan, M. B., Trzesniewski, K. H., Robins, R. W., Moffitt, T. E., & Caspi, A. (2005).
 Low self-esteem is related to aggression, antisocial behavior, and delinquency. *Psychological Science*, 16(4), 328–335. https://doi.org/10.1111/j.0956-7976.2005.01535.x

- Driskell, D. (2002). Creating Better Cities with Children and Youth: A Manual for Participation, David Driskell. London: Unesco.
- Durkheim, E. (1982). The Rules of Sociological Method. New York: Free Press.
- Ecclestone, K., & Hayes, D. (2008). *The Dangerous Rise of Therapeutic Education*. https://doi.org/10.4324/9780203870563
- Edwards, R & Mauther, M. (2002). *Ethics and qualitative research: Theory and Pracitice*. (2nd ed.; J. Birch, M., Miller, T., Mauther, M. and Jessop, Ed.). London: Routledge.

Ellingson, L. (2008). *Engaging Crystallization in Qualitative Research: An Introduction*. Thousand Oaks, CA: Sage Publications, Inc.

- Elliott, G. (2012). Policy, performativity and partnership: an ethical leadership perspective. *Research in Post-Compulsory Education*, *17*(4). Retrieved from https://www.tandfonline.com/doi/abs/10.1080/13596748.2012.738977
- Ellis, C. (2007). Telling secrets, revealing lives: Relational ethics in research with intimate others. *Qualitative Inquiry*, *13*(1), 3–29. https://doi.org/10.1177/1077800406294947
- Ellison, N. (1996). Consensus Here, Consensus There ... but not Consensus Everywhere: The Labour Party, Equality and Social Policy in the 1950s. In H. J. et Al. (Ed.), *The Myth of Consensus*. Palgrave Macmillan,.
- Engel, G. L. (1977). The need for a new medical model: A challenge for biomedicine. *Science*, *196*(4286), 377–396. https://doi.org/10.1521/pdps.2012.40.3.377
- Ennew, J and Plateau, D. P. (2004). *How to research the Physical and Emotional punishment of children*.
- Erickson, F. & Christman, J. (1996). Taking stock/making change: Stories of collaboration in local school reform. *Theory Into Practice*, *35*(3), 149–157.
- Erickson, F. (2006). Studying side by side: collaborative action ethnography in education research. In L. Spindler, G., & Hammond (Ed.), *Innovations in educational ethnography:theory, method and results.* Mahwah: Lawrence Erlbaum.
- Etherington, K. (2004). *Becoming a Reflexive Researcher*. London: Jessica Kingsley Publishers.
- Etzel, E., & Watson, J., I. (2006). Introduction to the special issue: ethics in sport and exercise psychology. *Ethics and Behaviour*, *16*(1).
- Eynon, B., Torok, J., & Gambino, L. M. (2013). USING E-PORTFOLIOS IN HYBRID PROFESSIONAL. *To Improve the Academy*, 109–125.
- Families, A. F. N. C. for C. and. (2022). Social and emotional skills.
- Flasher, J. (1979). Adultism. Adolescence, 13(51), 517–523.

- Flecha, R. (2009). The dialogic sociology of the learning communities. In L. Apple, M., Ball.S., & Gandin (Ed.), *The Routledge International Handbook of the Sociology of Education*. Routledge.
- Flutter, J. & Rudduck, J. (2004). *Consulting pupils: what's in it for schools?* London: Routledge Falmer.
- Foot, J., & Hopkins, T. (2010). A glass half-full: how an asset approach can improve community health and well-being. *Local Government Improvement and Development*. Retrieved from http://www.assetbasedconsulting.net/uploads/publications/A glass half full.pdf
- Foucault, M. (2001). Fearless Speech (J. Peareson, Ed.). Cambridge MA: Semiotext(e).
- Foulkes, L., & Andrews, J. L. (2022). The Prevalence Inflation Hypothesis: Are mental health awareness efforts contributing to the rise in reported mental health problems? *PsyArXiv*, 69(November 2022), 101010. https://doi.org/10.1016/j.newideapsych.2023.101010
- Francis, D., & Hester, S. (2012). *An Invitation to Ethnomethodology: Language, Society and Social Interaction* (online). https://doi.org/10.4135/9781849208567
- Freeman, J. G., Samdal, O., Klinger, D. A., Dur, W., Griebler, R., Currie, D., & Rasmussen, M. (2009). The relationship of schools to emotional health and bullying. *International Journal of Public Health*, 54, 251–259. Retrieved from https://www.research.ed.ac.uk/en/publications/the-relationship-of-schools-to-emotional-health-and-bullying
- Freeman, M. (1996). Children's education; a test case for best interests and autonomy,. In D.Davie, R. & Galloway (Ed.), *Listening to children in education*. London: David Fulton.
- Freiberg, H. J., & Stein, T. A. (1999). Measuring, improving and sustaining healthy learning environments. (H. J. Freiberg, Ed.). Philadelphia: Farmer Press.
- Freire, P. (1970). Pedagogy of the Oppressed. London: Penguin.
- Freire, P. (1974). Education for Critical Consciousness. London: Continuum.
- Fresnoza-Flot, A., & Shinozaki, K. (2017). Transnational perspectives on intersecting experiences: gender, social class and generation among Southeast Asian migrants and their families. *Journal of Ethnic and Migration Studies*, 43(6), 867–884. https://doi.org/10.1080/1369183X.2016.1274001
- Fricker, M. (2007). *Epistemic Injustice: Power and the Ethics of Knowing*. Oxford: Oxford University Press.
- Friedman, S. (2014). The Price of the Ticket: Rethinking the Experience of Social Mobility.

Sociology, 48(2), 352–368. https://doi.org/10.1177/0038038513490355

- Fry, C., Treloar, C., & Maher, L. (2005). Ethical challenges and responses in harm reduction research: Promoting applied communitarian ethics. *Drug and Alcohol Review*, 24(5), 449–459. https://doi.org/10.1080/09595230500263905
- Fuhrmann, D., Knoll, L. J., & Blakemore, S. J. (2015). Adolescence as a Sensitive Period of Brain Development. *Trends in Cognitive Sciences*, 19(10), 558–566. https://doi.org/10.1016/j.tics.2015.07.008
- Gajdics, J., & Jagodics, B. (2021). Mobile Phones in Schools: With or Without you?
 Comparison of Students' Anxiety Level and Class Engagement After Regular and
 Mobile-Free School Days. *Technology, Knowledge and Learning*, (0123456789).
 https://doi.org/10.1007/s10758-021-09539-w
- Galindo, C., & Sheldon, S. B. (2012). School and home connections and children's kindergarten achievement gains: The mediating role of family involvement. *Early Childhood Research Quarterly*, Vol. 27, pp. 90–103. https://doi.org/10.1016/j.ecresq.2011.05.004
- Gallagher, S. (2007). Pathologies in Narrative Structures. *Royal Institute of Philosophy* Supplement, 60(May 2007), 203–224. https://doi.org/10.1017/s1358246107000100
- Garber, J., Frankel, S. A., & Herrington, C. G. (2016). Developmental Demands of Cognitive Behavioral Therapy for Depression in Children and Adolescents: Cognitive, Social, and Emotional Processes. *Annual Review of Clinical Psychology*, Vol. 12, pp. 181–216. https://doi.org/10.1146/annurev-clinpsy-032814-112836
- Geldard, K, Geldard, D. & Yin Foo, R. (2020). *Counselling Adolescents: The Proactive Approach for Young People* (5th ed.). London: Sage Publications Ltd.
- Gergen, K. J. (1985). The Social Constructionist Movement in Modern Psychology. *American Psychologist*, 40(3), 266–275. https://doi.org/10.1037/0003-066X.40.3.266
- Gewirtz, S., Ball, S. J., & Bowe, R. (1995). *Markets, choice and equity in education* (*Buckingham, Open University Press*). Buckingham: Open University Press.
- Gibb, N. (2021). The importance of a knowledge-rich curriculum.
- Giddens, A. (1979). Central Problems in Social Theory. London: MacMillan Publising Company.
- Gillard, D. (2018). Education in England: a history. Retrieved from www.educationengland.org.uk/history
- Ginwright, S., & Cammarota, J. (2002). New Terrain in Youth Development: The Promise of a Social Justice Approach. *Social Justice*, *29*(4), 82–95. Retrieved from

https://www.jstor.org/stable/pdf/29768150.pdf?refreqid=excelsior%3A3e30b67e3a11d8 4d0a3624a22b9d4eaa

- Glassman, M., & Erdem, G. (2014). Participatory Action Research and Its Meanings. *Adult Education Quarterly*, 64(3), 206–221. https://doi.org/10.1177/0741713614523667
- Glazzard, J., & Stones, S. (2021). Supporting Young People 's Mental Health : Reconceptualizing the Role of Schools or a Step Too far ? 5(February), 1–5. https://doi.org/10.3389/feduc.2020.607939
- Gómez, A., Puigvert, L., & Flecha, R. (2011). Critical communicative methodology: Informing real social transformation through research. *Qualitative Inquiry*, 17(3), 235–245. https://doi.org/10.1177/1077800410397802
- Gorard, S. (2012). Experiencing fairness at school: An international study. *International Journal of Educational Research*, 53, 127–137. https://doi.org/10.1016/j.ijer.2012.03.003
- Gould, S. (2011). The Hedgehog, the fox and the magister's pox. Bleknap Press.
- Grace, G. (1996). School Leadership: Beyond Educational Management. London: Falmer.
- Graham, A., & Fitzgerald, R. (2010). Children's participation in research: Some possibilities and constraints in the current australian research environment. *Journal of Sociology*, 46(2), 133–147. https://doi.org/10.1177/1440783309355065
- Gregory I. (2003). Ethics in Research (1st ed.). https://doi.org/9780826464774
- Grenfell, M. & James, D. (1998). Theory, practice and pedagogic research. In M. & J. D.Grenfell (Ed.), *Bourdieu and education: acts of practical theory* (pp. 6–26). London,: Falmer Press.
- Grotevant, H. D., & Cooper, C. R. (2009). Individuation in Family Relationships. *Human* Development, 29(2), 82–100. https://doi.org/10.1159/000273025
- Groundwater-Smith, S., Dockett, S. & Bottrell, D. (2015). *Particapatory Research with Children and Young People* (1st ed.). London: Sage.
- Guyer, A. E., Caouette, J. D., Lee, C. C., & Ruiz, S. K. (2014). Will they like me?
 Adolescents' emotional responses to peer evaluation. *International Journal of Behavioral Development*, 38(2), 155–163. https://doi.org/10.1177/0165025413515627
- Habermas, J. (1984). *The theory of communicative action. VI: Reason and the rationalization of society.* Boston: Beacon Press.
- Habermas, J. (1987). *The theory of communicative action: Vol. 1. Reason and the rationalization of society.* Boston: Beacon Press.
- Haggard, P., & Chambon, V. (2012). Sense of agency. Current Biology, 22(10), R390-R392.

https://doi.org/10.1016/j.cub.2012.02.040

- Hall. G. (1904). Adolescence: Its psychology and its relations to physiology, anthropology, sociology, sex, crime, religion, and education. New York: Appleton.
- Halpin, A. W., & Croft, D. B. (1963). The organizational climate of schools.
- Hammond, L. D., & Harvey, C. C. (2018). Educating the whole child: Improving school climate to support student success. In *Learning Policy Institute*. Retrieved from https://bit.ly/2xd17CZ
- Harlow, L. & Oswald, F. (2016). Big data in psychology: introduction to the special issue. *Psychological Methods*, *21*(4).
- Harrison, N., & Waller, R. (2018). Challenging discourses of aspiration: The role of expectations and attainment in access to higher education. *British Educational Research Journal*, 44(5), 914–938. https://doi.org/10.1002/berj.3475
- Hart, C. & Chesson, R. (1998). Children as consumers. *British Medical Journal*, *316*, 1600–1603.
- Hart, A. (1992). *Children's participation: From tokenism to citizenship*. Florence: UNICEF International Child Development Centre.
- Harter, S. (2012). *The Construction of the Self:Second Edition: Developmental and Sociocultural Foundations* (2nd ed.). New York: The Guilford Press.
- Hattie, J. (2015). *What doesn't work in Education: The politics of distraction*. Retrieved from https://visible-learning.org/wp-content/uploads/2015/06/John-Hattie-Visible-Learning-creative-commons-book-free-PDF-download-What-doesn-t-work-in-education_the-politics-of-distraction-pearson-2015.pdf
- Hauser, S et al. (1991). Adolescents and their families: paths of ego development. New York: The Free Press.
- Haynes, N. M., Emmons, C., & Ben-Avie, M. (1997). School Climate as a Factor in Student Adjustment and Achievement. *Journal of Educational and Psychological Consultation*, 8(3), 321–329. https://doi.org/10.1207/s1532768xjepc0803 4
- Hein, G. E. (2007). Constructivist Learning Theory. Manachusetts.: Lesley College Press.
- Hendricks, C. C. (2017). *Improving schools through action research: A reflective practice approach*. New Jersey: Pearson.
- Henslin, J. M. (1999). *Sociology: A down-toearth approach*, (4th ed). Boston, MA: Allyn and Bacon.
- Herrman, H., Moodie, R., Saxena, S., Izutsu, T., & Tsutsumi, A. (2016). Mental Health Promotion. *International Encyclopedia of Public Health*, 93–105.

https://doi.org/10.1016/B978-0-12-803678-5.00286-1

- Hi Kim, K. (2017). Transformative educational actions for children in poverty: Sen's capa bility approach into practice in the Korean context. *Education as Change*, 21(1), 174–192. https://doi.org/10.17159/1947-9417/2017/2338
- Hill, M., Davis, J., Prout, A., & Tisdall, K. (2004). Moving the Participation Agenda Forward. *Children and Society*, Vol. 18, pp. 77–96. https://doi.org/10.1002/chi.819
- Hill, R. W. (2017). *Civility is Needed in College Classroom Now More than Ever Civility is Needed in College Classroom Now More than Ever.*
- Hilppö, J., Chimirri, N. A., & Rajala, A. (2019). Theorizing Research Ethics for the Study of Psychological Phenomena from Within Relational Everyday Life. *Human Arenas*, Vol. 2, pp. 405–415. https://doi.org/10.1007/s42087-019-00073-x
- Hipolito-Delgado, C. P., Stickney, D., Zion, S., & Kirshner, B. (2022). Transformative Student Voice for Sociopolitical Development: Developing Youth of Color as Political Actors. *Journal of Research on Adolescence*, 1–11. https://doi.org/10.1111/jora.12753
- Hirsch, D. and J. A. (2007). Experiences of poverty and educational disadvantage. *Education*,p. 10. Retrieved from http://www.jrf.org.uk/sites/files/jrf/2123.pdf
- HMSO. (1975). The Bullock Report: A Language For Life. London.
- Hoddinott, P., Pollock, A., O'cathain, A., Boyer, I., Taylor, J., Macdonald, C., ... Donovan, J. L. (2018). How to incorporate patient and public perspectives into the design and conduct of research [version 1; peer review: 3 approved, 2 approved with reservations]. *F1000Research*, 7. https://doi.org/10.12688/F1000RESEARCH.15162.1
- Hong, J., & Eamon, M. (2012). Students' perceptions of unsafe schools: an ecological systems analysis. *Journal of Child & Family Studies*, 21, 428–438. Retrieved from doi:10.1007/s10826-011-9494-8
- hooks, b. (2017). Confronting Class in the Classroom. In R. Darder, A; Baltodano, M; Torres (Ed.), *The Critical Pedagogy Reader* (3rd ed.). Abingdon: Routledge.
- Hoonaard, van den. (2011). Seduction of Ethics. University of Toronto Press.
- Horgan, D., Forde, C., Martin, S., & Parkes, A. (2017). Children's participation: moving from the performative to the social. *Children's Geographies*, Vol. 15, pp. 274–288. https://doi.org/10.1080/14733285.2016.1219022
- Houlders, J. W., Bortolotti, L., & Broome, M. R. (2021). Threats to epistemic agency in young people with unusual experiences and beliefs. *Synthese*. https://doi.org/10.1007/s11229-021-03133-4
- Huber, M., André Knottnerus, J., Green, L., Van Der Horst, H., Jadad, A. R., Kromhout, D.,

... Smid, H. (2011). How should we define health? *BMJ (Online)*, *343*(7817). https://doi.org/10.1136/bmj.d4163

- Huttunen, R., & Kakkori, L. (2002). *The hermeneutics of truth and selfhood heidegger's, gadamer's and ricouer's significance in the autobiographical research.* (December 2016).
- Ianni, F. (1989). A search for structure: a report on American youth today. New York: Free Press.
- Ignat, S., & Galatiu, C. (2022). INTERNET DEPENDENCE / SOCIAL NETWORKS AND THE DEVELOPMENT OF TEENAGE SOCIAL ANXIETY. *Agora Psycho-Pragmatica*, 15(1).
- Islam, G. (2014). Social identity theory. In T. Teo (Ed.), *Encyclopedia of Critical Psychology* (pp. 1781–1783). https://doi.org/10.5840/iabsproc19967101
- Issacs, W. N. (1993). Taking Flight: Dialogue, Collective Thinking and Organizational Learning. Organizational Dynamics, 22, 24–39.
- Jenkins, R. (2002). Pierre Bourdieu (London, Routledge). London: Routledge.
- Jenkins, R. (2004). Social Identity. 2nd edn. London: Routledge. (2nd ed.). London: Routledge.
- Jessop, B. (1980). The transformation of the state in post- war Britain. In R. Scase (Ed.), *The State in Western Europe*. London: Croom Helm.
- Jones, M. & Karsten, H. (2008). Giddens's Structuration Theory and Information Systems Research. *MIS Quaterly*, *32*(1), 127–157. Retrieved from https://www.jstor.org/stable/pdf/25148831.pdf?refreqid=excelsior%3A1ab4a0fb24b464 7f339ffdfc7aa11f5e&ab_segments=&origin=&acceptTC=1
- Jones A. (2004). Involving children and young people as researchers: Issues, ethics and practice. In et al. Fraser S, Lewis V, Ding S (Ed.), *Doing research with children and young people*. (pp. 113–130). London: Open University/Sage.
- Juvonen, J. (2007). Reforming middle schools: Focus on continuity, social connectedness, and engagement. *Educational Psychologist*, 42(4), 197–208. https://doi.org/10.1080/00461520701621046
- Kalliala, M. (2014). Toddlers as both more and less competent social actors in Finnish day care centres. *Early Years*, *34*(1), 4–17. https://doi.org/10.1080/09575146.2013.854320
- Kane M. and Bibby J. (2018). A Place to Grow Old. In *The Health Foundation*. https://doi.org/10.1093/geront/25.3.326

Katz Rothman, B. (2007). Writing Ourselves in Sociology. Methodological Innovation

Online, 2, 11-16. https://doi.org/10.4256/mio.2007.0003

- Keaten, J., Kelly, L., Keaten, J., & Kelly, L. (2008). Emotional Intelligence as a Mediator of Family Communication Patterns and Reticence Emotional Intelligence as a Mediator of Family Communication Patterns and Reticence. *Communication Reports*, 22(2), 104– 116. https://doi.org/10.1080/08934210802393008
- Keehn, J. (1982). *The ethics of psychological research* (J. Keehn, Ed.). Oxford: Pergamon Press Ltd.
- Kelley, R. C., Thornton, B., & Daugherty, R. (2005). Relationships between measures of leadership and school climate. *Education*, 126, 17–25. Retrieved from https://www.semanticscholar.org/paper/Relationships-between-Measures-of-Leadershipand-Kelley-Thornton/ad9ed990c9ba8be383e6f9238d88b7aa59ca7e85
- Kemmis, S., McTaggart, R. and Nixon, R. (2014). *The action research planner: Doing critical participatory action research*. https://doi.org/https://doi.org/10.1007/978-981-4560-67-2
- Kennedy, J. H., & Kennedy, C. E. (2004). Attachment theory: Implications for school psychology. *Psychology in the Schools*, *41*(2), 247–259.
- Kennedy, H. (2018). How adults change from facilitating youth participatory action research: Process and outcomes. *Children and Youth Services Review*, 94(June), 298–305. https://doi.org/10.1016/j.childyouth.2018.10.010
- Kenway, J., & McLeod, J. (2004). Bourdieu's reflexive sociology and "spaces of points of view": Whose reflexivity, which perspective? *British Journal of Sociology of Education*, 25(4), 525–544. https://doi.org/10.1080/0142569042000236998
- Kilkelly, U., Kilpatrick, R., Lundy, L., Moore, L., Scraton, P., Davey, C., Dwyer, C. andAlister, S. (2004). *Children's rights in Northern Ireland*. Retrieved from https://dera.ioe.ac.uk/9165/1/22323 Final.pdf
- Kim, B. (2006). Social Constructivism. New York: Routledge.
- Kim, Beaumie. (2000). Social constructivism. https://doi.org/10.1093/0199251614.003.0009
- Kim, J. M. (1998). Korean children's concepts of adult and peer authority and moral reasoning. *Developmental Psychology*, 34(5), 947–955. https://doi.org/10.1037/0012-1649.34.5.947
- Kim, M. S. (2014). Doing social constructivist research means making empathic and aesthetic connections with participants. *European Early Childhood Education Research Journal*, 22(4), 538–553. https://doi.org/10.1080/1350293X.2014.947835

King, A. (2010). The odd couple: Margaret Archer, Anthony Giddens and British social

theory. *British Journal of Sociology*, *61*(SUPPL. 1), 253–260. https://doi.org/10.1111/j.1468-4446.2009.01288.x

- King, J. (1991). Dysconscious Racism: Ideology, Identity, and the Miseducation of Teachers. *The Journal of Negro Education*, 60(2).
- Kitzinger, J. (1997). Who are you kidding? Children power and the struggle against sexual abuse. In A. James & A. Prout (Ed.), *Constructing and reconstructing childhood: contemporary issues in the sociological study of childhood* (2nd ed.). London: Falmer Press.
- Knight, C. (1990). *The Making of Tory Education Policy in Post-War Britain 1950-1986*.London: Falmer Press.
- Ko, M. L. M., & Krist, C. (2019). Opening up curricula to redistribute epistemic agency: A framework for supporting science teaching. *Science Education*, 103(4), 979–1010. https://doi.org/10.1002/sce.21511
- Koehn, A & Kerns, K. (2018). Parent–child attachment: meta-analysis of associations with parenting behaviors in middle childhood. *Attachment & Human Development*, 20(4), 378-405,.

https://doi.org/https://www.tandfonline.com/action/showCitFormats?doi=10.1080/14616 734.2017.1408131

- Kuperminc, G. P., Leadbeater, B. J., & Blatt, S. J. (2001). School social climate and individual differences in vulnerability to psychopathology among middle school students. Journal of School Psychology, 39, 141–159. *Journal of School Psychology*, 39, 141–159. Retrieved from https://people.unica.it/lauravismara/files/2012/04/medie-inferiori.pdf
- Kuperminc, G. P., Leadbeater, B. J., Emmons, C., & Blatt, S. (1997). Perceived school climate and difficulties in the social adjustment of middle school students. *Applied Developmental Science*, 1, 76–88. Retrieved from https://doi.org/10.1016/S0022-4405(01)00059-0
- Lafferty-jenkins, C. (2017). *Pupil participation in decision making and the role of school councils in primary schools : An exploration of the views of school council members and staff*. (Exeter). Retrieved from https://ore.exeter.ac.uk/repository/bitstream/handle/10871/29654/Lafferty-JenkinsC.pdf?sequence=2&isAllowed=y
- Lamsal, M. (2012). The Structuration Approach of Anthony Giddens. *Himalayan Journal of Sociology and Anthropology*, *5*, 111–122. https://doi.org/10.3126/hjsa.v5i0.7043

- Langhout, R. D., & Thomas, E. (2010). Imagining participatory action research in collaboration with children: An introduction. *American Journal of Community Psychology*, Vol. 46, pp. 60–66. https://doi.org/10.1007/s10464-010-9321-1
- Larson, R. W., & Richards, M. H. (1991). Daily Companionship in Late Childhood and Early Adolescence : Changing Developmental Contexts. Society for Research in Child Developmentesearch in Child Development, 62(2), 284–300. Retrieved from https://www.jstor.org/stable/pdf/1131003.pdf?casa_token=df8q3Wq0Hg0AAAAA:UCE dcE9Ye_KE3i2bCNUQqAw0D3MScvG8Q08HVvyMtES-tpfoXPS19PUZBgj6Zczt2-Ft1gEF4BGEZwQRImprNISBqtNxTqaOGhmbnFBV0jSoLxz1_HQ
- Lather, P. (1986). Research as Praxis. *Harvard Educational Review*, 56(3), 257–277. Retrieved from https://doi.org/10.17763/haer.56.3.bj2h231877069482
- Laura, L. (2007). "Voice" is not enough: Conceptualising Article 12 of the United Nations Convention on the Rights of the Child. *British Educational Research Journal*, 33(6), 927–942. https://doi.org/10.1080/01411920701657033
- Lawler, S. (1999). Getting out and getting away. Feminist Review, 63(63), pp.3-24.
- Lawrence, P. (2022). Hearing and acting with the voices of children in early childhood. Journal of the British Academy, 8s4(June), 077–090. https://doi.org/10.5871/jba/008s4.077
- Laws, S. (1998). *Hear me! Consulting with young people on mental health services*. London: The Mental Health Foundation.
- Leckie, G., & Goldstein, H. (2016). The evolution of school league tables in England 1992-2016: 'contextual value added', 'expected progress' and Dr George Leckie and Bristol Working Papers in Education The evolution of school league tables in England 1992-2016: 'contextual value. 1–34. Retrieved from https://www.bristol.ac.uk/media-library/sites/education/documents/bristol-working-papers-in-education/The evolution of school league tables in England 1992-2016.pdf
- Lee, T. (2020). *Three Questions on Social Constructs Influencing Curriculum Design*. Retrieved from
 - https://d1wqtxts1xzle7.cloudfront.net/64696559/Unit_1_Written_Assignment-withcover-page-v2.pdf?Expires=1652436917&Signature=GXurKeDqjf40zPLoJy5-Ad7xCLRfQC50LfkQmuMg2dUHnLRAIorkutAcbXdo4Yn5S7e5wpbM2Y5fKQGC6 A8uUoqpHzOqQI3nU3EeJjeWdaxkCOWzK6ZuQSHvtgVS1ibYtJ
- Leithwood, K. A., & Riehl, C. (2003). What we know about successful school leadership. Philadelphia: Laboratory for Student Success, Temple University. Philadelphia:

Laboratory for Student Success, Temple University.

- Lester, L., & Cross, D. (2015). The Relationship Between School Climate and Mental and Emotional Wellbeing Over the Transition from Primary to Secondary School. *Psychology of Well-Being*, Vol. 5. https://doi.org/10.1186/s13612-015-0037-8
- Lifestyles Team, N. D. (2020). *Mental health of children and young people in England 2020, Wave 1 follow-up to the 2017 survey (Health and Social Care Information Centre)* (pp. 1–4). pp. 1–4.
- Lingard, B., Hayes, D., Mills, M., & Christie, P. (2003). *Leading learning: Making hope practical in schools*. Buckingham: Open University Press.
- Lioncare. (2022). Lioncare specialist service for children. Retrieved July 18, 2022, from https://www.lioncare.co.uk/therapeutic-education/
- Lipsey, M.; Lowney, D.; Posner, G. & Lucas, M. (n.d.). Little Britain. United Kingdom.
- Lipsky, M. (1980). Street level bureaucrracy: Dilemas of the individual in public service (30th Anniv). New York: Russell Sage Foundation.
- Local Government Association. (2022). Children and young people's emotional wellbeing and mental health - facts and figures. In *Local Government Association*. Retrieved from https://www.local.gov.uk/about/campaigns/bright-futures/bright-futures-camhs/childand-adolescent-mental-health-and
- Lowe, R. (2004). New Labour, Old Labour: The Wilson and Callaghan governments, 1974-1979. Education Policy. (K. Seldon, A. and Hickson, Ed.). London: Routledge.
- Lundy, L., Journal, B. E., Convention, U. N., Model, T. L., Affairs, Y., & Strategy, N. (2020). The Lundy model of child participation The Lundy Voice Model Checklist for Participation. 10–12.
- Lyons, H. Z., Bike, D. H., Ojeda, L., Johnson, A., Rosales, R., & Flores, L. Y. (2013).
 Qualitative Research as Social Justice Practice with Culturally Diverse Populations.
 Journal for Social Action in Counseling and Pyschology, 5(2), 10–25.
- MacNeil, A. J., Prater, D. L., & Busch, S. (2009). The effects of school culture and climate on student achievement. *International Journal of Leadership in Education*, *12*(1), 73–84. https://doi.org/10.1080/13603120701576241
- Maffesoli, M. (1989). The sociology of everyday life (epistemological elements). Current Sociology, 37(1). Retrieved from https://journals.sagepub.com/doi/10.1177/001139289037001003
- Mann, C & Blakemore, S. (2021). *Adolescent Mental Health and Development*. Retrieved from

https://www.youtube.com/watch?v=2eKDFkpzKxc&ab channel=CambridgeUniversity

Mann, M., Hosman, C., Schaalma, H.,& de Vries, N. (2004). Self-esteem in a broad-spectrum approach formental health promotion. *Health Education Research Theory and Practice*, 19(4), 357–372. Retrieved from chromeextension://dagcmkpagjlhakfdhnbomgmjdpkdklff/enhanced-

reader.html?openApp&pdf=https%3A%2F%2Fwatermark.silverchair.com%2Fcyg041.p df%3Ftoken%3DAQECAHi208BE49Ooan9kkhW_Ercy7Dm3ZL_9Cf3qfKAc485ysgA AAr4wggK6BgkqhkiG9w0BBwagggKrMIICpwIBADCCAqAGCSqGSIb3DQE

- Manning, K. (1997). Authenticity in Constructivist Inquiry: Methodological Considerations Without Prescription. *Qualitative Inquiry*, *3*(1), 93–115.
- Manwell, L. A., Barbic, S. P., Roberts, K., Durisko, Z., Lee, C., Ware, E., & McKenzie, K. (2015). What is mental health? Evidence towards a new definition from a mixed methods multidisciplinary international survey. *BMJ Open*, 5(6), 1–11. https://doi.org/10.1136/bmjopen-2014-007079
- Marino, J. (2011). A study of school boards and their implementation of continuous improvement practices. *Journal for Quality & Participation*, *34*, 27–30.
- Marmot, M, Allen, J, Boyce, T, Goldblatt, P, Morrison, J. (2020). *HEALTH EQUITY IN* ENGLAND : The Marmot Review 10 years on.
- Marmot, M., Allen, J., Boyce, T., Goldblatt, P., & Morrison, J. (2020). *Health Equity in England*.
- Marrero Castro, J. F., & García, M. J. I. (2021). Ethics and Democracy in Access to Food. The Venezuelan Case. *Frontiers in Sustainable Food Systems*, Vol. 5. https://doi.org/10.3389/fsufs.2021.635033
- Marsh, D. (2010). Meta Theoretical Issues. In G. Marsh, D. & Stoker (Ed.), *Theory and Methods in Political Science*. (3rd ed., pp. 212-231.). Basingstoke: Palgrave Macmillan.
- Martin, A. J., & Dowson, M. (2009). Interpersonal relationships, motivation, engagement, and achievement: Yields for theory, current issues, and educational practice. *Review of Educational Research*, 79(1), 327–365.
- Martin, J., Sugarman, J., & Cane, J. (2003). *Psychology and the question of agency*. New York: Stae University of New York Press.
- Martinez-Vargas, C. (2022). Democratising Participatory Research: Pathways to Social Justice from the South. https://doi.org/https://doi.org/10.11647/OBP.0273
- Martínez, M. D., Martínez, C., & Martínez-marín, M. D. (2020). Subjective well being and gender typed attributes in adolescents : The relevance of emotional intelligence

Subjective well-being and gender-typed attributes in adolescents : The relevance of emotional intelligence. *Australian Journal of Psychology*, (71), 29304. https://doi.org/10.1111/ajpy.12247

- Maslow, A. H. (1943). (1943). A theory of human motivation. *Psychological Review*, 50(4), 370–396. Retrieved from
 https://www.academia.edu/9415670/A_Theory_of_Human_Motivation_Abraham_H_M
 aslow Psychological Review Vol 50 No 4 July 1943
- Mathie, A., & Cunningham, G. (2005). Who is driving development? Reflections on the transformative potential of asset-based community development. *Canadian Journal of Development Studies*, 26(1), 175–186. https://doi.org/10.1080/02255189.2005.9669031
- Matthews, G., Zeidner, M., Roberts, R. (2017). *The Handbook of Stress and Health: A Guide to Research and Practice*. John Wiley & Sons.
- Maynard, L., & Stuart, K. (2017). Promoting young people's wellbeing through empowerment and agency: A critical framework for practice. In *Promoting Young People's Wellbeing through Empowerment and Agency: A Critical Framework for Practice*. https://doi.org/10.4324/9781315676418
- McCashen, W. (2014). The strengths approach. Bendigo: St Luke's Innovative Resources.

McLeod, S. (2018). Maslow 's Hierarchy of Needs Maslow 's Hierarchy of Needs. Business, 3–5.

- Mcneil, B. & Stuart, K. (2022). A Framework of Outcomes for Young People 2.1. In *The Centre for Youth Impact*. https://doi.org/10.2307/j.ctv29sfz5s.9
- Mcneil, B., Millar, G., & Fernandez, S. (2019). A Framework of Outcomes Framework of Outcomes 2012. Retrieved from https://www.youthimpact.uk/sites/default/files/2020-11/outcomes_framework_report_final.pdf
- Mcneil, B., & Stuart, K. (2022). Socio-emotional skill updates for youth work and provision for young people. Retrieved from https://www.youthimpact.uk/sites/default/files/2022-03/SES paper FINAL.pdf
- McNiff, J & Whitehead, J. (2012). *All you need to know about Action Research* (2nd ed.). London: Sager.
- McPartlan, D., Burrus, A., Elder, K., Gregory, P., Hillary, M., McCrea, C., Bell, S., Greenup, C., James, C., Liddell, J., Liddell, K., Norwood, E., Rome, A., & Schollick, J. (2021). *The Benefits Of Young Researchers In A School YPAR Project*. Retrieved from https://www.socialpublishersfoundation.org/knowledge_base/the-benefits-of-youngresearchers-in-a-school-ypar-project/

McPartlan, D. (2019a). Positive mental health policy. *Mental Health Care*, Vol. 1.

- McPartlan, D. (2019b). Working to improve the wellbeing and mental health of pupils at William Howard School.
- McPartlan, D. (2019c). Working to improve the wellbeing and mental health of staff at William Howard School.
- McPartlan, D. (2021). Power in Schools: An ex-teacher returns as a researcher. *Social Publishing Foundation*. https://doi.org/10.7551/mitpress/12093.003.0008
- McPherson, G., Rodney, P., McDonald, M., Storch, J., Pauly, B., & Burgess, M. (2004).Working within the landscape: Applications in health care ethics. In P. R. & R. S. J.Storch (Ed.), *Toward a moral horizon: Nursing ethics for leadership and practice*.
- Mercer, G. (2002). Emancipatory disability research. In L. Barnes, C., Oliver, O., & Barton (Ed.), *Disability studies today* (pp. 228–249). Cambridge, MA: Polity Press.
- Meredith, M. (2020). Five ways to help children heal when schools reopen. Retrieved from https://marymered.wordpress.com/2020/04/14/five-ways-to-help-children-heal-when-schools-reopen/
- Mills, C. (2008a). Opportunity and resignation within marginalised students: Towards a theorisation of the reproductive and transformative habitus. *Critical Studies in Education*, 49(2), 99–111. https://doi.org/10.1080/17508480802040191
- Mills, C. (2008b). Reproduction and transformation of inequalities in schooling: The transformative potential of the theoretical constructs of Bourdieu. *British Journal of Sociology of Education*, Vol. 29, pp. 79–89. https://doi.org/10.1080/01425690701737481
- Mitra, D. & Serriere, S. (2015). *Civic education in the elementary grades: promoting engagement in an era of accountability*. New York: Teachers College Press.
- Mohamed, H. (2020). *What it takes it to make it in modern Britain: People like us* (1st ed.). London: Profile books Ltd.
- Mónaco, E., Schoeps, K., & Montoya-Castilla, I. (2019). Attachment Styles and Well-Being in Adolescents: How Does Emotional Development Affect This Relationship? *International Journal of Environmental and Public Health*, 16(14). https://doi.org/10.3390/ijerph16142554
- Moon, J. (2005). *Reflection in learning & professional development*. Retrieved from https://www.researchgate.net/publication/247374308_Reflection_in_learning_profession al_development/citation/download

Morrow, V. (1999). 'We are people too': children and young people's perspectives on

children's rights and decision-making in England. *International Journal of Children's Rights, 7*, 149–170. Retrieved from https://doi.org/10.1163/15718189920494318

- Moules, T. & Kirwan, N. (2005). *Children's fund Essex: evaluation of the non-use of services in Essex*. Chelmsford: Anglia Ruskin University.
- Muris, P., Meesters, C., & Fijen, P. (2003). The Self-Perception Profile for Children: Further evidence for its factor structure, reliability, and validity. *Personality and Individual Differences*, 35(8), 1791–1802. https://doi.org/10.1016/S0191-8869(03)00004-7
- Nash, R. (1990). Bourdieu on education and social and cultural reproduction. *British Journal* of Sociology of Education, 11(4), 431–447.
- Nassem, E. (2019). Zero tolerance policies cause anger not reflection. Retrieved December 22, 2021, from Birmingham City University website: https://www.bcu.ac.uk/newsevents/news/zero-tolerance-policies-cause-anger-not-reflection
- Nesi, J., & Prinstein, M. (2015). Using Social Media for Social Comparison and Feedback-Seeking-Gender and Popularity Moderate Associations with DepressiveSymptoms.pdf. *Abnormal Child Psychology*, 43, 1427–1438. Retrieved from chromeextension://dagcmkpagjlhakfdhnbomgmjdpkdklff/enhancedreader.html?openApp&pdf=https%3A%2F%2Flink.springer.com%2Fcontent%2Fpdf%2 F10.1007%2Fs10802-015-0020-0.pdf
- Newlove-Delgado, T., McManus, S., Sadler, K., Thandi, S., Vizard, T., Cartwright, C., & Ford, T. (2021). Child mental health in England before and during the COVID-19 lockdown. *The Lancet Psychiatry*, 0366(20), 19–20. https://doi.org/10.1016/S2215-0366(20)30570-8
- NHS Digital. (2018). Mental Health of Children and Young People in England, 2017. In *NHS Digital*.
- NHS England. (2021). Mental health support in schools and colleges.
- Noffke, S. & Somekh, B. (2008). Action Research. In C. Somekh, B. & Lewin (Ed.), *Research methods in the social sciences* (pp. 89–96). Los Angeles, CA: Sage.
- Nussbaum, M. (1990). *Love's knowledge: Essays on philosophy and literature*. New York: Oxford University Press.
- Nyandarai, Z. I., & Egbuonu, N. G. L. (2018). International Students' Interaction Processes in Relation to Their Learning Experiences at One University in Northern China. *European Journal of Education Studies*, 4(2), 62–80. https://doi.org/10.5281/zenodo.1168751
- O'Brien, N., & Moules, T. (2007). So round the spiral again: A reflective participatory research project with children and young people. *Educational Action Research*, 15(3),

385-402. https://doi.org/10.1080/09650790701514382

- O'Connor, T. G. O., Woolgar, M., Humayun, S., Briskman, J. A., & Scott, S. (2019). Early caregiving predicts attachment representations in adolescence : findings from two longitudinal studies. *Journal of Psychology and Psychiatry*, 9, 944–952. https://doi.org/10.1111/jcpp.12936
- O'Doherty, K., & Burgess, M. (2019). Developing Psychologically Compelling Understanding of the Involvement of Humans in Research. *Human Arenas*, 2(4), 433– 450. https://doi.org/10.1007/s42087-019-00066-w
- O'Neill, J. and Adams, P. (2012). "Damned to mediocrity": political targets, bureaucratic intent, classroom performativity and the child as cipher. *New Zealand Journal of Teachers' Work*, *9*, 1–5.
- Office, P. M. (2021). Prime Minister appoints Dr. Alex George as Youth Mental Health Ambassador. Retrieved July 18, 2022, from Gov.UK website: https://www.gov.uk/government/news/prime-minister-appoints-dr-alex-george-as-youthmental-health-ambassador#:~:text=Prime Minister appoints Dr.,Mental Health Ambassador - GOV.UK
- Ofsted. (2019). *William Howard School Ofsted Report*. https://doi.org/https://files.ofsted.gov.uk/v1/file/50078453
- OFSTED. (2015). Early help: whose responsibility? (March), 32.
- OFSTED. (2019). *Education_inspection_framework*. Retrieved from https://www.gov.uk/government/publications/education-inspection-framework
- Olivier, E., Morin, A. J. S., Leo, V., & Salmela-Aro, K. (2022). The interconnected development of depressive symptoms and school functioning from mid-adolescence to early adulthood: A piecewise growth mixture analysis. *Journal of Educational Psychology*. https://doi.org/10.1037/edu0000761
- Olson, D. (2015). A Case for Epistemic Agency. Logos & Episteme, 6(4), 449–474. https://doi.org/10.5840/logos-episteme20156435
- Olthuis, J. (2001). *The beautiful risk: A new psychology of loving and being loved*. Grand Rapids, MI: Zondervan.
- Onoda, K., Okamoto, Y., Nakashima, K., Nittono, H., Yoshimura, S., Yamawaki, S., ... Ura, M. (2010). Does low self-esteem enhance social pain? The relationship between trait self-esteem and anterior cingulate cortex activation induced by ostracism. *Social Cognitive and Affective Neuroscience*, 5(4), 385–391. https://doi.org/10.1093/scan/nsq002

- Onu W et al. (2016). *Social Identity, Equality and Education*. Retrieved from https://www.academia.edu/16628813/Social_Identity_Equality_and_Education
- Orben, A., Tomova, L., & Blakemore, S. J. (2020). The effects of social deprivation on adolescent development and mental health. *The Lancet Child and Adolescent Health*, 4(8), 634–640. https://doi.org/10.1016/S2352-4642(20)30186-3
- Oswell, D. (2013). *The Agency of Children: From Family to Global Human Rights*. Cambridge: Cambridge University Press.
- Overton, I. (2021, September). Etonocracy. *Byline Times*, p. 29. Retrieved from https://subscribe.bylinetimes.com/my-account/downloads/
- Oyserman, D., & Markus, H. (1995). Possible selves and delinquency. *Journal of Personality and Social Psychology*, *59*(1), 112–125. Retrieved from https://psycnet.apa.org/record/1991-01552-001?errorCode=invalidAccess
- Păduraru, M. E. (2019). Coping strategies for exam stress. *Mental Health: Global Challenges Journal*, 1(1), 64–66. https://doi.org/10.32437/mhgcj.v1i1.26
- Parsons, T. (1937). *The Structure of Social Action. Maidenhead: McGraw Hill*. Maidenhead: McGraw Hill.
- Patalay P & Fitzsimons E. (2018). Development and predictors of mental ill-health and wellbeing from childhood to adolescence. Soc. Psychiatry Psychiatry Epidemiol. Social Psychiatry and Psychiatric Epidemiology Volume, 53, 1311–1323. https://doi.org/10.1007/s00127-018-1604-0
- Patton, G. C., Sawyer, S. M., Santelli, J. S., Ross, D. A., Afifi, R., Allen, N. B., ... Viner, R. M. (2016). Our future: a Lancet commission on adolescent health and wellbeing. *The Lancet*, 387(10036), 2423–2478. https://doi.org/10.1016/S0140-6736(16)00579-1
- Peck, S. C., & Smith, C. (2020). Realist(ic) Evaluation Tools for OST Programs|: The Quality-Outcomes Design and Methods (Q-ODM) Toolbox [White Paper #3]. Retrieved from https://www.qturngroup.com/wp-content/uploads/2021/11/2021-11-24 WP3 QDM.pdf

Perry. (1908). The management of city schools (T. M. Company, Ed.). New York.

- Petrides K.V., Mikolajczak M., Mavroveli S., Sanchez-Ruiz M.-J., Furnham A., P.-G. J.-C. (2016). Developments in Trait Emotional Intelligence. *Emotion Review*, 8,(4), 335–341. https://doi.org/10.1177/1754073916650493
- Piaget, J. (1970). Genetic epistemology. (E. Duckworth, Trans.). New York: Columbia University Press. New York: Columbia University Press.
- Piaget, J. (1971). Psychology and epistemology. (A. Rosin, Trans.). New York: Grossman.

(Original work published 1970.).

- Pianta, R. C. (1999). Enhancing relationships between children and teachers. Washington DC: American Psychological Association. Washington DC: American Psychological Association.
- Pitzer, H. K. (2013). Deficit and Neoliberal Discourses, Urban Teachers' Work, and the "Blame Game." (April). Retrieved from

https://surface.syr.edu/cgi/viewcontent.cgi?article=1165&context=etd

- Pleasance, S. (2016). Student Voice and Its Role in Sustainability. In R. Summers, D. Cutting (Ed.), *Education for Sustainable Development in Further Education* (pp. 213–229).
 Palgrave Macmillan.
- Pollard, C. L. (2015). What is the Right Thing to Do: Use of a Relational Ethic Framework to Guide Clinical Decision-Making. *International Journal of Caring Sciences*, 8(2), 362–368. Retrieved from www.internationaljournalofcaringsciences.org
- Power, E. M. (1999). An Introduction to Pierre Bourdieu's Key Theoretical Concepts. Journal for the Study of Food and Society, 3(1), 48–52. https://doi.org/10.2752/152897999786690753
- Puigverta, L., Christoub, M., & Holfordc, J. (2012). Critical communicative methodology: Including vulnerable voices in research through dialogue. *Cambridge Journal of Education*, 42(4), 513–526. https://doi.org/10.1080/0305764X.2012.733341
- Purdy, N; Finn, P; McDade, B., & Harris, J; Winter, F. (2021). LOYALIST AND REPUBLICAN PERSPECTIVES ON EDUCATIONAL UNDERACHIEVEMENT IN NORTHERN IRELAND. Belfast.
- Raffo, C., Dyson, A., Gunter, H., Hall, D., Jones, L., & Kalambouka, A. (2009). Education and poverty: Mapping the terrain and making the links to educational policy. *International Journal of Inclusive Education*, *13*(4), 341–358. https://doi.org/10.1080/13603110802124462
- Råheim, M., Magnussen, L. H., Sekse, R. J. T., Lunde, Å., Jacobsen, T., & Blystad, A. (2016). Researcher-researched relationship in qualitative research: Shifts in positions and researcher vulnerability. *International Journal of Qualitative Studies on Health and Well-Being*, Vol. 11. https://doi.org/10.3402/qhw.v11.30996
- Ratcliffe, M,. & Broome, M. (2012). Existential phenomenology, psychiatric illness, and the death of possibilities. In *The Cambridge Companion to Existentialism* (pp. 361–382).
 Retrieved from https://doi.org/10.1017/CCOL9780521513340.018

Ratcliffe, M. (2016). Existential Feeling and Narrative. In T. Breyer (Ed.), Funktionen des

Lebendigen (1st ed., pp. 169–192). https://doi.org/10.1515/9783110499100-011

- Reason, P. (2006). Choice and Quality in Action Research Practice. *Journal of Management Inquiry*, 15(2), 187–203.
- Reay, D.; David, M & Ball, J. (2005). Degrees of Choice: Social Class, Race and Gender in Higher Education (Vol. 35). Trentham Books Ltd.
- Reay, D. (2004). Education and cultural capital: The implications of changing trends in education policies. *Cultural Trends*, 13(2), 73–86. https://doi.org/10.1080/0954896042000267161
- Reay, D. (2009). Identity Making in Schools and Classrooms. 1–18.
- Reay, D. (2017). Miseducation (1st ed.). Bristol: Policy Press.
- Reid, K., Flowers, P., & Larkin, M. (2005). Exploring lived experience. *Psychologist*, 18(1), 20–23.
- Reis Monteiro, A. (2014). Ethics of Human Rights.
- Reyes, C. & Due, E. (2015). What do parents look for in their child's school? *Infocus*, 51(May), 1–4.
- Reynolds, D. and Sullivan, M. (1987). The Comprehensive Experiment. *Journal of Social Policy*.
- Reynolds, K. J. (2009). Introduction : Part 2 IS TI IS TI (pp. 13–32). pp. 13–32. Retrieved from https://openresearchrepository.anu.edu.au/bitstream/1885/24530/8/04_Turner_The_story_of_social_id_2010 .pdf
- Richardson, L. (2000). Introduction-Assessing alternative modes of qualitative and ethnographic research: How do we judge? Who judges? *Qualitative Inquiry*, 6.
- Ricoeur, P. (1976). *Interpretative Theory: Discourse and the surplus of meaning*. Forth Worth: The Texas Christian University Press.
- Ridout, B., & Campbell, A. (2018). The use of social networking sites in mental health interventions for young people: Systematic review. *Journal of Medical Internet Research*, 20(12), 1–11. https://doi.org/10.2196/12244
- Rigby, K. (2000). Effects of peer victimization in schools and perceived social support on adolescent well-being. *Journal of Adolescence*, 23(1), 57–68. https://doi.org/10.1006/jado.1999.0289
- Riley P. (2009). An adult attachment perspective on student-teacher relationship & classroom management diffculties. *Teaching and Teacher Education*, *25*(5), 626–635.

Riseborough, G. F. (1981). Teacher Careers and Comprehensive Schooling: An Empirical

Study. Sociology.

- Roberts, K. M. (2019). An analysis of autobiographical tools in written reflection:
 implications for teaching critical thinking and goal-setting. *Reflective Practice*, 20(2), 201–217. https://doi.org/10.1080/14623943.2019.1575196
- Robertson, J. (2000). The three Rs of action research methodology: Reciprocity, reflexivity and reflection-on-reality. *Educational Action Research*, 8(2), 307–326. https://doi.org/10.1080/09650790000200124
- Rogers, C. (1957). The necessary and sufficient conditions of therapeutic personality change. *Journal of Consulting Psychology*, 21(2), 95–103.
- Romer, D., Reyna, V. F., & Satterthwaite, T. D. (2017). Beyond stereotypes of adolescent risk taking: Placing the adolescent brain in developmental context. *Developmental Cognitive Neuroscience*, 27(July), 19–34. https://doi.org/10.1016/j.dcn.2017.07.007
- Rosenberg, M. (1985). Self-concept and psychological well-being in adolescene. In R. Leahy (Ed.), *The development of the self*. New York: Academic Press.
- Rothİ, D. M., & Leavey, G. (2006). Mental health help-seeking and young people: A review. *Pastoral Care in Education*, 24(3), 4–13. https://doi.org/10.1111/j.1468-0122.2006.00373.x
- Roy, R., & Uekusa, S. (2020). Collaborative autoethnography: "self-reflection" as a timely alternative research approach during the global pandemic. *Qualitative Research Journal*, 20(4), 383–392. https://doi.org/10.1108/QRJ-06-2020-0054
- Rudasill, K. M., Snyder, K. E., Levinson, H., & L. Adelson, J. (2018). Systems View of School Climate: a Theoretical Framework for Research. *Educational Psychology Review*, 30(1), 35–60. https://doi.org/10.1007/s10648-017-9401-y
- Ruus, V. R., Veisson, M., Leino, M., Ots, L., Pallas, L., Sarv, E. S., & Veisson, A. (2007).
 Students' well-being, coping, academic success, and school climate. Social Behavior and Personality: An International Journal, 35, 919–936. Social Behavior and Personality: An International Journal, 35, 919–936. Retrieved from https://www.ingentaconnect.com/content/sbp/sbp/2007/00000035/00000007/art00007
- Sabri, M., Khoshbakht, F., & Golzar, H. (2015). Relationship between attachment styles and social adjustment: The mediator role of emotional intelligence. *Developmental Psychology: Journal of Iranian Psychologists*, 11(42), 181–194.
- Safari, P. (2020). Iranian ELT Student Teachers' Portrayal of Their Identities as an English Language Teacher: Drawings Speak Louder than Words. *Journal of Language, Identity* and Education, 19(2), 125–141. https://doi.org/10.1080/15348458.2019.1650279

- Saldana, J. (2013). Power and Conformity in Today's Schools. International Journal of Humanities and Social Science, 3(1), 228–232. Retrieved from http://www.ijhssnet.com/journals/Vol 3 No 1 January 2013/27.pdf
- Saldana, J. (2020). Autobiographical Case Study on Teaching, Learning Language, and Culture. Social Behavior Research and Practice – Open Journal, 5(1), 10–18. https://doi.org/10.17140/sbrpoj-5-122
- Sawyer, S. M., Azzopardi, P. S., Wickremarathne, D., & Patton, G. C. (2018). The age of adolescence. *The Lancet Child and Adolescent Health*, 2(3), 223–228. https://doi.org/10.1016/S2352-4642(18)30022-1
- Schäfer, N., & Yarwood, R. (2013). Involving young people as researchers: Uncovering multiple power relations among youths. *Doing Children's Geographies: Methodological Issues in Research with Young People*, 114–128. Retrieved from https://ore.exeter.ac.uk/repository/bitstream/handle/10036/49333/Involving young people as researchers.pdf?sequence=1
- Schelbe, L., Chanmugam, A., Moses, T., Saltzburg, S., Williams, L. R., & Letendre, J. (2015). Youth participation in qualitative research: Challenges and possibilities. *Qualitative Social Work*, 14(4), 504–521. https://doi.org/10.1177/1473325014556792
- Schrag, Z. (2010). Ethical Imperialism: Institutional Review Boards and the Social Sciences, 1965-2009. John Hopkins University Press.
- Schratz, M., & Walker, R. (1995). *Research as social change: New opportunities for qualitative research*. London: Routledge.
- Schütz, A. & Luckmann, T. (1974). Structure of the life-world. London: Heinemann.
- Searle, J. (2004). LANGUAGE AND SOCIAL SCIENCES: DIALOGUE BETWEEN JOHN SEARLE AND CREA (M. Soler, Ed.). El Roure Editorial.

Selman, R. (1980). The Growth of Interpersonal Understanding. New York: Academic Press.

- Sen, A. (1999). Development Freedom. In Development Freedom.
- Sharma, B. (2009). Voice, Accountability and Civic Engagement. *Overseas Development Institute*, (August), 28.
- Sherwin, S. (1998). A relational approach to autonomy in health care. In S. Sherwin. (Ed.), *The politics of women's health: Exploring agency and autonomy*. Philadelphia: Temple University Press.
- Shier, H. (2015). What does 'equality' mean for children in relation to adults? (October 2012), 1–13. Retrieved from http://www.harryshier.net/docs/Shier-What Does Equality Mean for Children.pdf

- Siddiqui, S. & Singh, T. (2016). Social Media its Impact with Positive and Negative Aspects. International Journal of Computer Applications Technology and Research, 5(2), 71–75.
- Silverberg, B. (2020). Too Much Go, Not Enough Slow: Navigating the Challenges of Adolescence. Primary Care - Clinics in Office Practice, 47(2), xv–xvi. https://doi.org/10.1016/j.pop.2020.03.002
- Silvestri, A., Oldfield, M., Squires, P., & Grimshaw, R. (2009). Young people, knives and guns: A Comprehensive Review, Analysis and Critique of Gun and Knife Crime Strategies. Retrieved from www.crimeandjustice.org.uk
- Simkins, T., Maxwell, B., & Aspinwall, K. (2009). Developing the whole-school workforce in England: Building cultures of engagement. *Professional Development in Education*, 35(3), 433–450. https://doi.org/10.1080/19415250902885102
- Simmons, M., et al. (2021). Reflection as pedagogy in action research. *Education Action Research*, 29(2), 245–258.
- Simon, B. (1991). Education and the Social Order 1940-1990. London: Lawrence & Wishart.
- Skiba, R. J. (2008). Are Zero Tolerance Policies Effective in the Schools?: An Evidentiary Review and Recommendations. *American Psychologist*, 63(9), 852–862. https://doi.org/10.1037/0003-066X.63.9.852
- Slattery, P., & Rapp, D. (2003). *Ethics and the foundations of education: Teaching convictions in a postmodern world*. Boston: Allyn & Bacon.
- Smith, N.; Marshall, L.; Albakri, M.; Smuk, M.; Hagell, Aeld,.; Stansfeld, S. (2021). Adolescent mental health difficulties and educational attainment: findings from the UK household longitudinal study. *BMJ Open*. Retrieved from https://www.researchgate.net/publication/353452207_Adolescent_mental_health_difficu lties_and_educational_attainment_findings_from_the_UK_household_longitudinal_stud y
- Smith, M., Cameron, C., & Reimer, D. (2017). From Attachment to Recognition for Children in Care. *British Journal of Social Work*, 47(6), 1606–1623. https://doi.org/10.1093/bjsw/bcx096
- Smyth, J. (2007). Teacher development against the policy reform grain: An argument for recapturing relationships in teaching and learning. *Teacher Development*, 11(2), 221– 236.
- Somekh, B. (n.d.). Inhabiting Each Other's Castles: towards knowledge and mutual growth through collaboration. In B. S. & R. W. C. Day, J. Elliott (Ed.), *Theory and Practice in Action Research: some international perspectives*. (p. 90). UK: Symposium Books.

- Soneson, E., Puntis, S., Chapman, N., Mansfield, K. L., Jones, P. B., & Fazel, M. (2022).
 Happier during lockdown: a descriptive analysis of self-reported wellbeing in 17,000
 UK school students during Covid-19 lockdown. *European Child and Adolescent Psychiatry*. https://doi.org/10.1007/s00787-021-01934-z
- Spencer, M. B. (1999). . Social and cultural influences on school adjustment: the application of an identity-focused cultural ecological perspective. *Educational Psychologist*, 34, 43– 57.
- Spyrou, S. (2020). Children as future-makers. *Childhood*, *27*(1), 3–7. https://doi.org/10.1177/0907568219884142
- Squirrell, T. (2017). Understanding Charles Cooley's "Looking Glass Self." Retrieved August 24, 2021, from https://www.timsquirrell.com/blog/2017/5/15/understandingcharles-cooleys-looking-glass-self
- Sroufe, A. (1995). Emotional development. Cambridge: Cambridge University Press.
- Steele, B. J. (2010). Of "witch's brews" and scholarly communities: The dangers and promise of academic parrhesia. *Cambridge Review of International Affairs*, 23(1), 49–68. https://doi.org/10.1080/09557570903548170
- Stefanou, C. R., Perencevich, K. C., DiCintio, M., & Turner, J. C. (2004). Supporting autonomy in the classroom: ways teachers encourage student decision making and ownership. *Educational Psychologist*, 39(97–110). Retrieved from https://www.tandfonline.com/doi/abs/10.1207/s15326985ep3902_2
- Steinberg, L., Albert, D., Cauffman, E., Banich, M., Graham, S., & Woolard, J. (2008). Age differences in sensation seeking and impulsivity as indexed by behavior and self-report: Evidence for a dual systems model. *Developmental Psychology, Https://Doi.Org/10.1037/A0012955*, 44(6), 1764–1778. Retrieved from https://doi.org/10.1037/a0012955

Steinberg, L. (2008). Adolescence. New York: McGraw Hill.

Steinberg, L. (2014). Adolescent Brain Development Inform Public Policy ? Court Review, 50(3), 70–76. Retrieved from https://drive.google.com/drive/folders/0B4e3FILdCIeReTNUb3M5Sk1adVU?resourcek ey=0-B2Ba3OzdBEw52ilhL-a3ug

Stenberg, K. (2011). Working with Identities—Promoting Student Teachers' Professional Development. Retrieved from

https://helda.helsinki.fi/bitstream/handle/10138/24379/Workingw.pdf?sequence=1 Stephen, B. (2017). *the Education debate* (3rd ed.). Bristol: Policy Press.

- Stoye, G. (2018). The NHS at 70: Does the NHS need more money and how could we pay for it? In *The Kings Fund*. Retrieved from www.soapbox.co.uk
- Stringer, E. & Ortiz Aragon, A. (2021). Action Research (5th ed.). Thousand Oaks CA: Sage Publications.
- Stuart, K. (2020a). Problematic participation: reflections on the process and outcomes of participatory action research into educational inequalities. *Educational Action Research*. https://doi.org/10.1080/09650792.2020.1851737
- Stuart, K. (2020b). Real World Impact : The Rise of the Scholar Activist (pp. 0–16). pp. 0–16.
- Stuart, K., Bunting, M., Boyd, P., Cammack, P., Hornbæk Frostholm, P., Thore Graveson,
 D., ... Walker, S. (2019). Developing an equalities literacy for practitioners working
 with children, young people and families through action research. *Educational Action Research*, 28(3), 362–382. https://doi.org/10.1080/09650792.2019.1593870
- Stuart, K., & Perris, E. (2017). Asset-based youth support—reclaiming the roots of youth work at the Foyer Federation. *Cogent Social Sciences*, 3(1). https://doi.org/10.1080/23311886.2017.1377989
- Suldo, S. M., Thalji-Raitano, A., Hasemeyer, M., Gelley, C. D., & Hoy, B. (2013).
 Understanding middle school students' life satisfaction: does school climate matter? *Applied Research in Quality of Life*, 8, 169–182. Retrieved from https://link.springer.com/article/10.1007/s11482-012-9185-7
- Sumara, D. and Luce-Kapler, R. (1993). Sumara, D. J., and Luce-Kapler, R. Action Research as a Writerly Text: locating co- labouring in collaboration. *Education Action Research*, 1(3), 387.
- Swearer, S. M., Espelage, D. L., Vaillancourt, T., & Hymel, S. (2010). What can be done about school bullying? Linking research to educational practice. *Educational Researcher, 39*, 38–47. Retrieved from https://www.jstor.org/stable/27764552?seq=1
- Tabibnia, G., Satpute, A. B., & Lieberman, M. D. (2008). The Sunny Side of Fairness. *Psychological Science*, 19(4), 339–347. https://doi.org/10.1111/j.1467-9280.2008.02091.x
- Tapal, A., Oren, E., Dar, R., & Eitam, B. (2017). The sense of agency scale: A measure of consciously perceived control over one's mind, body, and the immediate environment. *Frontiers in Psychology*, Vol. 8. https://doi.org/10.3389/fpsyg.2017.01552
- Tate, A. J. M. (2019). Anhedonia and the Affectively Scaffolded Mind. Ergo, an Open Access Journal of Philosophy, 6(20201214), 647–680. https://doi.org/10.3998/ergo.12405314.0006.023

Teague, R. (2000). Social constructivism and social studies. *Retrieved August*, 1–14. Retrieved from

https://moodle.org/pluginfile.php/401/mod_forum/attachment/112845/SocialCo.pdf

- Teixeira, S.; Augsberger, A.; Richards-Schuster, K.; & Sprague Martinez, L. (2021). Participatory Research Approaches with Youth: Ethics, Engagement, and Meaningful Action. *American Journal of Community Psychology*, 68(1–2), 142–153. https://doi.org/10.1002/ajcp.12501
- Teo, T. (2009). Philosophical concerns in critical psychology. In S. Fox, D., Prilleltensky, I., & Austin (Ed.), *Critical psychology: an introduction*. London: Sage.
- Teschl, M., and Comim, F. (2005). Adaptive Preferences and Capabilities: Some Preliminary Conceptual Explorations. *Review of Social Economy*, 63(2), 229–247. Retrieved from https://psycnet.apa.org/record/2005-07950-004
- Thatcher, M. (1995). The Downing Stree Years. London: Harper Collins.
- The Department for Education (DfE). (2022). *Pupil premium : overview*. Retrieved from https://www.gov.uk/government/publications/pupil-premium/pupil-premium/
- The Lancet. (2018). Half measures on children's mental health. *The Lancet*, Vol. 392, p. 360. https://doi.org/10.1016/S0140-6736(18)31697-0
- The National Commission for the Protection of Human Subjects of Biomedical and Behavioral Research. (1979). *THE BELMONT REPORT: Ethical Principles and Guidelines for the Protection of Human Subjects of Research.*
- Thompson, J. (1991). Editor's introduction. In *Language and symbolic power by P. Boudieu* (pp. 1-31.). Cambridge MA: Polity Press.
- Thornberg, R. (2008). School children's reasoning about school rules. *Research Papers in Education*, Vol. 23, pp. 37–52. https://doi.org/10.1080/02671520701651029

Thornbury, R. (1978). The Changing Urban School. London: Methuen.

- Thornton, H. (2007). Every child matters. *Journal of Diabetes Nursing*, *11*(6), 222–223. https://doi.org/10.4324/9781315710044-10
- Thorstensson Dávila, L. (2014). Representing Refugee Youth in Qualitative Research: Questions of Ethics, Language and Authenticity. *Diaspora, Indigenous, and Minority Education*, 8(1), 21–31. https://doi.org/10.1080/15595692.2013.803466
- Tomaselli, G., Buttigieg, S. C., Rosano, A., Cassar, M., & Grima, G. (2020). Person-Centered Care From a Relational Ethics Perspective for the Delivery of High Quality and Safe Healthcare: A Scoping Review. *Frontiers in Public Health*, Vol. 8. https://doi.org/10.3389/fpubh.2020.00044

- Tomlinson, J. (1998). Why so austere? The British welfare state of the 1940s. *Journal of Social Policy*, Vol. 27, pp. 63–77. https://doi.org/10.1017/S0047279497005199
- Treseder, P. (1997). In Empowering Children and Young People: Promoting Involvement in Decision-Making. London: Save the Children.
- Trevarthen, C. (2011). What young children give to their learning, making education work to sustain a community and its culture. *European Early Childhood Education Research Journal*, 19(2), 173–193. https://doi.org/10.1080/1350293X.2011.574405
- Ttofi, M. M., & Farrington, D. P. (2008). Reintegrative Shaming Theory, moral emotions and bullying. Aggressive Behavior, 34(4), 352–368. https://doi.org/10.1002/ab.20257
- Turner, H. A., Shattuck, A., Finkelhor, D., & Hamby, S. (2017). Effects of Poly-Victimization on Adolescent Social Support, Self-Concept, and Psychological Distress. *Journal of Interpersonal Violence*, 32(5), 755–780. https://doi.org/10.1177/0886260515586376
- Turner, J. (1982). Towards a Cognitive Redefinition of the Social Group. In H. Tajfel (Ed.), Social Identity and Intergroup Relations (pp. 15–40). Cambridge: Cambridge University Press.
- Tyler, I. (2013). *Revolting Subjects: Social Abjection and Resistance in Neoliberal Britain*. London: Zed Books.
- Tyler, P. (2014). Negative youth portrayal in the media. Retrieved from Shoutout UK website: https://www.shoutoutuk.org/2014/08/14/negative-youth-portrayal-media/
- U, B. (1979). *The Ecology of Human Development: Experiments by Nature and Design*. Cambridge MA: Harvard Press.
- UNCRC. (2019). The United Nations Convention on the Rights of the Child. https://doi.org/10.1163/9789460913020_012
- UNICEF. (2020). *Worlds of influence* | *UNICEF*. Retrieved from https://www.unicef.org/reports/worlds-influence-what-shapes-child-well-being-richcountries-2020
- van Harmelen, A., Hauber, K., Gunther Moor B., Spinhoven, P., Boon, A., Crone, E., Elzinga, B. (2014). Childhood Emotional Maltreatment Severity Is Associated with Dorsal Medial Prefrontal Cortex Responsivity to Social Exclusion in Young Adults. *Holistic Perspectives on Trauma*, 9(1), 151–182. https://doi.org/10.1201/b18313-12
- van Lieshoult, F., & Cardiff, F. (2011). Innovative ways of analysing data with practitioners as co-researchers. In D. Higgs, J., Titchen, A., Horsfall, D., & Bridges (Ed.), *Creative Spaces for Qualitative Researching: Living Research*. Rotterdam: Sense.

- van Manen, M. (1990). *Researching lived experience: Human science for an action sensitive pedagogy*. Albany, NY: State University of New York Press.
- Varpio, L., Ajjawi, R., Monrouxe, L. V., O'Brien, B. C., & Rees, C. E. (2017). Shedding the cobra effect: Problematising thematic emergence, triangulation, saturation and member checking. *Medical Education*, 51(1), 40–50. https://doi.org/10.1111/medu.13124
- Vervliet, M; Rousseau, C; Broekaert, E; Derluyn, I. (2015). Multilayered Ethics in Research Involving Unaccompanied Refugee Minors. *Journal of Refugee Studies, 28*(4), 468–485.
- Victoria Rideout, M. A., & Michael B, R. (2019). The Common Sense Census. In Common Sense Media. Retrieved from https://www.commonsensemedia.org/sites/default/files/uploads/research/2019-census-8to-18-key-findings-updated.pdf
- W. Shakespeare. (1978). The Winters Tale. In *The complete works of William Shakespeare* (pp. 323–352). London: Abbey Library.
- Waite P (University of Oxford). (2020). Supporting children and young people around attending school during the Covid-19 pandemic. Retrieved from Emerging Minds website: https://emergingminds.org.uk/wp-content/uploads/2020/05/Dr-Polly-Waite school-reintegration-webinar 22 05 20 FINAL.pdf
- Wang, M. T., Brinkworth, M. E., & Eccles, J. S. (2012). The moderation effect of teacher– student relationship on the association between adolescents' self–regulation ability, family conflict, and developmental problems. *Developmental Psychology*, 49(690–705).
- Wang, M & Degol, J. (2016). School Climate: a Review of the Construct, Measurement, and Impact on Student Outcomes. *Educ Psychology Review*, 28, 315–352. Retrieved from https://link.springer.com/article/10.1007/s10648-015-9319-1
- Wang, M. (2009). School climate support for behavioral and psychological adjustment: testing the mediating effect of social competence. *School Psychology Quarterly*, 24, 240–251. Retrieved from https://psycnet.apa.org/buy/2009-23923-003
- Wartella, E., Beaudoin-Ryan, L., Blackwell, C. K., Cingel, D. P., Hurwitz, L. B., & Lauricella, A. R. (2016). What kind of adults will our children become? The impact of growing up in a media-saturated world. *Journal of Children and Media*, 10(1), 13–20. https://doi.org/10.1080/17482798.2015.1124796
- Waters, T., Marzano, R., & McNulty, B. (2004). Leadership that sparks learning. *Educational Leadership*, *61*, 48–52.
- Way, A. K., Kanak Zwier, R., & Tracy, S. J. (2015). Dialogic Interviewing and Flickers of Transformation: An Examination and Delineation of Interactional Strategies That

Promote Participant Self-Reflexivity. *Qualitative Inquiry*, 21(8), 720–731. https://doi.org/10.1177/1077800414566686

- Way, N., Reddy, R., & Rhodes, J. (2007). Students' perceptions of school climate during the middle school years: Associations with trajectories of psychological and behavioral adjustment. *American Journal of Community Psychology*, 40(3–4), 194–213. https://doi.org/10.1007/s10464-007-9143-y
- Weale, S. (2020). Youth services suffer 70% funding cut in less than a decade. Guardian Newspaper. Retrieved from https://www.theguardian.com/society/2020/jan/20/youthservices-suffer-70-funding-cut-in-less-than-a-decade
- Weber, M. (1978). Economy and Society. Berkeley: University of California Press.
- Weedall, A. D., Wilson, A. J., & Wayte, S. C. (2019). The Basics of Brain Development. Br J Radiol, Vol. 92, pp. 224–231. Retrieved from chromeextension://dagcmkpagjlhakfdhnbomgmjdpkdklff/enhancedreader.html?openApp&pdf=https%3A%2F%2Fwww.birpublications.org%2Fdoi%2Fpdf %2F10.1259%2Fbjr.20190300%3Fdownload%3Dtrue
- Weinstein, C. S., Curran, M., & Tomlinson-Clarke, S. (2003). Culturally responsive classroom management: awareness into action. *Theory Into Practice*, *42*, 269–276.
 Retrieved from https://www.tandfonline.com/doi/abs/10.1207/s15430421tip4204_2
- Whiting, J. B. (2007). Authors, artists, and social constructionism: A case study of narrative supervision. American Journal of Family Therapy, 35(2), 139–150. https://doi.org/10.1080/01926180600698434
- WHO. (2014). Health for the world's adolescents: A second chance in the second decade. In World Health Organisation. https://doi.org/10.1016/j.jadohealth.2014.10.260
- WHO. (2019a). Fact sheet on mental health. Retrieved from https://www.euro.who.int/__data/assets/pdf_file/0004/404851/MNH_FactSheet_ENG.p df
- WHO. (2019b). World Health Organisation (WHO) definition of health. Retrieved from https://www.publichealth.com.ng/world-health-organizationwho-definition-ofhealth/#comments
- WHO Mental Health and Substance Use Team. (2022). Transforming mental health for all. In World Mental Health Report. Retrieved from https://www.who.int/publications/i/item/9789240049338
- Wickham, S., Whitehead, M., Taylor-Robinson, D., & Barr, B. (2017). The effect of a transition into poverty on child and maternal mental health: a longitudinal analysis of the

UK Millennium Cohort Study. *The Lancet Public Health*, 2(3), e141–e148. https://doi.org/10.1016/S2468-2667(17)30011-7

- Wöhrer, V., & Höcher, B. (2012). Tricks of the Trade—Negotiations and Dealings between Researchers, Teachers and Students. *FQS Forum Qualitative Socialforschung*, *13*(1), 6– 7. Retrieved from http://www.qualitativeresearch.net/index.php/fqs/article/view/1797/3319
- Wood, L. (2016). Action Research for the 21St Century: Exploring New Educational Pathways. South African Journal of Higher Education, 28(2). https://doi.org/10.20853/28-2-351
- Wood, P. (2012). The Development of pupil strategies. In P. Wood (Ed.), *Pupil Strategies: Exploration in the sociology of the school* (2nd ed., pp. 11–27). Abingdon: Routledge.
- World Medical Association. (2013). Declaration of Helsinki. Retrieved from https://www.wma.net/policies-post/wma-declaration-of-helsinki-ethical-principles-formedical-research-involving-human-subjects/
- World Medical Association. (2016). Declaration of Taipei on ethical considerations regarding health databases and biobanks. Retrieved from https://www.wma.net/policies-post/wmadeclaration-of-taipei-on-ethical-considerations-regarding-health-databases-andbiobanks/
- Wright, N. (1977). Progress in Education. London: Croom Helm.
- Wyse, D. (2001). Felt tip pens and school councils: children's participation rights in four English schools. *Children and Society*, *15*(4), 209–218.